

Office of the Prime Minister's Chief Science Advisor Kaitohutohu Mātanga Pūtaiao Matua ki te Pirimia

Title:

REPORT: Preventing food loss and waste in Aotearoa New Zealand: Evidence for action across the supply chain

Author:

OPMCSA

Output type:				
PDF				
Pages:				
pp 234				
Date:				
Jun-2024				
Language:				
English				
Review:				
Peer reviewed				
Versions				
Record number:	Version:	Date V1 created:	Date:	Printed version
PMCSA-24-6-5-V1	V1	27-Jun-2024	27-Jun-2024	Υ
DOI:	10.17608/k6.OP	MCSA.25321963		
ISBN:	978-0-473-71280-8 (PDF)			
	978-0-473-71279	9-2 (Paperback)		
Archive page link:				
https://dpmc.go	ovt.nz/our-program	<u>mmes/special-progra</u>	ammes/prime-ministers-c	<u>chief-science-</u>
advisor-archives	s/archive/gerrard-	<u>2021-2024</u>		
Notes: -				

Preventing food loss and waste in Aotearoa New Zealand Evidence for action

across the supply chain

The fourth report in the food waste series from the Prime Minister's Chief Science Advisor, Kaitohutohu Mātanga Pūtaiao Matua ki te Pirimia.



June 2024



This document may be copied provided that the source is acknowledged. This report and others by the Office of the Prime Minister's Chief Science Advisor are available at pmcsa.nz

June 2024

ISBN:

978-0-473-71280-8 (PDF)

978-0-473-71279-2 (Paperback)

Other reports in the OPMCSA food waste series:

Food waste: A global and local problem

Food rescue in 2022: Where to from here?

Beyond the bin: Capturing value from food loss and waste

Food loss and waste in Aotearoa New Zealand: Towards a 50% reduction



Office of the Prime Minister's Chief Science Advisor The University of Auckland

Private Bag 92019 Victoria Street West Auckland 1142 Aotearoa New Zealand

Email info@pmcsa.ac.nz | Reb pmcsa.nz

O Instagram @nz_chief_science_advisor | X @ChiefSciAdvisor

Images and graphics that are not credited are public domain, reproduced with permission, don't require attribution, or are owned by OPMCSA.

Front cover (top to bottom):

- 1. Vertical farming at The Greengrower.
- 2. A misshapen tomato.
- 3. Food leftover after a meal.
- 4. A typical 'best before' date on food packaging.
- 5. Food preparation in a household kitchen.

Mā tōu rou, mā taku rourou ka ora te iwi

Foreword

Kia ora koutou,

This is the penultimate report in our series on food loss and waste, which wraps up with our summary report, capturing all our recommendations, published simultaneously.

We began the series by trying to understand the challenge of food loss and waste in our national context, and have since explored food rescue and ways to capture value from food that would otherwise be wasted. We have left the biggest part of the puzzle for last: prevention. Prevention is by far the most important intervention because it has the greatest potential impact on mitigating the social, economic, and environmental harms of lost and wasted food. Preventing food from being lost or wasted saves money and the environmental costs associated with its production.

We look at prevention through the lens of the supply chain. The COVID-19 pandemic illustrated the limits of our 'just in time' supply chains, encouraging a shift to 'just in case', with more food kept in stock at different points in the system. This presents new dimensions to the already challenging problem of how to avoid the loss and waste of 40% of food produced globally.

In Aotearoa, the food loss and waste challenge needs to be understood in our context as an exporter of premium produce. Our economy is built around producing more food than we can eat. When our export supply chains break down, as exemplified during the pandemic, we need mechanisms to prevent the food involved becoming waste.

No single actor in the system can prevent food loss and waste alone. It is a system problem, and so we recommend coordinated changes across the supply chain that make it easier for everyone along the supply chain to prevent as much waste as possible.

As we bring this extended piece of work to a close, I'd like to thank our large and enthusiastic reference group of researchers and stakeholders who have generously given their time to support this work. We hope it provides a useful evidence base to support a systems change to one in which less food is lost and wasted.

Ngā manaakitanga,

Juliet Genrard

Professor Dame Juliet Gerrard DNZM HonFRSC FRSNZ Prime Minister's Chief Science Advisor Kaitohutohu Mātanga Pūtaiao Matua ki te Pirimia

Acknowledgements

This report was drafted by Rebecca Benson, Trixie Croad, Jacques de Satge, Lysea Haggie, Jessica O'Connor, George Slim, and Carolle Varughese. We thank interns Jackson McBreen and Megan McKenzie for their contributions to report appendices.

We thank the many academics, industry experts, entrepreneurs, and government agencies who formed our reference group. To those who met with the team, provided introductions, and generously contributed time, energy, and suggestions to this project, we are thankful for your contributions. We also acknowledge the contribution of members of the Chief Science Advisor Forum.

We have incorporated as much feedback as possible. Our acknowledgement of people who helped us with this project in no way indicates their endorsement of the final recommendations. We have done our utmost to keep track of everyone who has contributed to this work. Please accept our sincere apologies for any inadvertent errors or omissions. Any mistakes in the report are our own.

Particular thanks to

Alice Hume, Commerce Commission Kaitlin Dawson, NZ Champions 12.3 Amanda Wolf, Victoria University of Katherine Silvester, MBIE Wellington Lisa Busch, NZ Champions 12.3 Ashlin Chand, Meat Industry Association Liz Butcher, MfE Barry Wards, MPI Millie Porter, Woolworths NZ Brittany Rymer, MfE Mindy Wigzell, Foodstuffs NZ Ben Reddiex, DOC Miranda Mirosa, University of Otago Chris Houston, Beef + Lamb NZ Nicholas Jolly, Beef + Lamb NZ Diane Stanbra, Rescued Limited Phil Bremer, University of Otago Deepa Goswami, Porirua City Council Qinhua Shen, MPI Emily Levenson, HortNZ Regina Wypych, Beef + Lamb NZ Francesca Goodman-Smith, End Food Waste Australia Roger Cook, MPI Gerald Rys, MPI Sarah Reader, MPI Jacqui Todd, Plant & Food Research Sheila Skeaff, University of Otago Jeff Seadon, Seadon Consulting Vicki Burggraaf, AgResearch Jenny Marshall, previously at MfE Vince Arbuckle, MPI Jessica Broun, NZ Champions 12.3 Vince Galvin, Stats NZ Katie Bright, MfE Wender Martins, University of Auckland

Reference group

Ailsa Robertson, HortNZ Akari Otsuka, Dunedin City Council Alex Kirkham, Auckland Council Ali Reza Nazmi, University of Canterbury Alice Hume, Commerce Commission Alison Collins, MfE Alzbeta Bouskova, Ecogas Amanda Wolf, Victoria University of Wellington Amanda Yates, AUT Amber Walter, MoE Amir Sayadabdi, Victoria University of Wellington Analeise Murahidy, University of Auckland Andrew Dickson, Our Land and Water Andrew East, Massey University Andrew Fisher, Ecostock and Ecogas Andrew McCallum, MBIE Andrew Prest, Sustainability Systems Angela Calver, KiwiHarvest Angela Clifford, EatNZ Anton Drazevic, Nelson Environment Centre Antonia Miller, Plant & Food Research Antony Heywood, Vegetables NZ Asch Harwood, ReFED (US) Ashlin Chand, Meat Industry Association Awilda Baoumgren, MPI Bailey Perryman, Nōku te Ao Charitable Trust Barbara Annesley, MfE

Barry Wards, MPI Ben Baldwin, Queensland Department of Agriculture and Fisheries (Australia) Ben Elms, Biochar Network NZ Ben Reddiex, DOC Benje Patterson, Independent economist Benoit Guieysse, Massey University Betsy Kettle, City to Farm Bex Graham, Independent researcher Bill Kaye-Blake, NZ Institute of Economic Research Brendon Malcolm, Plant & Food Research Brent Kleiss. NZ Pork Brett Robinson, University of Canterbury Brian Cox, Bioenergy Association Bridget Murphy, MoH Brittany Rymer, MfE Brittney Evans, Manawatū District Council Brody Gilroy, Waste-Ed with Kate Bruce Middleton, Waste Not Consulting Cameron Crawley, Satisfy Food Rescue Cara McNicol, Queensland Department of Environment and Science (Australia) Carel Bezuidenhout, Massey University Caroline Dewstow-Blanch, The Warehouse Group Carolyn Lister, Plant & Food Research Cath Gledhill, Dunedin City Council Catherine Rosie, Auckland Council Cecilia Manese, Foodstuffs NZ

Chelsea Mintrom, MPI Chen Wu, previously at FlyFarm (Australia) Cherie Pugh, Pyrocal (Australia) Chloe Lynch, MoH Chris Daughney, Te Uru Kahika Chris Galloway, Massey University Chris Hewins, MPI Chris Henderson, Dunedin City Council Chris Houston, Beef + Lamb NZ Chris Purchas, Tonkin + Taylor Christiane Rupp, University of Auckland Christina McBeth, Nourished for Nil Claire Hanrahan, Compass Group Claire Mortimer, MBIE Cliona Ni Mhurchu, University of Auckland Craig Bunt, University of Otago Craig Cliff, University of Otago Cristina Cleghorn, University of Otago Dana Gunders, ReFED (US) Daniel Morrimire, Manawatū Food Action Network Daniel Yallop, Re.Group Danielle Kennedy, MfE Danielle LeGallais, Sunday Blessings Darrin Hodgetts, Massey University Dave Christie, Snotek Dave Perkins, Waste Management David Carlton, DOC David Howie, Waste Management David Jefferson, University of Canterbury

David Kettle, City to Farm David Whitehead, Manaaki Whenua Dawn Hutchesson, AFRA Deborah Manning, KiwiHarvest and NZ Food Network Debs Mclaughlin, Fair Food Deepa Goswami, Porirua City Council Denise Conroy, Plant & Food Research Des Flynn, The Warehouse Group Diane Stanbra, Rescued Kitchen Diane Mollenkopf, University of Canterbury Dinarie Abeyesundere, MSD Donnell Alexander, MPI Dorthe Siggaard Duncan Wilson, Eunomia Consulting Eleonora DeCrescenzo, MSD Eli Gray-Stuart, Massey University Elise O'Brien, Auckland Council Elodie Letendre, Dunedin City Council Emil Murphy, Deer Industry NZ Emily King, Spira Emily Levenson, HortNZ Emma Harding, Foodstuffs NZ Emma Richardson, Climate Change Commission Emma Taylor, MPI Enda Crossin, University of Canterbury Erin Breen, MPI Erin Leitao, University of Auckland Erin Young, University of Otago

Eva Gaugler, Scion Fiona Duncan, MPI Fliss Roberts, Greenback Frances Clement, NZ Pork Francesca Goodman-Smith, End Food Waste Australia Freya Hjorvarsdottir, MPI Gami Karunasena, End Food Waste Australia Garth Lamb, Re.Group Gavin Findlay, NZ Food Network Geoffroy Lamarche, PCE Georgina Langdon-Pole, Auckland Council Georgina Morrison, Environment Hubs Aotearoa Gerald Rys, MPI Glenn Wigley, MfE Grace Clare, University of Otago Gradon Diprose, Manaaki Whenua Grant Blackwell, Climate Change Commission Grant Verry, The FoodBowl Hadas Ore, Waiheke Resources Trust Hannah Blumhardt, The Rubbish Trip Hans Maurer, AgriChain Centre Harmony Ryder, KiwiHarvest Harry Livesey, MfE Heather Riddell, MPI Helen Darling, Sumfood Hilary Sharpe, MoH Iain Lees-Galloway, AFRA Ian Barugh, NZ Pork

lan Town, MoH Indrawati Oey, University of Otago Ingrid Cronin-Knight, Waste Management Ivan Chirino-Valle, MfE Ivy Gan, Plant & Food Research Jack Heinemann, University of Canterbury Jacqui Forbes, Para Kore Jacqui Horswell, MBIE Jacqui Todd, Plant & Food Research Jacqui Yip, Auckland Council Janet Cole, Kaipātiki Project Janet Lymburn, MPI Jarrod Haar, AUT Jeff Seadon, Seadon Consulting Jenny Marshall, previously at MfE Jenny Elliott, Wellington City Council Jenny Ford, Mynoke Jenny Grainger, Venerdi Jeremy Helson, Seafood NZ Jesse Nichols, MSD Jessica Broun, NZ Champions 12.3 Jim Jones, Massey University Jo Fountain, Lincoln University Jo Sharp, Plant & Food Research Jo Wrigley, Go Eco Waikato Environment Centre Joanna Langford, Wellington City Council Joanne Ferry, Tonkin + Taylor Joanne Hort, Massey University Joanne Kingsbury, ESR

Joanne Todd, University of Auckland Jocelyn Eason, Plant & Food Research John Bronlund, Massey University John Milligan, Foodbank Canterbury John Roche, MPI Jonathan Elms, Massey University Jonathan Hannon, Massey University Jonathan Middis, End Food Waste Australia Josie Lambert, Food Nation Joya Kemper, University of Canterbury Judith Goldsack, Nourished for Nil Juhi Shareef, Project Moonshot Julian Heyes, Massey University Julie Dickinson, Auckland Council Julie Harris Julie North, Foodcom Juliet Armstrong, MPI Julio Bin, Auckland Council Kaitlin Dawson, NZ Champions 12.3 Kang Huang, University of Auckland Karen Lau, MPI Karen Lee, Nelson City Council Karen Fernandez, University of Auckland Kate Meads, Waste-Ed with Kate Kate Parker, Scion Kate Porter, previously at Woolworths NZ Kate Springer, Commerce Commission Kate Walmsley, Kaicycle Katherine Silvester, MBIE Kathryn Pavlovich, University of Waikato

Kathy Voyles, Waiheke Resources Trust Katie Bright, MfE Katie Buller, Auckland Council Katy Bluett, Future Foods Aotearoa Kelly Drombroski, Massey University Kenny Lau, NZ Trade and Enterprise Kim Stretton, Waikato Regional Council Kim Hang Pham Do, Massey University Kirra Havemann, Sunday Blessings Kirsten Unger, Bakeworks Kristin Busher, Waiheke Resources Trust Kristin Joiner, The Connective Lance Williams, Kaibosh Lara Cowen, MfE Laura Hetherington, MPI Lauren Beattie, Gizzy Kai Rescue Lauren Simpson, Auckland Council Leanne Young, University of Auckland Lena Kovac, WasteMINZ Lesley Ottey, Eco Educate Liam Prince, The Rubbish Trip Libby Harrison, NZFSSRC Linden MacManus, MfE Lisa Bridson, Formerly Te Whatu Ora Lisa Busch, NZ Champions 12.3 Lisa Eve, Eunomia Consulting Lisa Te Morenga, Massey University Livné Ore, Waiheke Resources Trust Liz Butcher, MfE Liz Goodwin, World Resources Institute (UK) Logan Dingle, Living Earth Louise Lee, Independent researcher Luca Serventi, Lincoln University Lucy Pierpoint Madeline Shelling, University of Auckland Madelon de Jongh, WasteMINZ Madi Walter, NZ Food Network Manpreet Dhami, Manaaki Whenua Marc Gaugler, Scion Marco Morgenstern, Plant & Food Research Marian McKenzie, Plant & Food Research Marianne Lukkien, MPI Mark Barthel, End Food Waste Australia Mark Bell, Woolworths NZ Mark Casey, Foodstuffs NZ Mark Milke, University of Canterbury Marshall Bell, The FoodBowl Martin Workman, MfE Mary-Ann Carter, MoH Mat Walton, ESR Matiu Prebble, University of Canterbury Matthew Ashworth, ESR Meg Thorsen, University of Otago Meghan Cooper, Rotorua Lakes Council Meghan Hughes, AFRA Melanie Vautier, MfE Melissa Hodd, Foodstuffs NZ Michael Backhurst, Auckland Council Michael Brooks, NZ Feed Manufacturers Association

Michael Hall, University of Canterbury Michael Maahs, Waiheke Resources Trust Michael Macbeda, Waimate District Council Michael Quintern, Mynoke Michaela Coleman, MSD Michal Garvey, Foodprint Michelle Blau, Fair Food Michelle Gibbs, MPI Mike Beare, Plant & Food Research Mike Perry, DOC Milana Blakemore, MPI Millie Porter, Woolworths NZ Mindy Wigzell, Foodstuffs NZ Miranda Mirosa, University of Otago Mohan Dutta, Massey University Molly Chapman, End Food Waste Australia Monisha Wylie-Kapoor, Auckland Council Na Luo, Dongbei University of Finance and Economics (China) Nadine Wakim, Auckland Council Natalie Exeter, Nourish App Neil Birrell, University of Auckland Neill Ballantyne, MSD Nic Turner, Mainstream Green Nicholas Jolly, Beef + Lamb NZ Nick Hanson, NZ Pork Nick Lanham, Central Otago District Council Nick Loosley, Everybody Eats Nick Smith, Riddet Institute Nicole Banks, Tauranga City Council

Nigel Davenport, Venture Timaru Nigel French, Massey University Nitha Palakshappa, Massey University Noel Thomson, Snotek Olga Pantos, ESR Parul Sood, Auckland Council Paul Bennett, Scion Paul Johnstone, Plant & Food Research Petelo Esekielu, Auckland Council Peter Cressey, ESR Peter Landon-Lane, T&G Phil Bremer, University of Otago Phil Cumming, The Warehouse Group Phil Grainger, Venerdi Phillipa Hunt, Satisfy Food Rescue Phillippa Hawthorne, MPI Qinhua Shen, MPI Racheal Bryant, Lincoln University Raewyn Bleakley, NZ Food and Grocery Council Ray O'Brien, University of Otago Rea Kenkel, Healthy Families Waitākere Rebecca Culver. Just Zilch Rebecca Doonan, MPI Regina Wypych, Beef + Lamb NZ Renwick Dobson, University of Canterbury Ricardo Bello-Mendoza, University of Canterbury Richard Love, Massey University Richard O'Driscoll, NIWA

Rob Tinholt, Watercare Roderick Boys, MfE Roger Cook, MPI Roger Hurst, Plant & Food Research Sally Fraser, Waipā District Council Sam Beaumont, KiwiHarvest Sam Buckle, MfE Sam Oakden, End Food Waste Australia Sara Mustafa, University of Auckland Sara Smeath, CiRCLR Sarah Crisford, The Warehouse Group Sarah Gell, Dunedin City Council Sarah Grant, Magic Beans Sarah Knight, University of Auckland Sarah Pennell, Foodbank Australia Sarah Pritchett, WasteMINZ Sarah Reader, MPI Sean Connelly, University of Otago Serena Curtis, MSD Sharon McIver, Our Daily Waste Shaun Lewis, MfE Shawn Gardner, Enspire NZ Shawn Shepherd, ReFED (US) Sheila Skeaff, University of Otago Sheree Kearney, Whakatāne District Council Sheryl Ching, MoE Simon Lipscombe, Compass Group Simon Lockrey, End Food Waste Australia Sonya Cameron, Kore Hiakai Zero Hunger Collective

Sophie Mander, Queenstown Lakes District Council Sophie Percy, NZ Food Network Spring Humphreys, EnviroWaste Stef Van Meer, Satisfy Food Rescue Stephanie Hill, MfE Stephen Trebilco, MBIE Stewart Collie, AgResearch Stewart Donaldson, IRD Subh Ganguly, University of Auckland Sunshine Yates, Sunshine Yates Consulting Susanna Barris, MPI Susie Robertson, Kaibosh Susie Trinh, Auckland Council Taima Moeke-Pickering, Laurentian University (Canada) Talia Hicks, AgResearch Tane Leong, MfE Tava Olsen, University of Melbourne (Australia) Te Kawa Robb, Para Kore Tessa Brothersen, WasteMINZ Tessa Vincent, Climate Champions (UK) Thao Le, AUT Tim Garlick, MSD Timofey Shalpegin, University of Auckland Toine Timmermans, Wageningen University & Research (Netherlands) Tracey McIntosh, MSD Tracey Pirini, Fair Food

Tric Malcolm, Kore Hiakai Zero Hunger Collective Valerie Bianchi, Waikato Regional Council Veronica Shale, Zero Food Waste Aotearoa Vicki Burggraaf, AgResearch Victoria Egli, University of Waikato Vince Arbuckle, MPI Vince Galvin, Stats NZ Wai Woo, University of Auckland Wallis Greenslade, MfE Wayne Langford, Meat the Need Wender Martins, University of Auckland Wendy Zhou, Perfectly Imperfect Zoe Mack, Climate Change Commission

Contents

Foreword	ii
Acknowledgements	iii
Particular thanks to	iii
Reference group	iv
Contents	xi
Summary and recommendations	1
Executive summary	1
Recommendations	4
Theme 1: Systems problem, systems solutions	5
Theme 2: Measure and monitor	7
Theme 3: Prevent food loss and waste at source	8
Theme 4: Save good food for people	13
Theme 5: Capture value from unavoidable food waste	13
1. Preventing food loss and waste – the top of the food recovery hierarchy	14
1.1 Definition and scope	15
1.2 The food supply chain	15
1.3 Food supply chains are both global and local	17
1.3.1 Food export is a central part of New Zealand's economy	20
1.4 Reshaping the food system	21
2. Looking across the whole supply chain	22
Our strong exporting orientation creates challenges and opportunities	22
2.1 The drivers of food loss and waste across the chain	23
2.1.2 Market driven	24
2.2 Potential solutions	27
2.2.1 Research and innovation	29
2.2.2 Digitalisation of the supply chain	29
2.2.3 Shifting practice across the supply chain and in households	30
2.2.4 Finding ways to keep imperfect produce in the supply chain 2.2.5 Changes to nurchasing agreement structures	31
2.2.6 Collaboration	32
2.2.7 Implications for policy settings	34
2.3 Sticking points	35
2.3.1 There is an evidence gap	35
2.3.2 Relaxing quality specifications has potential benefits but would create new challenges	35
2.3.3 Voluntary agreements are voluntary	36
2.3.4 Adoption of innovation can be expensive and slow 2.3.5 Sharing data is sometimes perceived as a risk to competitive advantage	30
3. Production	38
Understanding food loss during production	38
3.1.1 New Zealand dairy sector	41

Annex 2: International examples of organic bans to landfill Africa	140 141
Annex 1: Loss and prevention in horticulture and arable production	134
8. Preventing food loss and waste requires a systems approach	133
7.2 Supporting research and innovation to prevent food loss and waste	131
 7.1 The research and innovation landscape for food loss and waste prevention 7.1.1 The contribution of business to food research and innovation 7.1.2 Examples of public sector organisations that do research into FLW prevention 7.1.3 Collaboration between industry and research is vital for sector impact 	129 130 130 130
7. Research for food loss and waste prevention	128
6.3 Households' role in preventing food loss and waste across the supply chain	127
 6.2 How can we prevent food waste in households? 6.2.1 Promoting voluntary behaviour change 6.2.2 Using policy as a tool to drive change at a system level 6.2.3 Identifying and prioritising effective ways to prevent household food waste 	120 120 124 125
6.1 Understanding household food waste6.1.1 How much food do New Zealand households waste?6.1.2 Drivers of food waste in households	114 114 116
 5.2.1 There are interventions that can prevent food waste within the retail and food service sectors 5.2.2 Preventing FLW upstream: the role of retailers and food service 5.2.3 Preventing food waste downstream: the role of retail and food service 5.2.4 Prevention efforts: involving retailers and food service providers in system-wide initiatives 6. Household 	99 103 107 110 113
 5.1 Understanding food waste in retail and food service 5.1.1 Understanding supply chains 5.1.2 What do we know about food waste in retail and food service? 5.1.3 What are the gaps in our knowledge? 5.2 How can we prevent food waste in retail and food service? 	90 91 93 98 98
5. Retail and food service	90
4.2 How can we prevent FLW during processing and manufacturing?4.2.1 What business strategies could be considered?4.2.2 What technological interventions are available?4.2.3 What policy levers should be considered?	70 72 74 83
 4.1 Understanding FLW during processing and manufacturing 4.1.1 Processing and manufacturing in New Zealand 4.1.2 What causes food loss during processing and manufacturing? 4.1.3 We need data to map FLW in food processing and manufacturing 	64 65 67 68
 4. Processing and manufacturing Levels of food processing Tensions and trade-offs in the processing and manufacturing of food 	61 62 63
3.2 How can we prevent food loss during production? 3.2.1 Potential solutions	53 53
 3.1.2 Meat production 3.1.3 Horticulture and arable 3.1.4 Fishing and aquaculture 3.1.5 What are the gaps in our knowledge? 	43 48 52 52

Asia	141
North America	142
Europe	144
Annex 3: Interventions to prevent food waste in retail and food service	149
Annex 4: Household food waste measurement methodologies	154
Annex 5 : Household food waste prevention interventions	157
Abbreviations	165
Glossary	168
References	173

Summary and recommendations

Executive summary

The food recovery hierarchy (below) is a framework for thinking about food loss and waste (FLW). Our previous reports <u>Food rescue in 2022: Where to from here?</u> and <u>Beyond the bin: Capturing value</u> <u>from food loss and food waste</u> describe many ways to solve some of the problems created by FLW at different levels of the hierarchy. This report focuses on the top level of the hierarchy, prevention. Preventing FLW across all parts of the supply chain is a more effective way to save money and mitigate climate change than simply managing wasted food at the end of its life, because this prevents the unnecessary financial and environmental costs that FLW incurs along the food supply chain, as well as the financial and environmental costs of its recovery or disposal.



When we talk about prevention in this report, we mean any efforts to stop food going to lower levels of the hierarchy. For further clarity, we borrow ReFED's framing:

"Prevention efforts focus on interventions at the root causes of food waste—they locate and address inefficiencies in the food system and food related practices before excess food is produced, transported to places where it cannot be utilized, or discarded rather than eaten."

Many people and businesses are involved in the pathway getting food from the place where it originates to the places where we eat it, often modifying or transforming it along the way. This pathway forms the supply chain. Emissions associated with FLW occur not only when it is landfilled, but in every stage along the chain, which is why prevention is at the top of our food recovery hierarchy.

The food supply chain is a complex system, and efforts to influence that system – like preventing FLW – need to take a systems view. The relationships between stakeholders across the chain, their relative power, and the different economic incentives they face all contribute to FLW. We can't understand

why we have FLW, much less design and implement ways to prevent it, without understanding these factors. Our context in Aotearoa, where food export makes a large contribution to the economy, is an important consideration in our efforts to prevent FLW.

While we present the systems view first, we also recognise the value in looking at each stage of the supply chain in depth. Production – where food is grown and harvested – is the first stage of the food supply chain. Food loss in production includes produce that is grown for human food use and not harvested, pre-harvest losses, losses through disease and weather impacts, and food rejected on farm because of safety or quality concerns. Globally, the production stage has been identified as accounting for 25% of total FLW, the second highest proportion of FLW after households. On farm food loss is, however, recognised globally as a significant data gap. This is partly due to the difficulty of accurately measuring food loss on farms because of logistical barriers and the variability between crops. FLW that occurs in production has a complex set of interacting drivers that are often outside of the control of producers themselves. Common drivers include labour shortages, weather events, and prohibitive cost of harvest. Power dynamics in the food supply chain combined with the variable nature of weather patterns, exacerbated by climate change, often mean that producers are left with significant financial risks, which can result in overproduction as a form of risk mitigation.

The first steps in preventing food losses during production are:

- Addressing market drivers.
- Exploring the potential of cooperative business models to improve farmers' market power.
- Supporting farmers to identify and share data on their food loss occurrence.
- Enabling access to viable secondary markets, for example, for rejected fruit.
- Adoption of technology that can prevent losses pre- and during harvest.

Processing and manufacturing, the next stage of the food supply chain, involves creating new ingredients or products that often have a longer shelf life and greater consumer appeal than whole foods. Food loss during this stage is not openly shared in detail. It can occur due to the removal of parts that are not typically used or eaten; difficulties in planning inventories and matching production with demand; machine faults or human errors; and the requirement to have samples on which to carry out quality control testing and meet product specifications. Increasing efficiencies in manufacturing processes through new business practices and technology integration can help prevent unnecessary FLW within this stage of the supply chain, as well as have an impact both upstream and downstream through interactions with suppliers, distributors, and consumers.

Key to facilitating the implementation and evaluation of preventative practices by food processing and manufacturing companies are:

- Policies around waste measurement and reporting for processors and manufacturers.
- Innovation in packaging and storage technology.
- Resources for the adoption of new digital technologies.

Retail and food service sectors are uniquely positioned in the food supply chain, interacting directly with consumers and suppliers of food. As such, we can think about FLW in these sectors in two ways: waste that occurs within retail and food service sectors, and FLW that occurs up and down the supply chain but is driven by these sectors. We know little about food waste generated within retail and food service in New Zealand; extrapolation from limited

studies suggests it's significant, but less than that which occurs in households. Food waste within the retail sector is driven by a combination of operational inefficiencies like poor demand forecasting, and market-driven practices, like overstocking.

In preventing FLW upstream in the supply chain, two key levers are available to retail sectors: addressing cosmetic specifications to reduce on farm losses and improving ordering practices to limit overproduction. Looking downstream, retailers have a role to play in educating consumers and can influence purchasing decisions through their marketing strategies to help reduce food waste in stores and in households.

In food service, international evidence suggests that plate waste from consumers is the primary driver of food waste in the sector, followed by operational inefficiencies like non-optimal inventory management. Changes to the way food is offered to consumers – such as portion sizes in restaurants and cafes or the availability of trays in self-service venues – may be effective at addressing plate waste, while digitalisation could improve operational efficiency in ways that prevent food waste.

At the end of the food supply chain, households in New Zealand generated an estimated 122,547 tonnes of food waste each year, the equivalent of \$872 million of edible food. This accounts for a significant portion of FLW across the New Zealand food supply chain; it is estimated to account for 40% of all FLW. However, without robust data from other parts of the supply chain, this estimate relies heavily on data extrapolated from international sources. Drivers of household food waste involve a complex interplay of behavioural, environmental, and socio-cultural factors. Interventions which empower people by giving them tools, knowledge and support, in combination with environmental modifications that promote waste prevention behaviours, tend to be more successful than information-based approaches in isolation. Policy makers can support evidence-based interventions by setting strategic direction, enacting legislation and regulation, and facilitating infrastructure and services that reduce food waste.

Scientists, engineers, and entrepreneurs, both in Aotearoa and abroad, are innovating in ways that could prevent FLW. From breeding crops with characteristics that promote shelf life, to smart packaging that can help consumers, to digitalisation that can make supply chains more efficient, and many other advances, there are abundant opportunities in this space. With the right support for both the research and innovation and food systems, we can capture these opportunities. However, some of the drivers of FLW are not amenable to technical improvements, but instead come from the structure of the market and the relationships between different actors across the supply chain. Consumer preferences and habits also play a role. These factors will need to be addressed alongside any gains we can make through scientific innovation.

There is no single cause of FLW in our food system, and so it is unlikely that there will be a single solution. There is also a lot we don't know about FLW and that is a big part of the mission, but uncertainty doesn't have to be a barrier to action. At the highest level, we should invest in technical and other solutions that optimise the operations of the supply chain, while also ensuring that the market is set up to incentivise prevention of FLW. Currently, economic incentives and supply chain power dynamics drive much of FLW and the costs of FLW are accrued where the FLW is realised, not where it is caused. Preventing FLW meaningfully requires a system-wide view and a coordinated and strategic approach, informed by a data-rich evidence base.

Recommendations

Recommendations are made under five themes. Each report in the series contains recommendations under these themes.^{*} The recommendations relating to this report are summarised below.

Systems problem, systems solutions		 P1: Utilise sector action plans (SAPs) to identify intervention opportunities that take a systems view while allowing for the unique contexts of different sectors. P2: Evaluate the effect of the Grocery Supply Code on trade term driven food loss and waste.
Measure and monitor		P3: Support the creation and adoption of a data platform for the sector.
Prevent food loss and waste at source	\bigotimes	 P4: Encourage novel and emergent models of food purchase by consumers. P5: Identify mechanisms to avoid food loss and waste caused by extreme weather events. P6: Update specification practices that lead to edible food being discarded. P7: Continue to explore alternatives to traditional date labelling. P8: Pursue opportunities to reduce food loss and waste through research and innovation. P9: Support evidence-based consumer communications campaigns.
Save good food for people	101	There are no recommendations under this theme in this report.
Capture value from unavoidable food loss and waste		There are no recommendations under this theme in this report.

Each recommendation contains detailed sub-recommendations. For each sub-recommendation, we provide an indicative timeframe for implementation.

- Next 12 months These recommendations should be considered for immediate implementation, to capture existing momentum and make the most of low-hanging fruit.
- **By 2027-2028** These recommendations might take a little longer to implement but should be pursued in the near term to keep Aotearoa on track to a future without food waste.
- **By 2030** The UN <u>Sustainable Development Goal (SDG) 12.3</u> calls for per capita retail and household waste to be halved by 2030, and for food loss to be reduced elsewhere in the food system. These recommendations should be considered for implementation by 2030, in pursuit of SDG 12.

^{*} We have maintained the term "food waste" in theme names and descriptions for consistency with previous reports, but this should be interpreted to include food loss.

Theme 1: Systems problem, systems solutions

Combatting food loss and waste requires people throughout the food system and in the waste management sector to work collaboratively towards a shared vision. To achieve this, we need a national food loss and waste strategy and reduction target, and coordination mechanisms that empower stakeholders to bring the shared vision to life.



Prevention recommendations for theme 1

P1: Utilise sector action plans (SAPs) to identify intervention opportunities that take a systems view while allowing for the unique contexts of different sectors.

Next 12 months	By 2027-2028	By 2030
a) Undertake a trial SAP for one key sector.	 b) Undertake process and outcome/impact evaluation for the SAP undertaken as part of <u>P1.a</u>. c) Using the learnings from <u>P1.b</u>, undertake SAPs for other key sectors. 	 d) Implement interventions identified in SAPs to support system change.

Considerations

This work could be led by the Ministry for the Environment (MfE), and informed by best practice overseas and work by the Kai Commitment in New Zealand.

It will be important to have meaningful partnership from organisations across each sector to ensure interventions identified are workable. End Food Waste Australia (EFWA) has undertaken <u>SAPs for several sectors</u>. These could be useful models for SAPs in Aotearoa. If resources are limited, an abbreviated process could be to seek input from sector stakeholders on the feasibility and local applicability of interventions identified in the Australian SAPs in New Zealand.

Sectors can be but don't have to be defined by food product. For example, EFWA has undertaken SAPs for <u>bread and bakery</u>, <u>cold chain</u>, <u>food rescue</u>, <u>dairy</u>, <u>food service</u>, and <u>horticulture</u>.

SAPs will be more useful if designed to include regular reviews and refreshing. <u>P1.d</u> will require a dedicated administrative body with industry buy in, as implementation processes will require coordination.

P2: Evaluate the effect of the Grocery Supply Code on trade term driven FLW.

Next 12 months	By 2027-2028	By 2030
 a) Commission an evaluation designed to understand the effects of the Grocery Supply Code (the Code) on FLW. 	 b) Receive and report interim findings. c) Identify areas where amendments to the Code may improve results. d) Signal willingness to use enforcement or stronger regulation if interim results suggests that the Code is not effective. 	 e) Receive and report final findings. f) Consider whether updates to the Code or new regulation is necessary in light of P2.e. g) Design and implement an ongoing monitoring mechanism.

Considerations

<u>The Code</u> is intended to promote fairness, transparency, and certainty in the grocery market. Although not its purpose, the Code has potential to prevent FLW.

P2 could be a joint workstream between MfE and Ministry of Business, Innovation and Employment (MBIE) in consultation with the Commerce Commission. The scope of this workstream could include topics of interest to the Commission beyond FLW.

<u>P2.a</u> should consider the whole supply chain, as some requirements of the Code may prevent food loss upstream.

The European Commission's <u>directive against unfair practices in the food supply chain (EU 2019/633)</u>, and the <u>review of the Australian Grocery Code of</u> <u>Conduct</u> published in 2024 that includes specific reference to FLW (page 58), may be useful background.

Theme 2: Measure and monitor

We need to know more about food waste in Aotearoa. Not just how much food is wasted, but where in the food system that waste occurs, current diversion practices, dominant food waste types, and geographic variation in waste volumes. Good data is crucial to articulating the challenge, galvanising action, designing well-targeted interventions, and monitoring progress.

Prevention recommendations for theme 2

P3: Support the creation and adoption of a data platform for the sector.

Next 12 months	By 2027-2028	By 2030
 a) Undertake engagement with relevant stakeholders to identify: functional needs of the data platform; and adoption and implementation support needs. b) Scope project requirements, taking into account findings from P3.a but also FLW and other goals, and existing technical capabilities and gaps. c) Identify options to support platform creation (e.g. direct commissioning, public private partnerships, support for adapting existing private sector solutions, etc.). 	 d) Pilot data platform. e) Design and implement a monitoring and evaluation plan for the platform. f) Design mechanisms to incorporate relevant advances in digital technology as appropriate. 	g) Fully roll out platform to all businesses in the New Zealand food supply chain.

Considerations

P3 is consistent with a <u>2024 report</u> from the Parliamentary Commissioner for the Environment on resource use and waste generation, which recommends "establish[ing] a formal set of material flow accounts for New Zealand".

P3 could be a workstream across central Government, to include Stats NZ, MBIE, and the Ministry for Primary Industries (MPI) and others. <u>The EU's data</u> <u>platform programme</u> may provide a model to follow. Locally, initiatives like the Trust Alliance New Zealand's <u>Digital Farm Wallet</u> or Stats NZ's accounting system that includes '<u>satellite accounts</u>', could be scaled up or integrated into the platform.

Theme 3: Prevent food loss and waste at source

Preventing food loss and waste at the source has scope to deliver the greatest environmental, social, and economic benefits throughout the food system, and everyone has a role to play. A high degree of connectivity means that New Zealanders can contribute to food loss and waste prevention not just at their stage of the food supply chain, but throughout the system.

Prevention recommendations for theme 3

P4: Encourage novel and emergent models of food purchase by consumers.

Next 12 months	By 2027-2028	By 2030
 a) Undertake scoping of models for consumer purchase outside of major supermarkets (e.g. refilleries, consumer supported agriculture, and farmers' markets) that for example can enable purchase of only desired quantities. b) Undertake or commission evaluation and/or evidence synthesis on the effects of these models on FLW. 	 c) Receive and report findings from <u>P4.b</u>. d) Informed by <u>P4.c</u>, consider mechanisms to promote a diversity of food purchase models. 	
Considerations		

This work could be led by the Commerce Commission in collaboration with MBIE as it would fit within their remit for increasing competition in the grocery sector.

P5: Identify mechanisms to avoid food loss and waste caused by extreme weather events.

 a) Commission research to understand the food loss implications of predicted increases in extreme weather events. b) Use the results of P5 a to identify and develop interpretions 	Next 12 months	By 2027-2028	By 2030
to prevent waste during extreme weather events.	 a) Commission research to understand the food loss implications of predicted increases in extreme weather events. b) Use the results of <u>P5.a</u> to identify and develop interventions to prevent waste during extreme weather events. 	 c) Implement actions identified in <u>P5.b</u>. 	

Considerations

Extreme weather and other disruptions bring into play other tiers of the hierarchy as well. See our report <u>Food Rescue in 2022: Where to from here?</u> that includes Foodbank New Zealand, and this report.

P6: Update specification practices that lead to edible food being discarded.

Next 12 months	By 2027-2028	Ву 2030
 a) Explore best practice for consumer communications to make blemished products more acceptable to consumers. b) Support stakeholder groups working in this space to evaluate FLW outcomes. c) Explore mechanisms to incentivise relevant stakeholders to reform specifications practices and undertake a review of best practice for specifications. 	 d) Support businesses to undertake consumer communications. e) Implement relevant mechanisms suggested by <u>P6.c</u>. 	

Considerations

<u>P6.a</u> would be part of <u>P9.b</u>.

<u>P6.d</u> and <u>P6.e</u> could be piloted on selected businesses or sectors before being rolled out in full.

Specifications cover appearance, taste and texture, and food safety. Taste and texture and food safety criteria are relatively fixed while appearance (size, shape, and colour) criteria are more dynamic. This recommendation is focused on the more dynamic criteria.

This recommendation will interact with recommendation **P8**, which promotes technological advances leading to more produce being 'in spec' and recommendation **P3**, which promotes all supply chain actors having access to demand forecast information.

P6 is in line with recommendations made in EFWA's <u>Horticulture SAP</u>. Consider replicating activities recommended in 'Action P3' of the aforementioned report.

The application of quality standards in export industries (e.g. kiwifruit), which protects the economic wellbeing of producers may have lessons for domestic markets.

P7: Continue to explore improvements to current date labelling.

Next 12 months	By 2027-2028	By 2030
 a) Commission or undertake work that explores: alternatives to current requirements, including standardisation of label formatting; 	 b) Support retailers to pilot, evaluate, and refine potential new labelling approaches, ensuring evaluation includes effect on FLW reduction. c) Informed by P7.a and P7.b, phase in new 	
 the likely implications of different alternatives for waste; 	education and/or labelling approach.	
iii. the likely implications of different alternatives for food safety;		
 iv. stakeholder views on uses of date labels; and 		
 v. evaluate the evidence base on the efficacy of consumer education programme around date labelling. 		

Considerations

There would need to be coordination with Australia if Food Standards Australia New Zealand leads this work.

Successful examples of date label reform are detailed in table 8.

A comprehensive review of international best practice for date label reform, building on <u>table 8</u> would be a good first step. Note that EFWA have a <u>current project</u> reviewing date labelling and food storage advice. Industry as well as consumer perspectives would need to be considered.

<u>P7.b</u> could involve supporting an interested national retailer to undertake a pilot.

Label reform could include messaging around date labelling. The 'Look, smell, taste, don't waste' campaign in the UK may be a good example, and Aotearoa could build on New Zealand Food Safety's 2023 campaign 'Check it, sniff it, taste it, don't waste it'.

P7 would be part of **P9.b**.

P8: Pursue opportunities to reduce food loss and waste through research and innovation.

 a) Prioritise funding for FLW prevention under: MfE's Waste Minimisation Fund; and/or Government research schemes. b) Support research for technology that can be used on farm to prevent FLW, including implementation research. c) Continue to support exploration of opportunities for agri-tech addressing FLW as an export product. d) Explore mechanisms to support adoption of innovation by small and medium enterprises (SMEs) to prevent FLW. e) Ensure that public investment in research and innovation in FLW. is evaluated to realise 	Next 12 months	By 2027-2028	By 2030
	 a) Prioritise funding for FLW prevention under: MfE's Waste Minimisation Fund; and/or Government research schemes. b) Support research for technology that can be used on farm to prevent FLW, including implementation research. c) Continue to support exploration of opportunities for agri-tech addressing FLW as an export product. d) Explore mechanisms to support adoption of innovation by small and medium enterprises (SMEs) to prevent FLW. e) Ensure that public investment in research and innovation in FLW is evaluated, to realise 	 f) Implement mechanisms to support adoption of innovation to prevent FLW by SMEs. g) Develop a strategy covering prioritisation of relevant research, mechanisms to support research and adoption of innovation, and ongoing evaluation of performance in this space. 	h) Review and refine research strategy to meet needs post-2030.

Considerations

<u>P8.a</u> should align with SAPs in **P1**, to ensure coordinated funding.

<u>P8.b-e</u> could be achieved through mechanisms such as the MPI-administered Sustainable Food and Fibre (SFF) Futures Fund.

Note that opportunities for research and innovation to reduce FLW and its related harms go beyond prevention of FLW.

P8.c would allow innovation and technology development that is unlikely to be sufficiently profitable at the New Zealand scale. There is a trade-off between the benefits of the technology to New Zealand producers and the benefits of exporting the technology to its developers on one hand, and the risk of losing competitive advantage by making our innovations available internationally on the other.

P8.b and **P8.d** will include innovations around digitalising operations to enable better inventory management, demand forecasting etc., as well as innovations related to food (e.g. breeding and shelf life) and the way we interact with food (e.g. robot harvesters and smart packaging).

<u>P8.e</u> and <u>**P8.f**</u> may align with <u>**P3.c**</u>.

Pilot studies of innovative tools could be a useful way to evaluate effectiveness before rolling out.

P9: Support evidence-based consumer communications campaigns.

Next 12 months	By 2027-2028	Ву 2030
 a) Scope existing work on consumer communications on FLW to identify: mechanisms to ensure their continuity and; gaps where additional work is needed. b) Identify new and planned activities that would benefit from consumer communications campaigns on FLW. c) Incentivise robust evaluation of activities in this space. 	d) Use evaluation findings to prioritise ongoing support.	

Considerations

P9.a should capture both national third sector campaigns like '<u>Love Food Hate Waste</u>' as well as actions taken in the private sector, for example, messaging in retail settings.

P9.a.ii should consider not only specific topics or issues, but also the geography of the campaign (local versus national) and the demographics targeted.

Activities captured under <u>P9.b</u> would include <u>P6</u> (date labelling) and <u>P7</u> (quality specifications).

To achieve <u>P9.c</u>, funding could be contingent on having robust evaluation mechanisms embedded.

Theme 4: Save good food for people

Good food is not a waste stream to be managed – it is a resource for nourishing people. Surplus food, imperfect but nutritious produce, and edible by-products are examples of food, not food waste. Resources, systems, and enabling conditions that promote food rescue and upcycling are crucial to ensuring edible food is never treated as waste. If nourishing people is not practical, using the food as feed resources for animals is the next best alternative.

Prevention recommendations for theme 4

There are no prevention recommendations under this theme. See our reports <u>Food rescue in 2022: Where to from here?</u> and <u>Beyond the bin: Capturing</u> <u>value from food loss and waste</u> for recommendations under theme 4.

Theme 5: Capture value from unavoidable food waste

There will always be some waste in our food system, which must be managed to capture value in alignment with circular economy thinking and the food recovery hierarchy. Diversion to animal feed and investment in material, nutrient, and energy recovery from food waste will ensure there are decent end-of-life options for unavoidable food waste. Landfilling food waste has no place in our waste management future.

Prevention recommendations for theme 5

There are no prevention recommendations under this theme. See our reports <u>Food rescue in 2022: Where to from here?</u> and <u>Beyond the bin: Capturing</u> value from food loss and waste for recommendations under theme 5.





1. Preventing food loss and waste – the top of the food recovery hierarchy

The food recovery hierarchy (see <u>figure 1</u>) is a framework for thinking about food loss and waste (FLW). We discuss the hierarchy – and the need for nuance in its application – in <u>Beyond the bin:</u> <u>Capturing value from food loss and food waste.</u>

Our previous reports <u>Food rescue in 2022: Where to from here?</u> and <u>Beyond the bin: Capturing value</u> <u>from food loss and food waste</u> describe many ways to solve some of the problems created by FLW at different levels of the hierarchy. This report focuses on the top level of the hierarchy, prevention. Preventing FLW across all parts of the supply chain is a more effective way to save money and mitigate climate change than simply managing wasted food at the end of its life, because this prevents the unnecessary financial and environmental costs that FLW incurs along the food supply chain, as well as the financial and environmental costs of its recovery or disposal.



Figure 1: Food recovery hierarchy. Adapted from Teigiserova et al. and Moshtaghian et al.^{1,2}

Preventing FLW in the New Zealand context differs from many other places because we export a very large percentage of the food we produce (see <u>section 1.3</u>), estimated at 80%-90%,³ and even higher for major sectors such as dairy at 95%.⁴ In less export oriented food systems, an important component of prevention will be minimising overproduction, but that is not always financially viable for Aotearoa, given the importance of food export for our economy and the lack of visibility of the fate of our produce in the export markets.

Preventing FLW in the New Zealand context differs from many other places because we export a very large percentage of the food we produce...

Additionally, because only part of the supply chain for exported food is domestic, prevention of food loss upstream in the supply chain (see <u>section 1.2</u>) plays a greater role than in some other countries. The distance between producers and consumers also introduces uncertainties of supply and demand

and economic drivers of FLW not present in countries where most food is produced for local consumption. Prevention is challenging because it involves considering less tangible outcomes such as using fewer resources, rather than reducing waste once food has been manufactured.

1.1 Definition and scope

A precise definition of FLW prevention is more complicated than it appears at first glance.^{5,6} In general, prevention means that surplus, lost or wasted food is not created. For this report, we sidestep definitional difficulties by referring to our hierarchy: when we talk about prevention, we mean any efforts to stop food going to lower levels of the hierarchy. For further clarity, we borrow ReFED's framing:

"Prevention efforts focus on interventions at the root causes of food waste—they locate and address inefficiencies in the food system and food related practices before excess food is produced, transported to places where it cannot be utilized, or discarded rather than eaten."⁷

We explore the reasons food is lost and wasted at each stage of the supply chain, while noting that in some cases, FLW is unavoidable. As we discuss more in <u>box 1</u>, the reason for FLW is not necessarily the same as the direct cause. There are many potential interventions at different points in the supply chain, but understanding the reasons for FLW will allow us to identify those interventions with the greatest likelihood of success. No step in the supply chain can operate in isolation, so we take a whole of system view.

As we discuss in <u>section 1.3</u>, there is a global component to our food supply chains, as we both export and import food. For this report, we focus only on FLW that can be prevented in Aotearoa. This means consumer behaviour of export customers is out of scope, although some interventions here – for example, breeding fruit for longer shelf life – will likely have downstream effects offshore. Similarly, FLW upstream, before it reaches New Zealand, is out of scope for this report, although local importers and downstream actors could in theory include requirements around meeting FLW standards in their terms of trade.

1.2 The food supply chain

In modern Aotearoa, even those of us who hunt, fish, raise our own animals, or grow our own fruit or vegetables mostly rely on the commercial food system for the majority of our diets. Many people and businesses are involved in getting food from the land or water where it originates to the places where we eat it, often modifying or transforming it along the way. This pathway, from where food – or the raw ingredients that will become food – originates to the places where we eat, forms the supply chain. More formally, researchers have defined the food supply chain as "the movement of products along the value-added chain ... that aim at realising better value for the customer alongside cost minimisation."⁸

The specific pathways differ for different kinds of food, but <u>figure 2</u> shows an illustration of the supply chain that can be adapted and applied for most food products. The first stage of the supply chain is production. Production usually involves growing or raising plants or animals in a somewhat controlled environment like a farm, but would also include wild fishing and hunting, and foraging.³ The next stage of the supply chain is processing and manufacturing. There is considerable variation in how much transformation food undergoes in processing. Some products, like fruit that will be eaten whole, need only to be sorted and packaged, while crops like wheat need to be milled to

become flour and then baked to become bread. Food that has been processed and/or manufactured moves along the supply chain towards consumer-facing businesses, including retailers, restaurants, and institutions that include food service, like rest homes or prisons. Depending on the product, there may be an intermediary between the producer/manufacturer and the retailer. Finally, consumers purchase food, either to eat immediately, as in a restaurant, or to take home and consume.



Figure 2: A simplified depiction of the food supply chain. Note that we have modified the figure since we introduced it in *Food waste: A global and local problem* to group food service with retail rather than with household.

In reality, the food supply chain is more complex than <u>figure 2</u> suggests. Firstly, the simplified model of the supply chain omits some actors and stages, including transport, storage, wholesale, and distribution. Secondly, the simplified version implies each actor interfaces only with other parties directly upstream or downstream, and that food products must 'touch' each stage. This is often not the case. Large supermarkets, for example, source their goods directly from producers, from processors and manufacturers, and from wholesalers and distributors. Wholesalers and distributors, in turn, can connect producers and processors or manufacturers, processors or manufacturers and retailers, or even two or more processors or manufacturers. Finally, the food supply chain features considerable vertical integration (see section 5.1). Figure 3 depicts the supply chain in a way that is closer to reality, although it is unlikely to capture every node and path through which food passes on its way to the consumer.



Figure 3: A detailed view of the food supply chain. Arrows indicate the movement of food along the chain. Darker colours indicate upstream and lighter colours indicate downstream.

This report is structured around the simplified model of the food supply chain. Section 3 looks at food loss during production; section 4 focuses on food loss in processing and manufacturing; section 5 examines food waste in retail and food service as well as wholesale; and section 6 considers food waste in households.

17

We precede these stage-specific examinations by taking a systems view in <u>section 2</u>. A systems view gives us a better understanding of the causes of and possible solutions to FLW.⁹⁻¹¹ This is because such a view shows where there are interdependencies, synergies, and trade-offs and illuminates 'hidden' drivers of FLW or barriers to change.^{11,12} <u>Section 2</u> also considers issues that affect FLW in all parts of the supply chain like transport and logistics.

In each of the following five sections, we describe what we know about FLW at each stage of the food supply chain and identify ways to prevent FLW on the basis of this information. What we know about FLW is uneven. We have drawn on many outstanding pieces of research from Aotearoa and abroad that give us deep insight into specific issues. Yet there are important gaps: we are unable to

estimate with any confidence how much FLW is occurring at various points in the supply chain in New Zealand. A forthcoming baseline measurement, commissioned by the Ministry for the Environment (MfE), will provide information on FLW around Aotearoa; however, discussions with the team producing the report at the University of Otago (UoO) suggest that some gaps in data have been identified.

...we are unable to estimate with any confidence how much FLW is occurring at various points in the supply chain in New Zealand.

There is significant academic and grey literature on FLW in the supply chain, which we review throughout the report. At a high-level, in common with other supply chains, the food supply chain faces challenges in coordination, logistics, and matching supply to demand.⁸ In the food supply chain, these challenges are exacerbated by perishability and food safety considerations.⁸

1.3 Food supply chains are both global and local

In Aotearoa, our food supply chains are globally connected. There is some food that is grown here and then exported (for example, premium grade kiwifruit and apples); food that is produced elsewhere and then imported (for example, processed products like some chocolate); food in which the entire supply chain is in New Zealand (for example, fresh lettuce); and food which is exported as raw ingredients and imported processed (for example, some brands of oat milk^{*}). <u>Case study 1</u> uses tomatoes to demonstrate the complexity of importing and exporting pathways for our food.

Q Case study 1: Tomatoes

Tomatoes are one of the most popular forms of produce in New Zealand.¹³ Around the world, tomatoes are widely eaten in many forms including fresh, processed into pastes, preserves, purees, soup, and sauces, as an ingredient in other processed foods (for example, curries, frozen pizzas), and cooked in dishes prepared in food service or by consumers. The tomato industry is global and interconnected, many stages and stakeholders are involved to deliver tomatoes and tomato-based products around the world. Tomato production is one of the most advanced and globalised horticultural industries. China, India, Turkey, and the US (California) are the largest tomato producers, and many of these and other countries grow tomatoes in open fields. However, large-scale greenhouse growing is gaining in popularity and occurs globally, particularly in Europe, for intensified tomato production.¹⁴

As of 2023, Tomatoes New Zealand reported 140 registered growers of fresh tomatoes.¹⁵ Growers choose varieties based on factors such as appearance, colour, texture, and flavour, as well as intended use and resistance to pests. The most popular varieties are vine, cherry, plum, and Roma (see <u>figure 4</u>).¹⁶ Tomatoes are typically grown year round in greenhouses with soil-

^{*} Although a factory is being built in New Zealand so that processing will eventually be domestic.

less hydroponic or semi-hydroponic systems for total environment control, increased yield, reduction in disease, and consistent quality. Greenhouse production is primarily around Auckland, the central North Island, Nelson, and Christchurch, with some open field production in Northland.¹⁷ Locally based greenhouse crops are also a key part of food resilience to climate change and extreme weather events.¹⁸



Figure 4: Tomato varieties. Image credit: Eric Rautenbach (via iStock).

Tomatoes are mainly grown for the domestic market in Aotearoa, but around 10% are exported, predominantly to countries in Asia and the Pacific, with 41% of exports sent to Japan in 2023.^{15,19} Fresh tomatoes currently can only be imported from Australia and must be irradiated to reduce the likelihood of fruit fly eggs or larvae – as well as be labelled as irradiated for consumer information. In 2023, 530 tonnes of fresh tomatoes were imported, compared to the 681 tonnes exported the same year.¹⁵ Imports have been significantly lower than prior to 2012 (approximately 1,181 tonnes in 2011 and 3,085 tonnes in 2009) since the use of dimethoate (a treatment for fruit flies) was banned.¹⁵ Exports have also decreased over the recent years (approximately 3320 tonnes exported in 2019) due to a pepino mosaic virus (known as PepMV) outbreak in 2021, as well as disruptions to sea freight, with growers not being able to export to the Australian market, resulting in an oversupply in the domestic market and reduced prices.¹⁹

Harvesting is done manually for fresh tomatoes grown in greenhouses, but in open field growing, typically for process tomatoes, harvesting is aided by machines.^{14,20} Sorting and grading also takes place to remove damaged or diseased specimens and sort for size, colour and ripeness, for consistency and quality. Tomatoes are then packaged in crates or cartons and transported to processors, wholesalers, or retailers in temperature controlled trucks. The appropriate temperature depends on whether the tomatoes need to be ripened or stored.²¹



Figure 5: Canned tomatoes. Image credit: Brent Hofacker (via Adobe Stock).

'Process' tomatoes in New Zealand, which are used for canning (see <u>figure 5</u>) and tomato paste, are grown mainly in the Hawkes Bay and Gisborne.¹⁷ Processing involves washing, blanching, peeling and cooking or combining tomatoes with other ingredients to turn whole tomatoes into a variety of products.²² New Zealand imports processed tomatoes from Italy (47%), China (19%),

the US (19%) and Australia (6%) and exports processed tomatoes to Australia (50%), Japan (34%) and Fiji (7%).²³

During processing, large quantities of by-product consisting of peels, seeds and pulp are generated, known as tomato pomace (see <u>figure 6</u>). Up to 30% of the tomato may be discarded and is typically sent to animal feed, compost, or landfill.²⁴ However, efforts at higher levels of the hierarchy to reduce FLW including upcycling and valorisation through the creation of new products or biofuels is taking place.²⁴ Tomato pomace contains fibres, oils, lycopene, and proteins and has been used in powder form as a nutritional and antioxidant additive to foods such as baked goods, meat products, dairy products, and oil products.^{25,26}



Figure 6: Tomato pomace. Image credit: Premium pet foods.²⁷

Imperfect, misshapen, or damaged tomatoes are also typically lost or wasted especially if they are being sold fresh. There are a handful of companies such as Perfectly Imperfect²⁸ and Wonky Box²⁹ which sell vegetable boxes to customers that include some of these misshapen tomatoes (see <u>figure 7</u>).

Unsold fresh tomatoes in supermarkets may be donated to local food rescue charities and food banks or, if not fit for human consumption, be used for animal feed or compost.³⁰ The UK supermarket, Waitrose, began selling packs of mixed tomatoes that naturally fell off the vine or are misshapen as early as 2014 in an effort to reduce FLW.³¹ Retailers can also provide information around handling, recipes, storage, and origin stories for consumer education. The average household likely has several fresh tomatoes or processed tomato-based products in their fridge or pantry and need to be able to make informed decisions around considerations such as ingredients, safety, portions, and quality regarding their food.

In Aotearoa, the upcycled food company Rescued has partnered with tomato producers including Heirloomacy and NZ Hothouse.^{32,33} These relationships allow tomatoes that don't meet product specifications (see <u>section 2.1.2</u>) to be used in various processed forms, including supplying over 11 tonnes of tomato sauce to a catering company.^{32,33}



Figure 7: Imperfect, misshapen or damaged tomatoes may be lost or wasted, though this can be prevented. Image credit: Katsiaryna Voitsik (via iStock).

1.3.1 Food export is a central part of New Zealand's economy

While Aotearoa is both an importer and exporter of food, it is food export that dominates our economy. As a nation, we export enough food to meet the energy requirements of 20 million people.³⁴ Figure 8 shows the value of our food and non-food exports since 1985. At the start of the series, food accounted for about half of the value of our exports. This has increased over time, with food accounting for 68% of the value of our exports in 2023, worth about \$48 billion.³⁵



Figure 8: Value of food and non-food exports over time. The values are not adjusted for inflation. Data source: Stats NZ.³⁵ Abbreviation: b = billion.

Although we are diversifying, it seems likely that food export will remain an important pillar of our economy for the medium- to long-term. Because we are a small country, access to global markets provides opportunities not available domestically for our producers, processors, and manufacturers. The Government's expectations of the continuing importance of this sector is further signalled by the Ministry for Primary Industries' (MPI) choice of food export as the topic of their Long-term Insights Briefing (LTIB), focusing on consumer demand to 2050.

New Zealand has long sought to portray itself as environmentally friendly, with the country employing branding like "clean, green" and "100% Pure New Zealand".³⁶ In food exports, we have been proactive in highlighting our low impact production, actively countering narratives about the environmental harms of food miles.³⁷ MPI's LTIB suggests that this will become increasingly important because by 2050, the global consumers whom exporters in Aotearoa will be targeting will be highly aware of sustainability, which will influence their purchasing decisions.³⁸

The primary sector has already started producing strategic work around sustainability.³⁹ The importance of exporting to our economy and the increasing expectations global customers have for sustainability makes it relevant to our continued prosperity that we are successful in our efforts to prevent FLW.
1.4 Reshaping the food system

In <u>Food waste: A global and local problem</u>, we described the social, economic, and environmental consequences of wasted food, including emissions and land and water use. However, the food system affects society in myriad other ways. The food system also provides livelihoods to tens of thousands of employees and small business owners in Aotearoa, employing one in five working people.⁴⁰

Domestically, our food system has challenges. For example, one estimate suggests 21.3% of New Zealand children live in households reporting that food runs out sometimes or often.⁴¹ Industrial agriculture, especially animal products, is associated with climate-changing emissions.^{42,43} Proliferation of some types of processed foods may contribute to environments that promote obesity.⁴⁴

We are cognisant that actions to prevent FLW may have broader impacts – positive or negative – in some of these areas. Preventing FLW is not just valuable in and of itself, but can play a role in improving the wider food system's outcomes. In a submission to the Commerce Commission, the industry body Horticulture New Zealand (HortNZ) noted that in recent years consumers have faced increasing prices at the same time as the amount growers earn for their produce has declined.⁴⁵ It is important to ensure that interventions to prevent FLW work to solve rather than exacerbate existing problems in the wider food system.

We lay out the alternatives and an assessment of the likely effects on FLW and some other food system outcomes, here and in our final report <u>Food loss and waste in Aotearoa New Zealand:</u> <u>Towards a 50% reduction</u>.

2. Looking across the whole supply chain

One approach to preventing FLW in food supply chains has been to measure and reduce waste at the various stages.^{*,46} Thinking of the problem in discrete stages is useful, but masks a lot:

 The root causes of FLW are often complex and interconnected. They are also often in a different part of the supply chain to where the FLW occurs. One example is growers leaving produce unharvested, or tilling it back into the soil, because the cost of sending the produce to the next stage of the supply chain exceeds the expected returns.⁴⁷



complex and ... in a different part of the supply chain to where the FLW occurs.

- Relatedly, the causes of FLW along supply chains are interrelated⁴⁸ and FLW can cascade across the supply chain.⁴⁹ Conversely, actions taken to prevent FLW at one part of the supply chain may have positive impacts elsewhere in the chain.⁵⁰
- The different parts of the supply chain are often bound together and subject to many of the same challenges, including extreme weather, inadequate infrastructure, and labour shortages, as well as the same opportunities as new technologies emerge.
- Thinking in terms of the supply chain, rather than its constituent parts, allows a more holistic picture to form. This is particularly useful for measurement and monitoring, where assembly of fragmented data may tell a different story than would have emerged with system-level data.

Our strong exporting orientation creates challenges and opportunities

Many New Zealand food businesses stand to make greater profits as exporters, because of the relatively small domestic consumer base. This creates economic incentives to produce as much as possible. Prevention strategies focusing on overproduction will therefore have limited effectiveness when much of the domestic production is targeting international markets. Further, the economic incentive to export premium produce creates a supply of produce that does not meet export grade, with its own FLW challenges. That is not to say the economic incentives are inherently at odds with preventing FLW: simulation modelling based on UK FLW data suggests the value of exports increase as FLW is reduced.⁴⁹

FLW can arise because of disruption to export pathways. Disruption can be because of events beyond our borders; as a geographically isolated country we are reliant on global supply chains to transport our exports to their sometimes-distant end markets. In recent years, disruptions to global supply chains have been caused by the COVID-19 pandemic, a ship getting stuck in the Suez Canal, and a shortage of shipping containers. Domestic events can also disrupt export pathways; for example, Cyclone Gabrielle damaged crops so they failed to meet export quality specifications.

FLW can also arise from the uncertainties associated with exporting. In addition to the uncertainties around yield and quality faced by all producers, our exporters face uncertainty from competing

^{*} We have intentionally prioritised a system view of the supply chain first, to emphasise the need for a systems approach to meaningfully prevent FLW. However, those who may be unfamiliar with the food supply chain components can jump ahead to read about each stage in <u>sections 3-6</u> and then return to this section.

export markets – a glut elsewhere in the world will increase global supply, making export markets more competitive and potentially resulting in food failing to be exported.⁵¹

Food that cannot be exported is at risk of becoming FLW if it cannot be absorbed into the domestic food system. However, there are barriers to diverting food intended for export to the domestic market. Economic factors are one consideration: increasing the supply of produce to the domestic market decreases domestic prices at the same time as producers are going without the premium prices their products would have attracted in the international market.⁵²

Our export orientation may create additional mechanisms for preventing FLW. The New Zealand-European Union Free Trade Agreement creates a platform for cooperation between Aotearoa and the EU on sustainable food systems.⁵³ This inclusion is the first of its kind and reflects the EU's focus on sustainability in trade and growth.⁵⁴ Among the topics covered is FLW with a view to achieving Sustainable Development Goal (SDG) 12.3. While the chapter of the agreement covering sustainable food systems has few specific actions, its inclusion is a signal that sustainability goals are being considered alongside economic ones in export markets. Moreover, it provides a mechanism for Aotearoa to benefit from progress made on FLW by EU researchers. As New Zealand pursues future free trade agreements for the benefit of our exporters, it may be worth exploring whether they could be a vehicle for further progress on FLW.

There are potential gaps in our knowledge around FLW in the domestic part of the export supply chain. The academic and grey literature has little to say about whether different strategies are needed to prevent FLW in domestic versus export supply chains. Additionally, we don't have data to form a sense of the scale of the problem in export supply chains.

2.1 The drivers of food loss and waste across the chain

Sox 1: Drivers, causes, and reasons

Much of this report attempts to unpick the 'why' of FLW. By identifying the 'why', we can identify opportunities for change that could reduce FLW. We talk about the 'why' in various ways. Largely we use 'drivers' and 'causes' interchangeably, which are those identified by experts in the research literature and in the sector. Where there are competing views on causality, we make this clear.

In a system as complex as the food supply chain, attributing causality is not straightforward. Sometimes, there might be a simple and direct cause and effect: a household bought more food than they needed and so some food was uneaten and went to waste. Often, however, the causes might be indirect: a supermarket promotion led to a household buying more food than they needed and so some food was uneaten and went to waste. At times, the pathway from the cause to the waste may depend on other factors: a supermarket promotion led to two households buying more food than they needed, and in one household food was uneaten and went to waste while in the other it was cooked and frozen for future use. The number of mediating steps and opportunities for different outcomes from the same starting point increase as the supply chain lengthens. Conventional wisdom suggests that the bulk of FLW, at least in high income countries like New Zealand, comes from consumers discarding food that is still good to eat.⁵⁵ Businesses are generally incentivised to maximise profits,⁵⁶ and minimise waste. However, we know in Australia – which is likely to be fairly comparable to Aotearoa – only 32% of FLW happens in households.⁵⁷ A narrow focus on consumers, neglecting the commercial supply chain, will be ineffective. It also suggests that individual consumers are responsible for the problem, when they have much less influence than

large commercial actors. Nonetheless, collective consumer awareness and purchasing power can drive change.⁵⁸ We discuss consumers in detail in <u>section 6</u>.

The loss of food along the supply chain provides insight into the operations of the food supply chain: it can be more profitable to individual actors to let food exit the chain than to keep it in. For example, growers may find it more profitable not to flood the market with a crop that is



in season, which might lower the price. This is especially apparent in production, where, for example, labour costs to harvest fruit can be up to two thirds of all input costs, making it economically unviable to harvest.⁵¹

In this section, we explore drivers of waste across the supply chain (see <u>box 1</u>). We modify thinking that emerged from a New Zealand case study in the kiwifruit industry, which identified two competing framings of the causes of loss in the sector: the first framing is that loss in the sector was driven by technical or operational failings and was described as 'supply chain driven'; in contrast, the second framing sees loss as the result of market driven quality specifications, where loss is an inevitable corollary of value.⁴⁷ We expand the scope of 'market driven' loss to include the power relationships between parties in the supply chain. We use 'supply chain driven' and 'market driven' as competing framings of the problem, we use them to categorise specific drivers of FLW that emerge from our evidence review and stakeholder engagement. Many of the drivers outlined in this section are explored in more detail in the rest of report in relation to specific supply chain stages.

2.1.1 Supply chain driven

Difficulty forecasting demand

A key driver of FLW across supply chains in high income counties like Aotearoa is overproduction; that is, more food being produced than will ultimately be consumed locally or exported. Some of this overproduction is market driven, which we discuss later in this section. But some overproduction arises from difficulty in accurately anticipating the amount of food needed. The challenge of demand forecasting is widely agreed to be an important driver of FLW across the supply chain.^{59,60,61}

Demand forecasting is challenging for a number of reasons. In the case of primary production, there is a lag time between when decisions are made that will determine production levels – for example, the size of herds, mix of crops, and the number of workers – and when demand is confirmed by contracts with upstream actors.⁴⁵ Further downstream in retail and hospitality, demand fluctuates quickly and often unpredictably, for example, around weather or local events.

Demand forecasting is made more challenging because of data siloing across the supply chain. Although data sharing could enable all supply chain actors to have more accurate demand forecasts, a mix of logistical challenges, such as data not being interoperable, the existence of relatively few tools for sharing data, and concerns around competitive advantage, mean this often does not happen.^{62,63}

Logistics

Significant planning is needed for food to move from production along the supply chain to where it is consumed. Among the considerations are: having suitable transport and storage available at the right times; ensuring all the components of a manufactured product or a food service menu are available at the same time; coordinating the use of machinery used for multiple products; and optimising stock levels. A 2021 literature review identified 49 studies that described logistics-related drivers of FLW across five categories: transport, warehousing, inventory management, packaging, and communication.⁶⁴

Breakdowns in logistics create conditions that promote FLW. Delays in transportation to perishable food use up more of its shelf life before it reaches its final destination, making it more likely to be discarded without being consumed.^{48,59} Failure to maintain proper temperature control can lead to FLW by shortening the shelf life of fresh produce, or by introducing food safety risks that require disposal.⁶⁵

The contribution of suboptimal logistics to FLW in Aotearoa is unclear, with a lack of available data. Globally, Boston Consulting Group (BCG) estimates that of US\$700 billion of FLW that could be prevented each year, about US\$270 billion could come from improvements to supply chain efficiency and infrastructure.⁶⁰ It is unlikely that such a high proportion of preventable FLW in New Zealand is due to logistics as BCG's global estimate will include low income countries. In the US, ReFED estimates that "enhanced product distribution" would avert about 1.7 million tonnes of lost and wasted food.⁶⁶ In Australia,

...of US\$700 billion of FLW that could be prevented each year, about US\$270 billion could come from improvements to supply chain efficiency and infrastructure.

423,000 tonnes of food is lost or wasted in transport and distribution.⁶⁵

The human element

In a study of the Portuguese food supply chain, researchers identified pathways through which human error can drive FLW.⁴⁸ These included: rough handling of products leading to physical damage and making products more vulnerable to microbial damage; storage at the wrong temperature leading to either physical damage or risk of food safety issues; and storage and inventory management decisions that shorten items' shelf life or do not make them available to move along the chain within



their shelf life.⁴⁸ We did not find any systematic data on the role of human error in the New Zealand food supply chain, although human decision making has been identified as a factor in the New Zealand retail sector.⁶⁷ In our stakeholder engagements, human error was often mentioned as a cause of FLW, albeit of varying significance according to different stakeholders.

The human element also comes in through stakeholder attitudes, which can affect inclination to undertake sustainable approaches to FLW.⁶⁸ The human element is particularly relevant to household food waste, which we discuss in <u>section 6</u>.

Data gaps

Data is a key component of FLW prevention.⁵⁰ As we have highlighted in our previous reports, there are gaps in FLW data in New Zealand. Sections 3, 4, 5 and 6 will describe available data and data gaps at each stage of the supply chain. But the supply chain is a complex system rather than a series of independent stages, and data needs to be captured in a way that supports optimisation of the system as a whole. At the supply chain level, data is important to prevent FLW in two ways. Firstly, to

identify 'hotspots'⁶⁹ for targeting, and secondly, so that the effects of actions taken at one part of the supply chain on FLW elsewhere can be detected. The forthcoming baseline measurement does take a systems perspective, which will be a valuable contribution to the data landscape.

However, data gaps will remain. There are parts of the supply chain where data was unavailable for the research. Moreover, while the forthcoming baseline measurement is an important milestone for New Zealand's progress towards SDG 12.3, further advancement is needed in data collection and reporting. Most obviously, the baseline uses secondary data sources, which are difficult to compare because of differences in the data collection methods. Moreover, the baseline is a snapshot of a single point in time. Longitudinal data is needed to monitor progress towards SDG 12.3 and the success of policies aimed at reducing FLW. Additionally, the Food and Agriculture Organization of the UN (FAO) suggests that data should capture not only the magnitude of the FLW and where in the supply chain it occurred, but also the underlying causes and drivers.⁷⁰

Data on topics other than FLW itself also has relevance. For example, understanding the degree of digitalisation in the supply chain would be useful. Data gaps here exist because downstream players are characterised generally as, for example, wholesalers, retailers, transport, which includes these types of businesses from non-food sectors; therefore, it isn't possible to look at food supply chains in isolation. More data is available upstream where agriculture is its own category, although this may also be too coarse a category to be useful.

A centrally commissioned research survey is not the only way to meet our data needs. We discuss how supply chain actors can use and share their own data in <u>section 2.2.2</u>, and barriers to them doing so in <u>section 2.3.5</u>.

2.1.2 Market driven

Terms of trade

The movement of food along the supply chain is governed by contracts between the various parties. The terms of these contracts incentivise overproduction, with purchasers often exerting their greater power so that a supplier's failure to supply enough product would be economically damaging to them. Meanwhile, the costs – both economic and the waste produced – of overproduction also lie with suppliers. We discuss this with regard to the retail sector's ordering practices in <u>section 5.2.2</u>.

One example of trade arrangements that cause FLW is 'take-back' agreements.^{*} These contracts are common internationally for the supply of bread to larger retailers like supermarkets. They are structured so that retailers do not pay for the quantity of bread ordered, only the quantity sold, with the supplier responsible for the costs of disposing of any unsold product.⁷¹ A Swedish case study found that up to 30% of bread ordered between 2011 and 2015 was returned.⁷² Our attempts to understand the role



A Swedish case study found that up to 30% of bread ordered between 2011 and 2015 was returned.

of take-back agreements in New Zealand's bread industry were hampered by confidentiality agreements for some of our key stakeholders. However, a premium Auckland baker describes the process of supplying supermarkets in a way that is consistent with a take-back agreement.⁷³ Additionally, public-facing communications from the industry talk about 'returns'.^{74,75} For one large bread manufacturer, roughly 10% of products are returned.⁷⁴ In total, one estimate puts the annual bread 'returned' to New Zealand manufacturers at nearly 10,000 tonnes.⁷⁵ Demand for the donation of bread in Aotearoa is limited so much of this ends up in landfill.

Quality specifications

Food must meet quality specifications to progress along the supply chain. Specifications exists for

^{*} Also known as 'sale and return' and 'scan based' agreements.

multiple reasons. Some are pragmatic, for example, size and shape requirements to facilitate packing and transport. Others relate to the eating experience, for example, an acceptable range of sugar concentration. Specifications also include cosmetic issues, like size, shape, (cosmetic requirements for size and shape are distinct from size and shape requirements for packing and transport reasons) or colour. For produce especially, cosmetic specifications can appear arbitrary and result in edible food being lost. More details on how such specifications contribute to FLW are described in sections <u>3</u>, <u>4</u>, and <u>5</u>. This issue's appearance in multiple sections of this report reflects its complexity.

The underlying reasons for cosmetic specifications leading to food being lost and wasted are contested, with one argument being that the supply chain is simply responding to consumer demand and the other being that consumer preferences are learned and responsive to both what is available and evidence-based communications campaigns. Box 2 explores how specifications are set in more detail.

Box 2: Who sets cosmetic specifications for food products?

Cosmetic specifications,^{*} which dictate the acceptable appearance of fruits, vegetables, and other food items, are considered an important driver of FLW throughout the supply chain.⁷⁶ The responsibility for these specifications is complex, as various actors along the supply chain influence the criteria that dictate the way our food looks. Retailers are at the forefront of setting cosmetic specifications, enforcing the criteria for the size, shape, and colour of produce to meet the demands of customer preferences⁷⁶ and maximise the appeal of products on their shelves. Consumer expectations and preferences play a significant role in the perpetuation of cosmetic specifications, as their purchasing patterns create demand for cosmetically perfect produce. However, it's important to note that consumer attitudes are learned and in part shaped by what's available to them in retail settings,⁷⁷ creating a feedback loop that reinforces stringent cosmetic specifications. There is some evidence that consumer acceptance and purchasing behaviours based on appearance, size, and weight are malleable. Research on children's perspectives has shown that recurring exposure through seeing, using, and consuming 'ugly' fruit and vegetables leads to greater acceptance,⁷⁸ and that targeted marketing and socialising of 'different' or non-uniform produce improves purchasing frequencies.⁷⁹

Cosmetic specifications can also be used by retailers as a tool to respond to supply and demand, as illustrated by the relaxation of cosmetic quality during times of short supply.⁸⁰ In order to sell their produce, farmers and producers have to reactively adapt to the specifications set by retailers and market demands, and, in doing so, can further entrench cosmetic specifications by selecting and cultivating crop varieties that are more likely to meet retail and consumer expectations.⁷⁶ Additionally, some cosmetic specifications may by influenced by industry bodies, national or international regulatory bodies, and certification agencies, which can relate to quality, safety, and trade specifications. This is particularly relevant in New Zealand's export oriented production sector (see <u>section 1.3</u>). While some regulations like the Overseas Market Access Requirements are in place to ensure food safety and export acceptance, they can indirectly contribute to establishing cosmetic norms in food production, and in turn FLW.

2.2 Potential solutions

We have identified a number of potential solutions to the supply chain level drivers of FLW. These solutions are described in detail in the remainder of this section, and table 1 summarises the drivers each solution addresses, as well as related recommendations. Additional solutions or applications of

^{*} Note, cosmetic specifications do not include standards or specifications that pertain to food safety.

these solutions that affect particular stages of the supply chain are presented in the relevant sections later in this report.

Table 1: Potential solutions to prevent food loss and waste and the drivers they address. Abbreviations: FLW
food loss and waste, SAP = sector action plan.

Solution	Driver(s) addressed	Related recommendation(s)		
Research and innovation.	 Logistics. The human element. Data gaps. Quality specifications. 	P8: Pursue opportunities to reduce FLW through research and innovation.		
Digitalisation of the supply chain.	 Difficulty forecasting demand. Logistics. The human element. Data gaps. 	P3: Support the creation and adoption of a data platform for the sector.		
Shifting practice across the supply chain and at home.	• The human element.	P1: Utilise SAPs to identify intervention opportunities that take a systems view while allowing for the unique contexts of different sectors.		
		P7: Continue to explore alternatives to traditional date labelling.		
		P9: Support evidence-based consumer communications campaigns.		
Finding ways to keep imperfect	Terms of trade.Quality	P4: Encourage novel and emergent models of food purchase by consumers.		
produce in the supply chain.	specifications.	P5: Identify mechanisms to avoid FLW caused by extreme weather events.		
		P6: Update specification practices that lead to edible food being discarded.		
Changes to purchasing agreement structures.	Terms of trade.	P1: Utilise SAPs to identify intervention opportunities that take a systems view while allowing for the unique contexts of different sectors.		
		P2: Evaluate the effect of the Grocery Supply Code on trade term driven FLW.		
Collaboration.	Logistics.Data gaps.Terms of trade.	P1: Utilise SAPs to identify intervention opportunities that take a systems view while allowing for the unique contexts of different sectors.		
		P5: Identify mechanisms to avoid FLW caused by extreme weather events.		

2.2.1 Research and innovation

Research and innovation has significant scope to prevent FLW. Specific applications at different stages of the supply chain are highlighted throughout the report, including <u>case study 8</u>, <u>case study 10</u>, and <u>annex 1</u>. Broadly, relevant research and innovation applications include innovating around the food itself, for example, breeding for specific traits; genetic editing (subject to legislative changes); and product formulation; innovation around the way people interact with food, for example, creation of new tools and packaging technology; and digital technologies (which we discuss in <u>section 2.2.2</u> and which has some overlap with innovations that change our interactions with food).

Harnessing research and innovation relies upon a suitably resourced research sector and close alignment between research agendas and the needs of businesses along the supply chain in Aotearoa. These needs include not only discrete problems that a new technology could solve, but also support with implementation across the supply chain. We explore research and innovation in more detail in <u>section 7</u>.

2.2.2 Digitalisation of the supply chain

A fully digitalised and integrated supply chain opens multiple opportunities to address FLW, including improving demand forecasting and logistics and reducing human error (see <u>section 2.1.1</u>). A detailed list of relevant digitalisation applications appears in <u>table 7</u>; some relevant elements of a fully digitalised supply chain are:

- Devices connected to the Internet of Things (IoT), for example, sensors that can allow monitoring of food in its journey across the supply chain. Sensors already exist to optimise growing and harvesting and time of slaughter decision making,⁸¹ and to monitor conditions like temperature and humidity of food as it travels along the supply chain.⁸² The information recorded in sensors can be reported to a human for action, or can automatically adjust systems such as temperature control using artificial intelligence (AI) or preprogrammed conditions. A fully digitalised and integrated supply chain with these features could prevent FLW through multiple pathways. Sensor technology can both prevent reduction in shelf life by creating an alert to avoid improper handling or storage, and enable accurate assessment of reduced shelf life arising from improper handling and storage to be a part of inventory management decisions.⁸³
- Distributed ledger technologies (see box 5) like blockchain ensure that all parties have access to the same information at the same time, for example, about the location of a given pallet as it moves within a warehouse or from a manufacturer to a retailer.⁸⁴ Blockchain potentially enables better coordination across the supply chain, for example, by ensuring a retailer is ready to receive a delivery at the right time; it also improves traceability.⁸⁵ Walmart in the US has integrated blockchain into its procurement.⁸⁴ Distributed ledger technologies can prevent FLW by, among other things, allowing precise targeting of recalls.⁸⁶
- A digitalised supply chain collects an abundance of data. With appropriate analytics, this data can offer insight that could improve the operation of the supply chain. 'Big data' approaches are well-suited to this kind of challenge, and AI could be useful when data is of different types and stored in different modes across the supply chain. Demand forecasting can also be improved with more sophisticated analytics.
- Data stored on a suitable platform can be shared appropriately allowing relevant stakeholders to access, analyse, and visualise the data without compromising commercially sensitive matters.
- Digitalisation allows greater automation, which can in turn prevent FLW caused by human error.

While various tools enabled by digitalisation have potential in preventing FLW within organisations, at the supply chain level, data sharing across the chain is essential. Shared data would allow upstream players to modify their operations in response to demand forecasts downstream. Analytics drawing on data from across the supply chain can identify hotspots in FLW, for example, specific

ingredients that are overproduced. Gains in efficiency in logistics from digitalisation rely on data flowing across the chain. A platform facilitating this data sharing would enable integrated supply chain management which has been identified by researchers as a key leverage point for prevention.⁸⁷

Internationally, governments at various levels are pursuing the creation of such platforms for sharing and analysing data. The EU has outlined plans for creating common data spaces for a range of sectors and topics, including



agriculture and also for their 'green deal'.⁸⁸ Various projects developing agricultural data spaces have been funded by the Horizon Europe programme, of which New Zealand is now an associate member.⁸⁹⁻⁹² In Aotearoa, Trust Alliance New Zealand, a non-profit industry consortium, is creating a platform for farmers to share data through a digital wallet, which may be a useful model in designing a data platform for sharing data across the supply chain to improve logistics.⁹³ To our knowledge, there is not as yet a platform spanning the whole supply chain.

A platform for sharing data has value for efforts to address FLW at many levels of the hierarchy, not just prevention. Similarly, a data sharing platform could benefit other food system outcomes like nutrition, food security, and local food business viability.

2.2.3 Shifting practice across the supply chain and in households

While not strictly part of the supply chain, consumers are a significant contributor to FLW. We discuss strategies to address household food waste in <u>section 6.2</u>.

Within the supply chain, improving employee practice has the potential to avert FLW. In Australia, Food Innovation Australia (FIAL) estimates that providing training around best practice to employees across the food supply chain could avert more than 265,000 tonnes if undertaken only by industry, or more than 325,000 tonnes if undertaken by industry and policy makers in combination.⁵⁷

There are a variety of models this could take. End Food Waste Australia's (EFWA) sector action plan (SAP) for the bread and bakery sector includes development of a guide for best practice and a toolkit for waste prevention, and the creation of micro-credentials around reducing FLW in the sector.⁶⁹ As described above, digitalisation of the supply chain offers opportunities to reduce or mitigate human error, as does automation of processes.

More generally, organisational culture has a role to play. There is evidence that organisational culture may affect employees' motivation and attitudes,^{94,95} and this in turn is related to employee and organisational performance.⁹⁵ Importantly, the relationship between organisational culture on the one hand, and organisational and employee performance and behavioural outcomes on the other, has been found not only for organisations' core business but for dimensions such as ethical⁹⁶ and environmental⁹⁷ behaviours. Interventions to raise awareness and change practice should be targeted not only at employees but also at leaders within organisations. More specifically, this could involve a combination of bottom-up culture change initiatives and top-down accountabilities for the organisational sustainability profile and leadership teams.

The success of communication led interventions to target organisational culture change may be low, especially for small and medium enterprises (SMEs), if there is not seen to be a benefit to business performance.⁹⁸ It has been suggested that SMEs are more likely to alter their behaviour when the issue is one of compliance,⁹⁸ and there are models where regulation has affected business culture and practice, such as health and safety.⁹⁹ However, such regulation may create unreasonable compliance burdens for SMEs.

2.2.4 Finding ways to keep 'imperfect' produce in the supply chain

One approach to reducing FLW arising from cosmetic specifications would be to simply widen those specifications. This is only likely to be possible for domestic, non-premium products. However, for this to actually prevent FLW, rather than simply moving the location of the FLW further down the supply chain, the produce captured by such a relaxation would have to be purchased by consumers.

The evidence is contradictory on whether this is likely to happen. On the one hand, New Zealand consumers have shown themselves willing to purchase 'imperfect' produce, as demonstrated by the success of Odd Bunch¹⁰⁰ and Wonky Box.²⁹ The success of lines like Odd Bunch suggests that consumers are willing to buy food explicitly branded as imperfect and at risk of going to waste, while the success of adjusting acceptable sizes of potatoes suggests that some status quo settings aren't strongly tied to consumer preferences.⁸⁰ Additionally, there is a long history of marketing effectively employing behavioural science to influence consumer preferences and behaviour.¹⁰¹ On the other hand, during our engagement, we heard that in the case of carrots, market research had found that consumers would accept a bag of carrots with a certain proportion of carrots that were misshapen, and any more than that proportion would be rejected. We also heard that even for frozen peas, a company that changed its specifications to accept yellow peas as well as green received negative feedback from its customers. These are observations based on anecdotes or confidential data, and this question would benefit from a rigorous analysis of systematic data sets, by sector. It is likely that any conclusions will be context dependent.

Consumer preferences are not fixed, and, as discussed in <u>section 6.2.1</u>, we have reasonable evidence about communication strategies that are effective in changing preferences and behaviours.^{*} An evidence-based consumer communications campaign will be crucial to the success of any widening of specifications aimed at reducing FLW.

For any chosen specifications, there would still be produce that falls outside while being safe to eat (see <u>figure 12</u>). Preventing this produce falling down the hierarchy (to food rescue as described in *Food rescue in 2022: Where to from here?* and upcycling, animal feed, and material, nutrient, and energy recovery as described in *Beyond the bin: Capturing value from food loss and waste*) requires secondary markets for these products. Barriers to these secondary markets include fluctuating and not completely predictable supply, and a lack of flexibility in operations to switch between, for example, fresh and processed supply chains. These barriers may be reduced further down the supply chain as the scale generated from combining products from multiple individual producers could reduce uncertainty about supply and make it more economically viable to invest in parallel operations. In <u>section 2.3.2</u> we discuss some of the barriers to trying to keep more imperfect produce in the supply chain.

Digitalisation of the supply chain (see <u>section 2.2.2</u>) could allow for a clearinghouse for produce failing to meet specifications – facilitating dynamic secondary markets and taking the burden off producers who don't have capacity to seek secondary markets for their produce. Woolworths Australia and BCG have developed such a clearinghouse, Refresh Food, a business-to-business platform for farmers to sell their excess produce.¹⁰²

^{*} While there is a broad literature on 'nudge' type methods to change behaviour, the degree to which such interventions are successful or appropriate is contested.

Based on our stakeholder engagement, a useful starting point to make progress in this space would be to review produce that is harvested but consistently becomes surplus, produce that goes unharvested if not purchased, and existing secondary markets. We also note the potential of upcycling as a destination for produce that doesn't meet specifications, which we discuss in more detail in our previous report *Beyond the bin: Capturing value from food loss and waste*.

2.2.5 Changes to purchasing agreement structures

Aotearoa is not alone in having purchasing agreements that fail to disincentivise waste. Take-back agreements for bread, for example, are common in Europe ^{71,72,103} and Australia.¹⁰⁴ Variation at short notice in both quantity and price of produce to be purchased is common in Germany and Italy.¹⁰⁵ The Grocery Supply Code (the Code) (see <u>box 8</u>) introduces a requirement for agreements to be in writing, but does not require that the agreement specify price or quantity.¹⁰⁶

Although contracts that make it harder for upstream actors to avoid waste are common, they are not

inevitable. One approach to addressing unfair terms is through legislation, as we have seen in New Zealand with (the Code). Narrower pieces of regulation can also be used to target specific practices – for example, in South Australia, bread take-back schemes are not permitted and the state produces less than half the amount of bread waste as other Australian states, with no evidence of higher prices.¹⁰⁴ However, take-back arrangements are recommended in some circumstances as a way to reduce FLW that arises from products with short shelf lives spending a lot of time in transit across the supply chain or in distribution centres.¹⁰⁷



schemes are not permitted and the state produces less than half the amount of bread waste as other Australian states, with no evidence of higher prices.

One innovative purchasing agreement is whole crop purchasing, which, as the name suggests, involves purchase of a whole crop rather than only the segment of that crop that meets quality specifications.^{*} Instead of intentionally creating food loss by growing a surplus to ensure they have enough product that is within specifications (and ploughing in crops outside of these specifications), producers are able to 'right-size' their crops. Purchasers are able to pay less per unit or weight of food because of the variable 'quality' while producers maintain their profitability because they can be paid for products that don't meet specifications. In a vertically integrated supply chain – as we have in Aotearoa (see section 5.1.1) – whole crop purchasing is viable because purchasers have the ability to use non-premium products in their own brand processed lines.¹⁰⁷

Contracts could also cover longer time periods. This would provide greater certainty for suppliers, enabling planning and purchasing of large equipment – including FLW reducing technology.

2.2.6 Collaboration

Collaboration across the supply chain is necessary for FLW prevention for several reasons. The causes of FLW are not necessarily in the same part of the supply chain as the FLW occurs. Further, identifying drivers requires all levels of expertise, from policy through to production floor stakeholders. Collaboration can ensure that interventions genuinely prevent FLW, rather than simply pushing it to another part of the supply chain. Collaboration also facilitates complementarity and efficiency in prevention efforts. Despite the potential to reduce FLW – and for benefits in other domains – there are often barriers to collaboration.¹⁰⁸

^{*} Quality specifications aren't inherently unfair trade practices, but they share many features with unfair trade practices that are relevant to FLW, namely, that loss is created to mitigate economic risk, which is unevenly spread between the contracting parties.

Internationally, FLW efforts have typically been industry led by a coalition of food sector companies, such as in Norway, Hungary, Spain, Germany, and the Netherlands, or third party organisations such

as the Waste and Resources Action Programme (WRAP) in the UK, or Too Good To Go, which was started in Denmark but now operates across several EU countries as well as in North America.¹⁰⁹ In the UK, the Courtauld Commitment is a voluntary agreement including businesses across the supply chain.¹¹⁰ Through the Courtauld Commitment, 315 food businesses have committed to the UK's Food Waste Reduction Roadmap, and 19% are described as "actively collaborating" with their supply chain partners to reduce FLW.¹¹¹ EFWA's Food Pact initiative¹¹² is another example of a voluntary agreement for industry organisations to measure and quantify FLW and FLW action, collaborate with each other and researchers, and apply innovative solutions across the supply chain. Government departments are also involved and EFWA's research projects and initiatives are on track to generate \$2.7 billion in industry profitability by 2033.¹¹³ New Zealand's



Internationally, FLW efforts have typically been industry led by a coalition of food sector companies... New Zealand's own Kai Commitment is a version of this, with data being collected and shared (anonymously) across the food industry and collaborative practices across the sector being developed.

own Kai Commitment¹¹⁴ (see <u>case study 2</u>) is a version of this, with data being collected and shared (anonymously) across the food industry and collaborative practices across the sector being developed.

One model of collaboration brings together stakeholders from across the supply chain to establish a unified plan of action for a sector as a whole. EFWA calls the output of this approach sector action plans (SAPs) and the Pacific Coast Collaborative in North America calls their analogous process 'sector summits'. Key strengths of this approach are diverse stakeholders and sector specific understanding of the problems.¹¹⁵ The Australian SAPs follow a process of 'review-plan-do', with opportunities for key stakeholders to come together over multiple phases of the project.¹¹⁶

Actions proposed in SAPs generally fall in one of five categories: policy levers; research, development and extension; implementation; building a community of practice; and monitoring, evaluation, reporting, and improvement. In some domains, members of the SAP, along with other interested stakeholders, can immediately implement a recommended action. For example, one outcome of the bread and bakery SAP⁶⁹ was the development of a toolkit for bakeries offering practical guidance on how to reduce their waste,¹¹⁷ which was launched at a national trade show in 2023. In other domains, the proposed action will take some time to happen: for example, monitoring waste across the dairy supply food chain is the subject of a current research proposal.¹¹⁵ Outcomes of the Pacific Coast Collective's sector summits include consumer messaging in the food service sector, which

improved consumer awareness of waste, and 'whole chain' projects that quantified waste across strawberry and potato supply chains.

To our knowledge, New Zealand has no equivalent to SAPs or sector summits. Kai Commitment may fill some of this gap by providing a community of practice, but there are key differences. Firstly, outputs of SAPs are public facing and involve calls to action both inside and outside the sector, while the work of signatories to the Kai Commitment is generally kept confidential and are actions to be taken only by signatories. Secondly, SAPs bring

...one outcome of the [Australian] bread and bakery SAP was the development of a toolkit for bakeries offering practical guidance on how to reduce their waste...

together stakeholders from specific sectors, for example, the dairy industry, or cross-cutting areas, for example, cold chain stakeholders. This allows identification of the causes and solutions relevant

to specific supply chains. Finally, SAPs can be used to identify drivers of solutions to FLW for the benefit of a wider set of stakeholders rather than just the participants involved, although this may not be the case in practice.

The Kai Commitment, the Australian Food Pact, the Pacific Coast Collaborative, the Courtauld Commitment, and SAPs are examples of formalised voluntary collaboration with the explicit purpose of addressing FLW, taking place outside of business as usual. These are not the only formats in which collaboration can impact FLW. Voluntary collaborative action can also be taken informally and on an ad hoc basis, as was the case when British supermarket Sainsbury's extended lamb season after poor weather delayed lamb growth.¹¹⁸ This averted food loss, but the reasons given for this action by Sainsbury's were related to valuing their relationships with suppliers.

Q Case study 2: Kai Commitment

The Kai Commitment in New Zealand is an initiative of the New Zealand Food Waste Champions 12.3 Trust, aimed at reducing FLW by 2030 in line with SDG 12.3.¹¹⁴ This multi-year programme involves a collaborative platform for businesses to maximise the value of the food they produce, distribute, and sell, by adopting sustainable practices. The programme is structured around a framework that encourages businesses to set targets for reducing FLW, measure their waste in a consistent manner, take action on reduction activities, and collaborate to share best practices and innovate in the field.¹¹⁴

Key participants in Kai Commitment include prominent New Zealand food businesses like Woolworths NZ, Foodstuffs NZ, Goodman Fielder, Fonterra, Silver Fern Farms, AS Wilcox, and George Weston Foods, along with Nestlé. These signatories have committed to targeting, measuring, and acting on FLW reduction. The initiative is supported by various partners and establishment supporters, including funding from MfE. Kai Commitment emphasises the importance of a collaborative approach to tackle the issue of FLW, recognising that this is a shared challenge that benefits from collective action and shared solutions across the supply chain. By bringing together different stakeholders in the food supply chain and facilitating the exchange of knowledge and best practices, Kai Commitment aims to create a more efficient, productive, and sustainable food system in Aotearoa (see <u>figure 9</u>).



Figure 9: The Kai Commitment launch event in Auckland, 2022.

2.2.7 Implications for policy settings

Policy settings can influence the drivers of FLW.¹¹⁹ There are a number of general levers available, from explicitly banning or requiring certain actions, to providing or supporting enabling infrastructure, to incentivising action through tax and other financial mechanisms. We explore the following policy actions that could be taken by government as follows:

- Disincentivising disposal by banning food waste to landfill (<u>section 4.2.3</u> and <u>annex 2</u>), and implementing 'pay as you throw' (see <u>section 6.2.2</u>).
- Reviewing date labelling requirements (see section 4.2.3 and box 9).
- Regulating against unfair trade practices (see box 8).
- Supporting research, innovation, and digitalisation (sections <u>3.2</u>, <u>4.2</u>, <u>5.2</u> and <u>7</u>).
- Supporting school programmes (see section 6.2.1).
- Undertaking consumer communications campaigns (section 6.2.1, and annex 5).

2.3 Sticking points

Most actions to address FLW – and any reductions in FLW – affect different actors in the supply chain differently.⁷⁰ Producers, for example, could profit from reducing on farm loss because they would have either reduced input costs for the same amount of product or more product for the same input costs. On the other hand, retailers' profits would reduce if consumers wasted less food at home and purchased less food. A reduction in oversupply may result in less food being available for food rescue organisations to distribute, requiring more investment in food security and food system resilience.

Although a net benefit to reducing FLW is widely accepted,⁷⁰ not all businesses will immediately benefit from some prevention efforts. Additionally, even interventions that don't obviously create winners and losers may be more accessible or useful to some businesses than others. In the remainder of this section, we explore some of the ways experiences of prevention interventions may differ along the supply chain. In our recommendations these aspects are captured under 'considerations'.

2.3.1 There is an evidence gap

Although the academic and grey literatures suggest many interventions to address the drivers of FLW across the supply chain, few of these have been formally evaluated. Many proposed interventions have not been implemented or have been implemented very recently, providing little opportunity for evaluation; robust quantitative data of the type necessary for evaluation are scarce; and it is challenging to convincingly evaluate the impact of interventions that aim to contribute to system change.¹²⁰ The exception to this may be some behaviour change interventions targeting consumers, although this is still challenging (see <u>section 6.2.1</u>).

Modelling has also been undertaken to estimate how different scenarios would affect FLW.⁴⁹ The Australian National Food Waste Strategy used cost benefit analysis to prioritise potential interventions and identify their potential impact.⁵⁷

Future interventions should be monitored and evaluated as rigorously as possible, to build an evidence base to inform system changes.

2.3.2 Relaxing quality specifications has potential benefits but would create new challenges

Much of the FLW that we attribute to quality specifications occurs because of financial incentives. Relaxed quality specifications and/or new secondary markets would need to change these incentives to be effective in preventing FLW. This may be challenging to achieve. For producers, the input costs are similar regardless of whether the product meets specifications or not. Conversely, consumers are willing to pay a premium for 'high-quality' products or expect to pay less for 'low-quality' products. Another consideration is that bringing more 'imperfect' foods to consumers could work in competition with secondary markets for these products.

Relaxing specifications as a solution to FLW assumes that items that do not meet current specifications, but would meet relaxed specifications, can be sold. Consumers might be persuaded to

purchase such products (see <u>section 2.1.2</u>) but this needs to be confirmed to prevent food previously lost on farms being wasted in supermarkets instead.

Finally, it may be challenging to get some stakeholders onboard. Retailers or industry brands who position their products as premium may be legitimately concerned about the impact of relaxed specifications on brand image. Additionally, operational adjustments like extra sorting or modifications to packaging may be needed, and the price achieved for affected products would have to compensate for expenditure arising from these adjustments.

2.3.3 Voluntary agreements are voluntary

Businesses can only be expected to voluntarily engage in practices that prevent FLW if branding and reputational considerations outweigh the costs. In our stakeholder engagement, we heard that during a SAP process in Australia, one stakeholder withdrew because the recommended action would be detrimental to their business. Voluntary agreements can also be prohibitive to small businesses if the cost of joining is too high.

The limits of voluntary agreements are tested when collecting and monitoring FLW data. There is an increasing appetite internationally for employing mandatory reporting of FLW to enable fairer, more

consistent, and robust reporting. Most recently, a coalition of over 30 major food businesses, including most major UK supermarkets, published an open letter to the UK Department for Environment, Food and Rural Affairs (DEFRA), urging legislation around mandatory reporting, resulting in DEFRA agreeing to reconsider ruling it out.^{121,122} An important consideration when implementing mandatory FLW reporting is to build in recognition that drivers for FLW often lie elsewhere in the supply chain from where FLW occurs. Data collected and communicated from any mandatory reporting should be carefully designed to capture this.

...a coalition of over 30 major food businesses, including most major UK supermarkets, published an open letter to the UK Department for Environment, Food and Rural Affairs (DEFRA), urging legislation around mandatory reporting...

2.3.4 Adoption of innovation can be expensive and slow

Part of the solution to our FLW problems includes innovation. We highlight novel business models and technologies and how big data could make a difference throughout this report. However, not all players in our food supply chain are equally well positioned to take advantage of new technologies.

Most New Zealand businesses, including businesses in the food supply chain, are SMEs.¹²³ SMEs face particular challenges to adopting innovation within the business, as they often lack specialised human resources and costs can be prohibitive.¹²⁴ In addition, in our conversations with experts in the Aotearoa supply chains, we heard that the demands of business as usual may limit resources – human and capital – available to consider the merits of specific innovation. This is consistent with research from Greece, where a small food business reported not having the resources to invest in digitalisation and being unsure of how doing so could benefit their operations.⁶¹ The limited capability of SMEs to digitalise is a barrier to the supply chain as a whole being able to capture the benefits of digitalisation.⁶¹

Even among businesses with the capacity to innovate, there will be variation in how feasible it is to incorporate new technologies into business operations and reducing FLW can take time. For example, using robots instead of humans to harvest fruit – which prevents loss by enabling harvesting with less labour resource and potentially by less damage to the fruit – requires that the vines or trees are

grown in specific configurations, meaning the technology is not useful to established orchards. See <u>section 3.2</u> and <u>table 4</u> for a deeper exploration of barriers to New Zealand producers' adoption of innovation.

Innovation in the form of digitalisation faces specific barriers. Digital product development is shaped by economies of scale, meaning smaller or more diversified businesses are less likely to find digital products meeting their needs.¹²⁵ The expected return on investment from digitalisation is different for upstream versus downstream players in the supply chain.¹²⁵

2.3.5 Sharing data is sometimes perceived as a risk to competitive advantage

Data sharing can be seen as risky due to the potential loss of competitive advantage and bargaining power.¹²⁶ For digitalisation to prevent FLW, all supply chain participants must be confident both in digitalising their own operations and in sharing their data across the supply chain. Data aggregated horizontally and vertically in the supply chain has value as both a public and private good.¹²⁷⁻¹²⁹ This can provide disincentives for data aggregators to share data. The potential benefits to businesses downstream in the value chain may also disincentivise actors upstream from sharing their data.¹²⁷⁻¹²⁹

Much of this evidence is international. A survey of primary producers conducted by AgriTech New Zealand in 2022 provides some insights for Aotearoa.¹³⁰ The majority of producers in Aotearoa are not interested in being early adopters: about 18% say they are "generally one of the first farmers/growers in [their] area to use new digital technology" compared to 82% who say they "prefer to let others try new forms of digital technology before I invest in it". On the wider issue of

seeing value in digitalisation, New Zealand producers are more evenly split: 59% see value in existing digital technology while 41% "don't see much value".¹³⁰ The split is even closer for getting relevant information about the potential of digital technology: 55% say that it is difficult to work out what may or may not be beneficial for [their] business" compared to 45% who say they "know where to get reliable and impartial information about digital technologies". On data sharing, respondents were evenly split between "being happy to share [their] data with others" and "only shar[ing their] on farm data when [they]



are ... required to do so".¹³⁰ Despite half of respondents saying they only share data when required, smaller proportions said "the risks of sharing outweigh the benefits" (24%) and they "worry about who has access to [their] on farm data" (36%).¹³⁰

Based on these results, the barriers to digitalisation on farm identified in the literature are issues in New Zealand, although low trust may be less common. It would be helpful to have similar information on the attitudes of actors elsewhere in the food supply chain towards digitalisation. Understanding attitudes to digitalisation will be necessary to support adoption.

Individual businesses within the supply chain can choose to adopt various digital tools within their organisation, but sharing data across supply chains presents a collective action problem. One way to support digitalisation of the supply chain in the context of potential lack of trust would be for government or a trusted entity to create a suitable platform.

Understanding attitudes to digitalisation will be necessary to support adoption.



Figure 10: A simplified depiction of the food supply chain highlighting the production stage.

Production – where food is grown and harvested – is the first stage of the food supply chain (see figure 10). We use the term 'food loss' when talking about food production in this report, aligning to MfE's definition.⁵ However, we acknowledge that not all food lost on farm is accidental; some loss is a deliberate choice, which is often caused by other stakeholders, policies, and regulations in the food supply chain.

Defining when crops or animals become 'food' on farm is highly subjective due to biological, cultural, and economic factors.¹³¹ This section adheres to the

broad definition outlined in an earlier report of this series (see <u>Food waste: A global and local problem</u>). At this stage of the food supply chain we focus on preventing 'agricultural waste' of food products intended for human consumption. This includes produce that is not harvested due to financial or operational reasons; pre-harvest losses; casualty animals; food rejected on farm because of quality or food safety concerns; and inedible (or not commonly eaten in New Zealand) components of crops and animals not eaten by the intended human



market/consumers including husks, cores, offal, lambs tails, and the like. This broad inclusion is reasonable given the ultimate goal is to prevent future food production contributing to further FLW.

Methods to prevent food loss during production can sometimes be confused with farming methods that increase yield potential. While both prevention of on farm food loss and increases in yield potential can have positive impacts for farmer profitability, they differ in their focus. Food loss prevention in production focuses on both technological and social drivers that prevent crops and animals that are already grown from being lost, whereas yield potential focuses on controlling conditions such as nutrient levels to realise the full genetic potential of the food being grown.¹³¹

Out of scope for this report are crops grown for seed production and food crops intended for animal feed due to their distinct production and processing chains. For example, maize grown for animal feed is usually harvested and turned into silage;^{*} fodder beet may be 'lifted' and stored in sheds or grazed like swedes using the practice of 'break fencing' where the animals eat the vegetables directly from the ground. This is not to say these crops aren't part of the broader FLW landscape, as the land and resources used to produce animal feed crops are arguably better used for food intended for people, but they are omitted here to deliver focused insights on food loss occurring during production of food intended for human consumption.

3.1 Understanding food loss during production

Globally, understanding of on farm food loss is constrained by a lack of data.¹³² This is likely due to the technologically difficult and logistically challenging nature of collecting quantitative data about on

^{*} Heaped into a pit for anaerobic fermentation to be fed out at a later date.

farm food loss.¹³³ The few data points that are available rely on farmer estimates – which may significantly underestimate the scale of the issue according to bureaucratic definitions that are challenging to apply in a farm context. For example, one study found Californian farmers' estimates of crop loss were lower than direct loss measurements by 15% (tomatoes) and 21% (peaches).⁹⁸

Despite the lack of comprehensive data, the production stage of the food supply chain has been estimated as accounting for the second largest volume (after consumers) of FLW globally.¹³⁴ Currently, there is no national FLW data for Aotearoa (although the baseline measurement work is forthcoming¹³⁵). A recent Australian study, where there is a broadly similar export oriented food production system, estimated that this stage is responsible for around 22% of total supply chain waste,^{*} and FLW at production was identified as a previously unrecognised hotspot currently unaddressed by policy.⁵⁷

Compounding the issue of data collection, there is an underlying assumption reported anecdotally by researchers that there is little to no loss happening on farm, particularly in high-income nations like New Zealand, due to access to quality infrastructure and technology, compared with low/middle income nations.¹³¹ This view may be because farmers feel that some waste is inevitable in producing food, and that waste practices (like leaving crops to go unharvested, or leaving food on the ground) are not problems to solve but good stewardship of the land.^{136,137} The assumption by farmers that there is little or no food loss on farm may also be because they



feel that they are not 'purposely' wasting this food as the drivers are outside of their direct control.

The discrete, and sometimes interacting, causes of food loss at the production stage presented in figure 11 include direct losses of crops and stock due to economic factors, crops not meeting cosmetic specifications, weather events, operational issues⁺ like labour shortages, and animal treatments.^{131,133} Many of these factors are already being addressed through incremental improvements in technology that simultaneously improve yields and financial outcomes on farm. For example, upgrading harvesters can address some operational drivers of loss; weather causes can be avoided by selecting species and cultivars that suit the weather patterns of the geographical location of the farm, like growing short-season wheat for summer dry areas; pest management practices, like using nets, can prevent cosmetic damage;¹³⁸ and farm hygiene practices can prevent the spread of diseases such as using sprays to prevent mastitis on cow udders.

^{*} This estimate, from Australia's National Food Waste Strategy Feasibility Study only accounts for FLW going to disposal destinations (for example, composting and on farm disposal).

[†] Operational drivers refers to events outside of weather that disrupt the usual process of growing and harvesting food. For example, labour shortages are an operational driver because less labour leads to food loss through being unable to efficiently harvest the food before it is overripe, equipment failure like power outages in dairy parlours disrupting the cooling systems, which is then rejected by the processor, and infrastructure damage preventing transport from farms to processors.



Figure 11: Drivers such as risk, economic, cosmetic specifications, weather, operations, and animal health direct the decisions that generate food loss on farm. Often these drivers are from other stakeholders in the food supply chain. Adapted from O'Connor et al.¹³¹

Often on farm food loss is caused by other stakeholders in the food supply chain.^{136,139} This is largely due to farmers being price-takers^{*,137} with relatively little power attempting to avoid risks such as negative outcomes for health, reputation, and finances.¹³¹ As described in section 1.3, *Food rescue in 2022: Where to from here?*, current market conditions encourage farmers and growers to practice

overproduction as a buffer against uncertainty in the environment (for example, volatile weather conditions) and markets (for example, consumer demand and price).¹³⁶ However, because of the power dynamics in the food supply chain, farmers are often left to dispose of and accept the risks of dumping food on farm. When robust quantitative data is available on quantities of food loss on farm, it would be useful to explore how different business structures within the food supply chain positively or negatively affect loss outcomes on farm. For example,

...current market conditions encourage farmers and growers to practise overproduction as a buffer against uncertainty in the environment ... and markets.

^{*} Farmers are 'price-takers' because agricultural produce is largely seen as a commodity (interchangeable undifferentiated products). Therefore, they are exposed to the overall supply and demand conditions downstream in the food supply chain and farmers have little power influencing the price they receive at the farmgate.

farmers and growers that are part of cooperative business models tend to have better negotiating power^{140,141} and may avoid some of the onus to dispose of food on farm.

Currently, some of the financial risk of food loss can be abated for some farmers through purchasing insurance when it is accessible or affordable for particular conditions such as hail, fire, and flood,¹⁴² but the tangible disposal of food loss is still the farmers' responsibility. The choice to dump produce is influenced by strong financial incentives due to the high costs of pursuing alternative options and transport requirements from farm to elsewhere.¹³³ As many of New Zealand's food producers focus

on the production of premium produce for high-value export, this also drives waste when some lower grade produce is not sold to protect the value of the premium material and the reputation of the grower/supplier.⁴⁷

As there is no comprehensive national dataset, here we share illustrative examples of food loss occurring during the production stage of the food supply chain for dairy, meat, and horticulture/arable. We discuss what causes food loss, what is currently used that prevents food loss, and future directions for prevention to further improve food loss outcomes during food production in Aotearoa. The choice to dump produce is influenced by strong financial incentives due to the high costs of pursuing alternative options and transport requirements from farm to elsewhere.

3.1.1 New Zealand dairy sector

New Zealand dairy farmers produce 21 billion litres of cow's milk annually – the equivalent of 4,098 litres per capita in New Zealand – of which 95% is exported to 140 countries.¹⁴³ There is an assumption that there is little to no food loss occurring in the production of dairy, but there are very few data points to draw on.¹³⁴

Specific data on sector

Currently, there is no New Zealand industry data that captures food loss on farm – however there is a PhD research thesis in preparation that will provide some data on this.¹⁴⁴ In the meantime, two peer reviewed studies based on dairy farmer estimates have reported between 1%-3% of dairy production is lost on farm in Scotland,¹⁴⁵ and across some European countries.^{57,146} These reported losses were largely attributed to milk being disposed of due to antibiotic residues from animal treatments for mastitis. In New Zealand, most waste milks are disposed of either by being fed to calves when seasonally available, or via the on farm manure management system and eventually irrigated onto paddocks.¹⁴⁴

While the proportion of milk loss is small compared to the milk produced, the volume of loss is still large given the scale of dairy production in Aotearoa. Extrapolating 2% (the midpoint of the two papers referred to above) loss to New Zealand national production (21 billion litres annually), equates to 420 million litres of milk – the equivalent annual production of approximately 98,000 cows^{*,4,143} representing unnecessary greenhouse gas emissions, a considerable waste of resources (land, fertiliser, water), and production of surplus calves (see <u>case study 5</u>).

While the proportion of milk loss [on farm] is small ... the volume of loss is still large given the scale of dairy production in New Zealand.

^{*} Average production /cow/year = 386 kilograms, milk solids ~ 9% of milk. Therefore, an average of 4,289 litres of milk produced /cow/year in New Zealand. Milk loss equivalent = 420,000,000 / 4,289 litres = 97,925 cows. Average methane production per cow per year = 98 kilograms, therefore 97,925 cows * 98 kilograms of methane = 9,596,650 kilograms per year.

Drivers of and issues related to dairy food loss on farm

Overseas markets impose strict requirements for access, setting high-quality standards for milk that processors require farmers to meet, such as low antibiotic residues in milk, and low somatic cell counts.¹⁴⁷ In Aotearoa, the primary treatment for clinical mastitis, an inflammation often caused by bacterial infection in the udder of dairy cows, remains the use of antibiotics due to its effectiveness.¹⁴⁸ In some cases where mastitis isn't caused by bacterial infection, the use of viable alternatives like non-steroidal anti-inflammatory drugs could be further explored.^{149,150}

Current factors that prevent dairy food loss on farm

Competition between dairy processors for suppliers, particularly in geographic areas where the smaller independent companies are, and the leading dairy processor being a cooperative, means farmers have more power in dairy supply chains compared to other food industries. Therefore, farmers aren't required to overproduce or deal with surplus milk because dairy processors sell and

buy other dairy companies' milk prior to processing in response to fluctuating supply and demand. Further, a memorandum of understanding (MOU) between some dairy processors means less food loss when infrastructure is damaged or interrupted (see <u>case study 3</u>).

The high value of milk at the global dairy trade means milk at the farm gate is valuable (\$7.80 per kilogram of milk solids, which is approximately \$7.80 per 11 litres).¹⁵¹ Therefore, farmers have historically aimed for greater efficiency to improve profitability, which subsequently prevents food loss.¹⁵¹



... a memorandum of understanding between some dairy processors means less food loss when infrastructure is damaged or interrupted.

There is also strong political and social pressure for the dairy industry to produce food in a way that enables them to maintain social licence to operate; for example, the implementation of risk management plans.¹⁵² This further incentivises efficiency and non-wasteful mindsets.

Strong relationships between farmers and processors allows liaison and case-by-case assessment as to whether milk would need to be dumped or is acceptable in specific circumstances. This means some food loss on farm can be prevented as some flexibility, facilitated by good communication channels, is built into the system. For example, a cooler breakdown may mean the processor gets the tanker to pick up milk from that farmer last so the milk is cooled by entering a tanker full of already cold milk.¹⁴⁴

${\bf Q}\,$ Case study 3: Memorandum of understanding between dairy processors prevents milk dumping on farm

There is an MOU between the South Island independent dairy companies (Oceania Dairy Limited, Synlait, Westland, Open Country, and Mataura Valley Milk) that outlines an agreement that if there is a reason that these dairy companies can't collect milk from their suppliers, or if they have a breakdown at their processing plant, the other dairy processors will take the milk. This is either through purchasing the milk (for a fair price as outlined in the appendices of the agreement) or through a 'swap' where at a later date the amount of milk taken in is returned. This MOU prevents FLW happening due to infrastructure interruption during adverse events. For example, in 2019, flooding on the West Coast of the South Island caused road closures, which meant that Westland Milk could not collect some of its farmers' milk. Supported by the MOU in place, Oceania Dairy Limited sent tankers from South Canterbury to collect this milk from farmers via the Haast highway for several weeks while roads were repaired preventing farmers from having to dump this milk.¹⁵³

3.1.2 Meat production

In New Zealand meat production is largely ruminant (sheep, cattle, and deer) animals, poultry, and pork. Exports account for 80%-95% of ruminant animals,¹⁵⁴ and less than 5% of poultry.¹⁵⁵ The national populations of the species farmed for meat are presented in <u>table 2</u>. Wild meat that is hunted or culled is technically not produced within agricultural systems; however, it is included here as hunted animals are a source of meat that is edible for humans. See <u>case study 4</u> for some discussion about prevention of wild meat waste.

Specific data on sector

On farm loss of meat refers to casualty animals; that is, animals that die on farm due to natural causes or euthanasia. There is a lack of meat loss data during production because many definitions of FLW do not include animals pre-slaughter/harvest. Consequently, animals that die on farm are often not included during quantification. However, this represents a potentially significant quantity of meat that has been produced but is ultimately lost to the supply chain, and an opportunity for preventing food loss on farm. Clearly farmers have strong incentives to maintain good animal health and diseased animals that go to slaughter can affect the ability of a

...[casualty animals] are often not included during quantification. However, this represents a potentially significant quantity of meat that has been produced but is ultimately lost...

processing plant to export its products, so losses are kept to a minimum as part of normal business practice.

Compounding the issue of data availability are public and consumer perceptions linking animals dying prematurely in our food system to poor animal welfare practices and food safety concerns. This raises concerns that if industries shared the data in the public sphere it could potentially expose these businesses to reputational risks. However, transparent data collection on casualty animals across the industry would provide public reassurance that animal welfare is being well managed. To this end, Beef + Lamb New Zealand have collected this data for sheep and cattle (see <u>table 2</u>).

Anecdotally, the number of animals that die on farm in Aotearoa is expected to be small because sending animals to the abattoir is less affected by time-sensitive decisions such as weather, maturity, and quality attributes than horticultural and arable products.^{*} Instead, the decision to send animals to the abattoir is primarily driven by when it is most profitable to do so.¹³¹ While meat farmers are still mostly price-takers, their ability to be flexible with the timing of animal slaughter gives them some power to choose a 'better' deal. Therefore, there is some competition between meat processing companies for farmers' animals. Some farmers will opt to use different processing companies due to space availability, price, whether they are shareholders, and positive experiences with personnel.¹⁴⁴ This, combined with the relatively high value of (particularly ruminant) animals, means that meat producers very rarely dispose of animals on farm. Surplus animals (see <u>case study 5</u>) are usually sent to the abattoir or processor, which on-sells these processed animals as food, or into secondary markets.^{156,157} New Zealand's high legislative standards for animal welfare and concerns for the public perception of on farm operations – particularly around the visibility of casualty animals and means of disposal – further incentivises prevention of meat losses on farm.^{158,159}

Cattle, sheep, and farmed deer

The number of casualty cattle, sheep, and farmed deer is not recorded consistently. The percentage loss (1.9%-3.6%) provided for cattle, calves, sheep, and lambs in <u>table 2</u> is based on a Beef + Lamb

^{*} With the exception of chickens for meat. Because of breeding selection for greater growth rates/efficiencies and yields, the breeds used in New Zealand (Cobb and Ross) need to be harvested between six and seven weeks of age after this, the chicken's physiology can cause animal health issues.

New Zealand survey of 484 farms in 2017/18.¹⁶⁰ The deer percentage loss is an estimate from a paper published in 2003.¹⁶¹

Aotearoa does have a national animal register called NAIT (National Animal Identification and Tracing) for cattle, sheep, and deer.¹⁶² However, the purpose of NAIT is to track movements of animals between farms and processing facilities to prevent and respond to disease outbreaks, so inconsistent reporting of animal casualties occurs, particularly of slinks (lamb and calf fatalities) that aren't recorded in the system.^{*} There is also the Dairy Industry Good Animal Database (DIGAD)¹⁶³ primarily for tracking breeding worth of dairy animals.

Chicken

In the poultry industry, chickens that die on farm have been anecdotally reported between $1.8 - 2.2\%^{164}$ (see <u>table 2</u>). Low mortality may reflect the high standards of animal welfare, and a lack of some poultry diseases in New Zealand.¹⁶⁵ It is unlikely that this estimate accounts for the one-off events that occur, suggesting greater mortality rates on farms that suffer disease outbreaks, barn failures, or operational issues.¹⁶⁶

Pork

There are approximately 610,000 pigs¹⁶⁷ in Aotearoa; however, access to data of on farm mortality in the pork industry is limited.

Table 2: National data estimates presented for each meat species on population number (based on a 12 month time frame), average liveweight per animal, percentage loss, estimated number of casualty animals, and estimated liveweight of these casualty animals. Abbreviations: kg = kilogram, t = tonnes.

Туре	Livestock quantity (millions)	Liveweight per animal (kg)	Casualties (% loss)	Estimated quantity of casualty animals (thousands)	Estimated liveweight of casualty animals (t)
Calves. ^{+,168,169,160}	5.1	35	2.35	120	4,195
Cattle. ^{170,171,160}	9.6	467	1.9	182	85,181
Chicken (meat). ^{172,164,173}	119	2.5	2	2,380	5,950
Farmed deer. ^{174,161,175}	0.83	85	3.4	28	2,400
Lambs. ^{*,176,160}	20.9	4.5	2.5	523	2,351
Sheep. ^{176,177,178,33}	25.5	60	3.6	918	55,080
Total.	181.5	-	-	4,151	161,180

^{*} There are also incidences of ear tags falling out of livestock ears adding to missing data issues.

⁺ Liveweight is at birth.

Q Case study 4: Preventing 'wild meat' waste

In the second report of this series – *Food rescue in 2022: Where to from here?*, case study 14 discusses recovery of wild deer meat from culled animals on crown land for food donation as venison mince, but also as a meat product for commercial avenues like Burger Fuel's venison burger. Currently, extraction of wild deer from recreational hunting is estimated at 135,000 animals per year,¹⁷⁹ and 20,000 for commercial use,¹⁸⁰ but is arguably an underutilised resource for meat that can be eaten by people. To recover carcasses from conservation land for meat processing at a MPI approved meat processor, a wild animal recovery operations permit must be obtained by operators.¹⁷⁹ Given the challenging topographies that wild animals are often culled from, usually this meat is recovered through the use of helicopters. Longer travel times by helicopter (15-20 minutes) to reach more remote hunting areas increase costs significantly,¹⁷⁹ making recovery of carcasses from some areas unviable.

Challenge: Costs can outweigh the benefits/value

- Shooting a single red deer from a helicopter in Fiordland costs approximately \$90-100, transport and processing adds up to approximately \$300-400.¹⁸¹ Because of this, and fluctuating prices achieved, farmed venison is a cheaper alternative for many customers and food businesses. Other introduced wild meat species such as fallow deer, sika deer, goats, tahr, chamois, and pigs that aren't valued as much or have lighter carcasses may be unviable for retrieval in this way.
- Recovering carcasses takes extra time compared to culling operations, meaning fewer animals are culled on meat recovery trips, which compromises the primary objective of ecosystem protection.
- Associated temperature and timing regulations, and access to a MPI approved meat processor can be prohibitive.
- Browsing pressures caused by high numbers of deer in some areas of conservation land has led to animals that are low weight and are therefore not valuable for carcass retrieval after accounting for costs.¹⁸¹
- Charitable suppliers value wild venison meat donations but cannot afford it from recovery operators.¹⁸¹

Challenge: Food safety risks

- The use of poisons to manage excessive numbers of introduced predator species are a health risk for meat consumption of animals from these areas, therefore carcasses cannot be retrieved from or near these 'buffer zones'. Buffer zones are generally two kilometres from the area in which poison is applied (depending on species being hunted) and can last up to three years.¹⁸² While mitigated by buffer zones, there is still a risk that any wild meat consumed contains harmful toxins from predator control poisons.
- Contamination of wild meat by bacteria such as Salmonella and E. coli are also a risk and could be detrimental to the consumer.¹⁸³

Opportunities:

- Wild deer that are recovered from private land are usually heavier weights and are closer to processing plants, which reduces transport costs. Farmers also may pay hunters to decrease populations of wild deer near their farmland.¹⁸¹
- Charitable suppliers value wild venison meat donations and do receive some from recreational hunters.^{179,181}
- There is strong consumer demand in international markets in Asia and the US for meatbased petfood and treats made from New Zealand wild animals such as deer, rabbits, hare, goats, and wallabies. The limit on becoming a significant export market is currently processing capacity, but new plants are being built. This increase in demand and capacity for processing may increase the scale of operations and decrease costs of retrieval as well

as improve prices achieved for the meat.¹⁸¹ However, this means meat that could be eaten by people is instead used by animals.

- Exploring if risks of negative outcomes can be mitigated while reducing buffer zone areas and duration around poison application would be a useful avenue.
- Financial incentives that enable reduced costs for carcass retrieval might make the process viable.

Unintended consequences

Supporting or encouraging the establishment of business models that rely on 'harvesting' wild meat from introduced species that can be detrimental to native ecosystems may disincentivise efforts towards controlling them to levels where ecological benefits can occur.¹⁸⁴ In some places people will need to be mindful of avoiding reliance on large populations of wild introduced species in their business models where the wider benefit to New Zealand's native and natural ecosystems will take priority.

Q Case study 5: Surplus/bobby calves, a future food loss issue?

Dairy farmers rely on cows giving birth to a calf every year to begin the milking season. Approximately 22%-28%^{156,157} of these calves are kept as replacement heifers; approximately 27%-30%¹⁵⁷ are reared for beef; approximately 5% are euthanised or die on farm;¹⁵⁷ and the remaining 37%-46% (approximately 1.9 million) surplus (bobby) calves are sent to abattoirs annually.^{156,157} While the calves processed at the abattoir in New Zealand do not go to 'waste' and contribute to premium export products,¹⁵⁶ public and consumer concerns over animal welfare and ethical implications of this practice are motivating the industry to review surplus calf production, management, euthanisation, and markets.¹⁸⁵

What has been done generally:

- To address public concerns over bobby calf welfare, animal welfare codes were amended in 2016 outlining transportation and facility requirements of calves younger than 14 days.¹⁸⁶ This has possibly reduced transport fatalities.
- Fonterra announced that changes to their 2023 'terms of supply' means they won't accept milk from farmers euthanising bobby calves on farm they must be sent to an abattoir.¹⁸⁷ This has likely reduced disposal of meat on farms.

Implications of bobby calves for farmers:

- Typically, farmers are only paid a small amount for each bobby calf (~\$30 each). Calves are picked up anywhere between four and 10 days after birth depending on space availability on transport routes and at the abattoir. It is an additional cost for farmers to feed and have shed space for these surplus calves.¹⁴⁴
- Currently, many bobby calves are bred from sires to decrease the risk of birth injuries and deaths for the cows (for example, Jersey breed sires). These genetics aren't suitable for the beef industry due to the slow growth and small finishing weight of these animals.¹⁸⁸

Rearing calves for beef is becoming less profitable with rising costs of inputs, intensive labour requirements, animal health risks, and fluctuating purchase and sale prices. A lack of supply chain integration also limits the calf rearing industry. Some dairy farmers report difficulties finding people to take calves to rear as beef animals and are sending them off as bobbies instead.¹⁴⁴

Solutions being explored:

- Developing a high-value market for New Zealand veal, for example, Pearl Veal.¹⁸⁹
- Using sexed semen that increases the incidence of heifer calves born subsequently decreases the number of cows that farmers need to mate to achieve the desired number of replacement heifer calves, therefore more cows can be mated to beef sires.¹⁹⁰
- Use of high breeding worth beef breed sires to improve the value of surplus calves for the meat industry.¹⁸⁸ For example, calves bred from Wagyu sires can attract a premium price thereby decreasing the number of bobby calves.¹⁹¹
- Alternative uses for some surplus calves, for example, serums for pharmaceutical use are being developed from 12,000 animals grazing at South Pacific Sera farms.¹⁹² Eventually the cattle are processed for beef.

Future solutions

- Extending lactation would reduce the number of surplus calves. Extended lactation of a
 proportion of animals is commonly used in the dairy goat industry, as high milk volumes
 can be maintained over two years without the need to produce offspring. Dairy cows are
 not as suited to this approach in pasture-based systems, but it has been explored in New
 Zealand. Appropriate cow genetics and feeding are required to maintain good levels of
 milk production.¹⁹³
- Cell-based production of milk may offer an ultimate solution as no cow means no surplus calf.¹⁹⁴ However, the nutritional value and environmental footprint across the supply chain needs further investigation and the technology is not yet producing quality product at scale.

Drivers of animal casualties

Newborn lambs are at risk of increased mortality because of selection for increased fecundity. Triplet lambs have lower birth weights and are more vulnerable to seasonally wet and cold conditions in spring.¹⁹⁵ Likewise, during calving, cold and wet conditions have been observed to affect casualty animal incidences.

Disease outbreaks, and injuries incurred on farm that are not treatable or are unresponsive to treatment can lead to euthanisation; for example, *Celostridium*,¹⁹⁶ *Camplybactor* and *Salmonella* infections, fractured bones,¹⁹⁷ broken limbs, or 'black mastitis'.¹⁴⁴ Operational issues or breakdowns can also cause animal death; for example a breakdown of a barn ventilation system can result in mass chicken casualties.¹⁶⁶

Current factors that prevent meat food loss on farm

Improved and audited standards of animal care have probably incrementally decreased casualty animals. There is ongoing industry support to improve stockmanship and animal health outcomes such as PigCare[™],¹⁹⁸ and New Zealand Farm Assurance Programmes.¹⁹⁹ Further, there has been cross-sector collaboration initiatives to improve parasite and disease management, which subsequently reduces animal casualties. These initiatives include Wormwise,²⁰⁰ Ovis Management,²⁰¹ and the Eliminating Facial Eczema Impacts programme.²⁰²

There is ongoing development and uptake of appropriate genetics for reduced susceptibility to disease, and animal health technology. Improvements have been made through adoption of vaccines,¹⁹⁶ improving parasite management practices, and uptake of cow monitoring collars on dairy farms. These technologies decrease the occurrence of diseases as well as enable early detection of health issues preventing on farm casualties.¹⁴⁴

Anecdotally, some animals are butchered on farm and some farmers will use the meat of euthanised animals for their own food¹⁴⁴ as well as a source of dog meat, or for trading through informal networks. Some farmers will also occasionally consume lambs' tails as a 'farm delicacy' (see <u>case</u> <u>study 6</u>).

Finally, improvements in the transport of animals between farm, saleyard, and processor have been implemented through the New Zealand Livestock Transport Assurance programme and may have reduced the incidence of casualty animals during transport and condemnation of animals on arrival at slaughter plants.²⁰⁴

Q Case study 6: Lambs' tails: a food opportunity?

Another source of meat that is lost on farms are lambs' tails, which are routinely removed during tailing (docking) to prevent flystrike – a serious animal health concern. In Aotearoa there are approximately 20.9 million lamb tails 'harvested' annually.¹⁷⁷

Currently lamb tails are not collected for sale as a food in New Zealand; however, some farmers enjoy eating them as a delicacy, cooking lambs' tails on a grill over a fire to burn off wool, and sometimes finished in a steamer. In South Africa, lambs tails (Skaapstertjies) are sold at supermarkets as a barbeque (Braai) snack.²⁰³ Could there be an opportunity for utilising this part of the animal as added income for sheep farmers through selling lambs' tails as food into appropriate markets?

3.1.3 Horticulture and arable

Fruit and vegetable growers in Aotearoa produce over 2.5 million tonnes of fruits and vegetables annually across over 50 different species including kiwifruit, grapes, apples, cherries, avocados, onion, and tomatoes²⁰⁵ (see <u>case study 1</u>). Some crop species including kiwifruit and apples are valuable as export products – kiwifruit annual exports are valued at about \$2.5 billion and apples at \$902 million. Most produce enters domestic retail channels via wholesalers, trading floors, or direct delivery to retailer distribution centres before being on-sold to consumers. New Zealand's arable industry produces 800,000 tonnes of grain annually^{*} including wheat and barley, which are milled into flours for use in both domestic and export products, as well as vegetable oil crops such as canola.²⁰⁶

Packhouses are an integral intermediary between producers, wholesalers, retailers, and exporters.²⁰⁵ These facilities ensure that fruit and vegetables meet food safety standards as well as quality

specifications.²⁰⁷ Generally, horticultural goods that arrive at a packhouse are washed, graded, and packed – in crates, boxes, or their final packaging – before being placed in cool storage.²⁰⁷ Some packhouses manually grade produce, while others may use advanced optic and robotic technologies to streamline the process.^{208,209} Graded out produce that is deemed unmarketable is then used either for further processing into secondary markets, donated to food banks, or disposed of through going to animal feed, composting, or landfill.^{51,210,211}



Specific data on sector

Outside of Aotearoa, horticultural food loss (fruit and vegetables) has been estimated as having the largest volume of waste on farm, estimated to be 23%-26% of total production.^{57,134} While there are

^{*} For this report we exclude seed production (for example, radish seed or carrot seed) and animal feed production.

many instances reported anecdotally of food loss happening on farm pre- and post-harvest, <u>table 3</u> shows some of the few New Zealand specific examples that are available. See <u>annex 1</u> for an overview of technological solutions and management practices for food loss prevention in horticulture and arable crops.

Table 3: New Zealand specific data on horticultural crop pre- and post-harvest losses. Rounded to the nearest whole percentage.

Сгор	Pre-harvest	Post-harvest	Context
	% losses	% losses	
Glasshouse	14	3	From direct measurement of unharvested
tomatoes.			tomatoes and rejected tomatoes at a
			glasshouse. ²¹²
Apples.	10	16	Estimates from a report that surveyed 15 growers
Apricots.	6	12	(accounting for 65% of orchard land) from Central
Cherries.	8	14	Otago in 2021. Pre-harvest losses are farmer
Nectarines	4	12	estimates of unnarvested fruit. Post-narvest
and peaches.			losses include narvest and process losses.

Causes

The few studies focusing on food losses on farm in Aotearoa have identified several causes including weather events – for example, rain events causing splits in fruit;²¹³ lack of labour; and the influence of product specifications on harvesting choices.²¹³

A grower's produce will vary in appearance and quality attributes due to genetic variation within some cultivars, pests, diseases, and weather or soil factors (see <u>figure 12</u>). The value of this varying

produce is determined by product specifications applied by retailers, wholesalers, and packhouses. These specifications outline taste and texture attributes, specific sugar content/brix percentage, and cosmetic specifications such as appearance, size, and weight. Product specifications are necessary in some instances as they ensure that produce is transportable, meets Overseas Market Access Requirements, is safe to eat, and of a reasonable taste and texture.^{*}



packhouses.

Similar to Australia's agribusiness environment,¹³⁷ New Zealand's horticulture sector consists of a concentrated and oversupplied market with few retailers and subsequently low market power of growers. Therefore, growers must accept limiting practices by other stakeholders and their stringent conditions to sell their produce, including specifications and a lack of access to the exact crate determined by the retailer (see <u>case study 7</u>). When these conditions aren't met, growers do not receive payment, and have the additional burden of discarding 'unsellable' product. Sometimes transferring unaccepted produce to other buyers or secondary markets (preventing food loss) is not an option for growers due to exclusivity clauses in some supplier contracts; although this practice is in theory no longer permitted under the Grocery Supply Code (see <u>box 8</u>). Anecdotally, wholesalers and retailers vary specifications with little notice depending on market fluctuations in supply and demand.²¹⁴ This can significantly affect the price growers receive for their produce and the quantities of crops that are wasted. Where prices achieved for the crop equal the cost to harvest, the farmer

^{*} Arable crops are not as sensitive to FLW causes due to cosmetic specifications as these crops usually require further processing before being sold to consumers.

will opt to either partially harvest or leave the crop in the field, and when available and cost efficient (or allowed) use the surplus as animal feed or contact gleaning services (see <u>figure 12</u>).²¹³



Figure 12: Fruit and vegetable crops exist on a spectrum of quality attributes due to genetic, biotic (pests/disease), and abiotic (soil/hail/sunshine) influences. The economic value of the produce is reflected by the standard of specification that the food achieves (as set by wholesalers and retailers); this can vary depending on supply and demand (market driven). When the cost to harvest is equal to or greater than the price received (\$ = 0), this drives decisions that cause on farm food loss.

Further, the portion of a grower's crop that meets specifications varies significantly between 'good' and 'poor' growing seasons causing significant quantities of waste on farms. In Central Otago, 'good' and 'poor' season yields for cherries have been identified as varying as much as 50%.²¹³ The main cause of 'poor' seasons were rain events that cause the cherries to swell and have split skins,

therefore no longer meeting fresh produce specifications and being left unharvested.²¹³ Currently when specifications are not met many types of fresh produce have few viable secondary markets like processing into other food products such as frozen, juice, or puree.

Growers face variable climates, low market power, stringent specifications, and variable prices. Given these uncertainties, growers are driven to produce surplus food (causing more food loss) as a financial risk mitigation strategy.¹³⁷ This locks our food system into the cycle of using valuable natural resources, emitting greenhouse gases, and wasting nutritious food.



${f Q}$ Case study 7: Returnable plastic crates and their impact on FLW

Returnable plastic crates are used in New Zealand for packing produce for transport from farms to wholesalers and retailers. These crates are an alternative to disposable crates and are an effective way to transport produce without generating plastic or cardboard waste.²⁰⁵ However, they can also generate food loss.

New Zealand's main supermarket retailers, Woolworths NZ (formerly known as Countdown) and Foodstuffs North and South Island, require suppliers to use produce crates from three specific companies, based on exclusive supply agreements with the crate companies. The crate companies are: Viscount FCC; Loscam; and CHEP. Foodstuffs NZ will only accept the green crates (see <u>figure 13</u>) and Woolworths NZ will only accept the black crates (see <u>figure 13</u>) whether the producer is supplying the store directly or via a wholesaler. Crate dimensions differ by only a few millimetres yet they are not allowed to be used interchangeably, and no alternatives (for example, cardboard boxes) are permitted.

In the busiest weeks for retail, there is always a shortage of crates from at least one of the crate companies. This problem is worse in the summer at the peak season for harvest for many crops. The 2023-24 Christmas and January period was the worst to date, with Viscount FCC closing their distribution centre for two weeks, creating a backup of crates to process (clean and redistribute to growers). In the meantime, Foodstuffs NZ's suppliers couldn't turn to the black crates as a substitute.

Growers anticipate the problem and stockpile crates before the busy season, ordering more than they need, knowing they'll only get a fraction of what they order from the crate companies. Meanwhile, retailers are not incentivised to return crates from stores to the crate companies for cleaning and redistribution. Ultimately, if a grower can't get the right crate at the right time, there's no alternative way to market. Therefore fresh produce that meets retailer specifications is left with the grower for disposal.²¹⁴





Lift Lock RPC - 23L

Reusable Plastic Crates (RPC) in three sizes specifically designed for the New Zealand market. The unique design of the crates meets a variety of handling demands, Module in design allowing column stacking when erect and have a consistent height when folded.

Size: 600 x 400 x 119 mm Tare weight: 1.36 kgs

Black Foldable RPC 21L - (for supply to Progressive Enterprises)

A foldable plastic crate designed to reduce cost and environmental impact. Crates can be column stacked, cross stacked or a combination of both when erected

Size: 597 x 398 x 110 mm Tare weight: 1.31 kgs

Figure 13: Growers supplying Foodstuffs NZ must use the green crates (left),²¹⁵ growers supplying Woolworths NZ must use the black crates (right).²¹⁶ Image credit: CHEP; permission pending.

Drivers of and issues related to horticulture food loss on farm

Variable factors such as weather events cause produce losses and damage,²¹³ and global and domestic market factors result in variable prices achieved for produce.¹⁹ There is also a lack of access or existence of secondary markets for produce that is not accepted by the packhouse or the processor due to cosmetic damage.⁵¹

Constrained access to labour availability/skill and operational factors affect pre- and post-harvest food loss occurrence, leading to damage during harvest due to rough depositing of fruit into bins. However, this has been somewhat abated with an increase in the cap on Pacific workers from 2023.^{51,217}

The low level of market power – particularly for producers of domestic goods – when selling produce to wholesalers or retailers leads to overproduction to mitigate financial risks.^{137,214} This is compounded by a lack of transparency or forecasting of prices affecting choices around timing of harvests or whether to harvest a crop at all.²¹⁴ Further, terms of supply with retailers and wholesalers sometimes prevent growers opting for secondary markets with their surplus produce.⁵¹

Current factors that prevent horticulture food loss on farm

Food loss is reduced by adoption of practices such as using greenhouses, installing nets, two dimensional (2D) orchards, or 'strip picking' where all the fruit is picked and automated systems at packhouses sort the fruit into their categories. Strip picking prevents on farm food loss as the fruit is now accessible for distribution to secondary markets.²¹³

Uptake of a wide array of technology types can improve yields and subsequently prevent losses. Technologies include mechanisation of harvesting methods like picking platforms,⁵¹ using specific cultivars that are bred for disease tolerance, and using genetically uniform cultivars.

Some growers use gleaning services and subscription boxes (for example, Wonky Box).²⁹ However, there are concerns around gleaners due to possible health and safety risks of having volunteers on farm and subsequent liability for the grower.²¹⁴

3.1.4 Fishing and aquaculture

We have previously produced an evidence synthesis on commercial fishing <u>The future of commercial</u> fishing in <u>Aotearoa New Zealand</u>. A source of preventable loss in commercial fishing is non-target fish being captured in nets. In that report's case study 6.3.4, we describe technology that can prevent this, including specialised nets. Additionally, our previous report in this series, <u>Beyond the bin:</u> <u>Capturing value from food waste</u>, discusses material recovery, which comes into play when prevention is impossible.

3.1.5 What are the gaps in our knowledge?

Across food types, quantities of food lost on farm are unknown because it isn't routinely measured or even identified as being waste. Farmer estimates may underestimate proportions unharvested. Direct quantification is technically difficult and time consuming, but increasingly possible through use of technology. The benefits of such data collection are not just a better understanding of the issue and its scale, but also the identification of opportunities to attract innovative solutions and entrepreneurship in addressing the issue. This issue is covered in detail in <u>The future of commercial fishing in Aotearoa New Zealand</u>.

As well as specific quantitative data, there are gaps in our understanding of the effect of the operations of markets on food loss on farm through factors like:

- Business structures (for example, cooperatives versus corporations).
- Relative market power.
- Competition and competitive behaviours among retailers and processers.

- Mechanisms for sharing risk and responsibility for FLW across the supply chain.
- The role of consumer preferences in setting specifications and the adaptability of these preferences.

Better understanding of these aspects of market operations may suggest policy levers for preventing food loss on farm.

3.2 How can we prevent food loss during production?

It is important to view any recommendations that may impact farmers through the lens of farmers' financial and market power capacity. Current economic climates for many farmers in New Zealand are difficult due to disruptions and labour shortages during COVID-19, and large weather events such as Cyclone Gabrielle. Further, the cost to produce food has increased, putting pressure on farmers' business margins because of increases in input costs like fertiliser and labour, and high mortgage interest rates. Imposing any additional burden on farmers to address FLW issues during production needs to be seen in the context of farmers' relatively weak position to make changes in the overall food system or add value to offset costs or investment into FLW prevention technologies and strategies.¹³⁰ Therefore, the following potential solutions for food loss on farm will often require support from other stakeholders to implement.

3.2.1 Potential solutions

Data collection and sharing: addresses all drivers

Adoption of agreements like an MOU – rather than a non-disclosure agreement – for growers and farmers to share FLW estimates with researchers would significantly benefit efforts to address on farm food loss through enabling this data to support quantitative and causative research and subsequent solution implementation. This could be undertaken through expanding the capacity of currently available platforms such as Trust Alliance New Zealand (see <u>box 3</u>), Integrape, DIGAD, and NAIT. Further, where data is currently not collected, real time data collection and sharing would strengthen the operation of secondary markets.

Box 3: Trust Alliance New Zealand

Trust Alliance New Zealand is a non-profit food and fibre industry consortium of farmers and growers, governed and operated by members,²¹⁸ which aims to use decentralised distributed ledger technologies (DLT) to enable farmers to be in control of their own data and store, protect and share it effectively.²¹⁹ Trust Alliance New Zealand is co-funded by members, and Sustainable Food and Fibre (SFF) Futures from MPI and aims to improve collective profitability.²¹⁸ The consortium is currently exploring the concept of a 'Digital Farm Wallet' to "unlock value from farm data" and be able to easily share and verify information between farmers, industry, and retail while ensuring trust and privacy. Transactions sent across the network need to be verified by other members (farmer owned organisations) allowing farmers and growers to retain control and ownership of their data.⁹³ Conceivably, expanding the capacity of a tool like a Digital Farm Wallet to incorporate FLW data supplied by farmers could be a valuable and accessible opportunity for identifying food loss occurrence and opportunities during production.

Secondary markets: addresses economic and cosmetic specifications

Supporting access and increased capacity of networks that enable secondary markets for foods that are not cost-effective to harvest for their primary market would enable this food to be better accessed by other potential customers, such as processors. This would also benefit efforts at other levels of the hierarchy, like upcyclers, subscription box services, and food banks.

Supporting the development of viable and flourishing secondary markets for all food types, such as rendering facilities for meat by-products, and juicing/spray-drying/pureeing facilities for edible produce that doesn't meet specifications would reduce FLW from the system as a whole.

Technology: addresses operational, animal health, and economic drivers

Supporting development and adoption of accurate real time data collection technologies (such as light detection and ranging (LiDAR), see <u>table 4</u>) for use on farm pre-harvest would enable accurate data collection on the scale of FLW occurrence, but also enable forecasting and food loss prevention decision making through agronomic analysis of crops as they grow.

Encouraging innovation and research could address key drivers of FLW on farm across food types and enable prevention and informed decision making around food loss. To be effective, innovation from research needs to be scalable for export markets so, for example, robotic technology can be produced and sold at scale, decreasing price barriers for adoption by New Zealand growers. Examples of technologies that could address food loss drivers are listed in <u>table 4</u>.

Policy actions and terms of trade: addresses economic, operational, and cosmetic specification drivers Encouraging whole crop purchasing to enable full harvests, therefore facilitating 'out-of-spec' food to be used in secondary markets (because it is already at the packhouse/processor) could make a large impact on reducing FLW.

Removing exclusivity clauses for crops that have a high incidence of food loss occurrence due to financial drivers would allow this food to have a chance to be sold to another market/customer. This is addressed in the Grocery Supply Code and influences the market power of farmers. It may have an affect (positive and negative) on price achieved for some grades of produce.

Some niche solutions

Vertical farming of some crops – grown entirely indoors under artificial light in urban areas – is emerging internationally and locally (see <u>case study 8</u>). Vertical crop growing decreases weather and cosmetic specification causes of food loss during production.²²⁰

Another niche solution gaining momentum in food production is cellular agriculture. Creating food using this method enables more precise control, avoids overproduction, and is currently in development for some foods such as dairy, meat, eggs, some plant products, and raw ingredients.¹⁹⁴

Q Case study 8: Innovative vertical farming techniques at Greengrower

Greengrower is a company that produces lettuce, microgreens and other salad leaves located in Hamilton. All of Greengrower's produce is grown indoors (see <u>figure 14</u>), and each 5,000 square metre farm produces as much as 150 hectares of conventional growing. Seeds are planted in specialised plugs containing growing medium. Newly planted seeds are kept in a 'nursery' for seven to 11 days and then either harvested for microgreens or transferred to the main growing area. The main growing stage requires a further two weeks, and the maximum production time from planting to harvest is four weeks, compared to six weeks for conventional growing.



Figure 14: One stage of lettuce production at the Greengrower indoor farm.

The innovative approach to growing reduces food loss in their own operations, and food waste for their retail and household customers. Because the produce is grown indoors, it is not vulnerable to pests or weather, providing more certainty around yield. The indoor growing conditions as well as the absence of pests also mean the leaves don't need to be washed before packing, preventing loss through damage and lengthening shelf life, potentially reducing retail and consumer waste. One of Greengrower's food service customers reports that certain conventionally grown salad leaves are wasted due to large unpalatable cores; Greengrower's leaves are larger and don't have cores, resulting in greater utilisation by the customer. Table 4: Technology currently available and the FLW drivers it addresses on farm. Each technology is presented with a simple SWOT analysis through the lens of food loss and waste prevention. Abbreviations: AI = artificial intelligence, CRISPR = clustered regularly interspaced short palindromic repeats, FLW = food loss and waste, GMO = genetically modified organism (here it is synonymous with GE = genetically engineered), IP = intellectual property, LED = light emitting diode, LiDAR = light detection and ranging, RGB = red green blue (camera), SWOT = strengths, weaknesses, opportunities, threats, UV = ultraviolet light.

Technology and FLW driver addressed	How it works	SWOT: Strengths (S), Weaknesses (W), Opportunities (O), Threats (T)	What needs to happen to improve FLW on farm:
Wearable health	Wearable	S Simple to implement, accurate data collection, the two main	The technology would need to
monitors.	monitors that	companies in NZ (Sensehub/Allflex and Halter) have effective user	become business as usual and
Driver	(feed intake)	health of animals and lower the chance of casualty animals ²²¹	scaled to suit more farmers.
Animal health.	temperature,	W The cost of subscriptions may make it unviable for some. Currently	
	movement, etc.	collars need to be in range with readers, which is difficult in some	
	and use this data	rural topographies. Animals having collars and trackers on them	
	to alert farmers if	could become an animal welfare issue due to added weight, chafing	
	an animal is sick,	on skin, and use for virtual fencing.	
	enough etc	for industry wide perspectives adopt for other animal types	
		T Adoption by farmers needs to balance costs versus value. Higher	
		resolution detection of subclinical issues may lead to greater	
		antibiotic use and more milk waste on farm.	
Development of plant	Breeders develop	S NZ unique varieties that are tailored to fit our way of producing	The development of cultivar
varieties through	varieties with	food and consumers in export markets. Genetically uniform cultivar	development efficiencies that
genomic selection	attributes to	create uniform produce and therefore more quantity should meet	shorten timelines (and cost) for
and genetic marker	increase	specifications. ²²²	growers. Breeding objectives
methods.	pest/disease	W Long timelines for developing varieties (10+ years for some species)	that focus on reducing FLW (for
	tolerance, shelf	is expensive, requires several sites around the country to capture	example, traits that prevent
Drivers:	life, and adaption	the genotype by environment interaction effects on phenotypes.	fruit fall during wind weather
Cosmetic	to climates.	Sometimes training populations can take years to phenotype due to	events).
specifications;		focused on povel traits (e.g. red kiwifruit) or improving vield	
operational.		potential rather than improving FIW outcomes on farm	
		O Development of unique and valuable IP that can be a valuable	
		export. Building a pool of unique, diverse (but well characterised)	
Technology and FLW	How it works	SWOT: Strengths (S), Weaknesses (W), Opportunities (O), Threats (T)	What needs to happen to
--------------------------	---------------------------	--	------------------------------------
driver addressed			Improve FLW on farm:
		 germplasm provides important resources for adapting economically important species when future issues arise.²²² T Because of biosecurity risks and concerns, it is difficult to import germplasm for some economically important species because of strict regulations and expenses associated with following protocols. However, some steps are being made by the Plant Germplasm Import Council to address this.²²³ 	
Use of GE methods to	Using modern	S GE can be faster than classical breeding. Novel trait development can	Adoption and scaling the use of
develop novel cultivars.	breeding	be attained when needed for resistance or tolerance of new diseases	GE in New Zealand would require
	technologies to	or pests.	assurance from overseas markets
Drivers:	fast track and	W Currently GE is expensive to implement and requires very specialised	that using GE in NZ will not have
Cosmetic	improve the	expertise and facilities, but development of new methods (like	a detrimental impact on NZ
specifications; and	genetics of plants	CRISPR) and protocols are decreasing costs. ²²³	exports. A change in policies that
weather.	to express	O GE is a tool that can address multiple issues (such as yield, nutrient	support research in NZ and
	neonaturas ²²⁴	develop unique and valuable ID and products, and anables future	growers is also peeded before
	phenotypes.	proofing against changing climate strosses	this tochnology can be applied to
		T Currently GE faces regulatory barriers, driven by concerns of its	ELW issues
		notentially negative impact on NZ food exports to overseas markets	
		It is also a politically controversial topic, with concerns about the risks	
		of unintended consequences for example what is the impact of	
		upregulating specific expressions in food plants and their impacts on	
		nutrition and health for consumers?	
UV or LED treatment	Using different	S UV or LED treatment can decrease the requirements for growers to	UV or LED treatment technology
of plants.	spectrums of	use agrichemicals to prevent blemishes on produce from diseases	would need further
	light at different	etc. Using these treatments may enable farmers to better match	development to fit more crop
<u>Driver:</u>	stages of a crop	supply with market demand through controlling rates of crop	types.
Cosmetic	induces plants to	growth. Subsequently, this would reduce market driven waste.	
specifications.	upregulate their	These treatments are available in NZ through businesses such as	
	natural disease	BioLumic. ²²⁶	
	resistance	W Currently these treatments are effectively applied for indoor	
	functions.	growing only, therefore inaccessible for most crops.	

Technology and FLW driver addressed	How it works	SWOT: Strengths (S), Weaknesses (W), Opportunities (O), Threats (T)	What needs to happen to improve FLW on farm:
		 O Combined with robotics, exploring if these technologies can be effectively used in the field may increase uptake by growers. T Other options, like agrichemicals and breeding for tolerance, are currently effectively used and accessible for pest/disease management, therefore uptake of UV or LED treatments may be slow. 	
Optical technologies that facilitate growing and harvesting for example, LiDAR and RGB cameras. <u>Driver:</u> Operational.	Using optical technologies as a real time data collection method to identify when crops are ready for harvest, forecast potential issues, early identification of pest and/or disease.	 S Optical technology enables accurate and fast collection of data with fewer labour requirements than manual quantification methods. W Optical technology cameras can be expensive. Further, this technology generates a lot of data that requires analysis and interpretation. Creating models that make sense of this data for specific crops/cultivars is complex. O Al models may make optical data easier to manage, analyse, and interpret. Costs of optical technologies are decreasing.²²⁷ T Crop specific optical technologies are expensive to develop for a small market such as NZ, therefore development and manufacturing costs makes these technologies expensive to attain and maintain, erode profitability, and are a significant barrier for startups in this space, and growers who would benefit from these technologies. 	A possible solution is through incentives that encourage the development of technologies locally that can also solve issues in bigger markets and can be exported, while enabling local growers access to adopt this technology.
Mechanisation of harvesting. <u>Drivers:</u> Operational; cosmetic specifications; economic; and labour	Mechanisation (different to robotics) that enables efficient use of labour for example, dairy parlours, picking platforms, and e- bins. ²²⁸	 S Mechanisation improves efficiencies of harvesting and milking procedures, reduces the time taken, and can reduce the incidences of fruit damage therefore causing less food loss. Mechanisation on farm is already extensively used in dairy parlours, and some options are developing for orchards.⁵¹ W Initial upfront cost to purchase and implement mechanisation technology can be prohibitive for farmers.¹³⁰ Only some technology is available and sometimes this is crop or production process specific, which would require major shifts in farm operations to incorporate. 	Addressing health and safety concerns and possible liabilities of using mechanisation on farm would improve confidence in utilising picking platforms for harvesting. For seasonal needs, businesses that offer leases for this equipment (such as e-bins) may improve the economic viability for growers and improve adoption.

Technology and FLW driver addressed	How it works	SWOT: Strengths (S), Weaknesses (W), Opportunities (O), Threats (T)	What needs to happen to improve FLW on farm:
		 O Over time, using mechanisation can decrease the cost of labour to harvest, therefore more of the crop is economically viable to harvest and less market driven FLW occurs. T There are some barriers to adoption of mechanised operations, such as picking platforms, due to health and safety concerns for staff and subsequent liability risks for farmers.⁵¹ 	
Robotic technologies. <u>Drivers:</u> Operational; cosmetic specifications; and economic.	Robotics that apply specific inputs and harvests fruit when it is ripe without damage.	 S Using robotics could decrease the amount of fruit that goes unharvested due to labour shortages. NZ has expertise developing these robotics (for example, at AgriTech, Callaghan Innovation, Plant & Food Research, AgResearch etc.). W Developing robotic technology requires large upfront investment, which can limit access and uptake by growers. Adopting robotics into farm systems would likely require significant changes to the operations on farms, like changing tree spacings to suit machine requirements. O Instead of individual farmers buying robots, contracting businesses could service geographic areas (like silage operators). Development of robotic technology products can also be valuable for export markets. T Development of robotics for specific plant species or cultivars for NZ growers' specific needs is limited as the domestic market is too small to support scale up of robotic businesses locally. 	Development of robotic technologies by NZ businesses that can also solve issues in bigger markets (i.e. can be exported) so that start-up robotic companies can scale, possibly making the technology affordable and accessible for local growers.

Technology and FLW driver addressed	How it works	SWOT: Strengths (S), Weaknesses (W), Opportunities (O), Threats (T)	What needs to happen to improve FLW on farm:
Farm data digitisation.	Data centres that	S Farmers are already collecting some types of data for various	Better articulating the value for
	enable collection	regulatory, business, and biosecurity reasons.	farmers to collect and share
Drivers:	and sharing of	W Farmers don't currently collect data on what FLW is happening on	their food loss data through
Operational.	data by farmers.	farm. Many would need support about what to measure, how to	data centres would incentivise
	These services	measure, and how to input food loss data appropriately.	adoption. Further training and
	aid decision	O Collating these data types can help with forecasting and decision	ongoing assistance would be
	making and	making for the farmer that may enable prevention of FLW. Some	needed to support farmer
	enable the	decentralised data platforms already exist in NZ such as Integrape	uptake of data digitisation and
	identification of	and Trust Alliance NZ (see <u>box 3</u>).	sharing. These data centres
	waste quantities	T There could be some concern around data sharing and data use.	could prevent food loss through
	and occurrence.	However, this has been somewhat addressed by Trust Alliance NZ	informing better decision
		where farmers have control over what data is shared and who with.	making on farm, and
		Currently there are no persuasive incentives or value for farmers to	highlighting opportunities for
		collect data on food loss occurrence on farm to be shared. Negative	FLW reduction.
		reputational risks may be a concern in sharing food loss data.	



Figure 15: A simplified depiction of the food supply chain highlighting the processing and manufacturing stage.

In the food supply chain, processing and manufacturing refers to the transformation of raw agricultural produce into a new ingredient, good or product (see figure 15). Food processing is any deliberate change to raw food, such as through cutting or drying, or mixing and combining ingredients.²²⁹ Manufacturing describes the whole process of making a new good or product, including packing, labelling, and storing as well as the equipment and labour involved. The methods used in processing and manufacturing depend on the specific good or product being produced and can depend on several actors. This section will focus on the actors carrying out sorting, processing, and packaging of food that has come off the farm, and larger-scale or commercial settings where processing or manufacturing of food typically occurs in bulk, according to predetermined formulae or recipes to be sold at a later date (i.e. not food service or restaurants – which are discussed in <u>section</u> <u>5</u>).

Food processing and manufacturing not only provide economic value through the creation of new goods or products but also have wide reaching environmental and social effects. Processing increases palatability, reduces the incidence of pathogens that could contribute to illnesses and usually extended the shalf life of feed to prove the action of the shalf life of feed to prove the action.

extends the shelf life of food to prevent spoilage, making it a key stage in preventing FLW.²²⁹ However, food processors may create large volumes of FLW due to the scale of their operations and specialised production, where only part of an ingredient might be used in their production resulting in large quantities of by-product(s).²³⁰ For manufacturers and processors, the benefits of preventing FLW are not only the positive environmental impacts, but also the potential for greater profitability and improved public perception of a company.²³⁰ There are opportunities to prevent FLW both through and within processing and manufacturing.

Advances in food processing and manufacturing over the last century have increased the availability of a variety of food ingredients and products through technologies such as cold chain logistics, additives and preservatives, and packaging. This has resulted in the development of a wide Processing...usually extends the shelf life of food to prevent spoilage, making it a key stage in preventing FLW.

There are opportunities to prevent FLW both through and within processing and manufacturing.

range of food products, an extension to the shelf life of food, and, for some people, increased the range of foods available to eat. The manufacturing and consumption of a final product can now take place at a great distance away from where the raw ingredients were grown, and involve many actors including suppliers, processors, and distributors.²³¹

Food processors and manufacturers can affect FLW not only in their own operations, but across the supply chain. For example, the way food is packaged and labelled can affect retailers' or consumers' behaviour and thus the likelihood of waste generation further down the supply chain. For example, fresh tomatoes typically go through a decontamination process and are packaged in plastic – a trade-off between protecting and extending the life of the product to prevent FLW and the creation of plastic waste (see <u>Rethinking plastics in Aotearoa New Zealand</u>) – and labelled with a best before or expiry date, which indicates to retailers and consumers when to stop selling or dispose of the product.

Digitalisation and the integration of new technologies can also help support the quantification of FLW and any prevention interventions at this stage. However, there is a need for sufficiently detailed data to understand the extent of FLW in processing and manufacturing as well as to be able to evaluate the effectiveness of interventions, including their impacts elsewhere in the supply chain.

Levels of food processing

As processed foods have become a more significant part of modern diets, food processing classification systems have been developed and are typically used in epidemiological or dietary research. The degree of processing ranges from sorting fruit or slaughtering animals to the production of ultra-processed foods (UPF). UPF can undergo transformation through multiple technological processes, typically to isolate chemical components such as sugars, proteins, or fats, and can have several highly manipulated ingredients.²³² There are several different classification systems used globally, typically using three to five groups to indicate the degree of processing.²³³ One common example is the NOVA system, which has four groupings (see figure 16). Group 1 captures minimally processed or unprocessed foods that have been sorted, dried, frozen or packed; group 2 includes food that has been processed for ingredients including refining NOVA group 1 foods to be used in seasoning or cooking such as pressing olives for oil, or milling and refining wheat for flour; foods in group 3 are created by adding salt, sugar or combining NOVA group 2 ingredients to make items such as bread, cheese, canned fruits and vegetables; foods in group 4 are ultra-processed and contain additives made using highly modified ingredients using specialised, industrial techniques such as hydrogenation, for cosmetic or sensory properties. Group 4 is a broad category ranging from biscuits, confectionary and ice cream to instant noodles, hot dogs, and sauces.²³²



Figure 16: NOVA classification of processed foods. Adapted from Crimarco, et al.²³⁴

Tensions and trade-offs in the processing and manufacturing of food

There are many trade-offs that occur with advances in food processing and manufacturing. Packaging improves durability of a product, extends shelf life and can also be a food safety requirement, but the current use of plastics contributes to a global waste issue (see section 1.1, <u>Rethinking plastics in</u> <u>Aotearoa New Zealand</u>). Food processing may preserve or enable the addition of vitamins and minerals necessary for human nutrition, but many UPFs have been widely criticised for their poor nutritional quality and are hypothesised to have contributed to increasing obesity prevalence.²³⁵ Processed food can be convenient but can also have less nutritional value.²³⁶ There is some evidence that suggests people most reliant on convenience food generate the most waste, and hints at links between the consumption of UPF with poorer mental wellbeing.^{237,238} These relationships do not necessarily suggest that convenience and UPFs cause high waste and poorer mental wellbeing respectively; it is plausible that people turn to convenience and UPF at times of their life when minimising waste is more challenging or their mental health is poorer.

Fresh foods are generally more nutritious, but are not always a practical option. Large-scale processing can be an efficient mechanism to meet consumer demand for convenient, delicious, and varied food. However, processing increases the use of resources like water and energy and so if food is lost or wasted, so are these resources. Processing and manufacturing can potentially prevent waste by increasing the shelf life of fresh foods but also contribute to waste being generated, for example, through the removal of constituent parts like tomato skin, seeds, and pulp to make tomato juice (see <u>case study 1</u>).

Aotearoa produces high-quality and safe food, which is assessed to meet high standards and exported globally. Processors and manufacturers are obliged to produce safe and suitable food that won't make people sick, as well as being suitable for intended use with labelling requirements around nutrition, dates, and ingredients under the Food Act 2014.¹³⁷ Meeting quality specifications and food safety requirements when producing a food product can contribute to FLW during processing and manufacturing (see sections 2.1.2 and 3.1.3). Quality control needs to be carried out to ensure food meets safety and quality standards, but also typically results in some product being destroyed in the process of testing, and further loss if products do not meet specifications. Ethical responsibilities or environmental protection can sometimes work against business' profit motive – for example, pork produced in New Zealand is more expensive due to high animal welfare obligations.²³⁹ This potential trade-off is addressed by requirements around food safety, but there are no similar requirements on reporting or preventing FLW. Figure 17 describes some tensions and trade-offs when attempting to reduce FLW in processing and manufacturing, though they do not always have to be in opposition and could be aligned. This chapter explores the potential areas to prevent FLW in the processing and manufacturing sector, while acknowledging that cost is a significant factor in decision making in any company, and there are sometimes trade-offs manufacturers must make.



Figure 17: Tensions and trade-offs in FLW prevention during production and manufacturing; they do not always have to be in opposition and could align.

4.1 Understanding FLW during processing and manufacturing

Processing and manufacturing are important for preventing food from falling to lower levels of the food recovery hierarchy since much of the purpose of processing is to keep fresh food edible for a longer time. FLW within processing and manufacturing can be broadly understood as either 'unavoidable' loss, or 'avoidable' loss. Unavoidable loss in food processing and manufacturing, in general, is the expected waste quantities used for foreseeable circumstances including by-products (edible and inedible); retention samples for quality control; cleaning or testing machinery; testing to meet product specifications or terms of trade or financially viable returns; and research and

development for new products. Avoidable loss, in contrast, is a result of errors or inefficiencies, such as damage to equipment or human errors²⁴⁰⁻²⁴² including difficulties in planning inventory and matching production with demand.²⁴³ Figure 18 shows the main causes of FLW in processing and manufacturing facilities. Appropriate strategies for prevention will be specific to subsectors or individual companies and depend significantly on factors like firm size, processing methods, and resource availability, all of which are highly variable within the food industry.

Appropriate strategies for prevention will be specific to subsectors or individual companies and depend significantly on factors like firm size, processing methods, and resource availability...



Figure 18: Causes of FLW in processing and manufacturing. Adapted from Raak, et al.²⁴⁰

4.1.1 Processing and manufacturing in New Zealand

Major processing companies based in New Zealand include: Fonterra, Sanford, Talley's, Tegel Foods, New Zealand King Salmon, and T&G^{*}. International companies such as McCain, Watties, and Nestlé also manufacture in New Zealand (see figure 19). These companies purchase raw produce or ingredients from farmers or other suppliers and manufacture food products to be sold by retailers locally or internationally. The dairy sector in Aotearoa (see section 3.1.1) is dominated by Fonterra, which is estimated to have 80% of the market share of milk processing.²⁴⁴ In meat processing (see section 3.1.2), four major processors – Alliance Group, ANZCO Foods, Silver Fern Farms, and AFFCO New Zealand – account for almost 85% of all industry revenue.²⁴⁵

^{*} Formerly called Turners and Growers, we use the name T&G to include T&G Global as well as any other T&G operations.



Figure 19: A sample of processors and manufacturers operating in New Zealand (i.e. by no means a complete list).²⁴⁶ Some companies may also produce food that fits into more than one category. Adapted from a 2012 Coriolis report, Driving growth in the processed foods sector.²⁴⁷

A large percentage of food processing and manufacturing in Aotearoa involves dairy and meat, a significant proportion of which is exported to overseas markets (see figure 20). Currently, New Zealand exports mainly minimally processed foods including beef, kiwifruit, and apples but also processed dairy products such as milk powder, butter, and cheese.²⁴⁷ The packaged food market has been growing and is expected to continue to grow globally, and currently a large proportion of our processed foods are imported from overseas.²⁴⁶ Processing can create high-value goods with high export prices such as infant formula and nutraceuticals, and is therefore of significance to our economy; this makes it an important domain in reducing FLW.²⁴⁷



Figure 20: New Zealand food and beverage (including alcohol) export values in 2018. Data source: Coriolis report Investor's Guide to the New Zealand Food and Beverage Industry.²⁴⁸

4.1.2 What causes food loss during processing and manufacturing?

food loss during processing and manufacturing occurs for a variety of reasons. FLW in processing can

be high due to removal of the parts of food that are inedible or typically not eaten, for example the husk and cob of corn or the head, bones, and offal of animals; byproducts arising during processing are a major part of built-in waste. In common with other parts of the supply chain, it is difficult to accurately predict demand for a product, so overproduction and subsequently excess stock is common, creating avoidable FLW particularly in products with a short shelf life.²⁴³

... it is difficult to accurately predict demand for a product, so overproduction and subsequently excess stock is common, creating avoidable FLW particularly in products with a short shelf life.

Human and machine errors also create loss at this stage of the food supply chain. Incorrect orders and poor inventory management are significant contributors.^{241,242}

Poor practice around storage, handling, or transportation can result in damage to products including contamination, insect infestations, mould, and deterioration.^{249,250} FLW can occur due to technical faults or machine transitions occasionally resulting in incorrect manufacturing or contamination.²⁵¹ For example, on machine start up, standardised volumes may not be met, leading to unmet product specifications; or changes to formulation can result in temporary contamination. Equipment or machine failures can also result in deformed or inaccurate products or packaging.^{241,242}

Safety standards and quality specifications can also create loss. Part of this is unavoidable, as safety standards and quality control can require product samples that are removed from the supply chain. A

product not meeting the specifications set by the manufacturer or purchaser can result in rejected and wasted stock.²⁴¹ Similarly, developing and trialling new products can result in food loss due to removing products for calibration and testing, or failure to meet specifications. Products that do not meet specifications can sometimes be remanufactured or sold to secondary markets (often at other levels of the hierarchy like food rescue (section 2 in *Food rescue in 2022: Where to from here?*) and upcycling (section 3 in *Beyond the bin: Capturing value from food waste*)).

Specifications set by the manufacturer may be driven by large buyers with purchasing power, industry standards, and/or expectations of wholesalers or purchasers further downstream.²⁵² The setting of specifications could also be influenced by bargaining power and the agreed upon terms of trade around ownership and liability.¹⁰⁷ Attempts to exert market power through practices like stockpiling to reduce supply, or promotions to influence demand and pricing, can lead to FLW.²⁴¹

As an example of the effect of specifications on FLW during processing, milk bought by the dairy company Fonterra is collected by tankers from farmers to deliver to

Specifications set by the manufacturer may be driven by large buyers with purchasing power, industry standards, and/or expectations of wholesalers or purchasers further downstream.

processing plants where it is tested for quality, before processing (i.e. pasteurisation) and manufacturing into dairy products (for example, cream and yogurt).²⁵³ Test sampling occurs throughout, with samples that fail the test resulting in an entire tanker (25,000 litres) being thrown away.²⁵³ Products for export must meet food safety requirements under the *Food Standards Code* as well as export eligibility requirements and other extra requirements from overseas markets. Non-conformance could mean refused entry by the government of the destination country, which might result in waste, unless the product has not been opened or changed and can be re-exported or resold domestically.^{254,255}

4.1.3 We need data to map FLW in food processing and manufacturing

Using a model based on landfill data, the wider commercial and industrial sector in Aotearoa was estimated to have generated over 103,000 tonnes of FLW in 2011,²⁵⁶ which is equivalent to approximately 16% of total FLW. This figure is similar to international estimates with Australia in 2023 reporting 17%,²⁵⁷ Germany in 2021 reporting 18%,²⁵⁸ and Japan in 2011 reporting 20% of total FLW occurring in processing and manufacturing.²⁵⁹ In contrast, some reports from the EU, UK, and Italy estimate that as much as 30%-55% of FLW occurs in the manufacturing stage.²⁶⁰ A more recent estimate from 2022 suggests that 2.2 million tonnes per year of FLW is generated from processing in New Zealand,²⁶¹ highlighting the variability between measures of FLW due to classification (what is counted as FLW), and quantification (how it is estimated or measured).

In Australia, processing and manufacturing is estimated to have the largest proportion of FLW with 63% of food exiting the supply chain.⁵⁷ This high figure is partly due to the inclusion of the loss of inedible parts that may be used in the co-production of animal feed or other materials, for example, the head and offal in meat processing being used for pet food or rendering of tallow for biofuels and soaps, and so does not necessarily reflect loss that is preventable.^{*,57} There is a need to establish sufficiently accurate and detailed baseline measurements within individual processors and manufacturers, as well as the overall subsector, to enable clear problem definition. Obtaining relevant and reliable data from producers can be challenging due to the lack of standardised

^{*} If data from Australia's National Food Waste Strategy Feasibility Study is limited to disposal destinations (which the study defines as commercial composting, home or on-site composting, anaerobic digestion, waste-to-energy, on farm disposal, landfill, and wastewater treatment), the manufacturing sector accounts for 17% of FLW across the Australian food supply chain (1,276 kilotonnes in the manufacturing sector, of 7,676 kilotonnes across the supply chain).

measurement procedures and mandatory industry disclosure – again related to the difficulty in defining what should be counted as FLW, as well as some documentation not being publicly available.^{240,262,263}

Data is scarce making it hard to identify causes of food loss and waste

Definitions of what constitutes FLW likely differ between sectors and manufacturers, making amalgamation of, and comparison between, data sets difficult. Manufacturers may also not share

data due to commercial sensitivity. Many manufacturers (particularly small or medium-sized ones) may be unaware how much they are wasting due to a lack of resources for measurement and quantification, especially in the absence of requirements for such measurements.²⁵⁰ Errors or mistakes that lead to food loss also likely go unrecorded, or are not made publicly available. Insight into the causes of FLW, as well as the extent of FLW within various subsectors, are still unknown, and research specific to New Zealand is lacking.²⁴² Collated data on the main areas of waste and their quantities within a processing plant would be necessary for identifying



between, data sets difficult.

opportunities for prevention. Figure 21 shows an example of data on loss areas by cost and volume for a convenience food manufacturer in the UK.²⁴³ The relative importance of different types of FLW will be specific to this example of a convenience food manufacturer, likely UPFs, and will be vastly

different from a processor of a minimally processed food such as meat or processed foods like canned tomatoes, which may have higher amounts of by-product or trim. This example highlights the importance and usefulness of detailed data for quantifying FLW, as well as the potential for the emergence of different patterns using different units of measurements (for example, costs versus volumes). How we define and use the data can determine the story we tell.

How we define and use the data can determine the story we tell.



Cost (£) Volume (Number of products)

Figure 21: Example of food production food loss data from a UK convenience food manufacturer over a one month period. Adapted from Darlington et al.²⁴³ Units indicate number of products.

4.2 How can we prevent FLW during processing and manufacturing?

Processing and manufacturing companies have commercial interest in reducing or upcycling waste as it contributes to lower spending on raw materials and could increase profit margins. Some loss may be classified as unavoidable by-product and may potentially be valorised through upcycling or nutrient recovery. Sealord, for example, produces fish oil and fishmeal, which in turn are used in a variety of other products.²⁶⁴ Section 2 of *Food rescue in 2022: Where to from here?* discusses opportunities to redistribute food through the food rescue sector in Aotearoa and section 3 of *Beyond the bin: Capturing value from food waste* discusses processing techniques to capture value from FLW such as through the upcycling of ingredients that would have been discarded or underused.

As valuable as these approaches are, they do not prevent FLW at source, and can contribute to a view that FLW is minimal because it is used in other products. The food recovery hierarchy (see <u>figure 1</u>) needs to be applied with nuance as some FLW is not preventable and different sectors have different circumstances (see section 5 of *Food waste: A global and local problem*). In general, the use of the entire resource in food production, for example, using wheat bran and wheat germ (by-products of flour production) in products for human consumption, such as breakfast cereal, is ideal.



to be applied with nuance as some FLW is not preventable and different sectors have different circumstances.

A 2021 study in Germany estimated that 10%-55% of FLW was preventable during processing,²⁵⁸ but more data and information around categories, quantities, and areas of FLW in New Zealand will be

needed to help businesses make decisions and optimise processes to prevent FLW. The sections below will explore how business strategies, technological interventions – particularly in preservation techniques and digitalisation – and policy levers can provide solutions to prevent FLW.

Food manufacturers could take a range of potential actions that impact upon FLW at different levels of the supply chain; examples of these are shown in <u>table 5</u>. However, monitoring and measurement across the manufacturing process is a necessary first step to understand where loss is occurring and will allow for strategies, interventions, and mitigation plans to be designed and implemented (see <u>section 2.2</u>). Collaboration across sectors and subsectors also need to be facilitated,¹⁰⁹ with more frequent communication (between suppliers and retailers, for example), updates, and audits that can also help with better information and planning (see <u>section 2.2.6</u>).²⁵¹



and mitigation plans to be designed

and implemented.

Table 5: Actions by manufacturers that can impact FLW up and down the supply chain.²⁵⁸ Abbreviations: FLW = food loss and waste.

Actions by manufacturers	Where will FLW be prevented?
Check to see if specifications increase supplier's FLW (for example, through supplier audits). ¹⁹¹	Suppliers.
Communication and cooperation with supply chain network for planning. ^{265,266}	
Monitor orders and quality with transparency. ^{267,268}	
Design better packaging for protection and shelf life. ^{269,270,271}	Internal organisation.
Develop and implement systems for measurement and monitoring. ⁵⁷	
Improve planning for processing times and batch sizes. ²⁷²	
Make sure FLW is a part of business culture, goals, and strategy. ^{243,273}	
Proper handling and storage of materials. ⁹⁰	
Train employees around best practices for preventing contamination and machine maintenance. ^{274,275}	
Communication and coordination around demand forecasting (including promotions). ^{276,204}	Distributors/purchasers.
Ensure proper transport and storage conditions. ²⁷⁷	
Reevaluate optimal or necessary specifications such as labelling, quality, and package sizing. ^{241,252}	
Provide clear information around storage instructions and meaning of date labels. ²⁵⁷	Consumers.
Conduct education, outreach, or public communication campaigns. ²⁷⁸	Other stakeholders.
Sector collaboration on data and best practices. ^{61,109}	

There is a lack of aggregated evidence on the comparative effectiveness of different interventions

There are many ideas, options, and strategies for FLW prevention during processing and manufacturing; however, there is still a lack of significant evidence around the effectiveness of different interventions. Interventions are also typically case study specific, making it difficult to compare effectiveness or transfer learnings. Choosing the most effective interventions is challenging, and a lack of substantial evidence can prevent organisations from taking a risk to change processes.²⁷⁹ Evidence on the outcomes of specific interventions ideally needs to be able to be aggregated for transferability within and possibly across subsectors.

4.2.1 What business strategies could be considered?

The size and scale of business operations, as well as costs and the specific situation, will greatly impact a processor or manufacturer's capabilities to adopt changes to address FLW. However, there are two general approaches processors and manufacturers can apply in preventing FLW. The first is optimising existing internal processes, for example, by developing protocols around preventing FLW such as reworking products that don't meet specifications; and the second is increasing the market value of otherwise wasted food, for example, by prolonging the shelf life to increase the value of a product, or enabling secondary markets to sell products that have minor imperfections such as misprinted labels.²⁸⁰

The use of by-products to create new products, for example, through upcycling has been covered in the previous report of this series (see section 3 of *Beyond the bin: Capturing value from food loss and food waste*). Both optimisation and valorisation approaches can involve low-level operational changes and high-level strategy changes to business practices. In our discussions while carrying out research for this report, we heard the importance of change being driven through top-down leadership.

Collaborative practices could aid in the prevention of FLW

Increased exchange and sharing of information, both internally and externally, can support better planning, decision making, and the development of best practice. Better data being shared between suppliers and purchasers can aid in order prediction, inventory management and equipment scheduling. Collaborative, cross-sector approaches can also be used to reduce human error, including through education and training.²⁷³ However, collaboration and competition can be in tension.²⁷³ Data may be commercially sensitive, and within sector collaboration must avoid breaching anti-competition laws.

Collaborative, sector-wide approaches, particularly in data collection practices, need to be developed.¹⁰⁹ There is a need for more collaborative structures and open-source information for the synchronisation of organisations working in the same sector. Sharing data within subsectors can also support establishing sector-wide FLW measures and best practice guidelines as well as national industry growth. Third party data infrastructure and platforms like the Global Data Synchronisation Network (GDSN) by GS1 New Zealand (see <u>box 4</u>) provide a means for organisations to promote their product, choose what information to share and subscribe to promotions or information from suppliers. Also, there are sector specific networks, which work to develop their own data sharing protocols, for example, Trust Alliance New Zealand's digital farm environment plan project and digital wallet project (see <u>box 3</u>).

Box 4: Global Data Synchronisation Network

Better data systems across the supply chain can enable better transparency, risk management and efficiency to reduce FLW through better quality assurance, shorter wait times at choke points and less spoilage.²⁸¹ The Global Data Synchronisation Network (GDSN) is the world's largest real time product data network, developed by GS1 New Zealand, to support high-quality, standardised data from authoritative sources.²⁸²

Manufacturers sell to hundreds of retailers and distributors, so accurate, up-to-date data is important for the smooth movement of products. Examples of relevant data include logistical, shipping, and regulatory information. The GDSN makes it possible for any company and any product to upload and access information to trade products on local and international markets.²⁸¹

The GDSN uses data pools to synchronise product information for trading partners. Data pools are interconnected and interoperable through a global registry and uses a publish/subscribe model,²⁸² where suppliers publish product information to a source data pool and a recipient's data pool synchronises relevant updates based on subscription.²⁸³ This is one example of a data network for products which can facilitate efficiency through digitalisation of the supply chain.

Food loss and waste should be a part of key performance indicators and high-level strategy

A sustainability focused strategy can enhance environmental and economic performance through more efficient exchange of materials and by-products, for example, if the waste from one process becomes input to another. It can also promote innovation and new business opportunities as industry and consumers shift in attitudes.²⁸⁴ Within an organisation, strategy must be aligned with specific targets or key performance indicators (KPIs).²⁸⁴ It is hard to "manage what you can't measure", and having quantifiable measurement towards a desired outcome is necessary to effectively manage progress.²⁸⁵ Capturing data to understand the baseline around inventory, manufacturing processes, and loss or waste is necessary to set targets, and budget must be available for operational change.

New business or contract models could be explored

The lean manufacturing model focuses on mapping value-adding steps, improving connections and flows in a system (such as by prioritising quick changeovers), and continual refinement of processes to cut down costs and reduce waste in production.²⁸⁶ It has been popular in mass production industries, originating from the automotive industry.²⁸⁶ In a case study of a UK vegetarian and vegan meal products manufacturer applying lean techniques, food losses in mixing, weighing, packing, and freezing were reduced in the range of hundreds of kilograms over a three month period.²⁸⁷

Adopting loss prevention strategies can be a part of a greater transformation of an organisation's existing business model towards sustainability, circular economy approaches, social enterprise, or industrial symbiosis, which explores how a company operates within its larger ecosystem (see <u>section 1.2</u>).^{284,288,289} This could also potentially be extended to business relationships and incorporated in contract models or terms,¹⁰⁷ such as in procurement processes, including evaluation of bids for contracts, or sharing responsibility and carrying out due diligence for sustainable practices with suppliers and sub-suppliers.²⁴⁶ Digitalisation can also enable this, as exemplified by CiRCLR, which is a business-to-business platform that helps companies turn potential FLW into new revenue streams for businesses (see <u>section 2.2.2</u>).²⁹⁰ The growth in consumer appetite for environmentally and socially responsible goods also provides a solid business case.²⁹¹

4.2.2 What technological interventions are available?

New technological methods could enable innovative new products that create a competitive advantage and increase profits. For example, in processing milk for cheese processing, whey is created as a by-product, which is popularly marketed as a fitness supplement.^{*}

Technological innovations in food science can aid in reducing food spoilage. Short shelf life products are at a greater risk of being wasted, so innovations in processing technology that increase shelf life can support the reduction of waste. Packaging technologies and labelling can also increase shelf life of food products as well as mitigate the use of plastics (see sections 3 and 4, <u>Rethinking plastics in</u> <u>Aotearoa New Zealand</u>), prevent damage to the product, provide information for better storage, and shape the way consumers interact with food products. Examples of these types of technologies that increase a product's shelf life are described in <u>table 6</u> and are one area of action processors and manufacturers could consider for preventing FLW at other points in the supply chain.

Meeting safety requirements and product specifications are a major trade-off against FLW generation; there is a need for manufacturers to optimise production while maintaining quality. Automated digital systems can increase efficiency, reduce the risk of human error, as well as provide data measurement, control, and oversight. Overall system design for monitoring, measurement, and communication can also be aided by digital technological innovations to increase efficiency. Examples of such hardware and software technologies and how they can prevent FLW are described in table <u>7</u>.^{292,267} Technological interventions sometimes are not implemented due to accessibility, costs, and market access requirements – where some markets may have specifications on packaging, or regulations on technologies that impact the food product directly.^{292,267}

Food processing prevents FLW

Commercial food processing techniques typically act by inhibiting the growth of microorganisms (for example, freezing, drying, fermenting), inactivating microorganisms (for example, pasteurisation, irradiation) or restricting the access of microorganisms to the food product (for example, packaging).²⁹³ The development of new food processing techniques not only prevents spoilage or increases the variety of products but can also aid in maintaining nutritional values and increasing the quality and value of food products.²⁹⁴ For example, expanding the life span of bread has been taking place for thousands of years through fermentation with lactic acid bacteria to increase acidity for antifungal activity (in the case of sourdough), and in modern times chemical preservatives (for example, calcium propionate) are commonly added to breads to extend shelf life.²⁹⁵ <u>Table 6</u> describes state of the art food processing techniques in the context of FLW.

^{*} Whey is also an example of the sometimes blurry boundaries between different stages of the food recovery hierarchy – in <u>Beyond the bin: Capturing value from food loss and waste</u> we used whey as an example of an upcycled product.

Table 6: Current and emerging food processing and packaging technologies that help to extend the shelf life of foods. Abbreviations: MPa = megapascals, nm= nanometres, PEF = pulsed electric field, UV = ultraviolet light, UV-C = short wavelength ultraviolet light.

Technique/Technology	Description	Example(s) of use	State of use
Active packaging.	Packaging with added materials such as antioxidant or	Nano-pack ²⁹⁶ was an EU funded project	Some adoption.
	antimicrobial compounds that interact with the product or	that developed an antimicrobial flexible	
	product environment to extend shelf life, positively impact	plastic food film shown to extend the shelf	
	food quality and safety, and deter spoilage.	life of breads, cheeses, and cherries. ²⁹⁷	
Aseptic processing.	The controlled sterilisation of a processing facility where	This process is mainly used for liquid-based	Widespread use.
	the sterile food product is in a sterile package, which is	foods like milk, juices, yoghurts, salad	
	then hermetically sealed so no air can pass in or out. ²⁹⁸	dressings, baby foods, soups, and desserts.	
Chemical preservatives.	Added compounds, which help extend shelf life by	Lactic acid is commonly added to	Widespread use.
	inhibiting oxidation, browning, or enzymatic reactions and	fermented meat and dairy products while	
	ensure safety by inhibiting or killing microorganisms.	citric acid can be found in jams, sauces,	
	Typically, an acid-sorbate, benzoate, sulfite, or nitrite; well	cheeses, baked goods, and canned	
	lactic, citric, and acetic acids are common as well.	vegetables. ²⁹⁹	
	Parabens, sulfur dioxide and sulfites are also used for their		
	antimicrobial activity, as well as antioxidants. ²⁹⁹		
Cold plasma.	Applying partially or totally ionised plasma gas, which	Has shown to be an effective	Research and
	carries chemically reactive species including photons, ions,	decontaminant for fruits and vegetables,	development.
	free electrons, and atoms at ambient temperature, is a	and potential meat preservative but has	
	promising method for attaining microbiological safety in	not yet been applied commercially. ³⁰¹	
	temperature-sensitive food. ^{300,301}		
Ethylene-removing	Ethylene is formed during ripening, bruising or disease of	Ethylene adsorption packaging, which	Some adoption.
packaging.	produce via microorganisms. Ethylene removing packaging	covers pallets is used in the transportation	
	is a form of active packaging which reduces ethylene	of fruits and vegetables to slow ripening. ³⁰³	
	concentration surrounding produce through adsorption,		
	absorption or chemical reactions during transportation and		
	retail. ³⁰²		
Flash pasteurisation.	Also known as high-temperature, short-time	Typically used to extend the shelf life of	Widespread use.
	pasteurisation, which heats beverages to 72 °C- 80 °C for	milk, but also used in beer. ³⁰⁰	
	15-30 seconds allowing microbial inactivation to occur,		
	followed by rapid cooling to 4 °C- 5 °C. ³⁰⁰		

Technique/Technology	Description	Example(s) of use	State of use
Freeze-drying.	Also known as lyophilisation, freeze-drying involves the	Fruits and vegetables are the most	Widespread use.
	freezing of the product, which inhibits microbiological	common foods to be freeze-dried, as well	
	processes, drying through ice sublimation (most often at	as confectionery and desserts.	
	reduced pressure), and then desorption drying (drying the		
	product to the required final humidity). ³⁰⁴		
High pressure	The application of high mechanical pressure (100-800 MPa)	Deli meats, dips, seafoods, sauces,	Widespread use.
processing.	for seconds to minutes, which inactivates microorganisms	beverages, jams, pet foods, baby foods,	
	and denatures enzymes with minimal effect on food	fruits, and vegetables products typically	
	quality and nutrition. ²⁹⁸	undergo high pressure processing. ³⁰⁵	
Intelligent packaging.	Chemical agents or sensors that monitor the state of the	Colourimetric indicators (which changes	Some adoption.
	product or environment and provide information that	colour due to enzymatic or chemical	
	indicates food quality to help consumers decide whether	interactions) are used to indicate the	
	to purchase or eat. ²⁷¹	freshness of fish and meat products. ²⁷¹	
Modified atmosphere	Providing the optimum conditions for a food product	Common foods that have modified	Widespread use.
packaging.	through modifying or altering the atmosphere inside a	atmosphere or vacuum packaging include	
	package, slowing the growth of microorganisms through	salads, fresh pastas, meat, seafood,	
	preventing chemical and biochemical reactions. ²⁹⁸ Also	cheeses, and bakery products. ³⁰⁶	
	includes vacuum packaging.		
PEF Technology.	Involves the application of high voltage short electric	There is some commercial use in juices and	Research and
	pulses, which inactivate enzymes and microorganisms. This	smoothies. Also, in wine making and	development.
	also can be used to restore quality or change properties of	potato processing – so they have a more	
	produce. ³⁰⁷	uniform colour and absorb less oil during	
		frying (see case study 2, <u>Beyond the bin:</u>	
		<u>Capturing value from food waste</u>).	
Ultrasonication.	Ultrasound application can be used as a preservation	Used with fruits and vegetables, cereals,	Widespread use.
	method through microbial inactivation and in the analysis	honey, and dairy products. Also, for the	
	or change of physico-chemical compositions of food. ³⁰⁸	tenderisation of meat products. ³⁰⁸	
UV-C radiation.	UV irradiation in the wavelength range of (200-280 nm)	Can be used to increase the shelf life of	Some adoption.
	disrupts the genetic material of microorganisms, also	milk, breads, fruits, and vegetables.	
	enables the preservation of nutrients.		

Improved packaging design and innovations can prevent food loss and waste

Packaging design and packaging innovations like active packaging can prevent FLW both at the at the processing and manufacturing stage, as well as further downstream in the supply chain.

Improved packaging design can prevent food being lost or wasted due to damage, as well as enhance its shelf life. This can be achieved in a number of ways: on-package information to provide consumers with instructions regarding ideal storage conditions; resealable packaging can prevent exposure to air or moisture; flexible and transparent packaging can increase accessibility and visibility; smaller or more varied packaging sizes can make it easier for consumers to purchase the right amount of food.³⁰⁹ It is also critical for food preservation and safety.³¹⁰

Active packaging can embed compounds like antioxidants and antimicrobial agents to combat degradation, prolonging the quality and safety of food products, while intelligent packaging uses sensors and indicators that react to microbiological or environmental changes like temperature or pH

level, to monitor food quality and safety.^{270,311} Active and intelligent packaging (see figure 22) can help consumers make decisions on food disposal as more accurate indicators of quality or safety rather than date labels (see box 9). Finally, the addition of radio frequency identification (RFID) tags and 2D barcodes such as quick response (QR) codes, can enable real time gathering and/or sharing of more information about products to various stakeholders across the supply chain, including manufacturing date and batch number, which could also potentially be useful for tracking or food recalls.^{312,313}



Active and intelligent packaging can help consumers make decisions on food disposal as more accurate indicators of quality or safety rather than date labels...



(c) Freshness indicator

(d) Freshness indicator

Figure 22: Examples of intelligent packaging systems (a) Temperature indicator; (b) ripeSense indicator; (c) Freshness indicator; (d) Freshness indicator. Image credit: Yan et al.³¹¹

Though there is an inherent tension between reducing food waste through packaging and the increased waste production from packaging, this could potentially be mitigated through innovations at the systems level or by new materials technologies such as bio-based plastics (see sections 3 and 4 of <u>*Rethinking Plastics in Aotearoa New Zealand*</u>). Life cycle analysis has shown that the environmental footprint of FLW can sometimes be greater than the impact of the packaging itself.³¹⁴

Technological solutions supporting data and management can aid food loss and waste prevention

Analysing data for more effective decision making in food processing and manufacturing can be aided through recent advances in digitalisation (see <u>table 7</u>). Aggregate planning and scheduling have always been necessary to be able to optimise resources and production. Overproduction and FLW typically occurs when planning or demand forecasting is poor.^{272,276} Demand forecasting models can be used to support production planning and inventory management, with recent developments in big data, machine learning (ML) and AI improving their accuracy.^{315,316} IoT monitoring systems and digital twin technology, which provides virtual simulations and data dashboards, can capture data during manufacturing in real time and even make it available to other stakeholders such as suppliers, regulators, or purchasers.³¹⁷ The direct monitoring and digitalisation of collecting FLW information through IoT architectures also seems to be a promising intervention (in one case, a ready meal factory reduced waste by 60.7%).³¹⁸ A food processor could also use improved demand forecasting technology to combine data from diverse sources, like social media trends or weather, to understand potential changes in demand or risks to supply and make real time decisions in response to events such as severe weather events (see <u>case study 9</u>).³¹⁹

Enterprise resource planning (ERP) software is also commonly used to help analyse shelf life, track orders, prevent cross-contamination, avoid overstocking and more easily manage an inventory.²⁵⁰ Recent developments in advanced planning and scheduling (APS) systems can integrate supply management, production scheduling, order processing, and demand forecasting for integrated planning and coordination.³²⁰ APS systems can be useful for adapting to fluctuating demand, manufacturing processes that make multiple products, and making to order rather than making to stock. However, SMEs are less

APS systems can be useful for adapting to fluctuating demand, manufacturing processes that make multiple products, and making to order rather than making to stock.

likely to use ERP software or advanced planning systems and are also typically less able to cope with disturbances, rush orders and breakdowns.³²¹ The use of barcode and/or RFID technology (see <u>box</u> <u>5</u>), which could be combined with sensors for environmental monitoring is also being researched widely for application in the food industry, likely due to its low cost, accessibility, and ease of integration into existing systems.^{271,312,322} These digitalisation solutions for data and management enable better visibility and operational improvements to support better planning and decision making.

Table 7: Summary of technologies that could be used in processing and manufacturing to support the prevention of FLW. These technologies are non-exhaustive and may be combined and developed in a myriad of ways by designers and engineers for specific use cases and system integration. General implementation considerations for all technologies are costs, integration with existing infrastructure, privacy and security, and a need for skilled people and training. Abbreviations: 1D = one dimensional, 2D= two dimensional, 3D = three dimensional, AI = artificial intelligence, APS = advanced planning and scheduling, AR = augmented reality, ERP = enterprise resource planning, FLW = food loss and waste, IoT = internet of things, ML = machine learning, RFID = radio-frequency identification, UoA = University of Auckland, VR = virtual reality.

Technology	Description	Potential impact on FLW	Example	Considerations
AR and VR.	Computer-generated models or environments that alter a user's perception through visual immersion or overlays.	AR and VR could be used to help train staff in processing and handling. They could also provide access to, and present data to enable better decision making.	A VR environment allows staff to interactively and step-by-step be guided through alerts and maintenance operations of an industrial ice cream machine. ³²³	Early adoption. AR and VR hardware is commercially available, but software development for useful applications in manufacturing is needed.
APS systems.	Software algorithms or models that support the allocation of resources and materials to maximise production capability. ³²⁴	APS systems can help with the planning and optimisation of production to meet demand and prevent overproduction or order errors.	A poultry processor in the US, Koch Foods, implemented an APS system to streamline operations through better planning and scheduling resulting in reduced change overs and packaging problems. ³⁰⁹	Widely available. There is a large range of commercially available APS systems.
AI.	Simulations and software that can perform complex tasks through learning.	Al can be embedded in software systems to help analyse data and make suggestions and predictions for better decision making.	Al can be used to optimise and control the drying process of fruits and vegetables using physical fields (e.g. infrared, microwave), for better efficiency as well as improve the quality of dried products. ³²⁵	Early adoption. Generative AI models are now easily accessible. Neural network and deep learning models have had recent exponential breakthroughs.
Big data.	Large amounts of data and information for analytics, evaluation, or prediction.	Data can be collected throughout the food supply chain to enable better measurement and understanding of where and how loss occurs.	Anticipatory shipping can use historical order and customer data to predict future orders and ensure products are located at the nearest distribution centres to optimise logistics. ³²⁶	Significant adoption. Cloud-based storage systems are relatively accessible and cheap. Big data is the backbone of other technologies like AI and ML.
Blockchain.	Distributed, decentralised database mechanism for	Blockchain could be used to enable fast, secure	IBM Food Trust is a blockchain- based platform to track food	Early adoption. Blockchain technology has relatively high

Technology	Description	Potential impact on FLW	Example	Considerations
	secure transactions across a network.	transactions. For example, smart contracts. ³²⁷	products from farm to table, enabling suppliers and retailers to identify any safety or quality issues quickly. ⁸⁵	computational and network requirements.
Computer vision.	Algorithms or software, typically using neural network models, which enables computers to interpret and identify images or videos.	Computer vision could help identify and sort food products by visual qualities such as colour and texture for improved quality control.	Compac Sorting uses computer vision capabilities to improve defect detection and categorisation for cherry sorting and grading. ²⁰⁹	Early adoption. There have been significant developments in computer vision and it is able to be integrated in manufacturing systems.
Digital twin.	Virtual representation of objects, products assets, and processes that can deliver real time structured information through aggregating data across a context. ^{323,328}	Digital twins could help food processing companies monitor equipment, products, and processes to detect issues early and prevent unplanned waste.	The food manufacturer Mars is deploying digital twins in manufacturing facilities to optimise production, improve margins and reduce FLW. ³²⁹	Early adoption. Digital twins need access to data sources.
Demand forecasting.	The use of analysis and models based on past data to predict future demand for a product.	Demand forecasting could result in more accurate predictions to aid better planning in manufacturing quantities and minimise overproduction.	The use of demand forecasting models in a study of three German bakery chains resulted in a waste reduction of 37%-89%. ³³⁰	Widely available. The accuracy and complexity of demand forecasting models vary.
ERP software.	Integrated system for the management of business processes such as materials, production, inventory, and orders.	ERP software helps manage inventory and orders for better planning and fewer errors.	Spoiler alert ³³¹ is a commercial platform for inventory management, providing data insights and accelerating transactions – particularly for short-dated inventory in the food and beverage industry.	Widely available. There are many business and inventory management software solutions commercially available.
IoT.	Network of devices integrated with sensors and	IoT can provide information to workers to help with	A case study of a ready-meal factory using an IoT system	Widely available. Many actuators, sensors, and devices can now be

Technology	Description	Potential impact on FLW	Example	Considerations
	software to connect and provide data or information.	measuring, monitoring, improve communication and reduce manufacturing errors.	architecture for the capture and digitisation of FLW data in real time showed a 60.7% reduction in FLW. ³¹⁸	connected, there is also scope for extension.
ML.	Statistics-based algorithms that learn from patterns in data.	ML algorithms can be used to better predict demand for forecasting.	Hyperspectral imaging with ML to analyse food composition enabling producers to ensure their product meets quality standards non- invasively, identifying issues early and minimising destruction of the product. ³³²	Significant adoption. Using ML to analyse big data and create automated systems and do predictive modelling is relatively common.
RFID tags.	A wireless radio system to provide automatic identification and tracking by a reader or scanner.	RFID can help with the tracking and monitoring of items in production and throughout the supply chain.	A RFID system was embedded with non-destructive quality analysis on single units of artisanal Italian cheese for quality and tracing information. ³¹²	Widely available. RFID technology has been widely used and implemented in a wide range of cases.
2D barcodes.	Encodes information in the horizontal and vertical dimension compared, with traditional 1D barcodes to be read by a camera-based imaging scanner. ³³³	2D barcodes can contain information such as serial or lot numbers, manufacturing or expiration dates, and website links.	Woolworths Australia (in collaboration with Hilton Foods and Inghams) trialled 2D barcodes on meat products in August 2019 to improve traceability and stock management. ³³⁴	Widely available. If already using 1D barcodes, GS1 NZ offers 2D barcodes which will still work with traditional scanners. ³³³
3D printing.	An additive manufacturing process that involves the layering of material filaments to create a product.	Can be used with edible materials and create food products with customisable shapes, sizes, and nutritional value. Off cuts, trimmings, or by-products, which would otherwise be wasted could be used. ³³⁵	The Creative Design and Additive Manufacturing Lab at the UoA Faculty of Engineering is using chocolate powder or meat waste in 3D printing to create new high- value products. ³³⁶	Early adoption. There is a commercially available 3D printer for chocolate. 3D printing for other food substances like meat or cake or pizza are also in development. ³³⁷

Research and innovation for technological solutions in food processing

There are many areas of research that could contribute to the development of innovative solutions to prevent FLW in processing and manufacturing such as: food science for novel processing techniques; operations research for increases to efficiency; and food science and psychology for improved product and packaging design. For example, flavournomics research, which studies the development of flavour profiles, could reduce FLW by enhancing the taste and quality of food through processing techniques.³³⁸ The application of emerging digital technologies, as well as the integration of data and digital technologies such as through multimodal data fusion to be able to combine multiple data sources will be a key area for innovation in the food sector.³³⁹ Research into quality control requirements and novel quality control tools (for example, spectroscopy, imaging, and modelling) could also prove to be less destructive and require less product than current techniques. Social science research will be able to shed light on the effectiveness of interventions, their impacts, and identify top priorities for implementation.

Box 5: Technology for traceability

Barcodes were first used by the food and grocery sector in 1974 and were quickly adopted in New Zealand in 1981.³⁴⁰ They are now used ubiquitously on products during transactions and for inventory management. The organisation GS1 New Zealand (see <u>box 4</u>) coordinates universal product codes (known as UPCs) to provide standardised company codes, product codes and check digits (for verification). Manufacturers register with GS1 New Zealand and are assigned a company code to use.³⁴¹ 2D barcodes are now offered, which also encompass the data currently used for linear barcodes. 2D barcodes, like QR codes, can be used to store and access more information about a product than traditional barcodes, including batch numbers, serial numbers, manufacturing dates, quality assurance, and more.³¹³ Also, they can provide a URL for consumer facing web information to link to ingredient lists, expiry dates or even promotions and product reviews.

RFID tags use radio waves to transfer data and can be read at a larger range with multiple tags at once compared to barcodes (which also need to be in the line of sight of a scanner).³⁴¹ RFID data can be updated or overwritten, as well as encrypted to make it difficult to replicate and provide security. Tags are usually more durable (for example, waterproof) than printed barcodes, but are also more costly.³¹³ RFID technology, like barcodes, are therefore also useful for identification and can provide traceability in real time. RFID systems have the potential to combine information in inventory and logistics management, for example, with sensors for monitoring environmental conditions (for example, linking temperature or humidity sensors in storage for safety or spoilage analysis). Barcodes and RFID can be used at single product item level and/or at case or pallet levels, for managing batches, stock rotations and inventory.

The international standard ISO 22005:2007, which covers food traceability, requires that companies keep a record of suppliers and customers, according to a 'one-up, one-down' principle.³²² Developing traceability is important for quality assurance throughout the whole supply chain and can be useful in reducing handling or storage errors. 2D barcodes and RFID tags enable more detailed information to make product recalls more specific so fewer unaffected products are thrown out; they also potentially automate capturing the batch code or expiry date at till for batch tracing and/or delivering alerts or dynamic pricing. 2D barcodes or RFID can also potentially be integrated in the future with AI vision systems.³²²

Q Case study 9: AI for demand forecasting and production planning in food processing

A global food manufacturer of grains and meat integrated AI technology for demand forecasting and production schedule optimisation. This involved the development of a user interface and platform that integrated ML algorithms for demand forecasting and optimisation algorithms for scheduling. Previously, the company had used a forecasting solution based on a common statistical approach with limited data, which was less useful because it took a long time to generate and resulted in an inflexible schedule that lacked in optimisation capabilities and scenario planning.

Food manufacturing of short shelf life products have the challenges of short lead times, lastminute order changes, and strict regulatory requirements, which sometimes resulted in missed order fulfilment or excess production and FLW. Sales orders arrived at the plant sometimes daily and when orders deviated from the forecast, planners lacked the time to modify the active demand forecasts and production schedules. Weekly demand forecasts would deviate greatly due to highly variable demand, and schedules would take 90 minutes per production line to generate.

The final AI digital solution was able to leverage more data sources to improve forecast accuracy, including historical production schedules, sales order history and demand forecasts, production capacity and manufacturing specifications, and raw materials and finished goods inventory. It was able to generate daily demand plans with forecasts 21 days into the future, as well as generating production schedules for different time horizons (for example, 2, 7, or 14 days) in advance. The new platform was able to generate a demand forecast and schedule more quickly (in 3.5 minutes), with significantly reduced time spent on manual work for scheduling, better transitions, and order fulfilments.³⁴²

4.2.3 What policy levers should be considered?

There are currently no FLW reporting or reduction related obligations for manufacturers in Aotearoa though there are industry groups beginning this work in a voluntary capacity (most notably, the Kai Commitment). In this sense, we lag behind other countries: in Norway, government ministries and organisations from industry and retail signed an agreement on FLW reduction in 2017 with a shift toward binding law becoming likely;¹⁰⁹ commercial and industrial premises in Singapore have had mandatory waste reporting since 2014 with reporting requirements around FLW coming into effect in this year (2024);³⁴³ and France has had legislation for the food

There are currently no FLW

reporting or reduction related obligations for manufacturers in Aotearoa though there are industry groups beginning this work in a voluntary capacity...

industry to donate food products fit for consumption since 2020.³⁴⁴ Recently the European Parliament voted to set legally binding FLW reduction targets, with a 20% target for the processing and manufacturing sector.³⁴⁵ FLW reduction in processing and manufacturing could be incentivised through certification or funding programmes, which promote sustainable business practices or the uptake of research and innovation. Legislative changes such as preventing organics in landfill and changes to food labelling requirements could also be considered in New Zealand to prevent FLW happening at multiple levels of the supply chain.

Certification programmes for businesses that demonstrate FLW reduction

Corporate and consumer behaviour can be influenced through certification programmes.⁷ Examples of this implemented in Aotearoa already include energy star ratings on appliances, or health star ratings on food products. The Eco Choice Aotearoa label³⁴⁶ is the most directly related independent certification programme and a part of the Global Ecolabelling Network;³⁴⁷ however, it is not currently applied to food products. A low waste criteria as a part of a food product certification could help

consumers make responsible product choices and is shown to have positive influences on trust and attitudes towards products.³⁴⁸ Sustainability certifications can also grow sales and increase long-term profitability.³⁴⁹ These can be third party or government endorsed for increased credibility. Internationally, in March 2023, the French Government launched a national 'anti-food waste' label to distinguish the performance of FLW reduction practices by actors across the supply chain;^{350,351} and the Danish Government in 2022 announced the development of a



climate label for food to promote environmentally friendly food production and help shoppers make socially responsible choices.^{352,353}

Grant funding or financial incentives for research and innovation and its adoption

There are significant costs associated with the development and acquisition of new technology, which can prevent widescale adoption, particularly in SMEs. Increased investment could support the research and development of innovative practices or products that will also create new economic value. Providing financial incentives such as grants or tax credits for companies that adopt technology and demonstrate FLW reduction could also facilitate innovation in the sector⁷ (see <u>section</u> <u>7</u>).

Reforming date label requirements

In New Zealand, most packaged foods have a date label. Usually manufacturers will include the words 'use by' or 'best before', where use by is indicative of the date when food could become a health risk to eat and best before relates to the food quality. In the case of bread, 'baked for' or 'baked on' may also be used. Labelling standards are set by the Food Standards Authority of Australia and New Zealand (FSANZ)³⁵⁴ as legislated by the *Food Standards Australia New Zealand Act 1991*.³⁵⁵ Standard 1.2.5³⁵⁶ outlines the date information that must be marked on food labels, and states that it is not required if the best before date of the food is two years or more after the date it is determined; special cases and exceptions include infant formula, individual portions of ice cream and small packets of confection such as gum.

Use by dates are set by processors and manufacturers to indicate when food is safe to eat and can be based on laboratory testing of microbial presence, modelling based on earlier studies or static testing where it is stored under 'typical' conditions and sampled periodically for chemical or physical changes.³⁵⁷ Best before dates are often arbitrary and at the discretion of the manufacturer so may be chosen for business reasons to do with perception of quality rather than safety.³⁵⁸⁻³⁶¹ Deciding whether to utilise use by or best before can therefore be subjective, particularly in products like cheese, milk, treated meats

Best before dates are often arbitrary

and at the discretion of the manufacturer so may be chosen for business reasons to do with perception of quality rather than safety. and ready meals – which have sometimes used both use by and best before.^{362,358} Also, loose goods, for example, tomatoes, have no date of expiry associated with them but when they are packaged will typically have a date label, highlighting the inconsistency within product types (though packaging also can alter shelf life – as discussed previously in <u>section 4.2.2</u>). Food in New Zealand cannot be legally sold past the use by date due to the health risk but can be sold past the best before date.³⁶³ There is some suggestion that consumers dispose of food on the basis of date labels even if it is still good to eat; we explore this in more detail in <u>box 9</u>.

Date labels are also used in inventory management. A manufacturer's decision to put a short or long best before date can affect retail actions. For example, putting a short best before date can result in retailers prioritising selling the product, even though a product with a long best before dates could be considered to have a competitive advantage. Retailers also occasionally discount products near their expiration dates, although this may signal





A manufacturer's decision to put a short or long best before date can affect retail actions.

decreasing quality or undesirability and affect brand image for manufacturers.³⁶⁴

Reform of date labelling has been suggested to prevent premature disposal by consumers or suboptimal inventory management. A survey by the Australian Institute has shown large support (78%) for reforming use by and best before date labelling.³⁶⁵ Reforms that could occur include: having

standardisation within subsectors of food, rephrasing the wording on the labels – instead of best before using 'often good after'; more descriptive dates like 'manufactured on' or 'freeze by'; removing the requirement of (or even banning) best before date labels on food, and rather than having foods exempt from having date labels; and having foods that are required to have only a single type of date label (for example, use by on fresh fish).³⁶⁵

In our engagement for this work, we heard directly from

food producers and processors as well as academic researchers that best before dates provide specific benefits to some food businesses. In the case of one producer/processor, the best before date was part of their ability to market the product as having a longer shelf life, which they saw as a point of differentiation from their competitors. In another example, the food would deteriorate before it became unsafe to eat, and removing the option of a best before date could lead to customers experiencing a poor quality product, which would reflect badly on the brand. It is clear that the value of best before dates is very context specific.

Alternative design options to date labels could be considered, for example, through active and intelligent packaging. Freshness indicators such as Mimica Lab's 'Bump' (see <u>case study 10</u>) have begun to be integrated into retail products, removing the need for date labels to indicate quality or safety. Changes to packaging and label





It is clear that the value of best before dates is very context specific.

design could also be effective, such as Too Good To Go's label initiative, which is active in 13 countries and has brands pledge to include a 'Look, Smell, Taste, Don't Waste' label (see <u>figure 23</u>).³⁶⁶ The aim of the label is to prompt the consumer to check the food and use their senses to decide whether to eat something, instead of the best before date.³⁶⁶



Figure 23: Too Good To Go food waste label on Nestlé's products in the UK. Image credit: Too Good To Go.³⁶⁶

Q Case study 10: Bump spoilage indicator

Mimica Lab's Bump (see <u>figure 24</u>) is an example of intelligent packaging aiming to reduce food waste by providing an accurate, real time indication of freshness for products like seafood, juice, and smoothies. It is also compatible with a wide range of other fresh food products such as dairy, meat, and seafood. The label uses a temperature-sensitive, plant-based gel as the indicator. When fresh, the label feels smooth and lets the consumer know the product can be consumed, over time the label will feel bumpy to indicate that the product is no longer fresh. Short shelf life foods are most commonly wasted compared to longer shelf life products, especially if it is past the date label.^{241,367} Bump acts as an alternative solution to date label system, and aims to reduce household and retail food waste of perishable products.³⁶⁸



Figure 24: Bump cap lid. Image credit: Mimica Lab.³⁶⁸

Discussions around label changes have been occurring internationally in Europe since the mid-2010s.³⁶⁹ Adoption of changes to date labels have mainly occurred due to industry led collaborations and third party initiatives like Too Good To Go,³⁷⁰⁻³⁷² or by independent supermarkets

(in the UK).^{373,374} France has led legislative changes to alter wording on date labelling and abolished requirements for long shelf life food including pasta, rice, and coffee.^{375,376,377} Table 8 shows recent international changes and trends in date labelling. Empirical studies, post label change, to explore the effectiveness on consumer behaviour and food waste are currently missing.

Separate charges or bans for organic waste and landfill Ideally, FLW would be diverted at higher levels of the food recovery hierarchy with disposal as the last resort. Disposal is currently discouraged in New Zealand through France has led legislative changes to alter wording on date labelling and abolished requirements for long shelf life food including pasta, rice, and coffee.

the waste disposal levy, the revenue of which is distributed among city and district councils to carry out waste minimisation, as well as to the contestable Waste Minimisation Fund, for projects that promote or achieve waste minimisation (see section 4, <u>Beyond the bin: Capturing value from food</u> <u>loss and food waste</u> for more detail).³⁷⁸ This could be modified through having separate charges for organic waste or segregating the organic waste being sent to landfill to raise awareness and incentivise manufacturers to prevent FLW. The organic waste can then be recovered for nutrient or energy recovery (see section 4, <u>Beyond the bin: Capturing value from food loss and food waste</u>). In Singapore since 2019, commercial and industrial sites have been required to separate their FLW for treatment and reporting; new buildings are also required to have on-site FLW treatment systems.³⁷⁹ States in the US including California, Massachusetts, and New York also prohibit sending FLW to landfill.³⁸⁰ Annex 2 provides an overview table of international policies on bans of organics to landfill.

Geography	Year	Action(s)	Evidence/outcome(s)
Denmark	2019	In collaboration with Too Good to Go,	Based on survey results, 70% of
		a number of producers such as Arla,	respondents reported that they
		Carlsberg, and Unilever used the words	would throw less food away
		<pre>'ofte god efter' ('often good after')</pre>	because the new labelling lets
		instead of the usual best before. ^{372,370}	them know that it is still edible
		Also, food company Danish Crown has	after the best before date. ³⁸²
		a <i>'se, duft, smag</i> ' ('see, smell, taste')	
		label. ³⁸¹	
EU	2014	Delegates from the Netherlands and	The most recent version of
		Sweden submitted briefings to the	regulation no 1169/2011 of the
		European Commission and European	European Parliament states that
		Parliament on food waste and best	dates of minimum durability, use
		before date labels, suggesting an	by, and date of freezing are not
		extension to the list of products that	required for fresh fruit and
		could be exempted from the	vegetables, wine, beverages
		requirement of a best before date	containing more than 10% volume
		label, including pasta, coffee, and rice;	of alcohol, bakery or pastry
		exploring possible actions to support	goods, vinegar, cooking salt, solid
		better understanding by consumers of	sugar, and confectionery. ³⁸³
		date labels was also suggested. ³⁶⁹	

Table 8: Recent international experiences of date label changes.

Geography	Year	Action(s)	Evidence/outcome(s)
France	2015	The French senate voted to abolish	Carrefour removed best before
		best before label requirements on non-	dates on products where these
		perishable foods such as dried pasta,	dates are not meaningful, such as
		fice, and sugar.	sweets, and extended use by
	2020	Large organisations and food	dates on some meat- and dairy-
		producers such as the Ministry of	based products. ³⁸⁴
		Agriculture, Intermarché, Nestlé,	
		Danone and Carrefour, on an initiative	
		led by Danish organisation Too Good	
		To Go, developed and signed an action	
		about how long products can be safe	
		to consume, including improving use	
		by and best before date labels. ³⁷¹	
	2022	A government ministry decree	
		added to the current best before labels	
		in France: 'pour une dégustation	
		<i>minimale</i> ' (for minimum standard	
		taste) and 'ce produit peut être	
		consommé après cette date' (this	
		product can be consumed after this	
Norway	2017	O-Mejeriene (O-Dairy) changed their	The Norwegian Food and Hygiene
,		date label to 'best før (dato) men ikke	Authority, Mattilsynet, was
		dårlig etter' ('best before (date) but	alarmed by the new label with
		not bad after'). ³⁸⁵	concerns around misleading,
			ambiguous information. Matvett,
			a lood industry organisation, was
			on supplementary date
			labelling. ³⁸⁵
	2018	Several Norwegian food producers	Q-Meieriene decided to the keep
		Norgesgruppen met and agreed on the	the industry consensus ³⁸⁵
		new script for supplementary date	
		labels to be 'best before (date), often	A market research survey showed
		good after'. This was applied to dairy	77% of the respondents agreed
		products, eggs, orange juice, flour, and	that the new script explains the
		pread.	meaning of the date label better
			to use out-of-date products due
			to the supplementary date label.
			However, 67% of the same
			respondents answered that they
			did not need the addition as they
			already understood the original

Geography	Year	Action(s)	Evidence/outcome(s)
			best before well enough (though people who are informed on the topic may also be more likely to fill in the survey). ³⁸⁵
Nordic Council (Denmark, Finland, Norway, and Sweden)	2017	The Council conducted a study on how regulations for date labelling are applied in Nordic countries. They also interviewed food industry actors on whether they use best before or use by date, and how they determine food durability. ³⁸⁶	The study uncovered large variations in the durability estimates within product categories. Also, there were differences in the selection of the use by or best before date label within product categories. ³⁸⁶
UK	2018	Major supermarkets including Aldi, Sainsbury, Waitrose, Tesco, and Marks & Spencer independently decided to remove best before dates on fresh produce and changed use by to best before labels on milk and yoghurt. ³⁷³ A note added on Sainsbury's product lines includes "no date helps reduce waste". ³⁷³	The move by Tesco came after market research found 69% of customers would welcome a switch to products without best before dates and 53% said they would keep food longer if it did not carry a best before label. ³⁸⁷
US	2016	Laws vary state to state. The <i>Food Date</i> <i>Labelling Act 2016</i> was a bicameral bill proposed in both houses of Congress to provide federal oversight of date labelling, reducing the labels allowed and removing regulations that prohibit food banks' use of food past quality dates. ³⁸⁸ It was proposed again in 2021. ³⁸⁹	The bill has faced opposition from an industry group funded by the Grocery Manufacturers' Association, Food Marketing Institute, and National Restaurant Association. They instead advocate for voluntary initiatives by industry. ³⁹⁰
	2023	The proposal of the <i>Food Date</i> <i>Labelling Act 2023</i> aims to establish two standard date labels 'best if used by' for quality dates and 'use by' for discard or expiration dates and effectively end 'sell by'. ³⁸⁹	



Figure 25: A simplified depiction of the food supply chain highlighting the retail and food service stage.

In the food supply chain (see <u>figure 25</u>), 'retail and food service' (RFS) refers to consumer-facing businesses, like retail grocers, restaurants, foodservice providers, and various institutions, but can also include wholesalers and their distribution centres.³⁹¹ Typically, these businesses are divided into two groups:

- Retail: supermarkets, grocery stores, convenience stores, markets, and other outlets where consumers purchase food to prepare and eat at home. Here, we include wholesalers under retail, as these are often integrated into large supermarket operations.³⁹²
- Food service: establishments that prepare and serve food for immediate consumption, either on-site, takeaway, or through delivery. This includes restaurants, cafes, bakeries, fast food outlets, school and workplace cafeterias, hospitals, retirement homes, prisons, catering services, and other dining venues.³⁹³ Businesses that sell food and drink to customers, such as bars, restaurants, and cafes, are often referred to as 'hospitality' and are included under food service.

5.1 Understanding food waste in retail and food service

In the food service sector, food waste is typically categorised into three types of waste: spoilage waste, preparation waste, and plate waste.³⁹⁴ While these terms are not typically used to discuss retail food waste, some of these categories can have application to both RFS waste:

- Spoilage waste refers to food that becomes damaged or cannot be eaten because it's past its use by date and is potentially unsafe.³⁹⁴ Similarly, in the retail sector, food may decline in quality for a variety of reasons (for example, poor handling or stock rotation) and become unmarketable, even if it remains safe to eat.
- Preparation waste refers to food that is discarded during the preparation of food for sale or consumption. For example, preparation of waste in hospitality may include vegetable peelings and trimmings or offcuts of meat (see <u>figure 26</u>).³⁹⁴ This includes both edible waste (such as apple peel) as well as inedible waste (such as bones). While preparation waste is commonplace in the food service sector, it can also occur in some retail settings, for example, in in-house bakeries or delis.
- Plate waste refers to food that is served to customers but that is not eaten. Common examples of plate waste include servings of fries and garnishes.³⁹⁴ For simplicity, we include 'serving waste'³⁹⁵ food that is wasted after being prepared and offered to the consumer but not taken, for example, table bread rolls or buffet waste under the definition of plate waste. Plate waste is a category that generally occurs in the food service sector, although some grocery retailers may have in-house cafes that generate their own plate waste.



Figure 26: Quantifying data on food waste is a challenge for small food service providers, as evidenced by the preparation and plate waste generated at the Claris Sports and Social Club over the course of an evening.

5.1.1 Understanding supply chains

The retail supply chain in New Zealand

Much of the retail industry and supply chain in Aotearoa is dominated by two retail groups, a situation not dissimilar to Australia.^{*} Foodstuffs NZ and Woolworths NZ held 80%-90% of the market share between them between 2015 and 2020.³⁹² IBISWorld data indicates these two operators make up "almost 75% of revenue for supermarkets, grocery stores, and convenience stores" in New Zealand.³⁹⁶ Woolworths NZ is a subsidiary of the Australian Woolworths chain and operates Woolworths (currently rebranding from Countdown), Fresh Choice and Super Value brands.³⁹⁶ Foodstuffs NZ consists of two cooperatives, Foodstuffs North Island and Foodstuffs South Island, owned by the 523 operators of the constituent stores operating under New World, PAK'nSAVE, Four Square, Raeward Fresh, and On the Spot brands.

The grocery supply chain consists of suppliers, wholesalers, and processors (see <u>figure 27</u>). Wholesalers act as go-betweens, buying produce from suppliers and selling it on to grocery retailers.

However, in Aotearoa, Foodstuffs North Island, Foodstuffs South Island, and Woolworths NZ have their own integrated wholesale functions and distribution centres.^{392,397} This means that, unlike in many other countries, major grocery retailers are not overly reliant on independent wholesalers for their products.³⁹² This 'vertical integration' means that these retailers wield greater influence over the food supply chain, and provides opportunities to reduce waste across the system.



There are also independent wholesalers that encompass a broad range of businesses, including those specialising in fresh produce, meat, seafood, dairy products, and a variety of packaged foods and

^{*} In Australia, the two largest supermarket chains—Coles and Woolworths—account for at least 70% of packaged grocery sales and 50% of fresh produce sales, figures that have remained largely unchanged over decades.

beverages.²⁰⁵ In New Zealand, wholesalers cater to major grocery retailers, smaller grocery retailers, and the food service industry (see <u>figure 27</u>). For example, BidFood³⁹⁸ supplies a range of food and kitchen consumables to the food service industry, while T&G³⁹⁹ supplies a range of fresh produce to businesses in Aotearoa and around the globe. Smaller wholesalers might specialise in niche markets, such as organic or locally sourced foods, catering to specific consumer preferences.



Figure 27: A high-level summary of the supply chain for the New Zealand grocery sector, adapted from a Commerce Commission report on the retail grocery sector.³⁹² The summary has been updated to reflect a change in branding for Woolworths NZ and both Foodstuffs North Island and South Island.³⁹² Note, the report included food service wholesalers in the retail supply chain as some of these wholesalers also supply grocery retailers, although this is not the focus of their business. For an overview of the food service supply chain, see figure 28.

The food service supply chain

We don't have as clear a picture of the food service supply chain in New Zealand as we do for grocery retailers, but we can approximate its general structure based on similar work from the UK (see <u>figure 28</u>). Waste within the food service sector focuses on food service outlets, including both hospitality and various institutions (such as hospitals, prisons, and universities). However, as will be discussed in <u>section 5.2</u>, interventions in the food service sector have the potential to reduce FLW up and down the supply chain.


Figure 28: A generalised picture of the food service supply chain. Adapted from WRAP's 2013 report Overview of Waste in the UK Hospitality and Food Service Sector.⁴⁰⁰

5.1.2 What do we know about food waste in retail and food service?

How much waste is produced?

International evidence suggests food waste in RFS is produced in significant quantities. Estimates of food waste in different regions vary, in part due to differences in how waste is measured and/or allocated to different stages of the supply chain. The UN Environment Programme (UNEP) estimated that globally in 2022, 290 million tonnes of food were wasted in the food service sector, and 131 million tonnes in the retail sector.⁴⁰¹ In 2021, FIAL estimated that food wasted in wholesaling, retail, and food services accounted for 11% of all FLW along the Australian food supply chain, the majority of which was disposed of to landfills.⁵⁷ A 2022 estimate⁴⁰² for the US suggested that 9.5% of all FLW occurs within the distribution, wholesale, and retail stages of the US food supply chain, with a further 15% wasted in household and food service combined.

Extrapolating data from their global study released in 2024, UNEP estimates high-income countries like New Zealand waste 26 kilograms of food per person per year in the food service sector, and 13 kilograms per person per year in the retail sector.⁴⁰¹ This is consistent with a study undertaken in Aotearoa. A 2020 mixed-methods study estimated that food not sold or utilised at the retail level in

Aotearoa amounted to approximately 13 kilograms per capita per year, with 3 kilograms per capita per year being sent to landfill.⁶⁷ For context, this is just less than half of the estimated 32 kilograms per capita per year wasted in New Zealand homes in 2018 (see <u>section 6.1.1</u>).⁴⁰³ The retail study identified fresh vegetables (27%), bakery items (23%), meat and fish (19%), and fresh fruit (17%) as the major contributors to total discarded food. The research was informed by on-site food waste audits (16 stores), interviews, and analysis of existing data from



three major retailers (Woolworths, New World, and PAK'nSAVE) across four major urban centres. The study extrapolated this data across all stores to estimate that 60,500 tonnes of unsold or unused product are produced annually, some 14,000 tonnes of which go to landfill. The study was the first – and remains the only – to quantify food waste at a retail level in New Zealand. A potential limitation is that the sample of stores was self-selected by participating retailers, which could introduce bias if retailers suggested stores with better waste management practices and result in an underestimation of retail food waste. Moreover, the sample was not representative of all supermarkets in Aotearoa (for example, there were no supermarkets in rural locations, or smaller businesses represented) so caution must be used in interpreting data scaled to a national estimate.

Analysis of New Zealand's hospitality sector by WasteMINZ in 2018 found that cafes and restaurants generate an estimated 24,400 tonnes of food waste each year, with an unknown proportion going to landfill.⁴⁰⁴ Of this, 61% is considered avoidable waste, including preparation waste (60%), plate waste (33%), and spoilage (7%). The primary types of food wasted are vegetables (28%), bakery items (26%), meat (13%), and fruit (9%). Data informing the study were collected through bin audits of businesses (2017: five restaurants and four cafes in six different locations around the country; 2018: one restaurant and ten cafes in Auckland). Notably, the study audited twenty businesses for only a single day, limiting the precision of its findings. Additionally, the study did not cover food waste for other food service providers – such as pubs, bars, and hotels, or institutions like prisons, or retirement homes – about which little is known.⁴⁰⁵

Data from two New Zealand hospitals indicates that institutions also have work to do in preventing food waste: over a two-week period in 2014, an unnamed hospital served around 1750 meals per day and generated a total of 350 litres of food waste, while another unnamed hospital served 2,420 meals per day and produced 200 litres.⁴⁰⁶ Additional research on plate waste in New Zealand hospitals indicates that approximately 400 grams of food waste is produced per plate, equating to 31% of served food going to waste.⁴⁰⁷ Most food waste in hospitals is sent to landfill; food donation from hospitals are limited by hygiene and food safety constraints,⁴⁰⁸ and only 10% of hospitals compost their organic waste.⁴⁰⁷ Overproduction, inaccurate forecasting, and portion sizing were considered key drivers of this waste.⁴⁰⁶ To help understand the scale of food waste in other parts of the food service industry, MfE is funding research in the food service sector, with ongoing projects evaluating food waste in retirement homes and hospitality.

While we have some data on food waste in RFS in Aotearoa, we do not have a good understanding of its share of FLW along our supply chain.^{*} If these sectors follow patterns seen in Australia⁵⁷ or the UK,^{400,409,410} it's likely to be a small proportion of our total FLW, albeit still in significant amounts. However, this is only reflective of waste that occurs within the sector. Importantly, practices and

standards set within the RFS sector can have a significant impact on FLW patterns up and down the food supply chain (see <u>section 5.2</u>).

What are the drivers of food waste in the retail sector?

According to ReFED, food waste in the retail sector can arise from a combination of consumer-driven and operational factors.³⁹¹ For example, retailers can build in wasteful practices like overstocking shelves (see <u>box 6</u>). On the operational side, silos within retail supply chains

...practices and standards set within the RFS sector can have a significant impact on FLW patterns up and down the food supply chain.

can inadvertently exacerbate food waste, as different parts of a business fail to manage complex inventories and ensure older products are sold before newer ones (i.e. poor stock rotation). Consumer and operational factors can also interact, for example, customer demands for variety and consistency in food products can put strain on retailers' inventory management and food purchasing. Moreover, cosmetic standards on produce, determined by retailers and reinforced by consumer expectations for freshness, can lead to perfectly good food being thrown out simply because it's nearing its best before date or doesn't look appealing.⁴¹¹ ReFED estimates that concerns around date labels account for more than 50% of food waste at the retail stage of the US food supply chain.³⁸⁹ (See section 4.2.3 for an overview of date labelling practices and potential solutions.)

Sox 6: Shelf overstocking – a retailer's dilemma

Retailers face a trade-off in how they stock their shelves, with 'overstocking' a key problem. Simply put, shelf overstocking is when retailers stock more products on store shelves than can be sold before products reach their expiration or become less desirable to consumers (see <u>figure</u> 29). Driven by competition, many retailers prioritise product availability and diversity, thus overstocking shelves to drive sales.⁴¹² However, while increasing a store's attractiveness to consumers, overstocking brings with it negative effects. Economically, overstocking can cause a direct financial loss due to unsold inventory[†] as well as additional costs for handling, passing on, and/or disposing of food waste. Environmentally, overstocking drives the waste of food itself, and in turn the resources used to produce, transport, and store overstock food. As highlighted in <u>table 15</u>, there are several interventions that can help reduce excess food in supermarkets, although reactive initiatives like price promotions are likely insufficient by themselves.⁴¹² Instead, more nuanced and demand-driven approaches such as incorporating better demand forecasting, tailoring product ranges to better match sales data, and varying stock availability among different products, are required to design out waste associated with a one-size-fits-all approach to stocking.⁴¹²

^{*} UoO (commissioned by MfE) are conducting a national baseline study of FLW; results are forthcoming.

[†] Depending on specific terms of trade agreements, sometimes this economic impact can fall on the producer or manufacturer of foods rather than the retailer, as is commonly observed in the bread industry (see <u>section</u> <u>2.1.2</u>).



Figure 29: While overstocking can make shelves look attractive, doing so for highly perishable items (for example, bread and other bakery items, minced meat, fish, takeaway snacks, and some fruits and vegetables) can exacerbate the dilemma of appealing to customers while trying to avoid food waste.

A recent systematic review of food waste causes and reduction practices in the (global) retail sector suggests that issues with procedures, equipment, and materials used in the industry, inefficient labour practices and problems with waste measurement, as well as wider challenges in the retail environment entrench food waste in the sector.⁴¹³ The drivers of food waste within these groupings are presented in <u>table 9</u> below.

Table 9: A diverse range of factors can generate food waste in the retail sector, presented here as identified and grouped by De Moraes et al.⁴¹³ Groups are ordered as listed in the study, giving weight to the frequency with which these groups are discussed in academic literature. It is worth noting that these drivers are reported from a generalised context from a broad review of academic literature on retail food waste; there is no equivalent research specific to New Zealand. Abbreviations: IT = information technology, NGOs = non-governmental organisations.

Group	Food waste driver
Environment.	 Overly restrictive laws, including general commercial standards, as well as hygiene and food safety standards. Uncertainties in supply chains caused by climate change; for retailers this creates issues with demand forecasting and food availability. Problems with seasonal foods that influence demand and supply, as well as changing specifications in relation to seasonal changes. People's eating habits can drive food waste. For example, unpredictable changes in buying preferences can result in unforeseen changes to demand. Additionally, the size of foods available for purchase for small or single-person households can influence waste in the home.

Group	Food waste driver
	• Excessive stakeholder pressure on retail shares by government, legislators,
	NGOs, and shareholders, which can increase or reduce food waste.
Equipment.	 Cold chain gaps, i.e. the lack of structure and equipment needed to maintain (lower) temperatures for perishable foods over a long period of time. This is made more complex by different food types requiring different temperatures. Poorly maintained, outdated, or poor in technical condition transport equipment that causes damage to food as it moves within retail supply chains, for example, from wholesalers to retailers.
	 Storage that is not strategically located, lacks capacity, and/or does not have adequate cooling and ventilation. Ineffective and problematic display of products on shelves. For example, stacking large piles of certain fruits can result in physical damage to food.
Human factors.	 Incorrect handling, including failures to package food and over-handling fruit and vegetables. Lack of training and/or knowledge on food waste mitigation practices. Lack of commitment to food waste practices.
Materials.	 Inadequate packaging, including during transport and storage (see section 4.1.2) as well as confusing and incorrect labelling of expiry dates (see section 4.2.3). Short shelf life, in particular for fruit and vegetables.
Measurement.	 Inadequate demand forecasting systems, i.e. the inability to accurately predict demand and supply, in particular for foods with a short shelf life. Overstocking inventory to cover uncertainties. Lack of (standardised) waste measurement. Sudden changes in orders, in particular for promotions that drive unpredictable demand and waste.
Procedures.	 Stringent specifications of appearance and shape of food, particularly fruit and vegetables (see section 2.1.2). Poor management and rotation of inventory. Lack of coordination and collaboration within retail supply chains. Lack of information sharing, including not sharing data openly and not adopting advanced demand forecasting techniques. Inappropriate procedures to manage food waste, for example, unclear approach to food donation. Lack of integrated IT systems, resulting in failure to share and quickly communicate information between supply chain links.

What are the drivers of food waste in the food service sector?

Like the retail sector, food waste in the food service sector is a complex issue influenced by a range of factors, spanning societal norms, business styles, procurement strategies, customer expectations, and more.⁴¹⁴ Key drivers include operational practices within the food service establishments, such as overproduction of food, poor storage and handling techniques, and large portion sizes that often result in leftovers.⁴⁰⁰ ReFED estimates that some 70% of food waste in the US food service sector is plate waste from consumers who don't eat all that they're served, or take from a buffet.⁴¹⁵ This may be lower in Aotearoa, with plate waste accounting for 33% of food waste in restaurants and cafes, as per data from a single study.⁴⁰⁴ Consumer behaviour and expectations also play a significant role, where customers' specific preferences and demand for variety and aesthetics in food presentation

can lead to more waste.⁴¹⁴ Furthermore, outlets like restaurants or cafes in the sector face the challenge of balancing menu complexity, portion sizes, and food waste.⁴⁰⁰ A wide enough range of menu items is needed to satisfy and attract customers, but can increase the likelihood of waste due to the need for more diverse ingredients, especially if outlets can't accurately predict demand for different menu items.⁴⁰⁰

Another significant factor is a lack of planning and inventory management, leading to over-ordering or overstocking, which can result in food spoilage.⁴¹⁶ Training and awareness of staff in food service establishments are crucial as well, as inadequate knowledge and skills in food waste reduction can exacerbate the problem. Regulatory constraints and health concerns sometimes lead to more food being discarded as businesses err on the side of caution.⁴¹⁴ Lastly, a lack of food waste tracking and measurement makes it difficult for businesses to identify and effectively address specific areas where waste is occurring.⁴¹⁰

5.1.3 What are the gaps in our knowledge?

There's a lot we don't know about food waste and prevention efforts in our RFS sectors. This means we have knowledge gaps that span a range of issues, including but not limited to:

- Much of our understanding of food waste volumes, types, and potential interventions in the RFS sector comes from examples and research overseas. Thus, in many instances we lack context specific data and must rely on inference to understand the potential effect of interventions on food waste reduction efforts in the sector.
- Generally, research in the sector has been more focused on considering the causes of food waste than on reduction practices.⁴¹³
- Much of our local understanding of food waste in the New Zealand RFS sector comes from a study of big retailers,⁶⁷ a hospitality-focused study on twenty restaurants and cafes,⁴⁰⁴ and some research into hospital food waste.^{406,407} There is little New Zealand specific data about food waste and/or reduction practices in small retailers, independent wholesalers, and a wide variety of food service providers (for example, bakeries, meal delivery services, meal kit companies, and various public institutions.
- Other than for some of our larger retailers,⁶⁷ we have little idea of where food waste in the sector is being diverted to.
- We do not know how much the RFS sectors waste food in relation to other parts of the supply chain, nor do we know the extent to which their practices (for example, cosmetic specifications and ordering contracts) cause FLW upstream in the supply chain.

5.2 How can we prevent food waste in retail and food service?

Despite challenges in assessing their efficacy, we have taken a deep dive into the interventions touted as useful for FLW prevention in the RFS sectors (see <u>section 5.2.1</u> and <u>annex 3</u>). Because RFS actors are uniquely positioned in the supply chain, bridging producers and consumers, their ability to influence and reduce FLW extends beyond their own sector. This is particularly true of retailers and wholesalers, whose practices and policies dictate the flow of significant amounts of food along the supply chain,³⁹² and can affect how food is produced, processed, distributed, and consumed (see <u>section 1.2</u>). Later in this section, we discuss interventions for the RFS sectors that may have positive effects on FLW elsewhere in the supply chain.

Using our case study of the tomato supply chain (see <u>case study 1</u>), we can explore the potential for retailers to influence FLW upstream in production (see <u>section 3</u>) and processing and manufacturing (see <u>section 4</u>), in their own sector, and downstream. Upstream, retailers can mitigate waste by selecting suppliers with accredited sustainable farming practices and creating transparent supply

contracts where tomato waste is a mutual consideration, employing more accurate demand forecasting to avoid over-ordering, and accepting cosmetically imperfect but edible tomatoes. Instore, effective strategies to reduce tomato spoilage include implementing a 'first in, first out' (FIFO), or alternatively 'first expired, first out' (FEFO)⁴¹⁷ (see <u>box 7</u>) system to ensure older tomatoes are sold first, handling tomatoes with care to avoid bruising and soft spots, maintaining optimal storage conditions, and promoting or discounting tomatoes that are near the end of their shelf life. For downstream waste mitigation, retailers can provide in-store education to consumers on typical shelf life, proper storage, and tips for using tomatoes in the household.

Sox 7: First in or first expired? Comparing FIFO and FEFO approaches to inventory management

FIFO and FEFO are two key inventory management strategies that differ primarily in their application and the nature of the inventory they manage.^{418,419} FIFO is typically used where products do not have an expiration date and is aimed at moving the oldest continuous stock first, thus ensuring that the inventory remains fresh and relevant. This method is particularly useful for non-perishable goods such as canned goods or household items.⁴¹⁹ It also helps in maintaining a lower inventory cost and less obsolescence since products that come in first are sold first.

FEFO, on the other hand, prioritises the dispatch of products based on their expiration dates and is crucial for industries like food and pharmaceuticals, where selling products past their expiration can lead to health risks and legal issues. By prioritising products based on expiration rather than purchase date, FEFO helps grocery retailers and food service providers manage products for both consumer safety and regulatory compliance.⁴¹⁸ In addition, FEFO inventory management has significant potential to reduce wasted food, and concurrently boost sales and enhances customer trust in fresh food options.⁴¹⁷ However, caution is needed not to create perverse incentives to label fresh produce with a short expiry date.

In short, both methods aim to optimise inventory turnover and reduce costs associated with excess, outdated, or expired stock. However, the choice between FIFO and FEFO largely depends on the specific needs of the business, including the type of products handled and their respective shelf lives.

5.2.1 There are interventions that can prevent food waste within the retail and food service sectors

Interventions that prevent and reduce food waste within the RFS sectors target spoilage, preparation waste, and plate waste – the most common types of waste in the sector. Because the sector is so varied, both in the businesses that it comprises and the opportunities for waste, interventions to prevent food waste are wide-ranging (see annex 3).

A detailed list of potential interventions to prevent food waste and the evidence for each is presented in <u>annex 3</u>. Note that much of the evidence for interventions in RFS provided in this table is international research and not specific to the Aotearoa context. As explained in <u>section 5.1.2</u>, New Zealand-specific research in this part of the supply chain is scarce. This is something we should aim to improve but should not let this delay action based on international evidence, ensuring robust evaluation of interventions.

According to ReFED data from the US (see <u>figure 30</u>), one of the most effective and financially beneficial approaches for retailers to reduce food waste is to prioritise sales of products nearing

expiry.⁴¹⁷ This includes using markdown alert applications or distress sale platforms and dynamic pricing of goods in accordance with their remaining shelf life. These strategies are also available to food service providers, with apps like Footprint⁴²⁰ in Aotearoa and ResQ Club⁴²¹ in Europe highlighting perfectly good food available at reduced prices from various local eateries. We discuss these interventions in more detail in <u>annex 3</u>, with available evidence suggesting these solutions can be highly effective in reducing food waste.

...one of the most effective and financially beneficial approaches for retailers to reduce food waste is to prioritise sales of products nearing expiry.

In addition to prioritising sales, a key operational step to minimising food waste in RFS environments is efficient inventory management. In a perfect world, the full inventory of food bought from suppliers by RFS would be sold on to consumers, in optimal condition. While perfection may be a stretch, a range of solutions for inventory management can help reduce food waste as food moves from producers to consumers. Advanced demand forecasting (sometimes called enhanced demand planning), which typically employs ML to predict consumer demand more accurately, has potential to significantly reduce spoilage waste in RFS sectors.⁴²² Pairing this approach with inventory management strategies like FEFO,⁴²³ as well as optimised storage and transport (see <u>section 2.1.1</u>) can provide further improvements to food waste reduction efforts and provide significant financial benefits to RFS sectors. For example, in the US retail sector, ReFED estimates that retailers could divert around two million tonnes of food from waste annually, with a net financial gain of some US\$5.5 billion if they enhanced product distribution methods and refined inventory management systems (see figure 30).⁴²⁴ As further discussed in annex 3,

available evidence for the effectiveness for such interventions is largely positive, although data is patchy.

Beyond managing their inventory effectively, food service providers can implement interventions that target preparation and plate waste. Providing diners with doggy bags, changing portion sizes, adjusting plate sizes, and tweaking the dining set up in buffet style services can all reduce plate waste, while improving menu planning and tailoring staff training can reduce preparation waste (see annex 3). ...ReFED estimates that retailers could divert around two million tonnes of food from waste annually, with a net financial gain of some US\$5.5 billion if they enhanced product distribution methods and refined inventory management systems.



Figure 30: The potential for annual food waste diversion (kilotonnes – green bars, left axis) and annual net financial benefits (US dollars, in millions – grey bars, right axis) of food waste interventions in the US retail sector, as estimated by ReFED.⁴²⁴ Data source: the ReFED Solutions Database.⁴²⁴ Data was sorted into three general categories of intervention (inventory management, sales strategies, and transport and logistics). Abbreviations: kt = kilotonnes, USD = US dollars.

The potential of a given intervention to prevent food waste cannot be narrowly measured in terms of 'effectiveness' – whether it works – but also in terms of the real world conditions, which will shape an intervention's implementation and impact. In 2018, ReFED considered five criteria in their assessment of interventions for retail (see figure 31) and food service (see figure 32) in the US: profit potential (expected net annual profit, not including upfront investment costs); feasibility (implementation effort and initial capital requirement); industry prevalence (estimated percentage of retail providers that have implemented a solution); diversion potential (portion of all food waste – by weight – that could be diverted from landfill through implementation of the solution); and economic value (the annual aggregate financial benefit of a solution to society minus all investment and costs). As figure 31 and figure 32 show, no interventions score highly on all criteria. For example, in 2018, enhanced demand planning had significant potential to improve profits and divert food from landfill, but was difficult to implement and had a low industry prevalence in the US (see <u>annex 3</u>). Conversely, changes to cosmetic specifications were deemed highly feasible, but had relatively low potential to improve retailer and food service profits, with a moderate predicted impact on food waste reduction. Importantly, ReFED's assessment is context and time specific. For example, enhanced demand forecasting was indicated to have low feasibility based on ReFED's 2018 data, but several retailers have started using it successfully in 2024.⁴²⁵ ReFED's Insights Engine⁴²³ provides a range of up-to-date impact metrics for solutions like enhanced demand planning, although it does seek to not quantify their feasibility.

Importantly, retailers and food service providers alike may struggle to prioritise interventions for food waste reduction because they lack quantifiable information about where and why food waste is occurring. As highlighted in <u>section 2.2.2</u>, improvements and innovation in supply chain digitalisation and data collection can help identify waste hotspots and lead to better insights around the flow of food, and waste, within supply chains. For example, within food service, identifying and tracking the types and volumes of food that are disposed of during preparation can inform and optimise operations, increase profits, and help form a business case for investment in other food waste solutions.⁴¹⁵ As is true across the supply chain, without robust measures of food waste, it's difficult to

select or design effective interventions for food waste reduction, set credible goals, or monitor progress.⁴²⁶ By enabling tailored interventions, measuring waste has enormous potential to reduce food waste. For example, in the US food service industry, ReFED estimates that technology-enabled tracking of food waste could help divert some one million tonnes of food waste from landfill, with a net financial benefit of almost US\$4 billion.⁴¹⁵ Additionally, in the retail sector, tracking waste is key to evaluating key performance indicators that target food waste reduction.⁴²⁷



Figure 31: A 2018 ReFED assessment of solutions to food waste in the US retail sector, according to five criteria: profit potential, feasibility, industry prevalence, diversion potential, and economic value.³⁹¹ Note, in this report, we introduce concepts of produce specifications in <u>section 2.1.2</u>, packaging solutions in <u>section 4.2.2</u>, and standardised date labelling in <u>section 4.2.3</u>. Image credit: ReFED.

In many instances, interventions to reduce food waste in retail stores and food service providers are complementary. For example, reducing minimum order quantities⁴²⁸ can be strategically coupled with increased delivery frequencies, especially where dwell times are reduced.⁴²⁹ Similarly, efforts to pair improved demand forecasting with dynamic pricing are sensible; the former can help stores get

just the right amount of food into stores, while the latter is a failsafe for when forecasting gets it wrong and too much food is on hand.³⁹¹ The same is true for interventions to shape consumer expectations; making people more aware of food waste, giving them tips on how to avoid it, and simultaneously changing the way food is presented in stores to influence buying habits can make a difference.⁴³⁰



Figure 32: A 2018 ReFED assessment of solutions to food waste in the American food service sector, according to five criteria: profit potential, feasibility, industry prevalence, diversion potential, and economic value.⁴¹⁶ Note, in this report we introduce concepts of produce specifications in <u>section 2.1.2</u>. Image credit: ReFED.

5.2.2 Preventing FLW upstream: the role of retailers and food service

The RFS service sector can inadvertently cause FLW earlier in the supply chain through several mechanisms. These practices, often rooted in operational efficiencies and market expectations, can have a cascading effect on suppliers, leading to inefficiencies and excess waste well before products reach the consumer. While a host of different drivers can cause food waste within RFS sectors (see <u>section 5.1</u>), here we discuss two broad areas for intervention – cosmetic specifications and ordering practices – and their effects on FLW upstream in the food supply chain.

Cosmetic specifications

One of the primary ways retailers and food service providers contribute to upstream FLW is through stringent cosmetic specifications for produce (as introduced in <u>section 2.1.2</u>).

Several studies indicate that cosmetic specifications can contribute substantially to FLW in retail and production alike. For example, a 2018 study of EU retailers and producers estimated that producers discarded roughly half of their out-of-spec produce, while waste within retailers varied substantially.⁷⁶ German retailers assessed that 2% to 40% of delivered fruits did not fulfil their cosmetic specifications, while in the Netherlands, retailers assessed their own wastage to be extremely low (1%).⁷⁶ However, Dutch producers suggested their on farm losses due to cosmetic specifications ranged from 2% (greenhouse production of tomatoes) to 25% (open field production of lettuce and cabbage).⁷⁶ Another 2018 study of food loss in fresh fruit and vegetable production in

the UK and Europe estimated that over a third of total farm production is lost for 'aesthetic reasons', with on farm cosmetic rejections of up to 4,500 kilo tonnes per year in the UK and 51,500 kilo tonnes per year in the European Economic Area⁷⁷ (see <u>section 3.1.3</u>). While these studies highlight that cosmetic specifications may cause FLW, they did not test the effectiveness of broadening cosmetic specifications as a tool to reduce FLW. In 2009, the EU removed cosmetic specifications for 26 of 36 product types, a decision which had reportedly little effect on FLW because retailers retained their own cosmetic specifications⁴³¹ to satisfy assumed consumer expectations. In a New Zealand context, where the Government regulates standards around food safety and packaging but not cosmetic appearance,³⁵⁵ changes to cosmetic specifications will likely need to be led by retailers and require buy in from all actors along the supply chain.⁷⁷

Retailers are not the sole creators of specifications (see <u>box 2</u>), but their unique position in the supply chain means they have considerable potential to reduce FLW driven by cosmetic specifications. Retailers can help increase consumer acceptance criteria for produce, promoting and marketing 'imperfect'^{*,432} fruits and vegetables to consumers (see <u>case study 11</u>). This can help prevent food loss on farms, as supermarkets buy a wider range of products. Taking on a greater variety of produce has implications for sales strategies, with novel approaches needed to ensure that FLW isn't simply passed from farm to retailer. Initiatives to market imperfect produce, such as

Woolworths NZ's 'Odd Bunch',[†] coupled with in-store consumer education and competitive pricing could be helpful tools in changing customer purchasing patterns,^{433-⁴³⁶ and in turn reduce FLW along the supply chain. A range of food service providers also use and market imperfect produce (see sections 2.2.4 and 3.1.3). These include retail produce box companies like Perfectly Imperfect,²⁸ Wonky Box,²⁹ and Farmers Pick,⁴³⁷ as well as customers in the food service sector,⁴³⁸ and others getting in on the act.⁴³⁹ In many instances, imperfect foods are well-suited to the food service sector because they are flavourful yet}

Retailers are not the sole creators of specifications, but their unique position in the supply chain means they have considerable potential to reduce FLW driven by cosmetic specifications.

cost-effective and are particularly useful for dishes where the appearance of ingredients is less important, such as soups, salads, or sauces.⁴⁴⁰ Relaxing specifications is not without its complications for retailers, with considerations around brand image, operational adjustments, and food safety misconceptions. However, emerging evidence suggests that retailers who do broaden their cosmetic specifications can be successful in selling imperfect produce, especially if retailers can overcome consumer concerns of quality and appropriate pricing.^{441,442}

Supply chain coordination: Leveraging analytics and collaborative relationships

As discussed in <u>section 2.1.1</u>, ineffective coordination across the supply chain is a major factor that stimulates food overproduction and can see food lost before it reaches market or result in spoilage due to misaligned delivery and usage timelines. A range of practices drive this waste,⁴⁴³ including analytical factors like inaccurate demand forecasting, as well as ordering practices that build in waste, including inflexible orders, short lead times, last-minute cancellations, a lack of commitment

^{*} Here the term imperfect is used to describe foods that deviate in appearance from consumer expectations, as influenced by cosmetic specifications. In academic literature, these foods are often called 'suboptimal'; an umbrella term, which describes food that consumers perceive of lesser value than other items of the same kind. Foods can be considered suboptimal for a variety of reasons: they're nearing their indicated expiration or best before date, they deviate in appearance (i.e. imperfect foods), or their packaging is damaged.

[†] Critics of this scheme note that that this line of produce is only available as a bulk buy, a marketing approach that can promote food waste in the home by encouraging over purchasing.

to contracts, and terms of trade like take-back agreements⁷² (see <u>section 2.1</u>). The power imbalance between smaller suppliers and the large retail chains (and their associated wholesale companies) is a significant contributor to poor ordering practices.³⁹²

Poor demand forecasting is largely a technical issue, and one that can be improved by leveraging better data analytics to predict demand and in turn enable more accurate ordering. As highlighted in <u>annex 3</u>, employing analytics to improve demand forecasting has considerable potential to alleviate food waste in the retail sector, although it's not straightforward to implement and is not widely practiced (see <u>figure 32</u>). Additionally, where demand forecasting tools are used, preventing FLW would be enhanced by sharing the information gained from such tools with suppliers. While issues

around order flexibility, timing, and contracts could be aided by technical improvements like supply chain digitalisation (see <u>section 2.2</u>), the crux of these issues come down to the nature of relationships between suppliers and retailers and/or food service providers.⁴⁴⁴ Given that a lack of coordination and collaboration between suppliers and retailers is one of the most cited causes of food waste,⁴¹³ this is clearly not a straightforward fix. Establishing stronger, more collaborative relationships with suppliers can encourage

...where demand forecasting tools are used, preventing FLW would be enhanced by sharing the information gained from such tools with suppliers.

more realistic lead times and contract terms,⁴⁴³ reducing the pressure to overproduce (see section 2.1 and section 3.1). Food retailers can also be more systematic in sharing forecast data for specific food items with farmers to optimise their production plans (see section 2.2.2).⁴⁴⁵

In addition, some authors have argued that retailers should take more accountability for FLW caused by their ordering practices through, for example, take-back agreements and shared financial risk of food losses with producers.⁴⁴⁶ There are examples of these arrangements in practice: a UK retailer guaranteed 98% of their orders in banana production, building tolerance into contracts that account for production delays and overproduction.⁴⁴⁷ Similarly, Sainsbury's in the UK has established a close working relationships with sheep and dairy farmers, resulting in practices like 'cost of production' pricing models that help avoid on farm losses of livestock, reflect the real costs of farming operations, and builds in profit.⁴⁴⁵ In Aotearoa, the recently introduced Grocery Supply Code (see box 8) has potential to bring more certainty and transparency to supplier-retailer relationships. It remains to be seen if the code has a positive effect on FLW reduction efforts.

Q Case study 11: How Tesco collaborates with suppliers to reduce food loss and waste

Retailers can collaborate with farmers to develop clearer and more flexible procurement policies to reduce pre-retail FLW, ensuring a market for a wider array of produce, providing growers with more certainty over orders, and prioritising more seasonal and local produce.448 UK retailer Tesco has openly made food waste reduction part of its wider business strategy, becoming the first UK retailer to report their food waste statistics publicly.⁴⁴⁹ Tesco has implemented several initiatives with suppliers to reduce pre-retail waste. For example, Tesco's 'Perfectly Imperfect' line of fruit and vegetables (see figure 33),449 launched in 2016, has provided an avenue to market for a greater cosmetic variety of produce. According to Tesco, this strategy has seen a significant reduction in food waste, with the 2024 estimate suggesting that 68 million packs of fruit and vegetables were saved from going to waste.⁴⁴⁹ Between 2016 and 2021, the company claims that 44,000 tonnes of produce were sold through the initiative, with potatoes (12,600 tonnes), carrots (10,500 tonnes), apples (8,700 tonnes), and strawberries (8,500 tonnes) being the most popular.⁴⁵⁰ In addition to purchasing and selling imperfect produce, Tesco has also worked with suppliers to manage and purchase bumper crops – a situation that arises when farmers produce more than they were expecting. For example, a UK heatwave in 2022 saw an unexpected surge in local strawberry and cherry production, with Tesco purchasing the excess to sell kilo boxes of summer fruits at discounted prices.⁴⁴⁹



Figure 33: Tesco's Perfectly Imperfect line of fresh produce comprises a variety of fruits and vegetables, such as strawberries (left) and carrots (right). Image credit: Tesco.

In addition to flexible purchasing, Tesco launched an online marketplace in 2022, called 'Tesco Exchange',⁴⁵¹ that matches suppliers who have too much product with other Tesco suppliers that need product. The idea behind the platform is to enable some 3,500 Tesco suppliers to cut production costs and reduce waste by selling or donating stock that may previously have gone to waste. At time of writing, data pertaining to the amount or types of stock sold on the platform have not been published in recent Tesco reports.⁴⁴⁹

Box 8: The Grocery Supply Code

A notable recent development for the retail supply chain was the introduction of the Grocery Supply Code (the Code) in 2023; a piece of legislation designed to help regulate relationships between suppliers and retailers.⁴⁵² While the Code is not designed around FLW specifically, its promotion of fair, certain, and transparent agreements between suppliers and retailers could influence, and hopefully benefit, FLW prevention efforts. For example, the need for retailers to provide 'clear and reasonable' specifications for fresh produce, documented reasons for rejecting produce, and written agreements over quantity and quality requirements could provide suppliers with more certainty in their trade dealings and production needs. Similarly, the Code's requirement for retailers to provide suppliers with adequate notice for delisting products or creating promotions, could help suppliers with their forecasting efforts and help cut back on FLW. However, given the Code only came into effect in late 2023, we don't yet know it's impact on FLW prevention within the retail supply chain.

5.2.3 Preventing food waste downstream: the role of retail and food service

Food waste in households: Opportunities for retailers

Retailers, given their direct and frequent interactions with consumers, can influence and prevent food waste in households through a combination of strategies aimed at educating consumers and modifying purchasing patterns.

A frequently discussed approach for retailers involves in-store education around making the most of food, with in-store signage, online resources, pamphlets, store magazines, on-pack stickers, and organising in-store events all commonly used approaches.⁴⁴⁵ Evidence from the UK suggests that communication from retailers to their customers around food waste can have a significant effect on

reducing food waste of consumers,⁴⁵³ with consumers looking to retailers for guidance and to set an example.⁴⁴⁵ By providing clear, accessible information on food preservation, storage, and waste reduction, retailers can empower customers to make more informed decisions that align with FLW goals. In <u>section 4.2.3</u>, we discuss some of the labelling options that can make this type of information accessible. It's also important to consider different types of customers because people have various reasons for wanting to reduce food waste. Understanding these differences can help come up with targeted strategies that are more likely to work.⁴³³

As discussed in <u>section 4.2</u>, there are many ways that improvements to packaging, and packaging information can prevent food waste. Retailers can make impactful changes by optimising date labelling practices, enhancing packaging design to extend the storability of food, and providing on-pack information that helps consumers improve their food management capabilities.⁴⁵⁴ For example, retailers like Tesco in the UK have already reformed many of their date labelling practices, removing best before dates from fruit and vegetable lines,⁴⁵⁵ and removing use by dates from yoghurt lines.⁴⁵⁶ To our By providing clear, accessible information on food preservation, storage, and waste reduction, retailers can empower customers to make more informed decisions that align with FLW goals.

Retailers can make impactful changes by optimising date labelling practices, enhancing packaging design to extend the storability of food, and providing on-pack information that helps consumers improve their food management capabilities. knowledge, the effect of these changes on consumer behaviour has yet to be evaluated. Retailers also have the potential to influence choices consumers have about amounts purchased. Given that larger food packages sold in retail settings can lead to more food waste in households, offering smaller packages could help prevent food waste.⁴⁵⁷ Retailers can also inadvertently cause customers to choose larger packaging over smaller alternatives in-store, with practices like "buy two – pay less" or the display of price per kilogram ratios potentially leading to consumers buying more or a larger package than they actually need.⁴⁵⁷ Although it is often package designers at processors and manufacturers who make decisions around package sizes, given the considerable market power wielded by major supermarkets In New Zealand, these operators could hold sway in advocating for changes to packaging that prevent further food waste.

Altering pricing strategies is another avenue through which retailers can discourage the overbuying of food. By moving away from promotions such as 'buy one, get one free', which often lead to excessive buying and eventual waste,⁴⁵⁸ retailers can adopt pricing models that incentivise the purchase of quantities that are more in line with consumer needs. Tailoring prices to discourage the overbuying of perishable products, for instance, has potential to significantly cut down on food waste at the consumer level.

Lastly, offering ready-made meals and recipe suggestions for products nearing their use by dates can encourage the use of food that would otherwise be discarded. This not only helps in reducing waste but also provides convenience for consumers looking for quick meal solutions.

Q Case study 12: Food marketing in retail – reaching consumers on food waste

A 2018 study by researchers from Scandinavia and the Netherlands highlighted how the actions and marketing strategies of retailers influence consumer behaviour around food waste.⁴⁵⁴ The study introduces three case studies to show how three different retailers changed their practices to reduce food waste in conjunction with their customers:

ICA Resurskocken, Sweden

ICA Sweden, with a significant market share, introduced an in-store kitchen at their ICA Tuna store in Lund in 2007, utilising products that might otherwise be wasted. The kitchen aimed to offer quality meals at lower prices, largely targeting students and academics. The initiative helped reduce food wastage significantly, improved employee involvement, enhanced the store's image towards quality and sustainability, and received favourable media attention.

Rema1000, Denmark

A Norwegian retail chain operating in Denmark, Rema1000, decided to phase out multi-item offers (for example, 'buy one, get one free') in 2008 to reduce food waste. Initially, sales dropped in categories previously promoted through multi-item offers. However, this action received positive feedback from customers and improved the retailer's reputation, potentially influencing competitors to adopt similar strategies.

Intermarché, France

In 2014, Intermarché, a large French retailer, launched a campaign promoting the sale of imperfect fruits and vegetables at a 30% reduced price. The campaign aimed to increase awareness about food waste. The campaign increased store traffic, especially in the fruit and vegetable section, and led to higher sales volumes. It also prompted competitors to undertake similar actions and received extensive publicity and social media attention.

In Aotearoa, an uptick in ready-made meals sold by major retailers can be partially attributed to the rise of meal kit providers like My Food Bag, Woop, Bargain Box, and Hello Fresh.³⁰ While the Commerce Commission has classed meal kit providers as retailers,³⁹² these businesses straddle the

line between being retailers and food service providers. On the one hand, they sell raw food products (and sometimes other groceries) directly to consumers, much like a retailer.^{*,392} On the other hand, they share characteristics with food service providers because they handle, prepare, and portion food ingredients, which enables easier at home cooking. Understanding both of these characteristics can help us think about food waste in relation to meal kit providers. There is little known about food lost during meal kit processing and packaging;⁴⁵⁹ we explore food waste within households using these services in section 6.1.2.³⁰

Food waste in households: the role of food delivery services

Getting takeaways has always been a way to bring hospitality into the home. In 2016, the dominant form of food delivery globally (approximately 90% market share⁴⁶⁰) was via the 'traditional' model, whereby the consumer places an order with their local pizza place or restaurant and waits for the food service provider to deliver food to the door. However, the emergence of third party delivery apps has been reshaping this market, with food delivery apps and platforms (henceforth 'food delivery services') taking up an increasing market share.⁴⁶¹ From a food waste perspective, this likely shifts more waste from restaurant plates into households. While we discuss interventions to prevent household food waste in detail in <u>section 6.2</u>, here we highlight how food delivery services could influence patterns of household waste.

Food delivery services have been a growing segment of the hospitality industry globally⁴⁶² and in New Zealand.⁴⁶³ In New Zealand, a number of digital platforms enable delivery of restaurant foods to households, with competition among apps like Delivereasy, DoorDash, Menulog,⁺ Uber Eats, and Yummi.⁴⁶⁴ As discussed in <u>section 5.1.2</u>, the food service sector generates food waste in New Zealand, but we know little about the relative contribution of different food service providers to our FLW problem. For food delivery services, this data gap reflects a global lack of study around food waste in this sector.⁴⁶⁵

The extent to which food delivery services contribute to overall FLW is a matter of debate, as it is not clear whether household food waste is counteracted by improved preparation efficiencies within restaurants or other food service providers.⁴⁶⁶ On the one hand, researchers suggest that food

delivery services can cause consumers to order more food than necessary. Over-ordering is influenced by several factors,⁴⁶⁶ including the ease of access and variety of options available, promotional strategies and discounts used by food delivery apps (for example, setting a minimum price for free delivery)⁴⁶⁷ and a lack of physical cues that might deter over purchasing in a traditional dining or shopping setting (i.e. consumers do not see physical portion sizes clearly when ordering digitally),⁴⁶⁵ as well as positive attitudes and social approval towards these apps.⁴⁶⁶ On the other hand, food delivery services may reduce preparation waste within service providers, as



The extent to which food delivery services contribute to overall FLW is a matter of debate, as it is not clear whether household food waste is counteracted by improved preparation efficiencies within restaurants or other food service providers.

^{*} According to the Commerce Commission, while major retailers perceive meal kit companies like My Food Bag and Hello Fresh as competitors, their respective turnovers are orders of magnitude apart, with retailers dominating the market. In late 2023, My Food Bag's annual turnover was an estimated \$175 million, a fraction of Foodstuffs North Island's \$4.3 billion for the same time period. The Commission suggests that meal kit providers are unlikely to provide strong competition to retailers for consumers' total grocery spend in the foreseeable future.

⁺ Founded in Australia in 2006, Menulog operated in New Zealand from 2012 to 2024.

apps enable more accurate order matching based on real time availability of menu items, better demand prediction from consumer preference and ordering pattern data, personalised recommendations, and optimised delivery schedules.⁴⁶⁸

Given the relatively recent emergence of food delivery services and concomitant lack of study, we are yet to have a good understanding of FLW related to these services, making it difficult to design interventions. This is a feature of FLW across many parts of the supply chain (see section 2.3.1). As such, there is a clear need for further study of FLW within and caused by food delivery services.



...there is a clear need for further study of FLW within and caused by food delivery services.

5.2.4 Prevention efforts: involving retailers and food service providers in system-wide initiatives

As we introduced in <u>section 2.2.6</u>, collaboration across the supply chain is fundamental to reducing FLW. Voluntary agreements and commitments that commit signatories to specific actions to reduce FLW are commonly used to foster collaboration among stakeholders, including retailers and the food service sector. Listed below are several examples of such initiatives, which typically provide frameworks for sharing best practices, monitoring progress, and publicly reporting on achievements in FLW reduction.

The Courtauld Commitment 2030 (UK)

This is a voluntary agreement aimed at improving resource efficiency and reducing waste within the UK food and drink sector. Building on the work by WRAP started in 2005,⁴⁰¹ the agreement involves various stakeholders, including major supermarkets, brand owners, and manufacturers. Participants commit to measurable goals such as reducing food and drink waste by 50% per capita by 2030 (relative to the UK 2007 baseline), as well as decreasing greenhouse gas intensity, and improving packaging design to minimise waste. The Food Waste Reduction Roadmap⁴⁶⁹ supports the delivery of the Courtauld Commitment 2030 food waste target and is supported by all the largest grocery retailers in the UK (whose cumulative business represents 97% market share). Based on data from seven major retailers,⁴⁶⁹ the average quantity of food going to waste per tonne of food handled in 2021 was 0.44% (i.e. 4.4 kilogram for every tonne handled), a reduction of approximately 18% on 2018 levels.^{*} For the fifteen retailers who had comparable tonnage data for 2018 and 2021, food waste fell by over 19,000 tonnes, a decrease of 8%.⁴⁶⁹

The US Food Loss and Waste 2030 Champions

The US Food Loss and Waste 2030 Champions are businesses and organisations from across the supply chain that have publicly committed to reduce FLW in their own operations in the US by 50% by the year 2030.⁴⁷⁰ Initiated by a coalition between the US Department of Agriculture (USDA) and US Environmental Protection Agency (EPA) in 2016,⁴⁷¹ the group consists of more than 45 organisations, many of whom are retailers (for example, Aldi, Amazon, Hello Fresh, Kroger, Walmart and more) and major food service businesses (such as Starbucks and Wendy's). The quantifiable successes for individual companies can vary widely, as each participant is at different stages of implementing their food waste reduction strategies.[†] Notable examples of success include Walmart's

^{*} Based on WRAP's guidelines for retailers, we assume these estimates meet the definition for FLW prevention as defined by this report, as diversion destinations in the UK such as compost and anaerobic digestion are classed as waste destinations, while food redistribution data are reported separately.

[†] Notably, the initiative does not frequently release detailed collective results, instead presenting individual success stories and case studies highlighted by the EPA. As such, data on the wider impact of this initiative is

2018 reduction of food waste in their fresh departments by 90 million units of food, Kroger's improvement of food waste diversion from landfill from 2017 (27% of food waste diverted) to 2018 (40% diverted), and Aldi's diversion of 74% of their operational waste in 2021.⁴⁷²

Australian Food Pact

Since its launch in 2021, the Australian Food Pact has seen 32 businesses commit to halving their food waste by 2030. Signatories to the Pact include major retailers Coles and Woolworths, who together hold around 65% of the grocery market share.⁴⁷³ In their first year of reporting, signatories sent more than 54 million meals to food rescue charities.⁴⁷⁴

Kai Commitment (New Zealand)

As highlighted in <u>case study 2</u>, the Kai Commitment is a voluntary agreement for New Zealand businesses to reduce FLW and related emissions across New Zealand's food supply chain.¹¹⁴ The initiative, launched in late 2022, currently has eight signatories, including Foodstuffs NZ and Woolworths NZ, whose combined business likely accounts for more than three quarters of the grocery market in Aotearoa.³⁹²

While voluntary agreements have seen success in reducing FLW, there are also instances where governments pass laws and institute regulations that specifically target food waste. Targeting the food service and retail sector, France's legislation to reduce food waste in supermarkets and restaurants by diverting food waste from landfill is one of the better-known examples. Often described as a landfill ban for food waste, the policy is in fact more nuanced, as described in <u>case study 13</u>. Whether such an approach in Aotearoa would prevent food waste is unclear; in the French case, there was an increase in food rescue, suggesting there remains scope for prevention activities. France is not the only country to consider or implement policies to prevent food waste going to landfill, with a host of countries, states, and cities adopting a range of approaches. We summarise these policies in <u>annex 2</u>.

lacking. Additionally, many of the successes listed by the initiative relate to food waste diversion, rather than prevention as defined by this report.

Q Case study 13: France takes a unique approach to retail and food service waste

In 2016, France made international headlines³⁷⁷ with the adoption of a groundbreaking law aimed at combatting food waste, particularly in the retail sector. The legislation, the first of its kind globally, prohibited^{*} supermarkets from destroying unsold food products, compelling them to donate these items instead. To enact this, supermarkets larger than 400 square meters were required to enter partnerships with food rescue organisations to facilitate the donation of surplus food. Under these partnerships, supermarkets are expected to sort produce and donate packaged items 48 hours before their expiration dates, improving the quality of donated food. ⁴⁷⁵ In 2019, these measures were extended to the mass catering businesses (who produce more than 3,000 meals a day) and food industry (whose annual turnover exceeds €50 million), with both sectors banned from destroying unsold food products fit for consumption. Under these partnerships, supermarkets are expected to sort produce and donate packaged items 48 hours before their expiration dates, improving the quality of donated food.⁴⁷⁵ In addition, commercial catering sectors were obliged to offer doggy bag options for their catering services. In 2020, fines for destroying unsold food fit for consumption were strengthened, increasing from €3,705 to the equivalent of 0.1% of a company's annual turnover. In addition, 'best before' dates and 'use by' dates had to be integrated into product codification to optimise logistics.⁴⁷⁶

The impacts of these laws have been notable. Following their implementation, food donations from supermarkets have increased significantly, with some reporting suggesting that donations rose by 30% in 2017, with more than 90% of supermarkets donating unsold food.⁴⁷⁷ However, the implementation of these regulations has not been without shortcomings. Critics argue that the increase in food donations was not matched by an influx of funding or infrastructure to handle the dramatic uptick in donations, posing a challenge for food rescue organisations and resulting in waste redistribution rather than complete diversion.⁴⁷⁷ Additionally, the law does not specify the quantities or proportion of unsold food that must be donated, making a 1% donation theoretically sufficient to meet legal obligations.⁴⁷⁶ However, companies are also legally obligated to publicly display their food waste commitments. Additionally, the law has sparked a broader conversation about the need to prevent unnecessary food production in the first place.⁴⁷⁵ As noted in <u>section 1.1</u>, food donation falls outside the definitional boundary of prevention used in this report. As such, in this context, French laws around RFS waste are more aligned with food rescue work (see <u>Food rescue in 2022: Where to from here?</u>).

^{*} The degree to which this the law is prohibitive is debated, with some arguing the law promotes responsible corporate behaviour and formalises the expectation of donating food, rather than outright banning of food waste.



Figure 34: A simplified depiction of the food supply chain highlighting the household stage.

In this section, we focus on the last stage of our simplified supply chain (see <u>figure 34</u>) and discuss food waste in households. The food supply chain exists to bring food to consumers. While this objective is met at the RFS stage (see <u>section 5</u>), a significant proportion of the food produced is intended to be prepared and eaten in homes.⁵⁷ A household can be one person who lives alone, or a group of people who share living spaces and may pool resources – including food – collectively.⁴⁷⁸ What a household looks like varies significantly depending on circumstances and a range of factors such as age, geography, socio-economics, and cultural norms. Whereas 'consumers' refers to the individual people who purchase goods or services for their personal use, we use the term 'households' to refer to the private dwellings where food is being stored, prepared, and eaten.

Much research on household food waste is dedicated to understanding and modifying decision making processes and subsequent behaviours of consumers. Consumers' behaviours are related to many factors, including cultural attitudes around 'food' and 'waste', knowledge and beliefs about

waste, and how easy or difficult avoiding waste is – or how easy or difficult consumers perceive avoiding waste to be.⁴⁷⁹ Food and food waste in households can be complex and deeply personal, while also subject to social norms and expectations. Importantly, some of the reasons for behaviour can be external to the individual, such as social norms, available waste management infrastructure, and retail environments.^{479,480} Additionally, behaviours and the reasons for them are not static; stability at home, stress, and difficulty can all influence how easy it is for an individual to maintain behaviour change.⁴³⁹ Importantly, any interventions aimed at changing behaviour are not always contextually and culturally transferable.

While behaviour change campaigns can produce results at the household level, which is considered the part of the supply chain where the most FLW occurs, an excessive focus on consumers can unfairly place responsibility for FLW on individuals and households and detract from more systemic action by actors with relatively more power to effect change. Therefore, consumer behaviour change interventions should be included as part of a suite of solutions, which holds stakeholders appropriately responsible for reducing FLW.

Food and food waste in households can be complex and deeply personal, while also subject to social norms and expectations.

...an excessive focus on consumers can unfairly place responsibility for FLW on individuals and households and detract from more systemic action by actors with relatively more power to effect change.

6.1 Understanding household food waste

In households, food waste includes parts of food that are bought, received, or harvested and possibly processed or stored, but are not consumed and eventually discarded. Not all parts of food brought into the home can or will be eaten, and so we categorise the resulting food waste according to whether it is avoidable, potentially avoidable, or unavoidable:

- Avoidable: edible food that could have been consumed at some point.⁴⁸¹
- Potentially avoidable: parts of food that are eaten by some people and not others (like bread crusts), or that is eaten in some food preparations and not others (like potato skins).⁴⁸¹
- Unavoidable: inedible parts of food such as eggshells and bones, as well as parts of food most members of a population would not eat, like banana skins.⁴⁸¹

As this report is focused on prevention, we are particularly interested in waste that is avoidable or potentially avoidable. However, there is some subjectivity around what parts of food may be edible or inedible and so nuance is needed categorising what waste is avoidable, potentially avoidable, or unavoidable – these are broad, general categories rather than fixed.

Good data is needed to identify areas that need intervention and can act as a benchmark to track the effectiveness of food waste prevention strategies, while standardisation can also help with comparison and monitoring progress over time. In the Food Waste Index Report 2021 the UNEP highlighted New Zealand and 13 other countries with high-quality data for household food waste estimates.⁴⁸² <u>Annex 4</u> summarises how data was collected in these countries.

6.1.1 How much food do New Zealand households waste?

A significant amount of FLW across the food supply chain occurs in households. Internationally, the percentage of total FLW that comes from households has been estimated at 32% in Australia,⁵⁷ and 40% (2019)-60% (2022) in the US.^{401,483} In Aotearoa, the best estimate is 40%.²¹¹ Because of differences in classifications and data collection methods, it can be challenging to make comparisons

between these numbers. However, it seems reasonable to say that attention to household food waste has the potential to contribute to a substantial overall reduction in New Zealand's FLW. A recent estimate suggests food waste is costing around \$1,500 per year per household in New Zealand, while in Australia, 71% of food that entered the market was intended for households, and approximately 18% of this purchased food was wasted.⁵⁷

In addition to the proportion of all our FLW that comes from households, we also have estimates of the amount of food that is wasted by New Zealand households. WasteMINZ has conducted two bin audits. In 2015, data from 1,402 household was used to conclude that approximately 230,000 tonnes of food is disposed of in kerbside collection across Aotearoa annually, or almost 150 kilograms per household.⁴⁸⁴ In 2018, the audit was repeated on households in about half of the territorial authorities that were included in the 2015 audit, resulting in a sample of almost 600 households.

A recent estimate suggests food waste is costing around \$1,500 per year per household in New Zealand...



what parts of food may be edible or inedible and so nuance is needed categorising what waste is avoidable, potentially avoidable, or unavoidable... This study estimated that nationally nearly 300,000^{*} tonnes of household food waste is sent to landfill annually, or about 164 kilograms per household.⁴⁰³ The increase in the amount of food waste per household was partly attributed to one of the territorial authorities that was included in 2015, but not 2018, having separate food scraps bins destined for compost that were not counted. This biased the 2015 overall estimate downwards compared to 2018 when this territorial authority was not included, and illustrates the challenges of gathering reliable data. A strength of these estimates is that they involved direct measurement of waste by bin audit. However, as the authors of the report on the 2018 study acknowledge, the estimates are based on extrapolating from a sample of unknown representativeness of the New Zealand population. However, these estimates are broadly consistent with an analysis of secondary data that estimated 224,000 tonnes of food waste was sent to landfill by New Zealand households in 2011.²⁵⁶

As described earlier, not all household food waste is avoidable (see <u>section 6.1</u>). In the 2015 bin audits, 35% of New Zealand households' food waste was unavoidable, while 54% was avoidable and 12% potentially avoidable.⁴⁸⁴ In 2018, 34% of household food waste was considered avoidable, 53% unavoidable, and 14% potentially unavoidable.^{†,403} Differences between the two years were not statistically different from zero.⁴⁰³ Similar to what we see internationally in Australia,⁵⁷ China,⁴⁸⁵ and the EU,^{479,486,487} in Aotearoa, self-reported data⁴⁸⁸ and bin audits⁴⁰³ show bread, fruit, and vegetables as the most prominent categories of avoidable food waste.

Bin audits are useful because they can potentially be linked to surveys and other household characteristic data, and they can provide detailed information on the contents of food waste, for example, the type of food wasted and whether its waste was avoidable. This level of detail can allow identification of potentially high impact targets for intervention and is also useful for evaluation. However, it is a resource intensive method of data collection, meaning it is infeasible for nationwide or ongoing data collection.

For ongoing monitoring, other data is available – territorial authorities are required to report to MfE the amount of food waste or food and organic waste combined that households leave for kerbside collection.⁴⁸⁹ This data is not as granular as bin audits, but provides a nationwide picture of the amount of food waste at the level of the territorial authority, which is useful for ongoing monitoring.

A gap in this data is that in territorial authorities where waste is collected privately rather than directly by the council, there is as yet no requirement for reporting.⁴⁹⁰ It is also theoretically possible to collect and report this data at a smaller geographic level than the territorial authority.

Another approach to studying household food waste is to ask people about the amount of food of which they dispose. This has the advantage of being far less resource intensive than actually measuring household waste. ... bread, fruit, and vegetables [are] the most prominent categories of avoidable food waste.

...territorial authorities are required to report to MfE the amount of food waste or food and organic waste combined that households leave for kerbside collection.

^{*} The most recent Food Waste Index Report (2024) from UNEP used the same bin audit to report the annual total amount of food disposed of by New Zealand households as just over 315,000 tonnes per year. It is not clear how this figure was derived by UNEP but one possibility is they used the bin audit's per capita or per household figures and applied them to Census population data that has been updated since the original bin audit was published.

⁺ Percentages don't add to 100% because of rounding.

However, it can be subjective, inaccurate, or both. In a survey carried out alongside WasteMINZ's 2018 bin audits, respondents were asked whether the food they threw away was none, hardly any, a small amount, some, a reasonable amount, or a lot.⁴⁰³ The measured waste, in kilograms per household per week, was about the same for those who responded "a small amount" and "some", and was more for these categories than households who responded "a reasonable amount". Subjectivity can be removed by asking households to estimate actual quantities – for example, one programme evaluation used the number of two litre ice cream containers⁴⁹¹ – but such measures may still be inaccurate due to respondents guessing rather than measuring, or reporting low levels of waste due to social desirability bias.

6.1.2 Drivers of food waste in households

Household food waste involves numerous behavioural, environmental, and socio-cultural factors.⁴⁸⁰ These factors are interconnected, as shown in <u>figure 35</u>.^{479,480} Importantly, the specific influences will vary from household to household.



Figure 35: Influences on consumer behaviour and household food waste happen at various interconnected levels. Image credit: Boulet et al.⁴⁸⁰

To simplify our discussion of the drivers of food waste in households, we look at the issue in two ways. Firstly, we look at household characteristics, like demographics and attitudes, and how these might be related to different patterns of food waste.⁴⁹² Then, we look at the different stages of food's journey through the household, focusing on three stages: 1) planning and purchasing, 2) storage and preparation, 3) consumption and leftovers.³⁹⁴

There is considerable diversity in the composition of New Zealand households. According to the 2018 census,^{*} 40% of households have children, 23% of households are one person, and 28% of households are couples or couples and other persons.⁴⁷⁸ The population also has considerable ethnic and cultural diversity, as well as diversity in the locations of our households from cities to towns to rural locations. These and other characteristics are related to patterns of household food waste.

Different cultural traditions can lead to different patterns of food waste. For example, some cultures tend to serve food 'family style' where everyone serves themselves from a shared bowl, while in other cultures individual portions are served. We don't know of any studies that look at whether these styles result in different levels of waste in the home, but in hospitality settings, eating family style may be associated with more waste than individual servings.⁴⁹³ Another example pertains to the social acceptability of finishing all food served; in Abu Dhabi it is considered inappropriate to finish all the food on a plate while in Japan it is the opposite.⁴⁹⁴

Figure 36 shows data from the Organisation for Economic Cooperation and Development (OECD) on how often different types of households – those with and without children, and earning high and low incomes – wasted food for various reasons. For each reason for waste, households with children are at least twice as likely to waste food as households without children.⁴⁹⁵ In contrast, differences in waste between households in the highest and lowest income quintiles were less pronounced, although high income households were more likely than low income households to report wasting food because "too little was left to save", they "did not like or tired of eating" the food, and they were "unable to store or save" the food.⁴⁹⁵ In general, there is some conflict in the literature over whether households at different incomes have different levels of food waste; although high income households do waste food and sometimes provide abundant food to avoid the stigma of being identified as 'poor'.⁴⁷⁹



Figure 36: Food waste in households when factoring for children and income. These survey responses are compiled from nine countries in the OECD from the Environmental Policies and Individual Behaviour Change Survey. Image credit: OECD.⁴⁹⁵

^{*} Data from the 2023 Census was not available at the time of writing.

An Australian study used survey data on food shopping, preparation, and waste habits to divide respondents into three categories in relation to food waste: 'considerate planners' who planned their shopping and cooking and rarely prepared food that went uneaten; 'over providers' who planned ahead but often prepared food that went uneaten; and 'under planners' who didn't generally plan ahead and produced more waste than 'considerate planners' but less waste than 'over providers'.⁴⁹⁶ One of the main predictors of being an 'over provider' was having children; parents provided more food than was needed in response to unpredictable preferences of children.⁴⁹⁶

As well as ensuring sufficient food for children, waste resulting from preparing too much food can result from cultural or social values that prioritise showing hospitality to guests by having abundant food available, and spending time with guests rather than preserving leftovers for future use.⁴⁹⁷ Providing more food than guests are likely to eat can be motivated by a desire to avoid embarrassment, guilt, or perceived stigma related to the possibility of running out of food. 498-500 More generally, some people value being perceived as a 'good provider' by offering abundant food both to their own household and to guests.⁵⁰¹

Finally, age and geography may play a role. Older people may be less likely to waste food because of moral concerns around wastefulness and experiences of food scarcity earlier in life;⁴⁷⁹ for example, resulting from rationing in the aftermath of the second World War. European research suggests a relationship between geography and household food waste, with households in urban locations tending to waste more food than those in rural areas.502

Stages where household food waste can occur

Planning and purchasing

We combine planning and purchasing because they are closely related – many of the actions related to planning go on to affect purchasing. Actions at this stage include meal planning, making shopping lists, and shopping. People who plan their weekly menus are less likely to overstock than those who don't,^{503,504} while those who don't check fridges, freezers, and pantries before shopping often buy food they already have.⁵⁰⁵ Shopping habits such as buying in bulk tend to create more food waste, whereas those who stick to a list produce less food waste.⁵⁰⁶

Lack of time, energy, and money in busy lives have been identified as barriers to planning.⁵⁰⁷ There is an ever increasing demand for convenience, with consumers wanting to minimise inconvenience while valuing flexibility and choice.⁵⁰⁸ This can sometimes cause trade-offs with waste preventative behaviours such as planning or keeping and eating leftovers.⁴⁷⁹

As well as planning, features of the shopping environment can shape waste behaviours. People who shop at large supermarket chains tend to have larger amounts of food waste than those who grow their own food or buy from small shops or local markets.⁵⁰⁹ As discussed in <u>section 5.2.3</u>, marketing strategies employed by retailers, such as 'buy one, get one free' can encourage consumers to buy

...waste resulting from preparing too much food can result from cultural or social values that prioritise showing hospitality to guests by having abundant food available...



more than they need,⁵⁰⁴ and the positioning of products on shelves for children's snack foods may also promote excess, unnecessary purchasing, and wasteful behaviours.⁵¹⁰

There is some evidence suggesting less food is wasted when cooking with a meal kit – that is, recipes and pre-portioned ingredients for a set number of meals delivered to the home – than when cooking with ingredients sourced from a supermarket.⁵¹¹ Using meal kits can aid with proper portioning, reducing over purchasing and impulsive buying. Survey data from 955 households from six counties

across Europe and North America suggests that meal kits can reduce food waste in comparison to traditionally cooked meals by up to 38%.⁵¹² Similarly, a life cycle assessment of meal kits in the US found they had lower average greenhouse gas emissions than equivalent grocery store meals.⁴⁵⁹ In addition, the study inferred that because grocery meals are not pre-portioned, they result in higher food waste in households. However, the study also noted that meal kits have higher packaging impacts than grocery meals. In addition to packaging problems, in New Zealand, meal kits are typically more expensive than

People who shop at large supermarket chains tend to have larger amounts of food waste than those who grow their own food or buy from small shops or local markets.

shopping for the equivalent products in a supermarket,^{392,513} although brands like Bargain Box⁵¹⁴ are trying to minimise this difference.

Storage and preparation

During storage and preparation, knowledge of food safety and cooking or preservation techniques can influence food waste generation. Proper storage, such as in the fridge or freezer, can prolong the shelf life of foods, while improper storage accelerates food spoilage.³⁹⁴ Tomatoes, for example, should be stored stem side down, and in a bag in the fridge if they are ripe.⁵¹⁵ Studies have highlighted a lack of knowledge around storage, such as the optimal fridge temperature, and low confidence in food management skills, as important factors.^{504,510} In a survey in Aotearoa, Gen Y and Gen Z respondents reported not knowing how to store food as a reason for their own food waste.⁴⁸⁸

Food waste in the preparation stage can result from discarding what is inedible or perceived as undesirable (such as broccoli stalks or potato skins), as well as from preparation techniques that may not make effective use of food (such as, by discarding a large amount of apple when removing the core).^{78,504} Food can also be rendered inedible or become unpalatable during preparation due to a lack of cooking skills (such as by burning or over-seasoning), resulting in food waste.

Consumption and leftovers

Food that is prepared but not eaten, referred to as leftovers, can be plate waste, surplus meals, leftover ingredients from food preparation, or food leftover from a hospitality setting brought home in a doggy bag.⁵¹⁶ Leftovers are not inherently food waste; cooking enough food for multiple meals is

a common practice to maximise efficiency and convenience in food preparation.⁴⁷⁹ However, the 2018 WasteMINZ bin audits estimated that leftovers accounted for 8.2% of the avoidable food waste by weight generated by New Zealanders and was the second largest type of avoidable food waste after bread.⁴⁰³

Wasted leftovers may be a sign of not knowing how to portion food appropriately.⁴⁷⁹ Some of the social and

Leftovers are not inherently food waste; cooking enough food for multiple meals is a common practice to maximise efficiency and convenience in food preparation. cultural considerations described above, such as a desire to be a good provider or a generous host, come into play at this stage.

6.2 How can we prevent food waste in households?

As a framework for thinking about ways to prevent food waste in households we draw on a theory of change (ToC) developed by MfE.⁵¹⁷ The ToC is not specific to food waste but is about waste-related behaviour change more broadly. As we have emphasised the importance of a systems view throughout this report we focus on the macro level of the ToC, but we acknowledge that meso level and micro level considerations (see <u>figure 35</u>) – including household and individual characteristics discussed in <u>section 6.1.2</u> – will affect the success of any macro level intervention.

6.2.1 Promoting voluntary behaviour change

In New Zealand, promotions to reduce household food waste began around 2014 when WasteMINZ launched the National Food Waste Prevention Project, the first action being the 2015 audits as well as a nationally representative survey of 1,365 households about behaviours and attitudes that lead to food waste.⁵¹⁸ This research was followed by a three year 'Love Food Hate Waste' behaviour change multimedia campaign.⁵¹⁹ This campaign was an application of the highly successful version that has now been running in the UK for over 20 years.^{111,520}

Information- and engagement-based approaches

Providing information is probably the most common intervention to promote behaviour change. In general, 'information deficit theory' – the assumption that people don't know their behaviour is suboptimal and that providing more information will change behaviour – is not supported by the evidence.⁵²¹ MfE's literature review on reducing household food waste concluded that, on its own, information sharing is ineffective.³⁹⁴ In contrast, in the US, ReFED estimates that consumer education campaigns can achieve the greatest net financial benefit of all consumer focused interventions, at US\$17 billion.⁴²⁴ Information can be made more effective by considering the way it is conveyed, who it is conveyed by, and the prevailing norms around 'common knowledge'.⁵²² To the degree that a lack of knowledge is the problem, providing information on how to store food and improving cooking skills are the knowledge gaps aligned with the drivers of food waste identified in <u>section 6.1.2</u>.

Though awareness campaigns have received the most attention internationally – particularly across Canada, the UK, and the US – Love Food Hate Waste in the UK estimates that information campaigns in combination with face-to-face activities such as cooking courses can achieve around a 15%

reduction in avoidable household food waste.^{523,524} Gamification has also been shown to reduce self-reported food waste, although it had no effect on food waste measured by bin audit.⁵²⁵ There is the potential limitation of only meaningfully engaging with those who voluntarily sign up and reaching self-selected people who choose to participate. Targeted recruitment efforts could be considered to address this.

---{\$

Love Food Hate Waste in the UK estimates that information campaigns in combination with faceto-face activities such as cooking courses can achieve around a 15% reduction in avoidable household food waste. Awareness building campaigns use planned activities or communications over a period of time, typically aimed to appeal to many people.⁵²³ Such campaigns can operate at a variety of scales, from global campaigns such as 'Think. Eat. Save' spearheaded by the UNEP and FAO, to council led challenges (see <u>annex 5</u>). For example, Germany's 'Too good for the bin!' campaign by the Federal Ministry of Food and Agriculture of Germany started in 2012 and runs across the food supply chain. The activities as part of the campaign include a yearly awareness raising week, web resource, prizes for projects across five different categories, an app, social media content, and restaurant leftover boxes. The effectiveness of 'Too good for the bin!' is being evaluated with annual measures of consumer knowledge and attitudes, as well as measurement and reporting of food waste.⁵²⁶ The

New Zealand version of the international Love Food Hate Waste campaigns have already established some food waste prevention activities like "Eat Me First" stickers and classes, and many councils are engaged and funding the programme.

As discussed in <u>section 6.1.1</u>, there can be a disconnect between how much food waste people think they produce and how much they actually produce. Measurement of food waste is effective at changing behaviour.³⁹⁴ Awareness raising campaigns may prompt people to consider their food waste, even if they have limited effects on actual food waste behaviour.^{525,527} The New Zealand version of the international Love Food Hate Waste campaigns have already established some food waste prevention activities like "Eat Me First" stickers and classes, and many councils are engaged and funding the programme.

School programmes, which provide students with

information and experiences outside of their own homes, present an opportunity to educate children and young people about food waste.⁵²⁸ Promoting and building familiarity with imperfect produce, portion sizes, or cooking can help to normalise and reinforce prevention techniques that can then continue at home.⁷⁸ In school-based programmes in Australia that used a mix of educational, skillsbased and whole school events, both students and parents were observed to display more of the targeted behaviours, showing that interventions in external settings can influence behaviours in homes.⁵²⁹ In New Zealand, examples of school-based programmes include Garden to Table, which delivers food education for children,⁵³⁰ and Enviroschools, where young people design and lead sustainability projects in their school and neighbourhood.⁵³¹

Incorporating active engagement, for example, goal setting, feedback, and commitment making, can be more effective at changing behaviour than simply presenting information.⁵²² In the UK, a pilot run by WRAP with monthly workshops over four months achieved a reduction in avoidable food waste of more than 50% on average, based on estimated volume data by participants collected in the first and last week of the programme.⁵²³ An example from Aotearoa is the Food Lovers Masterclass delivered by Waste-Ed with Kate with funding from councils, providing information, demonstration, and participation in food waste prevention techniques. With recent MfE funding to evaluate the classes' effectiveness, there are promising signals with almost half of participants reporting a reduction in the volume of food they waste since taking the class (see <u>annex 5</u>).⁴⁹¹

Asking participants to make a commitment to a particular action is thought to promote adherence to their commitment and thereby behaviour change, with the evidence being mixed to positive in the short- and long-term.⁵³² A meta-analysis of commitment research showed that commitment with other treatments, such as feedback or incentives, was more effective than commitment alone, particularly for long-term change.⁵³² Processes such as self-concept, attitudes, and social and personal norms, are also thought to underlie the effectiveness of commitment making, particularly when commitments are made publicly and there is social pressure to adhere.⁵³² In 2022, Love Food Hate Waste in New Zealand ran a 'Food Waste Warriors challenge' that involved people signing up to participate in a series of food saving challenges with tips and resources to promote better storage,

meal planning, and making the most of leftovers. An incentive of a pantry and fridge makeover, worth \$1,500, was provided. Engagement on social media platforms were also promoted with prizes and giveaways.⁵³³

Approaches that make it easier for people to prevent waste

Behavioural scientists can draw on sophisticated understanding of human decision making to design behaviour change interventions.^{534,535} Information and engagement-based interventions target our conscious decision making, but there are also interventions that target our unconscious or default decision making. Such interventions are known as 'nudges' and work by removing 'friction' – that is, things that make a given course of action harder – from the desired behaviour. While often effective, these types of interventions have been criticised for undermining personal autonomy, ^{536,537} although one study found participants generally approved of using nudges to encourage healthy eating. ⁵³⁸ Additionally, while policy makers may grapple with the potential ethical implications, some marketing companies and other commercial actors show no such compunction, creating 'sludges' – nudges that promote behaviour that is of no benefit or is even harmful to the individual.⁵³⁹

As discussed in <u>section 5.2.3</u>, retailers can have significant influence over consumers' purchasing decisions through the way products are laid out in-store and the promotions available. Retailers could use this influence to make it easier for consumers to purchase the right amount of food for their needs. This could mean, in collaboration with processors and manufacturers, making it simpler to buy individual pieces or small portions of some products with short shelf lives, as well as avoiding

promotions that make it better value for money to buy more products with short shelf lives than are needed (see <u>section 5.2.3</u>). Changes to these practices are business decisions with potential implications for revenue; voluntary agreements like those discussed in <u>section 5.2.4</u> may help with getting buy in from retailers.

Friction can also be reduced with tools and technologies. These can be relatively simple, such as Love Food Hate Waste's "Eat Me First" stickers, which can be placed on shelves or specific food items as a visual cue.⁵⁴⁰ They can also incorporate more advanced technologies. For example, smart fridges could prevent food waste by relaying images that enable consumers to confirm which products they already have to avoid duplicate purchases, [I]nterventions ... known as 'nudges'
 ... work by removing 'friction' – that is, things that make a given course of action harder – from the desired behaviour. While often effective, these types of interventions have been criticised for undermining personal autonomy...

or to send alerts about product expiry.⁵⁴¹ Research on smart fridge systems that guide users in food related activities such as storage and planning have been shown to increase awareness of food availability at home, use of food before expiration date, and have environmental and economic benefits based on life cycle analyses.^{542,543} Other tools can be either high or low tech – for example, preventing waste by preparing the appropriate portion could be achieved with the help of simple measuring devices⁵⁴⁴ or with technology-aided measurement as is used in smart scales.^{545,546} Modifications to food labelling and packaging may also promote FLW prevention (see <u>table 6</u> and <u>section 4.2</u>).

Multipronged interventions

Interventions that combine multiple elements are often more successful than single interventions,

for example, combining awareness-raising and face-toface activities are recommended by Love Food Hate Waste UK, discussed above.⁵²³ A randomised controlled trial using a technology-aided measurement approach with personalised coaching collected data on food behaviours and waste through photos and informational tags of receipts, preparation waste, consumption, and storage purges, and found a 24% reduction on food wasted during dining.⁵⁴⁶ Because there are multiple drivers of household food waste, it makes sense that multipronged interventions are more successful than those with a single focus.



Because there are multiple drivers of household food waste, it makes sense that multipronged interventions are more successful than those with a single focus.

Box 9: Date labelling

Date labels indicate when a product is either safe or best to eat.⁵⁴⁷ A key issue is the distinction between 'use by' and 'best before' dates being misunderstood by consumers, resulting in food being thrown away when it has passed its best before date but is still edible.^{405,359,548,357} European research suggests that between 10%-20% of food wasted is due to product date labelling and that only a third of EU consumers correctly interpret the meaning of the best before date.^{369,549} In New Zealand there is evidence the public generally understands that food is still safe to eat even after the best before date has passed. Surveys in New Zealand have found high proportions of consumers understand the meaning of best before date labels, with only around 10% of people mistakenly believing food to be unsafe to eat after the best before date.^{488,550} Similar proportions reported that they would dispose of food past its best before date.^{488,550}

Understanding of what date labels mean and self-reported behaviour may not match actual behaviour, but there is little data available on actual behaviour. One clue may come from the 2018 bin audits. Of food that was disposed of in unopened packaging, 68% had passed its best before date prior to the study and 16% passed its best before date during the study. Because the food was thrown away unopened, it is unlikely that it was evaluated for potential spoilage before being disposed of. However, bin audits cannot tell us the actual reason for disposal of food, nor can they give us any insight into the universe of food in people's homes that is past its best before date and not disposed of. There would be benefit in more research into how date labelling affects food waste behaviour in Aotearoa.

While removing best before dates could prevent early discarding of food, it relies on the household members to be well-informed consumers. A consideration here is the value of education to enable better judgement of food that is safe to eat. Much date label reform internationally focuses on changing phrases on the label. However, many studies suggest that the differences in the intent to discard based on the specific label phrase is relatively insignificant, with the willingness to consume or waste a food product mainly mediated by perception of safety requiring that food is sometimes wasted regardless of the wording on the date label.^{358,551-554} Consumers may also be more sensitive to the date rather than the wording, with an eye tracking study show consumers fixate significantly more on the date than on the phrase, and that 50% of discard decisions involving no visual fixation on the phrase. Evidence on the effectiveness of messaging strategies around date labels are still early in the research stages.⁵⁵⁵

6.2.2 Using policy as a tool to drive change at a system level

In Aotearoa, MfE has set strategic direction with the publication of a waste strategy for New Zealand

focusing on circular and systems thinking to reduce the amount of waste we generate.⁵⁵⁶ Overseas, similar strategies towards more sustainable models to prevent waste have been taken up by the EU,⁵⁵⁷ Australia,⁵⁵⁸ and Singapore.⁵⁵⁹ Overarching national frameworks can provide direction to align regional and local initiatives, as well as support more effective collaborations across public and private sectors (see <u>section 2.2.6</u>).

The MfE strategy includes two components that are relevant for household food waste prevention. The first is the call for household food scraps collection to be available to all urban households in New Zealand by

2030.⁵⁵⁶ While diverting food scraps from landfill is the primary purpose of food scraps collection and thus not itself prevention, there is some evidence that separating food scraps from other household waste may lead to prevention in the long-term as the process of separating makes food waste more

visible and may therefore motivate households to take other preventive actions.⁴⁷⁹ Whether this actually happens is contested.⁵⁶⁰

The Strategy also promotes legislation around planning and reporting on waste for central and local governments, with territorial authorities now obliged to report how much food waste is collected in dedicated food scrap containers, or combined food and garden waste collections.⁴⁸⁹ This will enable prevention of household food waste by making it easy to monitor the amount of food waste households are generating, but see <u>section</u> <u>6.1.1</u> for a discussion of the limitations of this source of data.

...there is some evidence that separating food scraps from other household waste may lead to prevention in the long-term as the process of separating makes food waste more visible and may therefore motivate households to take other preventive actions.

Another policy lever available is the use of financial incentives or disincentives to encourage households to prevent food waste. Various 'polluter pays' models could be considered, with some overseas evidence suggesting higher collection fees may be effective at reducing people's waste output.⁵⁶¹ For example, in Luxembourg, citizens who reduce their household waste are financially rewarded with reduced fees.⁵⁶² Weight-based billing systems appear to have significant effects on reducing waste, increased recycling rates has been observed in Sweden where municipalities implementing a weight-based billing system had 20% less household waste per capita than other municipalities.⁵⁶³ In Aotearoa, 'pay as you throw' schemes have been used for general waste for

residents in some Auckland suburbs but, residents in these suburbs produced more or less the same amount of waste as ratepayer funded rubbish collections.⁵⁶¹ In drawing lessons for policy making, it is important to note that these approaches were designed around overall household waste and not food waste specifically. However, in South Korea, a 'pay as you throw' scheme applies to food waste and anecdotal reports suggest the combination of the financial disincentive and the visualisation of the amount of waste produced has reduced food waste output.⁵⁶⁴





regional and local initiatives, as well as support more effective collaborations across public and private sectors. Policy makers can also lead on monitoring and evaluation.⁵²⁸ Public data and reporting can be improved for accountability purposes at all levels of the supply chain. Better establishment of data management protocols can potentially enable sharing with protected confidentiality, and data dashboards for broader access and use. Data sources could also be expanded to include reporting, audits and tracking systems (at household or regional level – for example, through the collection of kerbside collection data), citizen science projects and product lifecycle assessments. Tracking could highlight potential areas of inequity for more targeted interventions or funding for prevention projects although there are potential unintended consequences and a need for principles around ethics, privacy, and transparency.

Policy action does not have to be top-down. Policy makers can support participatory approaches to reducing food waste. The European Commission has previously convened a citizens' panel on food waste to engage actors, strengthen collaboration, and make recommendations to help and guide member states in achieving food waste targets.⁵⁶⁵ Policy makers can also support businesses and charities undertaking interventions that target voluntary behaviour change, as described in <u>section 6.2.1</u>. Involvement from policy makers may lead to better integration and coordination between high-level food waste policy, local or regional policy, and third sector actors, in turn resulting in better engagement, including the establishment of campaigns, knowledge hubs, and community led initiatives.⁵²⁸

MfE has recently continued to fund work on prevention of household food waste. The Waste Minimisation Fund provides financial support for projects that minimise materials, including food, being sent to landfill.⁵⁶⁶ For example, the Dunedin City Council was supported for work delivering improved kerbside services (including food waste, organics, collection, and processing)⁵⁶⁷ and the Whanganui Kai Hub⁵⁶⁸ for providing a centralised place to redistribute food and run educational workshops. National food waste reduction programmes aim to reduce food waste in three settings, one of which is household food waste (the others being businesses and Māori led settings, which includes households in its scope). Projects funded under the National Food Waste Reduction Programmes relating to household food waste include the WasteMINZ Love Food Hate Waste campaign and Para Kore's 'Para Kai' programme, a Māori led kaupapa Māori initiative driving the reduction of food waste.⁵⁶⁹

6.2.3 Identifying and prioritising effective ways to prevent household food waste

It can be challenging to assess the effectiveness of interventions and policies intended to target food waste, because there is often not a convincing counterfactual.⁵⁷⁰ Some of the more common barriers to creating a convincing counterfactual in interventions to prevent household food waste are:

- The lack of a control group without the intervention in some studies.
- Self-selection into interventions. People who are more motivated to reduce their food waste undertake the intervention and those who are less motivated don't, confounding the analysis of differences between the groups.⁵⁷¹
- Multiple interventions are undertaken at the same time, so that it is hard to unravel which components, or combinations of components, are effective.⁵⁷⁰

Additionally, sometimes the trade-offs of different data collection methods (described in <u>section</u> <u>6.1.1</u>) are evident in attempts to evaluate the success of interventions. While we may want to know the volume of food waste a household generated before and after an intervention, it is easier to find out if a household representative believes they waste less than they did before the intervention, or intends to reduce their food waste.

Even when robust evaluations provide evidence of an intervention's or policy's effectiveness overseas, policy makers and others will need to consider whether similar results are likely in New Zealand in the relevant local context.⁵⁷² Mixed methods evaluations that combine quantitative data and qualitative data can provide understanding of whether an intervention is effective and why it

was effective, making it easier to determine the applicability of the findings in other contexts, and providing richer information than either approach alone.⁵⁷³ Another consideration is whether it's feasible to scale up interventions⁵⁷² – for example, are there enough suitably trained people to deliver a class nationwide, or is it likely that a large share of households will purchase a smart fridge? As well as feasibility consideration, ReFED considers the potential of an intervention to make profit,⁵⁷⁴ although profit potential is relevant only to some interventions targeting consumers, like those where retailers alter the shopping environment.

Ultimately, policy makers and others choosing which interventions to pursue need to prioritise. OzHarvest and Behaviour Works Australia, within the Monash Sustainable Development Institute at Monash University, conducted a systematic review of desired food waste behaviours to determine likely impact on food waste and a survey of 1,600 consumers to determine likelihood of adoption and used the results to create an impact-likelihood matrix (see <u>figure 37</u>) to prioritise target behaviours for campaigns and policy development.⁵⁷⁵ The matrix divides behaviours into quadrants based on likelihood and impact, with high likelihood and high impact behaviours being priority behaviours, implying that policy attention and resource should be focused in this quadrant.⁵⁷⁵ Interventions targeting behaviours in other quadrants are likely to be less effective in reducing food waste because the impact of the intervention on waste is small, because the underlying behaviour change is unlikely, or both.

HIGH IMPACT & LOW LIKELIHOOD

Shop from a list Measure household food waste (audit) Make a shopping list Make a meal plan Involve kids Subscribe to a meal kit or recipe box service Buy pre-portioned food Batch cook Follow the storage instructions on food Online food shop Only bulk buy non-perishable foods Check food date labels weekly Read packet before buying Store food in well-sealed, clearly labelled

Check food date labels weekly Read packet before buying Store food in well-sealed, clearly labelled containers Rotate food in fridge Preserve perishable foods by pickling, saucing or stewing for later use Allow more time to eat during meals Shop at local speciality food stores Make a stock and freeze for future use

LOW IMPACT & LOW LIKELIHOOD

HIGH IMPACT & HIGH LIKELIHOOD

Once a week, make a meal with food that needs using up Have a 'use it up' shelf in the fridge and/or pantry for any food that needs to be eaten Freeze excess food Before cooking a meal, check how many household members will be eating Check stock before shopping Store leftovers correctly Plan for using close-to-date food Once a week cook a meal from food in the freezer Buy frozen vegetables Buy perishables as needed

Inspect food close to date Serve yourself at meal times Store bread in airtight container Use small plates and bowls Check fridge is at the correct temperature Share excess food with extended family and friends Before going shopping, eat a meal or snack

LOW IMPACT & HIGH LIKELIHOOD

Figure 37: Food waste reduction behaviour matrix. Image credit: OzHarvest. 575

6.3 Households' role in preventing food loss and waste across the supply chain

As described in <u>section 2.1</u>, the dominant narrative is that in high income countries like Aotearoa, food waste primarily occurs at the consumer end of the supply chain, in contrast with low income countries where FLW happens earlier in the supply chain because of a lack of resources to invest in mechanisation and other technologies that both promote efficient supply chains and prevent FLW.^{55,134} However, as we have described throughout this report, there are abundant opportunities to prevent FLW in the supply chain in addition to opportunities to prevent food waste in households. Importantly, some of the opportunities to prevent

household food waste are upstream, in decisions made by processors, manufacturers, and retailers.^{55,134}

While we should avoid holding consumers responsible for more than their fair share of FLW, we can recognise their potential power in preventing FLW. The supply chain exists to bring food to consumers. As such, consumers do have some power to prevent FLW, not only in their households, but across the supply chain. Consumer action has long been a tool to pursue social and political goals,⁵⁸ with

relatively recent examples including boycotting products made in Apartheid South Africa^{576,577} and the 'buycott' of certified Fair Trade products.⁵⁷⁸ In its LTIB, MPI recognised the implications for New

Zealand producers of growing consumer demand for sustainability in food supply chains.³⁸

Many of the actions we discuss in this section and throughout the report could strengthen consumer action. For example, robust data and monitoring systems would make it easier for consumers to identify products, which are aligned with their goals around FLW prevention, while awareness raising on the topic of FLW prevention could make it more likely that consumer action achieves a critical mass.

...some of the opportunities to prevent household food waste are upstream, in decisions made by processors, manufacturers, and retailers.

In its LTIB, MPI recognised the implications for New Zealand producers of growing consumer demand for sustainability in food supply chains.



7. Research for food loss and waste prevention

Research and innovation is a key part of developing new ideas, technologies, and practices for preventing FLW throughout all stages of the supply chain. FLW prevention efforts will benefit most from applied research and experimental development,

where innovations are developed, used, and evaluated for their efficacy.

Businesses have varying capabilities and resources for innovation and research. Collaboration and the sharing of innovative practices are also in tension with maintaining competitive advantages. In addition, there are challenges in adoption, implementation, and evaluation of innovations. Industry led research collaborations, such as Australia's Cooperative Research Centres (CRCs), for example, End Food Waste (formerly Fight Food Waste



FLW prevention efforts will benefit most from applied research and experimental development where innovations are developed, used, and evaluated for their efficacy.

CRC) (see <u>box 10</u>), have proved successful in creating partnerships between researchers, government, and industry to facilitate knowledge and data sharing and provide funding to look at innovation to reduce FLW.⁵⁷⁹

Sox 10: The Australian Cooperative Research Centre model

In the early 1990s, the Australian Government set up CRCs as a separate funding programme in addition to the government owned research agency, the Commonwealth Scientific and Industrial Research Organisation (known as CSIRO). CRCs are government supported research programmes that provide public-private links across research and industry to promote high levels of adoption and commercialisation of science and are typically hosted by research organisations with government funding for a limited time. Australian CRCs are industry led incorporated entities funded through a combination of grants, commercial industry, and government-administered funding. Around 30% of CRC programme is contributed through government grants.⁵⁸⁰ The governance structures of Australian CRCs have central government (Innovation and Science Australia) involvement in strategy, assessment, recommendations, and advice to ministers, while being actively managed with a significant degree of flexibility by a CRC management team from industry and research organisations.^{580,581} Australian CRCs aim to solve industry problems as well as improving the competitiveness, productivity, and sustainability of Australian industries. Education and training (PhD) programmes in CRCs are also industry focused – to build capacity and capability, increase R&D capabilities (in SMEs), and encourage industry take-up of research.582,579

CRCs are a collaboration between at least one business and one research organisation, although many include tens of partners ranging from all levels of government, international partners, not-for-profits, multinational corporations, SMEs, industry and community associations, and universities. What were originally Fight Food Waste CRC and Stop Food Waste Australia merged to form End Food Waste Australia, one of the world's largest public private partnerships involving more than 100 organisations.⁵⁸³

The CRC programme provides long-term funding for up to 10 years, and short-term funding, known as the CRC-P (project) stream, for up to three years. The CRC-P uses a matched funding model of grants between AU\$100,000 and AU\$3 million, and in collaboration with at least two Australian industry organisations, including one SME, as well as one Australian research organisation.⁵⁷⁹
Over 200 CRCs have been supported since 1991,⁵⁸⁰ and currently there are 24 active CRCs across a wide range of topic areas including End Food Waste CRC, Food Agility CRC, and Future Food Systems CRC.

New Zealand has had a funding scheme conceptually similar to the CRC programme although more limited, the Partnerships Scheme. Its funding has been reduced and there have been no new Partnerships funded since 2018.⁵⁸⁴

7.1 The research and innovation landscape for food loss and waste prevention

In 2023, R&D spending by the primary sector totalled \$139 million (just under 4% of New Zealand's total research spending) while \$89 million was spent on R&D by the food manufacturing sector, with over 100 entities engaging in R&D in both sectors (9% of entities performing R&D in New Zealand).^{*,585}

Given the small scale of New Zealand's research enterprise, leveraging existing research innovations, including from offshore, is essential.⁵⁸⁶ In FLW prevention (see <u>box 11</u>), both adopting innovations and developing new practices or technologies are useful at all stages of the supply chain. There can, however, be trade-offs between developing or tailoring new innovative technologies versus leveraging already existing innovations, including costs of development versus adaptation and the expertise required.

So 11: Government funding for research and innovation in food loss and waste prevention Although significant research funding in New Zealand is directed towards research into food in its many dimensions, relatively little has been focused on FLW prevention. Three funding streams are particularly relevant.

- MfE has a Waste Minimisation Fund to accelerate New Zealand's transition towards a low emissions and low waste economy. There was over \$120 million available in this fund for 2023 and 2024, which is funded through the Waste Disposal Levy. Funding is available for infrastructure and other enabling systems to reduce landfill emissions from organics waste, including diverting FLW. The minimum grant sizes are \$50,000 for R&D, business cases, or innovation projects and \$150,000 for all other project types.⁵⁶⁶
- MPI funds innovative projects in New Zealand's food and fibre sector through the SFF funding with about \$60 million available each year through partnerships and grants, and has minimum co-investment requirements but no cap.⁵⁸⁷ Commercially-oriented proposals will have a maximum of 40% of total cost contributed by MPI; for community-driven proposals MPI will contribute up to 50-60%; and proposals with public-good benefits may have up to 80% of costs contributed.⁵⁸⁸ Projects must also demonstrate sustainable benefits to New Zealand (environmental, economic, social, or cultural).
- Callaghan Innovation provides grants for commercialisation of technology, focused on developing an innovation ecosystem or deep technology (for example, Ārohia Trailblazer⁵⁸⁹ and deep tech incubators⁵⁹⁰). They also provide support for businesses to carry out R&D through early career researchers, to build their experiences in industry (covering 6 months of salary of graduate students Masters and PhD⁵⁹¹ or internships for tertiary-level students),⁵⁹¹ and grants for companies to develop R&D capability.⁵⁹²

^{*} As a comparator, the computer services industry was the sector with the largest R&D expenditure of over \$1 billion (20% of total R&D spending in New Zealand) in 2023.

7.1.1 The contribution of business to food research and innovation

Businesses contributed 59% of total R&D spending in New Zealand in 2022. This percentage has significantly increased since 2010, when businesses contributed 40% of total national R&D spending. 23% of food, beverage, and tobacco manufacturing firms reported engagement in R&D in 2013 and 58% reported engaging in innovative activities, both figures higher than the New Zealand average (8% for R&D and 46% in innovation).⁵⁹³ However, there has been a reduction in food manufacturing business reporting performing R&D to 18% in 2022.⁵⁸⁶

Large companies can spend significant amounts on R&D. Fonterra, for example, has had an R&D centre based in Palmerston North since 1927 and spends more than \$80 million a year on R&D,⁵⁹⁴ with plans to increase total annual R&D investment by over 50% to around \$160 million per annum in 2030.⁵⁹⁵

7.1.2 Examples of public sector organisations that do research into FLW prevention

University based research groups in Aotearoa, most notably the UoO's Food Waste Innovation research group, lead FLW initiatives and also work collaboratively across the public and private sector. The Food Waste Innovation research group has been involved in measuring FLW, developing reduction strategies, developing technology for producers to reduce FLW, and studying techniques to influence consumer behaviour change.⁵⁹⁶

AgResearch has a focus area on sustainable farming systems, climate change, and transformation (including a circular bioeconomy model for maximising value while minimising waste from production through to consumption by quantification and evaluation of resource flows through food systems), as well as piloting new technologies to prevent and repurpose waste.⁵⁹⁷

Plant & Food Research has a 'smart food systems' research theme that includes developing tools that track food along the supply chain through sensing technologies, and how by-products from production can be used in innovative ways.⁵⁹⁸ They have also carried out a mapping study on organic waste in the Marlborough, Nelson, and Tasman regions to identify the available biomaterial and the potential for bio-waste reuse or upcycling.⁵⁹⁹

7.1.3 Collaboration between industry and research is vital for sector impact

New Zealand's food and beverage industry accounts for almost half of all goods and service exports⁶⁰⁰ and software-as-a-service is currently growing by 16% annually and could become a major industry for Aotearoa.⁶⁰¹ Combining these industries could create an economic niche for New Zealand.⁶⁰¹ In the New Zealand agricultural sector, AgriTech New Zealand takes a key role in developing the technology ecosystem for agricultural applications – connecting researchers, innovators, investors, regulators, and enablers.⁶⁰² It is difficult for technology companies to reach a viable scale within Aotearoa due to the size of the market; therefore, catering to the export market is

generally necessary for New Zealand businesses. However, in agri-tech there is a trade-off between maintaining competitive advantage in New Zealand's food industry against overseas companies through innovation and technology, particularly in dairy and farming, versus the profits from exporting innovations or technology overseas. Agri-tech that is developed to prevent FLW, however, could be a technology that does not undercut New Zealand's own industry and can also be exported.

Agri-tech that is developed to prevent FLW, however, could be a technology that does not undercut New Zealand's own industry and can also be exported. The relevant technology transfer system in New Zealand encompasses industry bodies (such as DairyNZ, Beef + Lamb New Zealand, HortNZ, Seafood New Zealand, Aquaculture New Zealand, and the New Zealand Forest Owners Association), consultants, and commercial firms, as well as the technology transfer offices of research organisations. To reach farmers and growers, for example, technology transfer can involve working with farmers oneon-one, presenting to groups or being involved in large community events such as field days. However, surveys show technology transfer support for farmers and growers



is fragmented, thinly spread and insufficient.⁶⁰³ More connectivity between groups, or the development of a more formal system to aid technology transfer in the food system, could support better uptake of technology.

7.2 Supporting research and innovation to prevent food loss and waste

Within the primary industry sector in particular, Aotearoa has shown it can quickly adopt new innovations and technologies. New Zealand is well placed to lead the way in sustainable development and FLW reduction goals, but this is not a research agenda that has been prioritised.

There is also a pressing need to develop the systems and processes to generate and collate data, to be able to share evidence around FLW and practices that prevent FLW. The New Zealand Research Information System, developed by MBIE, has already begun databases of research projects focused on managing extreme weather, and COVID-19, and could potentially collate data on FLW projects.⁶⁰⁴

Collaborative partnerships between public and private sectors or internationally could benefit research and innovation in FLW.^{*,605-609} For example, the UoO has a research project funded through the Climate Emergency

There is also a pressing need to develop the systems and processes to generate and collate data, to be able to share evidence around FLW and practices that prevent FLW.

Response Fund as one of MfE's national food waste reduction programmes, which brings together a consortium of organisations in the retirement housing sector, including Arvida, Bupa, and the Retirement Villages Association. The project aims to measure and reduce the amount of food waste in commercial kitchens in the sector.⁵⁶⁹ Research in agricultural and environmental sustainability can also benefit from international research collaborations. For example, INRAE – France's research organisation for agriculture and the environment – and Science New Zealand have an agreement for researcher exchange programmes and collaborative projects including quality food products, carbon neutral and climate resilient agri-food, and the circular bioeconomy.⁶¹⁰ New Zealand's strength in agricultural and food research is internationally recognised and attractive to potential collaborators.

International private collaborations can also benefit New Zealand's food ecosystem. A new raspberry breed with early maturing has been developed by Plant & Food Research and Northwest Plant Company from Washington in the US, which extends the harvest window and is machine

^{*} For example, Plant & Food Research and Zespri funded a 50/50 joint venture Kiwifruit Breeding Centre, which commenced operating from October 2021 to create new cultivars such as RubyRed[™] Kiwifruit. Zespri also co-funded research with Kiwifruit Vine Health (known as KVH) to minimise impacts of the bacterium *Pseudomonas syringae pv. Actinidiae* (psa-*V*) infection on Kiwifruit and support growers with detection, management and technology transfer. Also, Zespri provides a \$2 million innovation fund with key areas including waste management, sustainable packaging, technologies for growers to reduce carbon footprint and increase efficiency of yield.

harvestable, as well as meeting the requirements of processors.⁶¹¹ Rabobank also carries out and collaborates on food and agribusiness market research, including a recent collaboration with KiwiHarvest investigating food waste at the consumer level.^{612,613}

Understanding and addressing the issue of FLW requires involvement from all levels of society. Citizen science approaches have been used to investigate various aspects of the food system, including FLW, internationally.⁶¹⁴⁻⁶¹⁶ Partnerships between industry and academia may be well placed to capitalise on the opportunities presented by citizen science.

8. Preventing food loss and waste requires a systems approach

Economic incentives and supply chain power dynamics drive much food loss and waste

In our present food supply chains, the costs of FLW are accrued where the FLW is realised, not where it is caused. This increases the costs of doing business for our growers and other upstream actors, while providing no incentive for more powerful downstream actors not to order more than they will need. While there is an argument that loss occurring upstream has fewer environmental impacts than FLW occurring downstream,¹³⁶ interventions that shift waste from one part of the supply chain to another, but do not reduce the amount wasted, are ultimately not preventing FLW. Mechanisms such as innovative contracting arrangements and the newly implemented Grocery Supply Code could help in moving the economic costs of lost and wasted food to the part of the supply chain where it was caused.

A 'both, and' approach will get us further than 'either, or'

There is no single cause of FLW in our food system, and so it is unlikely that there will be a single solution. At the highest level, we should invest in technical and other solutions that optimise the operations of the supply chain, while also ensuring that the market is set up to incentivise prevention of FLW. Similarly, we can implement prevention measures simultaneously across the supply chain, rather than prioritising a single stage or intervention.

In isolation, some approaches to preventing FLW may be in tension with our wider sustainability goals. Packaging can preserve shelf life, but may create more plastic waste. We will need to take a whole-of-system approach to reconcile these types of tensions.

Technology won't save us, but it can help

Scientists, engineers, and entrepreneurs, both in Aotearoa and abroad, are innovating in ways that could prevent FLW. From breeding crops with characteristics that promote shelf life, to smart packaging that can help consumers, to digitalisation that can make supply chains more efficient, and many other advances, there are abundant opportunities in this space. With the right support for both the scientific and food systems, we can capture these opportunities.

However, some of the drivers of FLW are not amenable to technical improvements, but instead come from the structure of the market and the relationships between different actors across the supply chain. Consumer preferences and habits also play a role. These factors will need to be addressed alongside any gains we can make through scientific innovation.

Uncertainty doesn't have to be a barrier to action

Our prevention efforts are hampered by uncertainty. There are significant data gaps across the supply chain, and few mooted interventions across the supply chain have been rigorously evaluated. We can do better in this space, by developing our data collection, sharing, and storage capabilities and by advocating for evaluation as an integral part of intervention. This will stop uncertainty being a barrier to action in the long-term.

In the meantime, we can act despite uncertainty. We can triangulate information we do have about the operations of food supply chains in New Zealand and qualitative reports of drivers of FLW here with international data on FLW. We can experiment in our policy and practice where we have reason to believe action might prevent waste, and be rigorous in our evaluation of new policies, while being open to a possible need to change course when we have more information.

Annex 1: Loss and prevention in horticulture and arable production

There are many innovations that could reduce food loss in the horticultural and arable sectors. In this annex we survey some of the possibilities by sector.

Table 10: A non-exhaustive overview of technological solutions and management practices for food loss prevention in horticulture (apples, kiwifruit, stone fruit, grapes, and tomatoes) and arable (grain) production, including challenges to implementation and examples from New Zealand. Abbreviations: CODC = Central Otago District Council, e-Bins = electronic bins, FLW = food loss and waste, IoT = internet of things, t = tonnes, 1-MCP = 1-Methylcyclopropene, NZFN = New Zealand Food Network

Loss type and causes	Solution	Description	Challenges	Examples from NZ
Apple pre-harvest	Automated	Replacing some of the need for manual	Automated harvesters require a large	Automated harvesters by Abundant
	harvesting and	harvesting through adopting automated	upfront cost so adoption by growers is	Robotics are being used in the Hawke's
Losses in 2012 in the Central Otago	precision	harvesting applications would enable	currently low.618	Bay area to harvest apples. ⁶¹⁹
region were estimated at 2,669 t of	agriculture.	greater efficiency and reduce labour		
apples. ²¹³ Causes of pre-harvest		shortage issues. ⁶¹⁷		
losses were largely attributed to an	Genetic	Genetically engineered apple varieties	Genetic engineering has had limited	No specific data identified.
export focus meaning it was un-	engineering.	can reduce the impact of severe	commercial success in horticultural	
profitable to harvest some fruit		weather events making them stress	crops to date. The commercial release	
that didn't meet specifications. ²¹³		tolerant and less likely to fail to meet	of genetically modified, and especially	
		quality specifications. Artic apples are a	transgenic, crops also has many	
Pre-harvest losses of apples are		good example, as they don't brown	regulatory challenges. ²²⁴	
also caused by labour shortages,		easily when bruised, allowing for fewer		
and disease/rots. ^{213,131}		rejections. ⁶²⁰		
	Gleaning.	Farmers could implement gleaning in	Gleaning has many liability concerns	Perfectly Imperfect is an example of a
		their apple orchards by enabling	which can affect farmer willingness to	food loss on form in NIZ ²⁸
		gleaning organisations access to pick	Act 2014 protocts doppers from sivil and	1000 loss on farm in NZ.20
		any remaining apples post-narvest.	criminal liability that results from	
			consumption of food donated by the	
			donor, as long as a few requirements	
			are met ^{112,622}	
	In-field sorting.	Utilising an in-field sorting machine is a	No specific data identified.	No specific data identified.
		way growers can reduce the cost of		
		post-harvest handling, making		
		harvesting more of the apple crop		
		economically viable. ⁶¹⁸		
	Protective	Protective netting in orchards has a	No specific data identified.	Protective netting is widely used for
	netting.	range of benefits for reducing fruit loss.		orchards across NZ. Companies offering
		They protect against intense sunlight		netting products and services include NZ
		and extreme weather events like hail		Canopies, ⁶²³ AGROW plastics, ⁶²³ etc.

Loss type and causes	Solution	Description	Challenges	Examples from NZ
		and strong winds and provide a physical barrier against pests. ¹³⁸		
Apple post-harvest	Automated harvesting.	Automated harvesting applications minimise bruising and the picking of unripe fruit, reducing FLW. ⁶¹⁷	No specific data identified.	No specific data identified.
The CODC estimated 124 t of apples were lost in 2021 after being harvested. ²¹³ Losses are	Donations.	Donations give a destination for food not consumed or sold and help minimise food loss. ⁶²⁵	No specific data identified.	No specific data identified.
caused by fruit deterioration during transport such as firmness loss, decay, internal disorders such as	Edible coatings.	Edible coatings can be applied and improve the shelf life of apples minimising post-harvest food loss. ⁶²⁶	No specific data identified.	No specific data identified.
scab, and bruising. ⁶²⁴	Good harvest practices.	Quality issues can be greatly reduced in the apple industry when good harvest practices are employed: ⁶²⁴ for example, avoiding harvesting fruit during wet weather, using padded buckets, gentle filling of bins, opting for air suspension in transport vehicles, choosing the smoothest road to the packhouse, training pickers to minimise damage, strategically placing waste bins for unwanted fruit, avoiding direct sunlight on harvested fruit with bin covers.	No specific data identified.	Hazel Technologies ⁶²⁷ works within the NZ horticultural sector across a range of produce to deploy innovations to extend and optimise shelf life and quality of produce, for example from the apple industry.
	In-field sorting.	In-field sorting integrated with harvest assist platforms can be used to minimise losses in post-harvest storage by avoiding contamination of diseased fruit with healthy fruit and allowing gentle filling of buckets, reducing post-harvest loss. ⁶²⁸	Currently there is minimal in-field sorting technology available to growers and upfront cost is high, which means adoption by growers is low. ⁶¹⁸	No specific data identified.
	IoT.	IoT can provide pest and disease information to farmers so they can quickly control them. ⁶²⁹	No specific data identified.	No specific data identified.
	Protected cultivation.	Protected cultivation by way of nets, foil, or glass means fruit is less affected by weather conditions and pests, which reduces food losses. ⁶³⁰	Upfront investment costs of netting solutions are high making them unviable for some growers. ⁶²⁴ For example, some growers have high operating costs due to needing to purchase costly licences to grow the fruit in NZ. Therefore, many	GROWTECH group is developing specific hail and wind resistant nets for the kiwifruit industry. ⁶³¹

Loss type and causes	Solution	Description	Challenges	Examples from NZ
			orchards opt to not cover their crops with netting solutions. ⁶³¹	
	Training of staff.	Training of staff to focus on gentle techniques for handling fruit when harvesting can greatly reduce rejections due to quality issues associated with bruising and cuts on the fruit. ⁶³⁰	Training staff can be challenging; for example, many horticultural workers come from overseas and so there may be a language barrier.	No specific data identified
Kiwifruit pre-harvest	E-bins.	E-bins help open up kiwifruit picking jobs to a wider range of people as it	Automated harvesters require a large upfront cost so adoption by growers is	The electronic e-Bin was developed at University of Waikato in collaboration
Losses (no specific data identified). Caused by labour shortages, low		reduces the need for pickers to carry around a bucket, which can be physically straining. ²²⁸	low. Many mechanical harvesters also require orchards to alter the shape of the apple tree. ⁶¹⁷ Variation of crops and environment makes automated	with Zespri to combat the labour shortage issue and is reported to be implemented in the 2023/2024 season ²²⁸
pests, birds, and weather. ⁴⁸			harvesters difficult to make commercially available. ⁶¹⁷	
	Gleaning.	Farmers could give access for gleaning services in their apple orchards to rescue produce that was not harvested. ⁶²¹	Gleaning has many liability concerns which can affect farmer willingness to implement this strategy. ⁶²¹ The <i>NZ Food</i> <i>Act 2014</i> protects donors from civil and criminal liability that results from consumption of food donated by the donor, as long as a few requirements are met. ^{112,622}	Perfectly Imperfect is an example of a company exercising gleaning to rescue food loss on farm in NZ. ²⁸
Kiwifruit post-harvest The kiwifruit industry reports an estimated 38,000-40,000 t of post-	Donations.	Donations give a destination for food not consumed or sold and help minimise food loss by giving to those in need. ⁶²⁵	No specific data identified.	A Tasman grower donated 35 T of damaged kiwifruit to NZFN's distribution hub in late 2020. ⁶³²
harvest food loss per year, amounting to approximately 8% of the total yield in NZ. ⁴⁷ Losses caused by pre-mature harvesting, improper handling, mould.	Training of staff.	Training of staff to focus on gentle handling of fruit when harvesting can greatly reduce rejections due to quality issues stemming from bruising and cuts on the fruit. ⁶³⁰	Training staff can be challenging; for example, many horticultural workers come from overseas and so there may be a language barrier.	No specific data identified.
perishability, or loss of firmness during storage causing rejection. ⁴⁸	Treatments.	Calcium treatments have success at improving the storage of kiwifruit to maintain their firmness levels for longer. ⁶³³ 1-MCP treatments and melatonin have been successful at delaying ripening, reducing weight loss, and maintaining fruit firmness. ⁶³⁴	No specific data identified.	Hazel Technologies is a company that works with NZ kiwifruit growers to provide solutions to extend shelf life, reduce rejections, and combat food loss. They have developed a sachet containing 1-MCP to extend post- harvest shelf life. ⁶²⁷

Loss type and causes	Solution	Description	Challenges	Examples from NZ
Stone fruit pre-harvest	Gleaning.	Farmers could give access for gleaning services in their orchard to rescue the	Gleaning has many liability concerns which can affect farmer willingness to	Perfectly Imperfect is an example of a company excising gleaning to rescue
In 2021 volumes of loss due to non-		remaining produce post-harvest.621	implement this strategy. ⁶²¹ The NZ Food	food loss on farm in NZ. ²⁸
harvested fruit in Central Otago			Act 2014 protects donors from civil and	
were estimated to be 1,086 t (9%)			criminal liability that results from	
of cherries, 163 t (7%) of apricots,			consumption of food donated by the	
233 t (4%) of peach/nectarines. ²¹³			donor, as long as a few requirements	
Losses were caused by labour			are met. ^{112,622}	
shortage, low market price,	IoT.	IoT has the ability to provide real time	Farmers and agricultural companies	Harvest is a NZ company which
overproduction, pests, birds, and		data for the orchard environment which	tend to only adopt new technologies	integrates IoT into their monitoring
weather. ²¹³		can assist in early pest detection,	based on financial assessment, which is	systems. ⁶³⁷
		maturity assessment and temperature	hard to demonstrate until the	
		control which can help the farmer make	technology has been adopted.030	
		Geosions to ensure quality of fruit.		
		nost detection systems 635		
	Drotactive covers	Installing covers allows protecting	Installation of covers requires a large	No specific data identified
	Protective covers.	against weather events nests and can	initial invostment and also has	No specific data identified.
		also improve rinening ⁶³⁶	reoccurring costs due to	
		also improve ripering.	maintenance. ⁶²⁵	
Stone fruit post-harvest	Treatments.	1-MCP treatments can extend shelf life	No specific data identified.	No specific data identified.
		in stone fruit. ⁶²⁷		
In 2021 volumes of harvested food	Value-add	Value-add products are a way to	Usually, some infrastructure required to	Eden's Orchards New Zealand makes a
loss in Central Otago were	products.	manage waste and give a premium to	make a value-add product which could	wide range of juices from imperfect fruit
estimated at 1,121 t (9%) of		the farmer. Stone fruit have a variety of	be costly in the beginning for the	including cherry juice. They are also
cherries, 76 t (3%) of apricots, and		beneficial properties and have many	farmer.	trialling freezing of processing grade
693 t (12%) of peach/nectarines.		valorisation options. ⁶³⁸		cherries preventing them from going to
Harvested fruit losses were caused				waste. ⁶³⁹
when harvesting the fruit was not				
economically viable, and/or the				
weather conditions impacted the				
achievement of cosmetic				
specifications. ²¹³				

Loss type and causes	Solution	Description	Challenges	Examples from NZ
Grape pre-harvest	Gleaning.	Grape growers could implement	Gleaning has many liability concerns	Perfectly Imperfect is an example of a
		gleaning onto their vineyards which can	which can affect farmer willingness to	company excising gleaning to rescue
Loss (no specific data identified).		rescue produce that has not been	implement this strategy. ⁶²¹ The NZ Food	food loss on farm in NZ. ²⁸
Main causes of loss labour		harvested by allowing volunteers and	Act 2014 protects donors from civil and	
shortages, low market price,		initiatives to pick remaining grapes post-	criminal liability that results from	
overproduction, pests, birds,		harvest. ⁶²¹	consumption of food donated by the	
weather. ⁴⁸			donor, as long as a few requirements	
			are met. ^{112,622}	
	Mechanised	Mechanised harvesters could be	Mechanised harvesting equipment can	No specific data identified.
	harvesters.	implemented to solve the issue of	end up been more expensive than hiring	
		labour shortages in vineyards and	workers and could cause the price of	
		reduce losses. ⁶⁴⁰	grapes to increase. ⁶⁴⁰	
	Precision	Precision agriculture has a place in	No specific data identified.	Croptide has developed a novel sensor
	agriculture.	viticulture and has the ability to reduce		which allows growers to read the plant
		risk of crop loss, disease incidence,		status directly from stem.642
		labour costs and time management.641		
	Waste reduction	NZ Wine reports that 98% of wineries in	No specific data identified.	No specific data identified, but see
	programmes.	NZ have waste reductions programmes		Beyond the bin: Capturing value for food
		in place and 75% of wineries have		loss and waste for ways that grape marc
		reduction initiatives. They have also set		is being used.
		a target for the NZ wine industry to		
		achieve zero waste by 2050.643		
Grape post-harvest	Donations.	Donations give a destination for food	No specific data identified.	No specific data identified.
		not consumed or sold and help		
No specific data identified.		minimise food loss.625		
	Treatments.	Ozone treatments have been shown to	No specific data identified.	Hazel Technologies works with grape
		reduce decay of cold store grapes and		growers and has 1-MCP treatments that
		are effective for controlling post-harvest		can extend shelf life. They have also
		losses. ⁶⁴⁴		developed a mat that absorbs excess
				moisture reducing risk of fungal
				infection. ⁶²⁷

Loss type and causes	Solution	Description	Challenges	Examples from NZ
Tomato production pre-harvest	Gleaning.	Farmers could implement gleaning onto	Gleaning has many liability concerns	Perfectly Imperfect is an example of a
		their orchard which can rescue produce	which can affect farmer willingness to	company excising gleaning to rescue
One report estimates between 71%		that has not been harvested by allowing	implement this strategy. ⁶²¹ The NZ Food	food loss on farm in NZ. ²⁸
and 84% of produced tomatoes		volunteers and initiatives to pick	Act 2014 protects donors from civil and	
were left in the field and not		remaining produce post-harvest.621	criminal liability that results from	
harvested by two commercial			consumption of food donated by the	
growers in Australia.41			donor, as long as a few requirements are met. ^{112,622}	
The main causes of loss include				
labour shortages, low market price,				
overproduction, pests, birds,				
weather. ⁴⁸				
Tomato post-harvest	Donations.	Donations give a destination for food	No specific data identified.	No specific data identified.
		not consumed or sold and help		
Estimated post-harvest loss for two		minimise food loss by giving to those in		
commercial tomato growers in		need. ⁶²⁵		
Australia was found to be between	Evaporative	Evaporative cooling has been shown to	No specific data identified.	Hazel Technologies works with tomato
40.3% (55.34 t) and 55.9% (29.61 t)	cooling.	be effective at increasing shelf life in		growers and has 1-MCP treatments that
of the total harvestable product. It		tomatoes. ⁶⁴⁶		can extend the storage potential of
was determined that between				tomatoes. ⁶²⁷
68.6% and 86.7% of undamaged,				
edible, harvested tomatoes were				
rejected as out grades and				
consequently discarded due to				
product specifications.645				
Grain/arable	Cameras.	Farm Wave has developed a camera	No specific data identified.	No specific data identified.
		that attaches to harvesters and lets the		
No specific data identified.		farmer know how much grain is left		
		behind during harvest so they can make		
		adjustments to reduce waste. ²²⁷		
	Integrated crop	Pests and diseases are the main reason	No specific data identified.	No specific data identified.
	management.	for losses in grain production.		
		Integrated crop management is a way to		
		reduce losses. An example of this is		
		introducing natural predators as a way		
		to control pests. ⁶⁴⁷		

Annex 2: International examples of organic bans to landfill

An increasingly common approach to FLW prevention is the implementation of landfill bans to incentivise other uses of surplus food further up the FLW hierarchy (see figure 1). A scan of international policy settings indicates a variation in ways that bans have been implemented, interpreted, and enforced. Since waste is generally managed locally across jurisdictions, there are cases where organics bans to landfills are in place in particular locations, but not country-wide (see table 11, table 12, table 13, and table 14). Some local authorities have enforced recycling laws or bans based on the availability of landfills, increasing costs of moving solid waste to other areas, or legislative requirements (for example, directives from the EU).

Food may not be the primary organic material targeted, and therefore the broader classification of 'organic' bans has been used here. Other classifications include compostable, biodegradable, or biological waste. Depending on the municipal solid waste (MSW) management plans, the extent of the ban may be also dependent on who the generator of FLW is. For example, regulations may only be applicable to those that produce over a certain quantity of FLW per week.

The examples below include – a range of organic bans implemented with the aim of reducing the total volume of organic material in landfills. Details across jurisdiction vary, with different rules applying to processed organic material, such as anaerobic digestate. Additionally, recycling laws that mandate source separation are not included.

The information in <u>table 11</u>, <u>table 12</u>, <u>table 13</u>, and <u>table 14</u> includes examples of organic bans in different contexts (tables are grouped by continent; examples within tables are alphabetically ordered by country), summarising the scope and outcomes of organics bans, as well as additional information regarding their implementation. These tables are included to illustrate the range of policies internationally but are not comprehensive.

Africa

Table 11: Example of organic waste ban from South Africa. South Africa does not have any country-wide organics ban to landfill, however the Western Cape province has enforced a ban and set targets. Abbreviations: FLW = food loss and waste, t = tonnes.

Location	Policy	Year	Scope	FLW specifics	Diversion results of waste	Additional notes
	instrument(s)	enacted				
Western Cape	National Environmental Management: Waste Act, 2008 (Act No. 59 of 2008). National norms and standards for disposal of waste to landfill (GN No. R. 636 of 2013).	2022 for 50% reduction, 2027 for 100% reduction. ⁶⁴⁸	All organic waste from households, factories, businesses, and restaurants. ⁶⁴⁹	No detail for FLW, but refers to abattoir waste, green waste, and liquid waste. ⁶⁴⁹	More than 600,000 t of organic waste diverted from landfill. ⁶⁴⁹	Ban requires to divert up to 50% of all organic waste from landfilling, with the aim to increase this to 100% by 2027. ^{648,650}

Asia

Table 12: Examples of organic waste bans from Indonesia and South Korea. Indonesia does not have a country-wide organics ban to landfill, however West Java province does have an enforced ban. South Korea has a country-wide ban specifically for sending FLW to landfills. Abbreviations: FLW = food loss and waste, t = tonnes.

Location	Policy instrument(s)	Year	Scope	FLW specifics	Diversion results of organic	Additional notes
		enacted			waste	
Indonesia:	02 /PBLS.04/DLH ⁶⁵¹	2024651	All organic	Includes FLW.652	No data available.	The Sarimukti landfill fire in 2023 has resulted in the ban
West Java			waste, no			of organic waste from Bandung City, Cimahi City,
			specific			Bandung Regency, and West Bandung Regency by the
			groups.			Governor of West Java. ⁶⁵² The ban has been enforced
						since 1 January 2024.

Location	Policy instrument(s)	Year	Scope	FLW specifics	Diversion results of organic	Additional notes
		enacted			waste	
South Korea	Waste Management Act: prohibition of direct landfilling of FLW. ⁶⁵³	2005 ⁶⁵³	All FLW generated by agriculture, fisheries, leftovers from production, distribution, and cooking. ⁶⁵³	FLW going to landfill is limited to residues and contaminants from treatment facilities. ⁶⁵⁴	Food waste generation from homes and restaurants had increased from about 13,500 t per day to about 15,500 t per day (14.5%) between 2010-2020.655	The prohibition of FLW to landfill did not lessen the amount of FLW generated, but rather enforced other government supported/funded options for waste diversion. Several other policies and legal instruments support the diversion of FLW from landfills. ⁶⁵⁴ In 2013 a weight-based food waste fee for consumers to use food diversion services was extended. The volume- based waste free system charges consumers based on the weight of their FLW. ⁶⁵⁶ Individuals reduce weight by removing excess water content from their food. ⁵⁶⁴

North America

Table 13: Examples of organics waste bans in North America. Canada does not have any nationwide organics ban; however, Nova Scotia (province) and Vancouver (city) have enacted policies. The US do not have a country-wide ban; however, a number of states and smaller territories have taken the additional step of banning FLW at various levels and generators. Abbreviations: CMR = Code of Massachusetts Regulations, FLW = food loss and waste, MSW = municipal solid waste, t = tonnes.

Location	Policy	Year	Scope	FLW specifics	Diversion results of	Additional notes
	instrument(s)	enacted			organic waste	
Canada: Nova Scotia (Province)	Solid Waste- Resource Management Regulations, enabled under the Nova Scotia Environment Act. ⁶⁵⁷	1998 ⁶⁵⁷	All compostable organic material from industrial, commercial, institutional, and residential sources. ⁶⁵⁷	FLW and other organic material to be composted. ⁶⁵⁷	No current diversion data readily available.	In addition to the landfill ban, Nova Scotia has promoted its bans through education (to establish waste-separation standards) and funds a third party to deliver. ⁶⁵⁸
Canada: Vancouver (City)	Solid Waste By- Law No. 8417. ⁶⁵⁹	2015	All organic material, including food scraps. ⁶⁵⁹	All FLW, residential and non- residential. ⁶⁵⁹	In 2021, more than 400,000 t of yard and FLW have been diverted from landfill. ⁶⁶⁰	Owners and occupiers of non-residential properties producing FLW (or organic waste) must have a diversion plan for FLW. ⁶⁵⁹ Waste is inspected when it is delivered to a regional disposal facility, where excessive amounts of food results in the hauler paying a 50% surcharge on the cost of disposal.

Location	Policy	Year	Scope	FLW specifics	Diversion results of	Additional notes
	instrument(s)	enacted			organic waste	
US: Massachusetts (State)	310 CMR 19.000: Solid Waste Facility Regulations. ⁶⁶¹	First implemente d in 2014, lowered limits in 2022. ⁶⁶¹	Commercial food/organic wastes from facilities generating 0.5 t of waste per week. ⁶⁶¹	Food/organic waste may be donated, turned to animal feed, composted, or anaerobically digested. ⁶⁶²	Prior to the lowered 2022 limit, Massachusetts had already diverted 360,000 t of FLW in 2022. ⁶⁶²	Food/organic waste is banned from disposal or transport for disposal. ⁶⁶¹ This is part of a wider Massachusetts Department of Environmental Protection's goals to reduce waste by 30%. ⁶⁶³ This ban is not specifically for landfills, as Massachusetts also incinerates and transfers waste out-of-state. There is a local food action plan to support FLW reduction and rescue. FLW reduction is actively tracked in businesses and institutions. ⁶⁶²
US: Seattle (City)	Seattle Municipal Code 21.36.082– 21.36.083. ⁶⁶⁴	2015 ⁶⁶⁴	All households and businesses. ⁶⁶⁴	FLW and other recyclables must not be disposed as MSW. ⁶⁶⁴	Even though Seattle's population has increased by 35% between 2000 – 2020, commercial and residential waste trends are promising. Residential data shows the amount of waste generated per person per day decreased by 16% (apart from COVID- 19 time periods). ⁶⁶⁵	Households are liable to pay a \$50 fine, and businesses may need to pay a fee if more than 10% of waste contains prohibited material, including FLW. ⁶⁶⁴ Food and yard waste services must be subscribed to, with monthly costs relating to the size of the cart. ⁶⁶⁶
US: Vermont (State)	Universal Recycling Law (Act 148). ⁶⁶⁷	2020 ⁶⁶⁷	All households, businesses, and institutions. ⁶⁶⁷	All FLW; pre- and post-consumer. ⁶⁶⁸	Estimates place organic waste diversion at nearly 70,000 t in 2020. Additionally, more than 4,000 t of food were also rescued. ⁶⁶⁸	General recycling law was passed in 2012 and implemented in stages to reduce landfill disposal. ⁶⁶⁸ Residents may dispose of meat and bones in residual waste if they compost at home. Businesses may dispose of small amounts of FLW, especially in packages that are too small for de-packaging machines. ⁶⁶⁷

Europe

Table 14: Examples of organic bans to landfills across Europe. Included here are examples from across the EU, Austria, Belgium, Czechia, Denmark, Estonia, Finland, France, Germany, Hungary, Lithuania, Luxembourg, Netherlands, Norway, Poland, Slovenia, Sweden, Switzerland, and Scotland. Abbreviations: BMW = Biological municipal waste, FLW = food loss and waste, kg = kilogram, m = million, MBT = mechanical-biological treatment, MJ = megajoules, MSW = municipal solid waste, Mt = megatonne, PAYT = pay as you throw, t = tonnes, TOC = total organic carbon.

Location	Ban	Policy instrument(s)	Year	Scope	FLW specifics	Diversion results of	Additional notes
	status		enacted			organic waste	
EU Member States (27+)	No ban, targeted reduction	Landfill directive 1999/31/EC European waste framework directive 2008/98/EC European new waste framework directive 2018/851	1999	All EU member states are to reduce the amount of biodegradable MSW sent to landfill.	Obligations include preparing FLW prevention programmes, implementing related actions, monitoring, and reporting on progress achieved at various supply chain levels. ^{669,669}	A 2020 report stated that diversion of FLW was more than 9.5 Mt per year in the EU. ⁶⁷⁰	An amendment passed in March 2024, raised legally binding FLW reduction targets to 40% for the consumption level (retail, distribution, restaurants, and food services as well as households) and to 20% for food processing and manufacturing by 31 December 2030. ³⁴⁵
Austria	Ban	Landfill Ordinance Law Gazette II No. 164/1996. ⁶⁷¹	2004	MSW with TOC content >5% is banned. There are exceptions for MBT treated waste. ⁶⁷¹	There is no specific FLW regulation.	Austria achieved almost 0% BMW landfilled by 2006. ⁶⁷¹	In addition to the landfill ban, Austria uses a landfill tax, incineration tax, producer responsibility measures, PAYT and mandatory separate collection systems to decrease the incentive for landfilling organic waste. ⁶⁷²
Belgium	Ban only in states of Flanders and Wallonia.	VLAREMA- Flemish Regulation on Sustainable Materials Management and Waste. ⁶⁷³ Walloon Waste- Resources Plan. ⁶⁷⁴	2007	All biodegradable waste streams. ⁶⁷⁵	Wallonia – Management of FLW using ladder (hierarchy) model, with specific prevention measures. ⁶⁷⁴ Brussels Capital Region does not have landfills.	The reported rate of landfilled biodegradable municipal waste has been reported as 0% in 2017, 2018, and 2019. ⁶⁷⁵	The landfill ban added to the existing landfill tax in Flanders and Wallonia, there is also an incineration tax in all three regions of Belgium encouraging a waste hierarchy. ⁶⁷⁵

Location	Ban	Policy instrument(s)	Year	Scope	FLW specifics	Diversion results of	Additional notes
	status		enacted			organic waste	
Czech Republic	Aiming for a ban in the future	Act no. 541/2020 Coll. on Waste (Waste Act) ⁶⁷⁶	2020	All MSW. ⁶⁷⁶	There are plans to expand the separate collection of FLW. ⁶⁷⁶	The Czech Republic generated about 1.5 Mt of BMW in 1995. The amount of biodegradable MSW landfilled in 2020 was 28.3% of the reference value in 1995. ⁶⁷⁶	Currently policy instruments include PAYT, a landfill ban for separately collected waste streams suitable for re-use or recycling and a landfill tax. The landfill tax will gradually increase, and a landfill ban for recyclable, recoverable and mixed MSW (including dry matter with a calorific value of 6.5 MJ/kg) from 2030 onwards will be introduced. ⁶⁷⁶
Denmark	Ban	Environmental Protection Act and the Statutory Order on Waste. ⁶⁷⁷	1997	All organic and combustible MSW. ⁶⁷⁸	The separate collection of FLW is not covered by all municipalities (56%). ⁶⁷⁷	The landfilling rate of Denmark in 2020 was slightly below 1% of the BMW generated in 1995, and for many years has been 1%- 2%. ⁶⁷⁷	Denmark has had a landfill tax in place since 1987. As of 2023, non- household waste generators will be obliged to sort their waste for separate collection. ⁶⁷⁷
Estonia	Partial ban	Regulation of the Ministry of the Environment 29.04.2004, No 38 (RTL 2004, 56, 938). ⁶⁷⁹	2008	All untreated and unsorted waste. The share of biodegradable waste in MSW landfilled cannot exceed 20% by weight. ⁶⁸⁰	The Estonian Waste Act aims for door-to-door collection of bio-waste, or home composting to become mandatory for all residents by the 31st December 2023. ⁶⁸⁰	Estonia landfilled 9% (29,071 t) of biodegradable MSW in 2019 as proportion of the BMW reported in 1995 (317,000 t). ⁶⁸⁰	The limit to biodegradable content in landfilled MSW gradually decreased from 45% in 2010, 30% in 2013 and 20% in 2020. Other fiscal policies include a landfill tax, municipal waste user charge, PAYT, packaging tax, penalties, and fines. ⁶⁷⁹
Finland	Ban	Government Decree on Landfills 331/2013 – Prohibiting landfill of organic waste. ⁶⁸¹	2016	All organic and biodegradable MSW with TOC >10%. ⁶⁸²	If TOC >10%, FLW included in ban (no specific FLW regulation).	Finland landfilled 3% BMW in 2016, and 1% for 2017, 2018, and 2019, of the total amount of BMW produced in 1995. ⁶⁸²	Statistics suggest that the most significant reduction in landfilling between recorded 1997 and 2016 is from landfill tax. ⁶⁸¹
France	Ban	Code de l'environnement Loi No 92-646, 1992. ³⁴⁴	2002	All untreated MSW.	Separate bio-waste collection will become mandatory by 2025. ³⁴⁴	France reported 15% biodegradable waste being landfilled for 2016 of the total amount BMW produced in 1995. ³⁴⁴	Landfill tax differs by classification, and there are plans to increase its landfill tax by 2025. ³⁴⁴

Location	Ban	Policy instrument(s)	Year	Scope	FLW specifics	Diversion results of	Additional notes
	status		enacted			organic waste	
Germany	Ban	Landfill ordinance. ⁶⁸³	2005	Waste with a TOC >3% (18% for mechanical- biologically treated waste) is banned. ⁶⁸³	Separate bio-waste collection was made mandatory from 2015. ⁶⁸³	Germany achieved 0% BMW landfilled in 2016. ⁶⁸³	In addition to the landfill ban, Germany uses PAYT and mandatory separate collection systems to decrease the incentive for landfilling. ⁶⁸³
Hungary	Partial ban	Decree No. 385 of 2014 (XII. 31.) Korm of the Government concerning the conditions of providing waste management public service. ⁶⁸⁴	2003	All untreated MSW is banned since 2002, a partial ban on organic wastes since 2003. ⁶⁸⁴	No specific FLW regulation.	Hungary reported 28% biodegradable waste landfilled in 2019, as a percentage of the total amount of biodegradable MSW produced in 1995. ⁶⁸⁴	Hungary also has a landfill tax and PAYT scheme implemented in over 80% of the country. ⁶⁸⁴
Lithuania	Ban	Law on Waste Management <i>(No. VIII-787)</i> . ⁶⁸⁵	2003	All untreated MSW and all biodegradable waste from gardens, parks, and green areas. ⁶⁸⁶	No specific FLW regulation.	Lithuania reported 3% BMW landfilled in 2019, as a percentage of the total amount reported in 1995. ⁶⁸⁶	In addition to the landfill ban, Lithuania uses a landfill tax, producer responsibility measures, PAYT and some mandatory separate collection systems to decrease the incentive for landfilling organic waste. ⁶⁸⁶
Luxembourg	Ban	Specific policy name not found.	Specific year not found.	Ban on untreated MSW and organic waste (TOC > 5%). ⁶⁸⁶	No specific FLW regulation.	Luxembourg reported 5% BMW was landfilled in 2016 as a percentage of the total amount reported in 1995. ⁶⁸⁶	Door-to-door separate collection is the main system in cities, towns and suburbs, and rural areas for bio-waste. Bio-waste is also collected at civic amenity sites. ⁶⁸⁶
Netherlands	Ban	Decree on Landfill and Waste Disposal Bans (Besluit stortplaatsen en stortverboden afvalstoffen) ⁶⁸⁷	1995	Mixed MSW is banned. 35 waste streams are captured, including all combustible and biodegradable waste, with TOC >5%. ⁶⁸⁸	No specific FLW regulation.	Netherlands reported 2% BMW landfilled in 2016, as a percentage of the amount in 1995. ⁶⁸⁸	There is also a Landfill Tax (1995), which has been marginally increasing since its implementation. Since 2018, there are over 60 waste streams captured in the landfill ban. ⁶⁸⁸

Location	Ban	Policy instrument(s)	Year	Scope	FLW specifics	Diversion results of	Additional notes
	status		enacted			organic waste	
Norway	Ban	Waste Regulations (2004) – Chapter 9, Landfill bans for waste types (includes landfill ban on biodegradable waste – introduced 2009). ⁶⁸¹	2009	All biodegradable waste streams with a TOC >10% or organic matter >20%. ⁶⁸⁹	Up to 70% of municipalities offer separate door-to-door collection of bio-waste. ⁶⁸¹	Landfilling decreased from 25% of MSW in 2001 to 6% in 2010. ⁶⁸⁹	The landfill ban joined the existing policy initiatives, Landfill Tax (1999) and Incineration Tax (1999). ⁶⁸¹
Poland	Ban	Specific policy name not found.	2013	All biodegradable waste collected separately (2013). All combustible waste with >5 % TOC, >8% LOI, Calorific value > 6MJ/kg (2016). ⁶⁹⁰	Door-to-door separate collection of biodegradable waste is mandatory. ⁶⁹⁰	Poland reported 13% BMW landfilled in 2016 and 11% in 2018, as a % of the total amount in 1995. ⁶⁹⁰	Poland has a PAYT system, but it is targeted only for non-household waste producers. ⁶⁹⁰
Slovenia	Ban	Decree on waste to landfill (Official Gazette of the Republic of Slovenia, no. 10/2014 and 54/15). ⁶⁹¹	2011	All BMW streams based on calorific content and TOC. ⁶⁹⁰ (No mention of limits for calorific content and TOC found)	No specific FLW regulation.	Slovenia reported 15% BMW landfilled in 2019 (four-year derogation period), as a percentage of the generated amount in 1995. ⁶⁹⁰	Slovenia has a landfill tax and PAYT system based on waste container volume and the frequency of collection. ⁶⁹⁰
Sweden	Ban	Ordinance (2001:512) on the disposal of waste Sveriges riksdag, 2001, Förordning (2001:512) om deponering av avfall (SFS). ⁶⁹² As of 2024, a new Swedish law declares that everyone must separate their FLW. ⁶⁹³	2005 ⁶⁹⁴	All organic and combustible waste streams. ⁶⁹⁵	The Swedish government has set targets for FLW such as by 2018, at least 50% of food waste from households, commercial kitchens, shops and restaurants will be sorted out and treated biologically, ⁶⁹⁶ and to decrease by 20% weight per person from 2020 to 2025.	Over the past five years, the overall landfilling rate in Sweden is less than 1% and reported zero biodegradable waste landfilled in 2018. ⁶⁹⁵	In addition to the landfill ban, Sweden uses a landfill tax, producer responsibility obligations and PAYT to decrease the incentive for landfilling organic waste. ⁶⁹⁵
Switzerland	Ban	Ordinance on the avoidance and the disposal of waste. ⁶⁹⁷	2000 ⁶⁹⁸	All combustible, including biodegradable waste streams. ⁶⁹⁸	No specific FLW regulation. ⁶⁹⁸	No BMW has been landfilled in the country since 2006. ⁶⁹⁸	A level of zero BMW landfilling has been achieved by the combination of incineration and recycling. ⁶⁹⁸

Location	Ban	Policy instrument(s)	Year	Scope	FLW specifics	Diversion results of	Additional notes
	status		enacted			organic waste	
Scotland	Aiming for	Landfill (Scotland)	2025 ⁶⁹⁹	All BMW streams with a	FLW included in ban (no	No data.	Ban was initially planned to be
	a ban in	Regulations 2003,		TOC >5%. ⁶⁹⁹	specific FLW		enforced in 2021 but has since
	the future	Regulation 11(3).699			regulation). ⁷⁰⁰		been pushed back to December
							2025.701,702

Annex 3: Interventions to prevent food waste in retail and food service

Interventions to prevent food waste in RFS are wide-ranging, reflecting the diversity of causes of food waste in these sectors, as well as an array of sector specific challenges in managing flows of food. In this annex, we highlight commonly cited interventions designed to reduce food waste in RFS settings, as derived from ReFED's Solution's Database⁴²⁴ and MfE's literature review³⁹⁴ of organic waste reduction initiatives in households and businesses. For clarity, we have grouped interventions into four general categories: 1) sales strategies, 2) solutions specific to food service, 3) transport, logistics, and inventory management, and 4) waste measurement and tracking. It is worth noting that much of the evidence around these interventions is not specific to New Zealand, so care would need to be taken when applying interventions to the New Zealand context to achieve effective implementation.

Table 15: An overview of interventions to prevent food waste in the RFS sectors, the type of waste they address, how they're used, and available evidence around their effectiveness in reducing food waste. Interventions are grouped by categories (sales strategies; solutions specific to food service; transport, logistics, and inventory management; and waste measurement and tracking). Abbreviations: 2D = two dimensional, EECP = European Cluster Collaboration Platform, EPA = Environmental Protection Agency, m = million, MfE = Ministry for the Environment, Mt= megatonne, RFS = retail and food service; t = tonnes.

Intervention	Type of waste	How it is used	Evidence of effectiveness
Sales strategies			
Assisted distressed sales.	Retail: spoilage waste.	The use of third party companies or apps to sell salvaged, out-of-spec, overstocked, or out-of- date (but safe) food at a discounted rate. ⁷⁰³ Examples of overstock food buyers in the US include Natural Choice Foods, ⁷⁰⁴ Marvell Foods, ⁷⁰⁵ and Mid-America Food Sales. ⁷⁰⁶ In NZ, companies like Surplus Direct ⁷⁰⁷ buy and sell a range of surplus stock, including some food products.	ReFED estimates that greater uptake of assisted distressed sales across US retailers could divert up to 306,000 t of food waste annually. ⁷⁰³ The scale of operations at US company Natural Choice Foods ⁷⁰⁴ is indicative of the potential for this intervention; the company handles about 200,000 cases of overstock food every week, and are connected to 1,000 secondary market outlet stores across 35 states. However, there is little to no scientific study evaluating the efficacy of this intervention. Self-reported data from markdown applications – a strategy for assisted distress sales (see below) – indicates that this intervention can have a significant impact on food waste.
Dynamic pricing. ⁷⁰⁸	Retail: spoilage waste.	This automates the process of marking down products in-store, in particular to provide discounts on products near their expiration date.	Modelling suggests that dynamic pricing outperforms static pricing in reducing food waste and increasing profit for fresh produce retailers, ⁴¹¹ however, limited real-world study is available. ReFED estimates that implementation of dynamic pricing in the US retail sector could divert 352,000 t of food waste annually. ⁷⁰⁸ In Finnish retailers, early evidence indicates that end-of-day product markdowns can both reduce food waste and increase gross profit margins. ⁷⁰⁹ In Italy, a store introduced dynamic pricing technology by Wasteless, ⁷¹⁰ and self-reported an in-store waste reduction in their fresh product categories of 39%. At Tesco, a large UK retailer, a dynamic pricing approach resulted in a 2% reduction in produce waste and annual saving of £30m. ⁷¹¹

Intervention	Type of waste	How it is used	Evidence of effectiveness
Markdown alert applications. ⁷¹²	RFS: spoilage waste.	It lets consumers know that safe but short shelf life foods can be purchased at a discount. Examples include Too Good To Go, ResQ club ⁴²¹ and Foodprint in Aotearoa. ⁴²⁰	There is a shortage of independent, quantitative research on the efficacy of markdown applications, but self-reported data are encouraging. ReFED estimates that implementation of markdown apps in the US RFS sectors could divert 1 Mt of food annually. ⁷¹² According to their 2021 impact report, ⁷¹³ Too Good To Go has saved over 52m meals since 2016, with a user base of 20m people and 82,000 stores across 17 countries. Marketing statistics from ResQ club (primarily in Finland) suggest that food service outlets can increase revenue by 2%-6% and sell half on of their surplus food. ⁴²¹
Specific to food se	ervice		
Doggy bags.	Food service: plate waste.	This allows customers to take food home with them.	Several studies indicate that doggy bags can be an effective tool for reducing plate waste. ⁷¹⁴ For example, a Dutch study found that an opt-out strategy for doggy bags in food service settings increased doggy bag uptake to 74% of all leftovers (versus 27% for an opt-in approach). ⁷¹⁵ However, their research suggests that an opt-out approach can have detrimental effects on restaurant or service staff evaluations, which may require changes to how doggy bags are presented to consumers. Given the scale of plate waste in NZ food service, an estimated 33% of food waste in the sector, ⁴⁰⁴ addressing doggy bag uptake is important. However, uptake of doggy bags can depend on cultural attitudes towards leftovers ⁷¹⁶ and social stigma-related factions, ⁷¹⁴ as well as how take aways are approached within the establishment. ⁷¹⁵ For example, survey data from Italy suggests that just 30% of restaurant goers ask for the remainder of their food in a doggy bag, while 70% don't, citing reasons such as embarrassment or shyness (28%) or a lack of available containers (18%). ⁷¹⁷ A Japanese study of attitudes towards doggy bags suggests that appealing more to customers concern for the environment by highlighting the sustainability benefits of using doggy bags avould help uptake. ⁷¹⁸ There is little specific data on how much food taken home in doggy bags actually gets eaten.
Challenges or competitions.	Food service: plate waste, preparation waste, spoilage waste.	Businesses can sign up to measure their waste and receive support to reduce it over time ³⁹⁴ (e.g. Nashville Mayor's Food Saver Challenge, ⁷¹⁹ Denver's Food Waste Restaurant Challenge, ⁷²⁰ and the Worldchef's Food Waste Challenge). ⁷²¹	Challenges can be effective, but the evidence base for their long-term efficacy isn't as well developed as for other interventions. ³⁹⁴
Menu planning.	Food service: plate waste, preparation waste, and spoilage waste.	Menu planning requires quantifying and measuring of food that is required or adjusting the menu for food not often sold. ⁷²² Optimising the use of ingredients across multiple dishes, aligning with customer demands, and reducing surplus inventory.	Menu planning is a logical step in reducing food waste within food service, although there has been little scientific study linking the two specifically. To optimise menu planning, ReFED recommends that food service providers track waste operations (see row on waste tracking and measurement below).
Messaging for diners.	Food service: plate waste.	Includes initiatives like smaller plates at buffets, removal of dining trays at 'all-you-can-eat' diners, better menu descriptions, smaller portioning of side orders, etc.	As highlighted by an MfE review, ³⁹⁴ messaging interventions have been effective at reducing food waste in numerous food service environments, including university canteens, staff cafeterias, dining halls, buffets, and restaurants. For buffet style dining, ReFED estimates that buffet signage in 'all-you-can-eat' food service business in the US could divert up to 47,000 t of food waste annually. ⁷²³

Intervention	Type of waste	How it is used	Evidence of effectiveness
Plate size,	Food service:	Changing the size, colour, or shape of plates in	Changing the size and shape of plates is evidenced to reduce over portioning in university dining
colour and	plate waste.	food service settings can reduce over portioning	contexts. A US study found switching from a large round plate to a smaller oval plate reduced
shape.		by consumers. ⁷²⁴⁻⁷²⁶	plate waste from 15.8% to 11.8%. A study of NZ tertiary institutions efforts to reduce food waste
			shows at least one university dining hall facility has reported successful food waste reduction
			from switching to smaller plates. ⁷²⁶ Additionally, a study of restaurants in China found the colour
			of the plate could also influence plate waste, noting that plate waste was reduced when the
			plate was a warm colour. It should be noted, however, this study does not distinguish whether
			this was a result of reducing over portioning or of over eating by consumers. ⁷²⁵
Reduced	Food service:	Creating smaller size options on the menu can	Decreasing portion sizes can significantly reduce plate waste, as observed in several studies in a
portion sizes. ⁷²⁷	plate waste.	reduce over portioning.	variety of food service settings. ⁷²⁸⁻⁷³² While effective at preventing food waste, businesses may
			be reluctant to implement this measure, considering it a risk to their brand image. ⁷²⁷ ReFED
			estimates that reduced portion sizes across US food service business could divert up to 2.4 Mt of
			food waste annually. ⁷²⁷ In the context of buffet style venues, ReFED estimates that using smaller
			plates to reduce portion sizes could divert up to 47,000 Mt of food waste in the US annually. ⁷³³
Trayless dining	Food service:	Removing trays from 'all-you-can-eat' dining	There is mixed evidence on the efficacy of going trayless, with most studies focusing on this
facilities.	plate waste.	facilities to reduce over portioning by	intervention in university cafeteria settings. Several studies indicate that going trayless in
		consumers. ⁷³⁴	cafeterias can reduce food waste, ^{724,735,736} although this isn't always the case ⁷³⁷ and outcomes
			can vary by food type. ⁷³⁸ ReFED estimates that going trayless in 'all-you-can-eat' food service
			businesses in the US could divert up to 99,000 t of food waste annually. ⁷³⁴
Transport, logistic	s, and inventory ma	anagement	
Decreased	Retail:	Many retailers and food service businesses have	ReFED estimates that decreases in minimum order sizes across US retailers could divert 120,000
minimum order	spoilage waste.	minimum order quantities built into their	t of food waste annually. ⁴²⁸ However, there is little to no scientific study evaluating the efficacy
quantity.		ordering and replenishment systems to	of this intervention, nor available real-world data.
		streamline operations, but this can result in	
		sending slightly more food than meets demand	
		to every location. ⁴²⁸ Trimming order sizes can	
		help avoid overbuying and subsequent	
		wastage. ⁴²⁸ This is particularly relevant for highly	
		perishable products like meat, fresh produce,	
		and seafood.	

Intervention	Type of waste	How it is used	Evidence of effectiveness
Enhanced demand planning. ³⁹¹	RFS: spoilage waste.	This applies advanced forecasting and planning methods (e.g. machine learning software systems like Afresh ⁷³⁹) to accurately predict consumer demand for products and efficiently manage inventory levels. This approach can integrate various data sources, such as historical sales data, current market trends, promotional calendars, and even weather forecasts, to create a more accurate and dynamic forecast of product demand. ⁴²²	This is viewed as a key approach to reduce food waste in the RFS sectors, ⁴⁴⁴ although there is limited independent research quantifying its impact. ReFED estimates that implementation of dynamic forecasting in the US retail sector could divert 258,000 t of food waste annually. ³⁹¹ Modelling from the Netherlands ⁷⁴⁰ suggests that dynamic forecasting performance varies greatly across products and scenarios, but consistently outperforms 'naïve' forecasting (where the forecast equals sales from the last period) in reducing food waste. Financially, this can have a notable real-world benefits; a decade of improvements to Tesco's dynamic forecasting system have saved the company a reported £100m a year. ⁷⁴¹
Improved storage conditions.	RFS: spoilage waste.	This ensures better ventilated warehouses and increased cold storage capacity.	ReFED estimates that improved temperature monitoring for food service in the US has the potential to divert 742 t of food waste. ⁷⁴² A Swedish study focused on cheese, dairy, deli, and meat products found that a reduction of supermarket storage temperatures 4°C to 2°C could lead to 19% reduction in meat waste, albeit at an increased cost due to energy requirements. ⁷⁴³
Increased delivery frequency.	RFS: spoilage waste.	This reduces dwell time in distribution centres between suppliers to RFS outlets.	A study ⁷⁴⁴ comparing four retailers in Norway found that those with more frequent deliveries had lower levels of food waste, with notable impacts on meat, dairy, and convenience food waste (60%-100% reduction) as well as on fruit and vegetable waste (50% reduction). ReFED estimates that increases in delivery frequency across US retailers could divert up to 49,000 t of food waste annually. ⁴²⁹
Inventory management tools for traceability.	RFS: spoilage waste.	Labelling products to track their progress along the supply chain is key to identifying waste hotspots, supply chain disruptions, and aligning supply and demand. Important technology in this space includes 2D barcodes ⁷⁴⁵ (see <u>section 4.2</u>).	GS1 has reported that Woolworths' trials of 2D barcodes in Australia has improved traceability and reduced food waste by 40%. ⁷⁴⁶ As this is emerging technology, data are limited.
Minimised on hand inventory.	Retail: spoilage waste.	Reduces the amount of stock held in distribution centres or the back of stores. While 'safety stock' is needed to account for unanticipated spikes in demand, reducing the amount of product sitting in storage (and losing shelf life) can improve product freshness and reduces waste. ⁷⁴⁷	ReFED estimates that decreases in minimum order sizes across US retailers could divert 61,000 t of food waste annually. ⁷⁴⁷ However, there is little to no scientific study evaluating the efficacy of this intervention, nor available real-world data. Considerations such as the impact of smaller orders on transport and transport related emissions would need be understood before rolling out such an intervention. Note that on-hand inventory in distribution centres or out the back of stores is different to overstocked shelves. Overstocking is also an issue but has different drivers and therefore different solutions. Overstocking is addressed in <u>box 6</u> .
Stock rotation.	Retail: spoilage waste.	Organising and managing inventory ensures older stock is sold before newer stock.	Stock rotation within stores and on shelves is an expected routine to ensure that expired stock is not on shelves. ⁷⁴⁸ ReFED estimates that 350,000 t of food waste could be diverted in the US if products that expire first are sold first. ⁴¹⁷ This method does become challenging for products that may not have an expiry label and are highly perishable (e.g. fresh produce) Generally, empirical evidence for the specific effect of stock rotation on food waste is lacking.

Waste tracking an	Vaste tracking and measurement						
Intervention	Type of waste	How it is used	Evidence of effectiveness				
Apps and	Retail:	Apps and programmes can measure, visualise,	According to the EECP, restaurants using Waste Master ⁷⁴⁹ can decrease their food waste by				
programmes ³⁹⁴	preparation	and track waste within businesses. E.g. Waste	30%-50%, sometimes more. ⁷⁴⁹				
	waste and	Master, ⁷⁴⁹ Leanpath, ⁷⁵⁰ Smart Kitchen, ⁷⁵¹ and	Analysis of 735 hotels, restaurants, and canteens in Europe found that simply measuring food				
	spoilage waste.	Winnow. ⁷⁵² Retailers can also build or modify	waste led 61% of organisations to minimise their waste. ⁷⁵³ ReFED estimates that waste tracking				
		their own in-house inventory software to track	in food service has the potential to highlight areas to optimise menus and reduce cost. ⁴¹⁵ In the				
	Food service:	food waste and prioritise actions.	US, keeping track of waste could help divert over 1 Mt of food waste ⁴¹⁵ (but by no means only				
	plate waste,		through improved menu planning, the impact of which is unquantified).				
	preparation						
	waste, and						
	spoilage waste.						
Training and	Retail:	Typically provide guidance on food waste	Evidence from real-world trials suggests that training and toolkits can effectively reduce food				
toolkits for food	preparation	measurement and reduction techniques.	waste. Results from San Diego Smart Kitchens pilot programme saw a 42% reduction in food				
waste	waste and	Examples include: Unilever Food Services' 'Wise	waste. ⁷⁵⁷ Training of food service workers in 15 US schools resulted in significant food waste				
measurement	spoilage waste.	Up On Waste' Toolkit, ⁷⁵⁴ New South Wales' EPA	reductions across fruit (14%), vegetables (7%), and milk (4%). ⁷⁵⁸ Similarly, training of kitchen				
and		'Your Business is Food' programme, ⁷⁵⁵ and the	staff in restaurants ⁷⁵⁹ and hospitals ⁷⁶⁰ has also proven effective at reducing pre-consumer				
reduction.394	Food service:	Smart Kitchen Initiative in the United States. ⁷⁵⁶	waste.				
	plate waste,						
	preparation						
	waste, and						
	spoilage waste.						

Annex 4: Household food waste measurement methodologies

Methods for calculating household food waste can vary between countries and can make household food waste data comparisons challenging.⁷⁶¹ Table 16 features countries identified to have better household food waste data by the UNEP *Food waste index report 2021*.⁴⁸² Solid waste studies are typically conducted for municipal solid waste, where categories of waste (for example, organic or food waste) are physically separated from general waste to be quantified and a proportional value is derived.^{762,763} Direct measurement through food waste bin audits is a more detailed method than solid waste studies as it separates food waste specifically from municipal waste. Sorting can include discrete categories such as type of food (for example, fruit, vegetable, meat, etc.), specific food items (for example, tomatoes), and whether the waste was avoidable or unavoidable.^{403,764} The methodology column in <u>table 16</u> classifies items into solid waste studies or food waste bin audits, based on information that was readily available and identifies whether mixed methodologies or additional sources (such as surveys and diaries) were utilised for household food waste data.

Table 16: Country comparison of household food waste data and how it is measured. Abbreviation: kg = kilograms, Mt = megatonne, m = million, n = number of participants for whom data is reported, pp= per person, t = tonnes, UNEP = UN Environment Programme.

Country	Household food waste data	Methodology	Description of methodology	Factors to consider
Australia	Produced the largest proportion of food waste (38%), around 3.1 Mt (based on the reference year 2018-2019). ⁵⁷	Unknown.	Methodology was unclear; however data for estimated food loss was across the entire supply chain. Data from 169 sources across seven types of information routes were used for the overall analysis. Data quality was scored as part of the management system and the average score was 3.8/5. ⁵⁷	The rounded figure of 38% is in comparison to sectors like primary production (20%), processing (15%), distribution (3.1%), retail (6.3%), hospitality (15%), and institutions (3.0%). ⁵⁷
		Survey, diary, and bin audit.	Western Australia published their household food waste measurement methodology for data collected between 2020 and 2021 using a survey (n=324), an electronic diary (n=177), and a bin audit (n=103). ⁷⁶⁵	
Austria	Avoidable food waste ranged between 9.5-25.5 kg per capita. Rural areas produced less waste than urban areas. ⁷⁶¹	Food waste bin audit.	130 samples were collected from 9 representative rural municipalities and 1 urban municipality over 20 days in 2009. Food waste was manually sorted from residual waste and weighed.	Researchers notes that the state of degradation did make it difficult to categorise and identify items. Additionally, the values only give a snapshot of food waste in time.
Canada	79 kg per capita (estimated for 2016).482	Solid waste study.	There is no published national statistical data on household food waste. National residential food waste estimates were based on 56 studies conducted between 2008-2018 and the known quantity of waste provided by Statistics Canada for 2016. ⁷⁶³	Surrogate data was used for regions that did not have data available, and this could have applied to the calculation of food waste. ⁷⁶³

Country	Household food waste data	Methodology	Description of methodology	Factors to consider
Denmark	Produced 103 kg per household per year of avoidable food waste and 80 kg of unavoidable food waste (based on data collected between 2011 and 2012). ⁷⁶⁴	Food waste bin audit.	This analysis is based on 1,474 households across 5 municipalities. Samples were collected three times to account for any periodic variation in waste.	Recent publication from 2017 was not accessible in English.
Germany	59% of the food waste generated in the supply chain is from households, as a proportion of 10.9 Mt (based on data for 2020). ⁷⁶⁶	Unknown.	Unknown, but has been stated to be within the scope of EU reporting requirements, ⁷⁶⁷ which could include waste compositional analyses or direct measurements. Questionnaires and surveys are not used as measurement method for households. ⁷⁶²	59% is in comparison to processing and manufacturing (15%), primary production (2%), restaurants and food services (17%), and distribution (including retail at 7%). ⁷⁶⁶
Ghana	Food waste was differentiated between high- and middle-class income areas; however, both were around 0.2 kg pp per day (based on data collected between 2013 and 2014). ⁷⁶⁸	Solid waste study.	A large number of samples of general waste were collected from ten regions (in some regions over 1000 samples). These were manually sorted and weighed at a sorting centre.	This was a national municipal waste study. Food waste comprised of 44% (high-income) and 51% (middle-income) of the general household waste collected.
Malta	52% of general household waste contains food. ⁷⁶⁹	Solid waste study.	No methodology stated; however data is based on a 2011- 2012 survey by the National Statistics Office.	The survey is not accessible, however a publication from the EU has classified this survey as a waste compositional analysis. A value of 0.38 kg of food waste pp per day is implied for households. ⁷⁷⁰
Netherlands	Produced around 34 kg of solid food waste per capita (based on data collected in 2019). ⁵⁷³	Surveys, solid waste study, and food waste bin audit.	Multiple method study. Surveys (n=1000) captured self- reported data of solid food waste (solid, thick liquid, and dairy) and liquid waste. Municipal waste was collected from 130 households across 13 municipalities. Food waste was manually separated into categories.	Solid and liquid food waste is calculated separately. The total of 34 kg of solid food waste includes approximately 27 kg of food waste from municipal waste audits and an estimation of approximately 8 kg of food waste through other routes.
Norway	Produced 80.2 kg of total food waste per capita, of which almost 58% was edible food waste (based on data collected in 2011). ⁷⁷¹	Solid waste study and food waste bin audit.	Waste study of 220 individual households. Data was collected from 4 municipalities where food waste was not separated and assumed to be collected as municipal waste. Sampling areas were informed that samples would be collected anonymously on a random week and households could opt out of being sampled. Municipal waste was weighed and then food waste was separated into edible/non-edible categories and other sub-categories (fruit and vegetables, bread, leftovers, etc).	The study noted that a third of the edible food was still in the original packaging.

Country	Household food waste data	Methodology	Description of methodology	Factors to consider
Saudi Arabia	Produced 105 kg of food waste per capita. ⁴⁸²	Solid waste study.	The methodology for measuring household food waste is unclear.	The UNEP Food Waste Index Report 2021 indicated that the sample size for households and methodology was unclear, but categorised that a waste compositional analysis was carried out. ⁴⁸²
Sweden	Produced 94 kg of food waste pp, and 68 kg of food pp if excluding waste down the drain (based on data collected in 2018). ⁷⁷²	Solid waste study.	The methodology for measuring household food waste is unclear.	94 kg accounts for approximately 71% of the total food waste from the supply chain. Avoidable food waste measured at 26 kg pp disposed via the drain and 19 kg of avoidable food waste pp is either residual waste, separated waste, or in home composts. ⁷⁷² Sweden has been collecting data on national food waste since 2012. ⁷⁷³ It is likely that Sweden would be using the food waste measurement methodology set out by the European Commission.
UK	Produced 70% of food waste, around 6.6 Mt. Around 4.5 Mt of food waste was avoidable (based on 2018 data). ⁴⁰⁹	Solid waste study.	Compositional analysis of waste collected by 129 local authorities from general waste, kerbside food waste, food wate contaminated recycling, and waste recycling centre. ⁷⁷⁴	70% is a high value as the proportion of food waste generated is calculated post-farm gate (i.e. waste generated after primary production). ⁴⁰⁹
US	Produced 153 kg of residential food waste per household per year ⁷⁷⁵ (59 kg per capita ⁴⁸² and food waste accounted for 17% of total household waste). ⁷⁷⁵	Unknown.	The enhanced generation methodology infers household food waste using data estimates from 12 studies published after 2007. Extrapolated data is generalised based on assumptions and multiplied by a 'generation factor' to calculate estimates. ⁷⁷⁶	Households in the US are estimated to contribute to 24% of the food waste in the supply chain compared to 39% in the industrial sector. ⁷⁷⁵ The UNEP Food Waste Index Report 2021 suggests that US waste per capita values are only comparable to Australia and the UK if both residential and food service are combined (estimated at 123 kg per capita per year). The methodology is not recommended by the UNEP report for comparing data over time. ⁴⁸²

Annex 5: Household food waste prevention interventions

There is a growing number of studies on household food waste,⁷⁷⁷ and many interventions that have been proposed as solutions.^{778,779} However, selecting appropriate interventions for food waste prevention can be challenging. <u>Table 17</u> provides an introductory analysis of a few types of interventions. <u>Table 18</u> describes multipronged interventions combining multiple approaches. <u>Table 19</u> lists several tools that also play a role in household food waste prevention. These tables have been adapted from Auckland Council's review of interventions to reduce household waste in general, of which a significant number of interventions were targeted at food waste.⁷⁸⁰

Table 17: Singular intervention strategies and evidence for their use. Adapted from Auckland Council.⁷⁸⁰ Abbreviations: LFHW = Love Food Hate Waste, m = million, MTurk = Mechanical Turk.

Type of	Example	Goal	Was it successful?	Context	Conclusion and considerations
intervention	study				
Community workshops	Canadian multi-method study. ⁷⁸¹	To raise awareness about food waste and reduce food waste generation.	Some increase in participants' motivation to reduce food waste. ⁷⁸¹ This did not translate reduced food waste: there was no difference in the amount of food wasted compared with before the intervention ⁵²⁵	This study tested multiple interventions over 12 weeks and included workshops for one of the targeted groups. Workshop participation rates were low, even when childcare was provided to encourage participation.	Empirical evidence for community workshops is not strong. Community engagement strategies that build capabilities may be more effective, especially for individuals who are already motivated to do so. ³⁹⁴
Consumer challenges	NZ Food Waste Warriors Challenge in 2022. ⁵³³	Not published.	Social media reach of 15.1m, but no detail on food waste outcomes. ⁷⁸²	Campaign of online content in partnership with the One NZ Warriors. There was an incentive to win a pantry and fridge makeover worth \$1,500. ⁵³³	Challenges may work, but there is insufficient evidence on their effectiveness. ³⁹⁴ It may be valuable to undertake further research in this space, and to collect data on outcomes related to food waste as well as campaign reach.

Type of	Example	Goal	Was it successful?	Context	Conclusion and considerations
intervention	study				
	Brisbane's	Not published.	Not published.	Ongoing campaign. Residents are	
	Save Money			encouraged to text a free number to	
	Save Food			receive 10 text messages with	
	text			suggestions about how to reduce their	
	challenge.783			food waste over a three-week period.	
				The challenge is designed over the three	
				food categories that were recognised as	
				Brisbane's most wasted foods; meat,	
				salad greens, and bread. A survey at the	
				end of the challenge gives residents a	
				chance to win a food saver pack that	
				includes seeds, 'use it up' tape, a	
				cookbook with food waste tip, and food	
				covers. ⁷⁸³	
	Dutch 100-	To test if	There is some evidence here that	This study did not specially focus on food	
	100-100	informational	waste prevention behaviours were	waste but challenged 100 households to	
	challenge.784	strategies	improved compared to the control	live 100% waste free for 100 days in	
		reduced volume	group. Measured waste findings	2015. This included households needing	
		of residual	were however limited.	to actively engage with weekly	
		waste. This		assignments and waste comparisons.	
		includes food		Food related assessments related to	
		waste.		preparation waste, shopping, list making,	
				and recording about leftovers.	
Cooking classes	No data.				No evidence for cooking classes alone as a
					primary food waste reduction intervention,
					however an exploratory study suggested they
					can help with acquiring new skills. ⁷⁸⁵ Cooking
					classes have been utilised in tandem with
					other strategies (see <u>table 18</u>). Trials of cooking
					classes for food waste prevention may find it
					valuable to evaluate efficiency, effectiveness,
					and scalability.

Type of	Example	Goal	Was it successful?	Context	Conclusion and considerations
intervention	study				
e-Newsletter	Asda – retail influence on consumer behaviour in the UK. ⁷⁸⁶	Reducing reported food waste.	Self-reported reduction in food waste was observed in all arms of the study, including the control group, suggesting the e-newsletter was not effective.	A UK retailer compared the effectiveness of food waste messaging in e-newsletter, social media, and print/digital magazines with a control group comprising people who reported not seeing any of the interventions. The trial ran over six months and included 2,018 participants across all the groups.	The Asda study suffers from a low response rate, with 2,018 respondents fully completing all data collection activities out of 20,000 customers targeted. Further investigation may be useful. Auckland Council suggested that e-newsletters may be more suited as reminders rather than a primary intervention.
Food diary	Literature review on measurement methods (no specific location). ⁷⁸⁷		Not applicable.	Records how food is handled and usually includes a measurement component, for example, cups of rice discarded. Variations can also include app-based diaries or taking photos of the waste.	There is no empirical evidence of the use of diaries causing measured food waste reduction. ⁷⁸⁷ It is plausible that a diary has potential to reduce food waste as it makes the waste quantity visible and raises awareness about the issue. ²⁷⁹ Separately to their use as an intervention, diaries may also be useful for measuring food waste in evaluation of other interventions. They may underreport food waste, ⁷⁸⁸ but they are thought to be more reliable than surveys. ⁷⁸⁷
Meal kits	No data.				The use of meal kits and meal kit services may show some potential for reducing food waste at home; ⁴⁵⁹ however this is still in the early stages of research. There is a gap in the data on how much food waste occurs at the supplier itself. There is also a tension with excess packaging to consider; however one US study has reported that a meal kits' life cycle environmental impacts were lower than comparative meals made with store bought items. ⁴⁵⁹

Type of	Example	Goal	Was it successful?	Context	Conclusion and considerations
intervention	study				
Online quizzes and games	Canadian multi-method study. ^{781,525}	To raise awareness about food waste and reduce food waste generation.	In common with all arms of the study including the control group, there was no difference in the amount of food wasted compared with before the intervention, but the gamification arm had higher levels of participation and greater self-reported food waste reduction. ⁵²⁵	This study utilised a simple game design and the design of the game may have impacted engagement with quizzes.	The Canadian study was relatively small and did not report whether it was sufficiently powered to detect differences between the treatment arms. There is scope for more research here. Studies on apps using games and quizzes for food waste prevention are emerging, some of which are student led projects and focus on the development of the software rather than measured food waste reduction. ⁷⁸⁹
Prompts	US study on consumer willingness. ⁷⁹⁰	To determine if a person was willing to reduce food waste.	There was no difference in willingness to reduce waste between those who received various prompts and those who did not receive a prompt, although recipients of some of the prompts increased their perception of food waste as a social problem compared to the control group.	Participants (n=261) were recruited from the US using Amazon's MTurk crowdsourcing website, and randomly assigned to see one of three visual prompts or to a control condition.	Prompts are likely to be more useful for people who already engage in food waste reduction practices, or are likely to make behavioural changes to reduce food waste at home. ⁵²⁴ There are other types of prompts such as magnets or signs that act as reminders about particular food waste preventions behaviours, ⁷⁸¹ and placed where the prompt is useful. ⁷⁷⁹ For example, an "eat me" sign on leftovers in the fridge. Auckland Council supported exploring the delivery of prompts as an intervention. ⁷⁸⁰

Type of	Example	Goal	Was it successful?	Context	Conclusion and considerations
intervention	study				
	Literature review of different types of interventions. ⁷⁷⁹	Scan of practical and academic evaluations of prompts.	Two interventions showed measured food waste reductions; however, these were at a buffet restaurant and at a university dining facility. No data was provided on households.		
Social media	Asda – retail influence on consumer behaviour in the UK. ⁷⁸⁶	Reducing reported food waste.	Self-reported reduction in food waste was observed in all arms of the study, including the control group, suggesting the social media delivery was not effective.	A UK retailer compared the effectiveness of food waste messaging in e-newsletter, social media, and print/digital magazines with a control group comprising people who reported not seeing any of the interventions. The trial ran over six months and included 2,018 participants across all the groups.	The retail study did not find any effect of soci media on food waste, which the authors note was in contrast with lab-based studies. Eat Well Tasmania viewers found that learnin to set up a 'use it up' area, learning new cooking skills, and leftover food management practices as quite helpful. The authors concluded that social media might be a usefu tool to raise awareness.
	Eat Well Tasmania. ⁷⁹¹	Reducing food waste and eating well.	No, average food waste reduction based on survey data was not statistically significant.	This initiative utilised three 10-minute videos aimed at younger households (18- 35 year olds) targeting specific behaviours such as eating old food first; eating/storing leftovers; and storing food correctly.	Decision makers may need to consider the efficiency and effectiveness of a social media campaign if it is used as the only intervention for food waste prevention. The use of social media may be valuable to support a wider campaign to raise awareness. Auckland Council also recommended not to pursue social media campaigns as a primary intervention tool.

Name of the intervention	Components of the intervention	Food waste reduction data	Context	Conclusion and considerations
Food Lovers Masterclass ⁷⁹²	Workshops; emails; video content; incentive packs.	Nearly half of respondents to a survey at six weeks after the workshop reported less household food waste than in a survey before the workshop. ⁴⁹¹	This intervention is delivered by Waste-Ed with Kate across NZ. Alongside the workshop there are also surveys, emailed video content, and text surveys. The workshops have been well attended across the country by attendees that sign up at varying costs based on council funding. Some locations also have incentive packs that include tools like storage packs or "eat me" stickers. Food waste measurement strategies have included using an ice cream container.	Because people self-select into the workshops, participants are likely to be highly motivated to reduce their food waste. More evidence is needed on whether these types of workshops can be effective more broadly.
HomeLabs ⁷⁹³	Information; kitchen caddies; food growing kits; food boxes.	Households were able to reduce food waste by 28% over the course of the intervention.	This was a small, coproduced intervention which included just 5 households in Dublin. Different elements of the intervention were delivered each week for a total of 5 weeks.	It is not possible to draw conclusions of the programme's effectiveness on the basis of this small study. The study also did not include information about food waste reduction after the intervention. An individualised intervention such as the HomeLabs would be challenging to deliver at a large-scale as it is resource and cost intensive.
Reduce Food Waste, Save Money ⁷⁹⁴	Information; website; emails; postcards; fridge magnet; food storage container; freezer stickers; grocery list pad.	Avoidable food waste decreased by 30% and was statistically significant when compared to the control group.	This study had multiple interventions that were used in a two week period for 54 single family households in a Canadian city (London) with no kerbside food collection programme. The methodology directly collected garbage samples and manually sorted and weighed food waste.	The study was focused on single family household units and so excluded other types of family units or those in apartments. Understanding whether the food waste practice was sustainable over time was out of scope for the study.

Table 18: Multipronged interventions and evidence for their use. Adapted from Auckland Council.⁷⁸⁰

Name of the	Components of the	Food waste reduction data	Context	Conclusion and considerations
intervention	intervention			
Waste Watchers ⁷⁹⁵	Educational website; weekly reminders; laboratory visits.	The intervention showed a 28% decrease in household food waste.	In this study, 53 individual families (targeting one person in the household, primarily female) were recruited to participate in a five-week educational intervention delivered through a website ⁷⁹⁶ with downloadable content, podcasts, infographics, videos, and text. Each week, participants were reminded to check the website for new content, and regular surveys. Participants also conducted a laboratory visit pre-and post-intervention to observe and assess food-based habits at home.	The study suggests that policy interventions such as date labels, packaging for longer shelf life, and retail purchasing practices may have an impact. Those who dropped out of the study were from lower income households.
Weeknight Supper Savers	Online cooking classes; text messages; cookbook; child friendly knife; meal and shopping planners; 'Eat first' container; food storage guide	Waste audits suggested total food waste decreased by 7.4%. ⁷⁹⁷	This was a 4 week Canadian study focusing on 18 families with children (9-12 years) aimed at the whole family. The aim of the study was to reinforce food literacy and also reduce food waste.	The study did not include a control group and therefore the effectiveness of the interventions are harder to determine. The sample size is also small, and the households were mainly white and had high incomes. Additionally, 17 mothers and 12 fathers participated, but the text messages were targeted at mothers and children.

Table 19: Tools used in New Zealand or internationally to prevent food waste and evidence for their use. Adapted from Auckland Council.⁷⁸⁰ Abbreviations: AI = artificial intelligence.

Example of tool	Stage(s) of	How it is used	Factors to consider	When will it be available
	intervention			to scale?
Al, apps, and websites	Planning. Preparing.	There are recipe generators that can find recipes based on the ingredients available. PAK'nSAVE have Savey Meal-bot website that utilises OpenAI's ChatGPT to create recipes. ⁷⁹⁸	These tools are new, and there is not yet evidence that they have contributed to food waste reduction. ⁷⁹⁹ Uses of these tools will also require a warning to avoid glitches if the app recommends using non-grocery items. ⁸⁰⁰	Currently available.
			Apps that monitor and track food waste may need manual input of produce, which may be a barrier to use.	

Example of tool	Stage(s) of	How it is used	Factors to consider	When will it be available
Ethylene scavengers	Storage.	Ethylene scavengers slow down the ripening process and increase the shelf life of some fruits and vegetables. ⁸⁰¹ They are mostly used to preserve produce after harvest and before it reaches consumers (see <u>table 6</u>). There are some commercial examples for households such as: Bluapple [®] , ⁸⁰² Fridge Friend [®] , ⁸⁰³ and Purifie [®] FridgeMate. ⁸⁰⁴	There is a lack of experimental research on the effectiveness of ethylene scavengers at a household level, and furthermore its ability to prevent food waste at this scale. ⁸⁰⁵ It has been used as part of one wider food reduction study, but its use was not the major focus. ⁸⁰⁶	Currently available, but there is little empirical evidence about household use.
Kitchen products or practices	All stages.	There are many examples of simple products such as storage containers or pouches, food vacuum sealers, dehydrators, and portion measuring tools. There are also traditional techniques to preserve food at home like fermenting and canning.	People wanting to try fermenting and canning may need support to ensure that the food remains safe.	Currently available.
Smart bins	Disposal.	There is a case study of 'BinCam', a disposal bin equipped with a camera that shares images to a social media group in order to influence food waste behaviour. ⁸⁰⁷ Another product on the market is Mill, an in-home food recycling system that converts food waste into dry animal feed. ⁸⁰⁸ This is more applicable for capturing value from food waste (see <u>Beyond the bin:</u> <u>Capturing value from food waste</u>).	BinCam was experimental. There are privacy and data issues that arise from this method and it is not recommended to prevent food waste. ⁷⁸⁰	Mill is available internationally but not in NZ; limitations include cost to scale and requirements for power and internet.
Smart fridges	Storage. Planning. Preparation.	Smart/intelligent fridges can work in tandem with apps to support food management practices at home. ⁴⁷⁹ This could also incorporate cameras that can relay images back to the user about the contents in the fridge. Recent models also claim to use AI to support with preparation. ⁸⁰⁹	The use of gamification to teach people how to sort their fridges to reduce food waste are still in early stages of research. ⁷⁸⁹ We still do not know about the effectiveness of smart fridges on food waste prevention. Furthermore, the cost of owning a smart fridge is a barrier to implementation.	Currently available, costly to scale.
Abbreviations

1D	One dimensional
1-MCP	1-Methylcyclopropene
2D	Two dimensional
3D	Three dimensional
AI	Artificial intelligence
APS	Advanced planning and scheduling
AR	Augmented reality
BCG	Boston Consulting Group
BERD	Business enterprise sector research and development
BMW	Biological municipal waste
CODC	Central Otago District Council
CoRE	Centre of Research Excellence
CRC	Cooperative Research Centre (Australia)
CRISPR	Clustered regularly interspaced short palindromic repeats
DEFRA	Department of Environment, Food and Rural Affairs (UK)
DIGAD	Dairy Industry Good Animal Database
DLT	Distributed ledger technologies
E-bins	Electric bins
EECP	European Cluster Collaboration Platform
EFWA	End Food Waste Australia
EPA	Environmental Protection Agency
ERP	Enterprise resource software
FAO	Food and Agriculture Organization (of the UN)
FEFO	First expired first out
FIAL	Food Innovation Australia Limited
FIFO	First in first out
FLW	Food loss and waste
FSANZ	Food Standards Australia New Zealand
GDP	Gross domestic product
GDSN	Global data synchronisation network
GERD	Gross domestic sector research and development
GMO	Genetically modified organism
GovERD	Government intramural research and development
GPT	Generative pre-trained transformer (as in ChatGPT)

Higher education research and development
Horticulture New Zealand Ahumāra Kai Aotearoa
Internet of things
Intellectual property
Information technology
Kilograms
Key performance indictor
Light emitting diode
Light detection and ranging
Long-term Insights Briefing
Million
Ministry of Business, Innovation and Employment Hikina Whakatutuki
Mechanical-biological treatement
Ministry for the Environment Manatū Mō Te Taiao
Megajoules
Machine learning
Memorandum of understanding
MegaPascal
Ministry for Primary Industries Manatū Ahu Matua
Municipal Solid Waste
Megatonne
Mechanical turk
National Animal Identification and Tagging
Non-governmental organisation
Nanometres
New Zealand Food Innovation Network
New Zealand Food Network
Pay as you throw
Per person
Quick response
Research and development
Radio frequency identification
Retail and food service
Red, green, blue (cameras)
Risk management plan

Sector action plan
Sustainable Development Goal
Sustainable Food and Fibre (Futures Fund)
Small and medium sized enterprises
Strengths, weaknesses, opportunities, threats
Tonnes
(MfE's) Theory of Change
Total organic carbon
UN Environment Programme
University of Auckland Waipapa Taumata Rau
University of Otago Ōtakou Whakaiu Waka
Universal product codes
Ultra-processed foods
US dollars
Ultra-violet (light)
Short wavelength ultraviolet (light)
Virtual reality
Waste and Resources Action Programme

Glossary

2D barcode	A graphical image that stores data in two dimensions (vertically and horizontally), for example, a QR code. In the food supply chain, these codes are used to store product information and improve product traceability and inventory management.
Aotearoa New Zealand	The terms Aotearoa New Zealand, Aotearoa, and New Zealand are used interchangeably in this report.
Asexually propagated	Also known as vegetative propagation, where growing new plants does not require seeds. Plants are grown instead from cuttings, tubers, roots, or other plant material. For example, new potato plants are grown from potatoes.
Brix	A measure of dissolved solids in a liquid. Commonly used to measure sugar proportions of fruit.
Bobby calves	Surplus calves from the dairy industry that are sent to abattoirs/processors usually at four days old.
ChatGPT	An interactive generative AI interface which can have human-like conversations.
Cold chain	Temperature-controlled supply chain that keeps products – in this context, food – at the correct temperature while being transported and stored. Maintaining correct temperatures throughout the chain requires significant logistical capacity.
Colostrum	Milk from the first four days after calving, which has a different composition to 'normal' milk.
Cosmetic specifications	Specific requirements for fruit and vegetables regarding their colour, size, and shape. Produce must meet these specifications in order to be considered marketable. See also imperfect produce.
Counterfactual	An alternative state. In research and evaluation an ideal counterfactual is identical to the actual state in all but the variable of interest. When there are differences between the counterfactual and the actual state, outcomes may not be due to the variable of interest but to these differences.
Cultivar	A cultivated variety of a plant species that has been bred selectively to produce desirable attributes such as increased yield, disease resistance, colour, flavour, size etc.
Casualty animals	Animals that die or are euthanised on farm, includes all ages of animals (i.e. slinks).
Demand forecasting	The process of predicting future customer demand over a specified period of time.
Digitalisation	Converting operations and processes from analogue to digital. Not to be confused with digitisation.
Digitisation	Converting specific pieces of data from analogue to digital. Not to be confused with digitalisation.
Distributor (or distribution)	A business that has a relationship with processors and manufacturers, responsible for getting their products to other parts of the supply chain, often to retailers. Distinct from wholesalers (who also purchase from processors/manufacturers and sell to retailers) because they are

	usually associated with single brands of a given product while wholesalers may offer multiple competing brands.
Downstream	Parts of the food supply chain that are closer to the consumer end relative to the part of the supply chain being discussed. For example, retail is downstream of processing and manufacturing (also see upstream).
Dwell time	The amount of time goods spends sitting idle at a given point in the supply chain. This can occur at various stages, such as at a manufacturing site, within a warehouse, during transportation stops, or at a port waiting for customs clearance.
Food	In this project, food is intended to capture both food and beverages. Unless specified, we are referring to food intended for human consumption.
Food delivery services	In the context of this report, food delivery services refer to third party applications and digital platforms that enable delivery of food from food service providers direct to households.
Food loss	Food that is discarded during the production or processing stages of the supply chain. Food discarded after this stage (i.e. in retail, food service, and household settings) is referred to as food waste.
Food loss and waste	For the purposes of this project, food loss and waste is defined broadly and inclusively. Any food or drink that isn't utilised according to its original purpose, as well as by-products and non-edible components of food are included. We give regard to the variable understandings of food and food waste. The entire supply chain is in scope.
Food recovery hierarchy also referred to as 'the hierarchy'	A framework for thinking about solutions to FLW, prioritising interventions according to which types of solutions are likely to deliver the most environmental and social good. The food recovery hierarchy is a modified version of the waste management hierarchy, specific to food. There are many different versions of the food recovery hierarchy. Also known as the 'food waste hierarchy' or the 'food waste management hierarchy'.
Food rescue	The process by which surplus food is captured for human consumption, typically as part of a charity model – but this isn't inherent in the term.
Food service	A part of the food supply chain that prepares food for immediate consumption on-site, takeaway, or delivery, for example, restaurants, catering services, and cafeterias.
Food supply chain	The pathway from where the raw ingredients that will become our food originate, to the places where we eat, form the food supply chain. Specific pathways differ for different kinds of food.
Food system	While food supply chains are a part of food systems, food systems is a more holistic term. The OECD definition of food systems captures the meaning well: "Food systems refers to all the elements and activities related to producing and consuming food, and their effects, including economic, health, and environmental outcomes." ⁸¹⁰
Food waste	Food that is discarded after the processing stage of the food supply chain, i.e. in retail, food service, and household settings. Food that is discarded during the production and processing stages of the supply chain is referred to as food loss.

Genotype	The genetic makeup of an organism.
Germplasm	Genetic resources such as seeds, DNA, rootstock etc. usually used for research or breeding cultivars.
Gleaning	Gathering leftover crops from commercially harvested fields or on fields where the producer has decided not to harvest (for example, because it would not be economically profitable).
Imperfect produce	Fruits and vegetables that do not meet cosmetic specifications for size, shape, or colour, but are remain nutritious and safe to eat. See also cosmetic specifications.
Intervention	A deliberate action or set of coordinated actions designed to change specific behaviours or practices to achieve a particular outcome. In this report, we focus on interventions which aim to prevent FLW.
Manufactured or manufacturers	Involves machinery, automation, and assembly lines to produce packaged foods like snacks or ready-to-eat meals. Manufacturers are the actors who carry out manufacturing.
Mastitis	Inflammation of mammary glands in this report specifically of cows udders. May be caused by bacterial infection and require antibiotics.
Milk solids	The components of dairy after water has been removed. Used as a standard measure of mass/yield in the dairy industry.
Overproduction	In this report, we are referring to the overproduction of food. This is when the volume of food produced far exceeds the demand in the market. This often occurs as a strategy to avoid risk.
Overstock and overstocking	Overstocking is the practice of purposely keeping inventory levels higher than what is needed. In this report we apply this term to both household and retail food waste. In retail, it means stock is surplus to what is realistically expected to sell within a given timeframe in order to fill the shelves. Overstock is surplus product that results from overstocking.
Packhouses	These are sites of post-harvest sorting and grading of produce. Depending on the industry and the size of the operation, packhouses may be located on farms or at a second location. Packhouses are considered part of the production stage of the supply chain (also see post-harvest).
Phenotype	Observable traits, for example, size. Phenotypes are a result of the interaction between genotypes and the environment.
Plate waste	Food that is served to customers but not eaten.
Post-harvest	A sub-stage of the supply chain within production, where produce is sorted and graded, often in packhouses (also see packhouses).
Precision agriculture	A farm management technology that optimises the growth and harvest of crops.
Preparation waste	Food that is discarded during the preparation of food for sale or consumption.
Price-taker	Stakeholders that have lower market power in a stakeholder relationship. For example, farmers are identified as price-takers as they have little influence over the price they receive for the commodities they produce from processors, wholesalers, and retailers.

Processed and processors	Any modification to food, including the removal of inedible parts or the addition of other ingredients. Processors are the actors who carry out the processing.
Producers	People that use natural resources including land, water, seeds, animals, to generate food or ingredients of food. Also known as farmers or growers.
Quality specifications	The set of criteria that food products must meet to ensure they are suitable for consumption, sale, or further processing and maintain food safety across the food industry. Specifications can vary by food type and regulatory requirements, but typically include physical attributes (for example, size, weight, colour), safety standards (for example, pesticide residues or microbiological limits), chemical composition (for example, nutrient or moisture content), and sensory characteristics like taste, texture, or aroma.
Research and development (R&D)	Activities carried out by public or private institutions that seek to resolve scientific or technological uncertainty, creating new knowledge, improved processes, services, or goods.
Retailers	Businesses that sell food products directly to consumers. This can include a variety of outlet types, from large supermarket chains and specialty food stores to local grocers, farmers' markets, and online food retailers.
Secondary market	A marketplace where food products, which are not sold through primary channels (typically retailers), find a different route to consumers. This can include, but is not limited to, overstocked, returned, imperfect, or near-expiration products that are still safe and usable but may not meet the primary market's stringent criteria for aesthetics or quality.
Serving waste	Food that is wasted after being prepared and offered to the consumer but not taken (for example, table bread rolls).
Slinks	Calf and lamb casualties on farm.
Spoilage	Food that is unfit for human consumption due to damage or deterioration.
Stock rotation	An inventory management practice governing the placement of stock in warehouses and retail settings to determine the order in which stock is sold.
Subclinical	Pertains to a disease that does not show obvious symptoms. For example, subclinical mastitis in a dairy cow may not show differences in the visual attributes of the milk, but changes are detected when testing milk composition (i.e. increase somatic cell counts).
Surplus food	Quality, safe, edible food that exceeds the need or demands of a population and is at risk of being wasted if it isn't used. It is distinct from food that is spoiled, damaged, contaminated, past its use by date, or otherwise no longer fit for human consumption.
Take-back agreements	Common terms of trade between retailers or wholesalers and suppliers that are structured so that retailers do not pay for the quantity of product ordered, only the quantity sold, with the supplier responsible for the costs of disposing of any unsold product. These are also sometimes called 'sale and return' agreements or 'scan based trading contracts'.

Terms of trade	In this context, the conditions and agreements under which trade is conducted between parties involved in the procurement, production, and distribution of food.
Topographies	The surface features of an area, for example, hilly, flat, mountainous etc.
Total organic carbon	The concentration of carbon in a compound. Often used as a proxy for water quality.
Trim or trimmings	By-products or waste material that are generated when a primary food product is cut, shaped, or processed.
Unprocessed	Plants or animals that are harvested directly from the land or sea.
Upcycling and upcycled food	Creating new food products from by-products or unmarketable foods such as stale bread, offcuts, or damaged produce. Upcycling is the process whereas upcycled food is the output from upcycling.
Upstream	Parts of the food supply chain that are closer to the production end relative to the part of the supply chain being discussed. For example, processing and manufacturing is upstream of retail (see also downstream).
Vertical integration	A business model whereby a single company (for example, a large retailer or wholesaler) owns and controls multiple stages of the supply chain.
Wholesaler	An intermediary between producers (such as farmers, growers, or manufacturers) and retailers or other businesses that sell directly to consumers. Wholesalers purchase large quantities of products and on sell them in smaller quantities. Distinguished from distributors (who also purchase from processors/manufacturers and sell to retailers) by purchase of large quantities and coverage of multiple competing brands.
Yield and yield potential	Yield is the mass of food or food ingredient generated during production by producers. For example, the tonnes of apples harvested in an orchard. Yield potential is the expected yield when there are no limiting factors to growth other than genetics.

References

- 1. Teigiserova, D. A., Hamelin, L., & Thomsen, M. (2020). Towards transparent valorization of food surplus, waste and loss: Clarifying definitions, food waste hierarchy, and role in the circular economy. *Science of The Total Environment*, *706*. <u>https://doi.org/10.1016/j.scitotenv</u>.2019.136033
- 2. Moshtaghian, H., Bolton, K., & Rousta, K. (2021). Challenges for upcycled foods: Definition, inclusion in the food waste management hierarchy and public acceptability. *Foods*, *10*(11). https://doi.org/10.3390/foods10112874
- 3. United States Department of Agriculture Foreign Agricultural Service, & Global Agricultural Information Network. (2023). *Market fact sheet: New Zealand*. <u>https://apps.fas.usda.gov</u> /newgainapi/api/Report/DownloadReportByFileName?fileName=Food%20Processing%20Ing redients Wellington New%20Zealand NZ2023-0005.pdf
- 4. *The New Zealand dairy industry*. Dairy Companies Association of New Zealand. Retrieved 27 March 2024 from <u>https://dcanz.com/the-new-zealand-dairy-industry/</u>
- 5. Ministry for the Environment Manatū Mō Te Taiao. (2023). *Ngā tikanga ngaronga kai me te para mō Aotearoa | Food loss and waste definition for Aotearoa New Zealand*. <u>https://environment.govt.nz/assets/publications/Waste/Food-loss-and-waste-definition-for-Aotearoa-New-Zealand.pdf</u>
- 6. Redlingshöfer, B., Barles, S., & Weisz, H. (2020). Are waste hierarchies effective in reducing environmental impacts from food waste? A systematic review for OECD countries. *Resources, Conservation and Recycling, 156.* <u>https://doi.org/10.1016/j.resconrec.2020</u>.104723
- 7. Broad Leib, E., Beckmann, J., Ardura, A., DeBode, S., Oto, T., Becker, J., Hanel, N., Nalbantoglu, A., Carbrera, Y., Collins, A., Hoover, D., Keating, M., Sevilla, N., Goerger, S., Gunders, D., Cappa, S., Nichols-Vinueza, A., & Pearson, P. (2022). *Opportunities to reduce food waste in the 2023 Farm Bill*. Harvard Law School Food Law and Policy Clinic, US Natural Resources Defense Council, ReFED, and World Wildlife Fund. <u>https://chlpi.org/wpcontent/uploads/2022/04/2023-Farm-Bill-Food-Waste.pdf</u>
- 8. Chauhan, C., Dhir, A., Akram, M. U., & Salo, J. (2021). Food loss and waste in food supply chains. A systematic literature review and framework development approach. *Journal of Cleaner Production*, 295. <u>https://doi.org/10.1016/j.jclepro.2021.126438</u>
- 9. Ericksen, P. J. (2008). Conceptualizing food systems for global environmental change research. *Global Environmental Change*, *18*(1). <u>https://doi.org/10.1016/j.gloenvcha</u>.2007.09.002
- 10. Galli, F., Cavicchi, A., & Brunori, G. (2019). Food waste reduction and food poverty alleviation: a system dynamics conceptual model. *Agriculture and Human Values*, *36*(2). https://doi.org/10.1007/s10460-019-09919-0
- 11. Ingram, J. (2011). A food systems approach to researching food security and its interactions with global environmental change. *Food Security*, *3*(4). <u>https://doi.org/10.1007/s12571-011-0149-9</u>
- 12. Béné, C., Oosterveer, P., Lamotte, L., Brouwer, I. D., de Haan, S., Prager, S. D., Talsma, E. F., & Khoury, C. K. (2019). When food systems meet sustainability Current narratives and implications for actions. *World Development*, *113*. <u>https://doi.org/10.1016/j.worlddev.2018.08.011</u>
- 13. *New Zealanders love fresh tomatoes*. TomatoesNZ. Retrieved 26 March 2024 from <u>https://www.tomatoesnz.co.nz/</u>
- 14. Costa, J., & Heuvelink, E. (2005). Introduction: the tomato crop and industry. In *Tomatoes* (pp. 1). CABI Publishing Wallingford UK. <u>https://www.cabidigitallibrary.org/doi/pdf</u> /10.5555/20053224743#page=12

- 15. *Statistics*. Tomatoes NZ. Retrieved 26 March 2024 from <u>https://www.tomatoesnz.</u> <u>co.nz/industry/statistics</u>
- 16. *Tomatoes Tōmato*. vegetables.co.nz. Retrieved 26 March 2024 from <u>https://www.vegetables.co.nz/vegetable-a-z/vegetables-a-z/garlic-karika-3/</u>
- 17. *Industry overview*. TomatoesNZ. Retrieved 26 March 2024 from https://www.tomatoesnz. co.nz/industry/industry-overview/
- 18. Tomatoes NZ. (2023). *Key concerns of tomato growers*. <u>https://www.tomatoesnz.</u> <u>co.nz/about/key-concerns-of-tomato-growers/</u>
- 19. Steele, M. (2024, March 18). Low prices hurting local tomato growers. *Radio New Zealand*. https://www.rnz.co.nz/news/country/512043/low-prices-hurting-local-tomato-growers.
- 20. Nichols, M. A., & Bussel, W. T. (1980). The history of process tomato production in New Zealand. Acta Horticulturae 100: Symposium on Production of Tomatoes for Processing. https://doi.org/10.17660/ActaHortic.1980.100.9
- 21. *Tomatoes*. Cargo Handbook. Retrieved 26 March 2024 from <u>https://www.cargohandbook.</u> <u>com/Tomatoes</u>
- 22. Tomato Wellness. (2023, May 16). Farm to table: how tomatoes are processed. <u>https://tomatowellness.com/farm-to-table-how-tomatoes-are-processed/</u>
- 23. New Zealand | Imports and Exports | World | Tomatoes prepared or preserved otherwise than by vinegar or acetic acid | Value (US\$) and Value Growth, YoY (%) | 2012 - 2023. Trend Economy. Retrieved 26 March 2024 from <u>https://trendeconomy.com/data/h2?commodity</u> =2002&reporter=NewZealand&trade flow=Export,Import&partner=World&indicator=TV,Yo Y&time_period=2012,2013,2014,2015,2016,2017,2018,2019,2020,2021,2022,2023
- 24. Carmona-Cabello, M., Zorpas, A. A., & Dorado, M. P. (2022). Chapter Eleven Biorefinery concept for the industrial valorization of tomato processing by-products. In M. Jeguirim & A. Zorpas (Eds.), *Tomato Processing By-Products* (pp. 371). Academic Press. <u>https://doi.org/10.1016/B978-0-12-822866-1.00006-5</u>
- Méndez-Carmona, J. Y., Ascacio-Valdes, J. A., Alvarez-Perez, O. B., Hernández-Almanza, A. Y., Ramírez-Guzman, N., Sepúlveda, L., Aguilar-González, M. A., Ventura-Sobrevilla, J. M., & Aguilar, C. N. (2022). Tomato waste as a bioresource for lycopene extraction using emerging technologies. *Food Bioscience*, 49. <u>https://doi.org/10.1016/j.fbio.2022.101966</u>
- 26. Rajan, A., Kumar, S., Sunil, C. K., Radhakrishnan, M., & Rawson, A. (2022). Recent advances in the utilization of industrial byproducts and wastes generated at different stages of tomato processing: Status report. *Journal of Food Processing and Preservation*, *46*(11). <u>https://doi.org/10.1111/jfpp.17063</u>
- 27. *Tomato pomace, tomato skins and tomato seeds*. Feedipedia. Retrieved 16 May 2024 from <u>https://www.feedipedia.org/node/689</u>
- 28. *About us Perfectly Imperfect*. Perfectly Imperfect. Retrieved 25 March 2024 from <u>https://www.perfectlyimperfect.org.nz/</u>
- 29. *Why Wonky*? Wonky Box. Retrieved 25 March 2024 from <u>https://www.wonkybox.nz/why-wonky</u>
- 30. Junn, J. (2019, May 6). New Zealand creates tonnes of food waste. Supermarkets are trying to close the loop. *The Spinoff*. <u>https://thespinoff.co.nz/food-waste/06-05-2019/</u>supermarkets-create-tonnes-of-food-waste-every-week-heres-what-happens-to-it
- 31. Smithers, R. (2014, May 29). Fancy fallen tomatoes? Coming in a bag in a supermarket near you. *The Guardian*. <u>https://www.theguardian.com/environment/2014/may/29/fallen-tomatoes-bag-waitrose</u>
- 32. *Heirloomacy and Rescued Kitchen*. Grower2Grower. Retrieved 28 April 2024 from https://www.grower2grower.co.nz/heirloomacy-and-rescued-kitchen
- 33. *Communication with Diane Stanbra, Rescued Limited.* (15 April 2024). Personal Communication.

- 34. Rush, E., & Obolonkin, V. (2020). Food exports and imports of New Zealand in relation to the food-based dietary guidelines. *European Journal of Clinical Nutrition*, 74(2). <u>https://doi.org/10.1038/s41430-019-0557-z</u>
- 35. Exports Summary data Key statistics table 7.01 Value of exports by HS chapter group SD (Annual-Jun). Stats NZ Tatauranga Aotearoa,. Retrieved 15 May 2024 from https://infoshare.stats.govt.nz
- 36. Ministry for the Environment Manatū Mō Te Taiao. (2001). *Our clean green image: What's it worth?* <u>https://environment.govt.nz/publications/valuing-new-zealands-clean-green-image/our-clean-green-image-whats-it-worth/</u>
- 37. Saunders, C., & Barber, A. (2008). Carbon footprints, life cycle analysis, food miles: Global trade trends and market issues. *Political Science*, *60*(1). <u>https://doi.org/10.1177</u>/003231870806000107
- 38. Ministry for Primary Industries Manatū Ahu Matua. (2023). *The future of Aotearoa New Zealand's food sector: Exploring global demand opportunities in the year 2050.* <u>https://www.mpi.govt.nz/dmsdocument/56350-The-Future-of-Aotearoa-New-Zealands-Food-Sector-Exploring-Global-Demand-Opportunities-in-the-Year-2050</u>
- 39. Ministry for Primary Industries Manatū Ahu Matua. (2020). *Fit for a better world: Accelerating our economic potential*. <u>https://fitforabetterworld.org.nz/assets</u> /publications/fit-for-a-better-world-roadmap.pdf
- 40. Coriolis (Prepared for Ministry of Business, Innovation and Employment Hīkina Whakatutuki). (2019). *Regional growth opportunities in food and beverage processing employment in New Zealand*. <u>https://www.mbie.govt.nz/dmsdocument/7249-regional-</u> growth-opportunities-in-food-and-beverage-processing-employment-in-new-zealand
- 41. *New Zealand health survey annual data explorer July 2022 to July 2023*. Ministry of Health Manatū Hauora. Retrieved 11 April 2024 from <u>https://minhealthnz.shinyapps.io/nz-health-survey-2022-23-annual-data-explorer/_w_ad28fd5c/#!/home</u>
- 42. Howden, S. M., Soussana, J.-F., Tubiello, F. N., Chhetri, N., Dunlop, M., & Meinke, H. (2007). Adapting agriculture to climate change. *Proceedings of the National Academy of Sciences*, 104(50). <u>https://doi.org/10.1073/pnas.0701890104</u>
- 43. Horrigan, L., Lawrence, R. S., & Walker, P. (2002). How sustainable agriculture can address the environmental and human health harms of industrial agriculture. *Environmental Health Perspectives*, *110*(5). <u>https://doi.org/10.1289/ehp.02110445</u>
- 44. Drewnowski, A. (2004). Obesity and the food environment: Dietary energy density and diet costs. *American Journal of Preventive Medicine*, 27(3). <u>https://doi.org/10.1016/j.amepre.2004.06.011</u>.
- 45. Horticulture New Zealand Ahumāra Kai Aotearoa. (2022). *Submission on New Zealand grocery code of conduct*. <u>https://www.hortnz.co.nz/assets/About-Us/Submissions/HortNZ-Submission-nz-grocery-code-of-conduct.pdf</u>
- 46. Richards, C., Hurst, B., Messner, R., & O'Connor, G. (2021). The paradoxes of food waste reduction in the horticultural supply chain. *Industrial Marketing Management*, *93*. <u>https://doi.org/10.1016/j.indmarman.2020.12.002</u>
- 47. Croad, T., Campbell, H., & Mirosa, M. (2024). Investigating systemic and social dynamics of food loss and waste: An application of waste regime theory to food production in Aotearoa New Zealand. *Cleaner Waste Systems*, 7. <u>https://doi.org/10.1016/j.clwas.2023.100125</u>
- 48. Magalhães, V. S. M., Ferreira, L. M. D. F., & Silva, C. (2021). Using a methodological approach to model causes of food loss and waste in fruit and vegetable supply chains. *Journal of Cleaner Production, 283*. <u>https://doi.org/10.1016/j.jclepro.2020.124574</u>
- 49. de Gorter, H., Drabik, D., Just, D. R., Reynolds, C., & Sethi, G. (2021). Analyzing the economics of food loss and waste reductions in a food supply chain. *Food Policy*, *98*. https://doi.org/10.1016/j.foodpol.2020.101953

- 50. Cattaneo, A., Sánchez, M. V., Torero, M., & Vos, R. (2021). Reducing food loss and waste: Five challenges for policy and research. *Food Policy*, *98*. <u>https://doi.org/10.1016</u> /j.foodpol.2020.101974
- 51. *Meeting with Peter Landon-Lane, T&G Global.* (26 March 2024). Personal Communication.
- 52. Steele, M. (2024, 26 January). 'Perfect storm' for avocado growers during dismal season industry body. *RNZ*. <u>https://www.rnz.co.nz/news/country/507646/perfect-storm-for-</u> <u>avocado-growers-during-dismal-season-industry-body</u>
- 53. New Zealand Foreign Affairs and Trade Manatū Aorere. (2023). *NZ-EU FTA Key outcomes*. <u>https://www.mfat.govt.nz/assets/Trade-agreements/EU-NZ-FTA/NZ-EU-FTA-Key-Outcomes-.pdf</u>
- 54. European Commission. (2023, June 30). EU-New Zealand Trade Agreement: Unlocking sustainable economic growth. <u>https://ec.europa.eu/commission/presscorner</u>/<u>detail/en/IP_22_4158</u>
- 55. Gille, Z. (2012). From risk to waste: Global food waste regimes. *The Sociological Review*, 60(2). <u>https://doi.org/10.1111/1467-954X.12036</u>
- 56. Carbaugh, R., & Prante, T. (2011). A primer on profit maximization. *Journal for Economic Educators*, *11*(2). <u>https://digitalcommons.cwu.edu/cgi/viewcontent.cgi?article=1122&context=cobfac</u>
- 57. Food Innovation Australia. (2021). *National food waste strategy feasibility study final report*. <u>https://afccc.org.au/images/news%20nat%20food%20waste%20feas</u> %20study/FIAL%20NFWS%20Feasibility%20Study%20Report_FINAL.pdf
- 58. Hawkins, R. A. (2013). Boycotts, buycotts and consumer activism in a global context: An overview. *Management & Organizational History*, 5(2),123. <u>https://doi.org/10.1177/1744935910361644</u>
- 59. Dora, M., Biswas, S., Choudhary, S., Nayak, R., & Irani, Z. (2021). A system-wide interdisciplinary conceptual framework for food loss and waste mitigation strategies in the supply chain. *Industrial Marketing Management*, 93. <u>https://doi.org/10.1016/j.indmarman.</u> 2020.10.013
- 60. Hegnsholt, E., Unnikrishnan, S., Pollman-Larsen, M., Askeldottir, B., & Gerard, M. (2018). *Tackling the 1.6-billion-ton food loss and waste crisis.* Boston Consulting Group. <u>https://www.bcg.com/publications/2018/tackling-1.6-billion-ton-food-loss-and-waste-crisis</u>
- 61. Annosi, M. C., Brunetta, F., Bimbo, F., & Kostoula, M. (2021). Digitalization within food supply chains to prevent food waste. Drivers, barriers and collaboration practices. *Industrial Marketing Management*, *93*. <u>https://doi.org/10.1016/j.indmarman.2021.01.005</u>
- 62. Kosior, K. (2018). Digital transformation in the agri-food sector–opportunities and challenges. *Roczniki (Annals), 2018*(1230-2019-3703). <u>https://doi.org/10.22004/ag.econ.293647</u>
- 63. Jonkman, J., Badraoui, I., & Verduijn, T. (2022). Data sharing in food supply chains and the feasibility of crosschain data platforms for added value. *Transportation Research Procedia*, 67. <u>https://doi.org/10.1016/j.trpro.2022.12.031</u>
- 64. Surucu-Balci, E., & Tuna, O. (2021). Investigating logistics-related food loss drivers: A study on fresh fruit and vegetable supply chain. *Journal of Cleaner Production, 318*. <u>https://doi.org/10.1016/j.jclepro.2021.128561</u>
- 65. Australian Food Cold Chain Council, Stop Food Waste Australia, & Australian Department of Agriculture, Water and the Environment. (2022). *Sector action plan summary 2021-2023: Food cold chain*. <u>https://endfoodwaste.com.au/wp-content/uploads/2023/11/Summary-Cold-Chain-Sector-Action-Plan-.pdf</u>
- 66. *Key action area: enhance product distribution*. ReFED. Retrieved 24 March 2024 from <u>https://refed.org/action-areas/enhance-product-distribution/</u>
- 67. Goodman-Smith, F., Mirosa, M., & Skeaff, S. (2020). A mixed-methods study of retail food waste in New Zealand. *Food Policy*, *92*. <u>https://doi.org/10.1016/j.foodpol.2020.101845</u>

- 68. Peira, G., Bollani, L., Giachino, C., & Bonadonna, A. (2018). The management of unsold food in outdoor market areas: food operators' behaviour and attitudes. *Sustainability*, *10*(4). https://doi.org/10.3390/su10041180
- 69. Stop Food Waste Australia, Australian Department of Climate Change, Energy, the Environment and Water, New South Wales Environment Protection Authority, & Rawtec. (2022). Sector action plan summary 2022-2025: Bread & bakery. <u>https://endfoodwaste</u> .com.au/wp-content/uploads/2023/11/Bread-and-Bakery-Sector-Action-Plan_Full-Report-1.pdf
- 70. Food and Agriculture Organisation of the United Nations. (2019). *The state of food and agriculture: Moving forward on food loss and waste reduction*. <u>https://www.fao.org/3/ca6030en/ca6030en.pdf</u>
- 71. Eriksson, M., Ghosh, R., Mattsson, L., & Ismatov, A. (2017). Take-back agreements in the perspective of food waste generation at the supplier-retailer interface. *Resources, Conservation and Recycling*, *122*. <u>https://doi.org/10.1016/j.resconrec.2017.02.006</u>
- 72. Ghosh, R., & Eriksson, M. (2019). Food waste due to retail power in supply chains: Evidence from Sweden. *Global Food Security*, 20. <u>https://doi.org/10.1016/j.gfs.2018.10.002</u>
- 73. *Crumbs, we waste a lot of bread!* Bread Politics. Retrieved 18 March 2024 from https://breadpolitics.com/2019/10/16/crumbs-we-waste-a-lot-of-bread/
- 74. Breathing new life into unsold bakery products. George Weston Foods Limited. Retrieved 18 March 2024 from <u>https://www.gwfbaking.co.nz/breathing-new-life-into-unsold-bakery-products/</u>
- 75. *Bread bags from bread returns*. Baking Industry Research Trust. Retrieved 18 March 2024 from <u>https://www.bakeinfo.co.nz/bread-bags-from-bread-returns/</u>
- 76. de Hooge, I. E., van Dulm, E., & van Trijp, H. C. M. (2018). Cosmetic specifications in the food waste issue: Supply chain considerations and practices concerning suboptimal food products. *Journal of Cleaner Production*, *183*. <u>https://doi.org/10.1016/j.jclepro.2018.02.132</u>
- 77. Porter, S. D., Reay, D. S., Bomberg, E., & Higgins, P. (2018). Avoidable food losses and associated production-phase greenhouse gas emissions arising from application of cosmetic standards to fresh fruit and vegetables in Europe and the UK. *Journal of Cleaner Production*, 201. <u>https://doi.org/10.1016/j.jclepro.2018.08.079</u>
- 78. Makhal, A., Robertson, K., Thyne, M., & Mirosa, M. (2021). Normalising the "ugly" to reduce food waste: Exploring the socialisations that form appearance preferences for fresh fruits and vegetables. *Journal of Consumer Behaviour*, *20*(5). https://doi.org/10.1002/cb.1908
- 79. Qi, D., Penn, J., Li, R., & Roe, B. E. (2022). Winning ugly: Profit maximizing marketing strategies for ugly foods. *Journal of Retailing and Consumer Services*, 64. <u>https://doi.org/10.1016/j.jretconser.2021.102834</u>
- 80. Rooke, T. (2023, 7 December). Potato supply chain needs a pragmatic approach. *NFU*. <u>https://www.nfuonline.com/updates-and-information/potato-supply-chain-needs-pragmatic-approach/</u>
- 81. Sishodia, R. P., Ray, R. L., & Singh, S. K. (2020). Applications of remote sensing in precision agriculture: A review. *Remote Sensing*, *12*(19). <u>https://doi.org/10.3390/rs12193136</u>
- 82. Aguiar, M. L., Gaspar, P. D., Silva, P. D., Domingues, L. C., & Silva, D. M. (2022). Real-time temperature and humidity measurements during the short-range distribution of perishable food products as a tool for supply-chain energy improvements. *Processes*, *10*(11). https://doi.org/10.3390/pr10112286
- 83. Zhong, X., Zhang, M., Tang, T., Adhikari, B., & Ma, Y. (2023). Advances in intelligent detection, monitoring, and control for preserving the quality of fresh fruits and vegetables in the supply chain. *Food Bioscience*, *56*. <u>https://doi.org/10.1016/j.fbio.2023.103350</u>
- 84. Blockchain in the food supply chain What does the future look like? Walmart. Retrieved 16 May 2024 from <u>https://tech.walmart.com/content/walmart-global-tech/en_us/blog</u> /post/blockchain-in-the-food-supply-chain.html

- 85. *7 benefits of the IBM food trust Food waste*. IBM. Retrieved 27 February 2024 from <u>https://www.ibm.com/blockchain/resources/7-benefits-ibm-food-trust/#food-waste</u>
- 86. In wake of romaine E. coli scare, Walmart deploys blockchain to track leafy greens. Retrieved 16 May 2024 from <u>https://corporate.walmart.com/news/2018/09/24/in-wake-of-romaine-e-coli-scare-walmart-deploys-blockchain-to-track-leafy-greens</u>
- 87. Priefer, C., Jörissen, J., & Bräutigam, K. R. (2016). Food waste prevention in Europe A cause-driven approach to identify the most relevant leverage points for action. *Resources, Conservation and Recycling, 109*. <u>https://doi.org/10.1016/j.resconrec.2016.03.004</u>
- 88. *Common European data spaces*. European Commission. Retrieved 25 March 2024 from <u>https://digital-strategy.ec.europa.eu/en/policies/data-spaces</u>
- 89. *Project DIVINE demonstrating value of agri data sharing for boosting data economy in agriculture*. DIVINE project. Retrieved 25 March 2024 from <u>https://divine-project.eu/</u>
- 90. *Cracking the data to combat fruit cracking*. Crack Sense. Retrieved 25 March 2024 from <u>https://cracksense.eu/</u>
- 91. Upscaling agricultural sensor data for improved monitoring of agri-environmental conditions. ScaleAgData. Retrieved 25 March 2024 from <u>https://scaleagdata.eu/en</u>
- 92. *Building a European framework for the secure and trusted data space for agriculture*. Agri Dataspace. Retrieved 25 March 2024 from <u>https://agridataspace-csa.eu/</u>
- 93. *Digital Wallet project*. Trust Alliance New Zealand. Retrieved 13 March 2024 from <u>https://trustalliance.co.nz/projects/digital-wallet-project/</u>
- 94. Yusof, H. S. M., Munap, R., Badrillah, M. I. M., Hamid, N. R. A., & Khir, R. M. (2017). The relationship between organizational culture and employee motivation as moderated by work attitude. *Journal of Administrative and Business Studies*, *3*(1). <u>https://doi.org/10.20474/jabs-3.1.3</u>
- 95. Gregory, B. T., Harris, S. G., Armenakis, A. A., & Shook, C. L. (2009). Organizational culture and effectiveness: A study of values, attitudes, and organizational outcomes. *Journal of Business Research*, *62*(7). <u>https://doi.org/10.1016/j.jbusres.2008.05.021</u>
- 96. Lee, D. (2020). Impact of organizational culture and capabilities on employee commitment to ethical behavior in the healthcare sector. *Service Business*, *14*(1). <u>https://doi.org/10.1007</u> /s11628-019-00410-8
- 97. Al-Swidi, A. K., Gelaidan, H. M., & Saleh, R. M. (2021). The joint impact of green human resource management, leadership and organizational culture on employees' green behaviour and organisational environmental performance. *Journal of Cleaner Production*, 316. <u>https://doi.org/10.1016/j.jclepro.2021.128112</u>
- 98. Williamson, D., Lynch-Wood, G., & Ramsay, J. (2006). Drivers of environmental behaviour in manufacturing SMEs and the implications for CSR. *Journal of Business Ethics*, 67(3). <u>https://doi.org/10.1007/s10551-006-9187-1</u>
- 99. Vickers, I., James, P., Smallbone, D., & Baldock, R. (2005). Understanding small firm responses to regulation. *Policy Studies*, *26*(2). <u>https://doi.org/10.1080/01442870500127626</u>
- 100. *The Odd Bunch*. Woolworths. Retrieved 11 April 2024 from <u>https://www.countdown.</u> <u>co.nz/info/save/odd-bunch</u>
- 101. Gunter, B. (2016). *Food Advertising: Nature, Impact and Regulation*. Palgrave Macmillan Cham. <u>https://doi.org/10.1007/978-3-319-40706-7</u>
- 102. Refresh: Food. Refresh: Food. Retrieved 08 May 2024 from https://refreshfood.com.au/
- 103. Dri, M., Canfora, P., Antonopoulos, I., & Gaudillat, P. (2018). *Best environmental management practice for the waste management sector*. European Commission. <u>https://greenbestpractice.jrc.ec.europa.eu/sites/default/files/2019-03/Unsold%20bread</u> <u>%20waste%20reduction.pdf</u>
- 104. South Australia Office of Consumer and Business Affairs. (1999). *National competition policy review of the Prices Act 1948: Final report*. <u>http://ncp.ncc.gov.au/docs/SA%20review%</u> 20of%20the%20Prices%20Act%201948%2C%20Final%20Report%2C%20March%201999.pdf

- 105. Pietrangeli, R., Herzberg, R., Cicatiello, C., & Schneider, F. (2023). Quality standards and contractual terms affecting food losses: The perspective of producer organisations in Germany and Italy. *Foods*, *12*(10). https://doi.org/10.3390/foods12101984
- 106. Grocery industry competition (Grocery Supply Code) amendment regulations 2023, New Zealand Government. <u>https://www.legislation.govt.nz/regulation/public/2023/0220/latest/whole.html#LMS881069</u>
- 107. Consumer Goods Forum, & ECR Retail Loss. *Contractual terms for reducing food waste: Possibilities and potentials within fresh grocery supply chains.* https://www.theconsumergoodsforum.com/wp-content/uploads/ECR-Report-2020-v4.pdf
- 108. Bhattacharya, A., & Fayezi, S. (2021). Ameliorating food loss and waste in the supply chain through multi-stakeholder collaboration. *Industrial Marketing Management, 93*. https://doi.org/10.1016/j.indmarman.2021.01.009
- 109. Szulecka, J., & Strøm-Andersen, N. (2022). Norway's food waste reduction governance: From industry self-regulation to governmental regulation? *Scandinavian Political Studies*, *45*(1). https://doi.org/10.1111/1467-9477.12219
- 110. *The Courtauld Commitment 2030*. Waste and Resources Action Programme. Retrieved 30 April 2024 from <u>https://wrap.org.uk/taking-action/food-drink/initiatives/courtauld-commitment</u>
- 111. Waste and Resources Action Programme. (2023). *The Food Waste Reduction Roadmap: Progress update 2023*. <u>https://wrap.org.uk/sites/default/files/2023-11/WRAP-Food-Waste-Reduction-Roadmap-Progress-Update-report-2023-V1.0_0.pdf</u>
- 112. Food Act 2014. <u>https://www.legislation.govt.nz/act/public/2014/0032/75.0</u> /DLM2995811.html
- 113. *Our impact*. End Food Waste Australia. Retrieved 21 February 2024 from <u>https://endfood</u> waste.com.au/our-impact/
- 114. *Kai Commitment*. Kai Commitment. Retrieved 26 March 2024 from <u>https://kaicommitment</u>.org.nz/
- 115. Sector Action Plan documents shared by email by Carolyn Cameron, End Food Waste Australia. Personal Communication.
- 116. Sector action plans leading industry-wide and systemic change to end food waste. End Food Waste Australia. Retrieved 22 April 2024 from https://endfoodwaste.com.au/sector-action-plans/
- 117. Baking Association of Australia. (2023). *Don't waste your dough*. <u>https://endfoodwaste</u>.com.au/wp-content/uploads/2023/11/Dont-waste-your-dough.pdf
- 118. Sainsbury's makes British lamb commitment. Sainsbury's. Retrieved 27 March 2024 from https://www.about.sainsburys.co.uk/news/latest-news/2016/06-05-2016-a
- Herzberg, R., Schneider, F., & Banse, M. (2023). Policy instruments to reduce food loss prior to retail – Perspectives of fruit and vegetable supply chain actors in Europe. Waste Management, 170. <u>https://doi.org/10.1016/j.wasman.2023.09.019</u>
- Skivington, K., Matthews, L., Simpson, S. A., Craig, P., Baird, J., Blazeby, J. M., Boyd, K. A., Craig, N., French, D. P., McIntosh, E., Petticrew, M., Rycroft-Malone, J., White, M., & Moore, L. (2021). A new framework for developing and evaluating complex interventions: Update of Medical Research Council guidance. *British Medical Journal*, *374*. <u>https://doi.org</u> /10.1136/bmj.n2061
- 121. United Kingdom Department for Environment, Food and Rural Affairs. (2023). Summary of responses and government response: Improved food waste reporting by large food businesses in England. https://assets.publishing.service.gov.uk/media/64c2413ea4ba200010f1ef0b/Summary of responses and government response food/waste_reporting_July_2023.pdf

- 122. Goodall, B. (2024, March 19). Open letter from UK companies demands mandatory food waste reporting. *Resource.co.* <u>https://resource.co/article/open-letter-uk-companies-</u> demands-mandatory-food-waste-reporting
- 123. New Zealand business demography statistics: at February 2022. Stats NZ Tatauranga Aotearoa. Retrieved 27 March 2024 from <u>https://www.stats.govt.nz/information-</u> releases/new-zealand-business-demography-statistics-at-february-2022/
- 124. Al Qirim, N. (2005). An empirical investigation of an e-commerce adoption: Capability model in small businesses in New Zealand. *Electronic Markets*, *15*(4). <u>https://doi.org/10.1080/10196780500303136</u>
- 125. Kosior, K. (2022). The advancement of digitalization processes in food industry enterprises in the European union. *Zagadnienia Ekonomiki Rolnej*, *371*(2). <u>https://doi.org/10.30858</u> /zer/146782
- 126. Arias Bustos, C., & Moors, E. H. (2018). Reducing post-harvest food losses through innovative collaboration: Insights from the Colombian and Mexican avocado supply chains. *Journal of Cleaner Production*, 199. <u>https://doi.org/10.1016/j.jclepro.2018.06.187</u>
- 127. Kosior, K. (2021). Towards a common agricultural data space in the European Union: A sustainable development perspective. Wieś i Rolnictwo, 191(2). <u>https://doi.org/10.53098/wir022021/03</u>
- 128. Kosior, K. (2020). Smart manufacturing Potential of new digital technologies and big data in the food industry. In M. Wigier & A. Kowalski, "The CAP and national priorities within the EU budget after 2020" Multi-Annual Programme 2015-2019, Warsaw. <u>https://doi.org/ 10.30858/pw/9788376587516</u>
- 129. Kosior, K. (2019). Towards a new data economy for EU agriculture. *Studia Europejskie: Studies in European Affairs, 23*(4). <u>https://doi.org/10.33067/SE.4.2019.6</u>
- 130. AgriTech New Zealand. (2022). *Baseline of digital adoption in primary industries*. <u>https://agritechnz.org.nz/wp-content/uploads/2023/05/AgriTechNZ-Digital-Adoption-in-Primary-Industries-Report-2022.pdf</u>
- 131. O'Connor, J., Skeaff, S., Bremer, P., Lucci, G., & Mirosa, M. (2023). A critical review of onfarm food loss and waste: Future research and policy recommendations. *Renewable Agriculture and Food Systems, 38*. <u>https://doi.org/10.1017/S1742170523000169</u>
- 132. World Wildlife Fund United Kingdom. (2022). Hidden waste: The roadmap to tracking and reducing food surplus and waste on UK farms. <u>https://www.wwf.org.uk/sites</u> /default/files/2022-10/WWF-UK%20HIDDEN%20WASTE%20ROADMAP%20REPORT %202022_2.pdf
- 133. Johnson, L. K. (2020). Produce loss and waste in agricultural production. In C. Reynolds, T. Soma, M. Springmann, & J. Lazell (Eds.), *Routledge Handbook of Food Waste* (pp. 81). <u>https://doi.org/10.4324/9780429462795</u>
- 134. World Wildlife Fund United Kingdom. (2021). Driven to waste: The global impact of food loss and waste on farms. <u>https://wwfint.awsassets.panda.org/downloads</u> /wwf uk driven to waste the global impact of food loss and waste on farms.pdf
- 135. Mirosa, M. (2023). He taonga te kai An Aotearoa where food is valued not wasted. *New Zealand Economic Papers*, *57*(2). https://doi.org/10.1080/00779954.2023.2189157
- 136. Gillman, A., Campbell, D. C., & Spang, E. S. (2019). Does on-farm food loss prevent waste? Insights from California produce growers. *Resources, Conservation and Recycling, 150*. <u>https://doi.org/10.1016/j.resconrec.2019.104408</u>
- 137. Messner, R., Johnson, H., & Richards, C. (2021). From surplus-to-waste: A study of systemic overproduction, surplus and food waste in horticultural supply chains. *Journal of Cleaner Production*, 278. <u>https://doi.org/10.1016/j.jclepro.2020.123952</u>
- 138. Manja, K., & Aoun, M. (2019). The use of nets for tree fruit crops and their impact on the production: A review. *Scientia Horticulturae*, 246. <u>https://doi.org/10.1016</u> /i.scienta.2018.10.050

- 139. Beausang, C., Hall, C., & Toma, L. (2017). Food waste and losses in primary production: Qualitative insights from horticulture. *Resources, Conservation and Recycling, 126*. https://doi.org/10.1016/j.resconrec.2017.07.042
- 140. *Understanding cooperatives*. Cooperative business New Zealand. Retrieved 24 April 2024 from <u>https://nz.coop/understanding-co-operatives/</u>
- 141. Beverland, M. (2007). Can cooperatives brand? Exploring the interplay between cooperative structure and sustained brand marketing success. *Food Policy*, *32*(4). <u>https://doi.org</u>/10.1016/j.foodpol.2006.10.004
- 142. *Horticulture & viticulture. Orchard fruit.* FMG. Retrieved 27 February 2024 from <u>https://www.fmg.co.nz/what-we-cover/horticulture-and-viticulture/orchard-fruit</u>
- 143. Livestock Improvement Corporation Limited, & DairyNZ Limited. (2022). *New Zealand dairy statistics 2021-22*. <u>https://www.dairynz.co.nz/media/uzeekwgr/nz-dairy-statistics-2021-22-web.pdf</u>
- 144. O'Connor, J. (Unpublished) Towards a circular bioeconomy. Food loss, waste, and opportunities in New Zealand beef and dairy production.
- 145. March, M., Toma, L., Thompson, B., & Haskell, M. (2019). Food waste in primary production: Milk loss with mitigation potentials. *Frontiers in Nutrition*, 6. <u>https://doi.org/10.3389</u> /fnut.2019.00173
- 146. Guzman-Luna, P., Nag, R., Martinex, I., Mauricio-Iglesias, M., Hospido, A., & Cummins, E. (2022). Quantifying current and future raw milk losses due to bovine mastitis on European dairy farms under climate change scenarios. *Science of The Total Environment*, 833. https://doi.org/10.1016/j.scitotenv.2022.155149
- 147. More, S. (2009). Global trends in milk quality: Implications for the Irish dairy industry. *Irish Veterinary Journal*, *62*(4). <u>https://doi.org/10.1186/2046-0481-62-S4-S5</u>
- 148. Williamson, J. H., & Lacy-Hulbert, S. J. (2014). Lack of efficacy of homeopathic therapy against post-calving clinical mastitis in dairy herds in the Waikato region of New Zealand. *New Zealand Veterinary Journal, 62*(1). <u>https://doi.org/10.1080/00480169.2013.796435</u>
- 149. Li, X., Xu, C., Liang, B., Kastelic, J. P., Han, B., Tong, X., & Gao, J. (2023). Alternatives to antibiotics for treatment of mastitis in dairy cows. *Frontiers in Veterinary Science*, *10*. https://doi.org/10.3389/fvets.2023.1160350
- 150. *NSAID use in mastitis*. Vetlife. Retrieved 24 April 2024 from <u>https://vetlife.co.nz/nsaid-use-in-mastitis/</u>
- 151. Fonterra increases FY24 forecast farmgate milk price. Fonterra Co-operative Group Ltd. Retrieved 18 March 2024 from <u>https://www.fonterra.com/nz/en/our-</u><u>stories/media/fonterra-increases-fy24-forecast-farmgate-milk-</u><u>price.html#:~:text=Fonterra%20Co%2Doperative%20Group%20Ltd,from%20%247.00%20%2 D%20%248.00%20per%20kgMS</u>
- 152. *Risk management programmes for farm dairies*. Ministry for Primary Industries Manatū Ahu Matua. Retrieved 18 March 2024 from <u>https://www.mpi.govt.nz/agriculture/dairy-farming/risk-management-programmes-for-farm-dairies/</u>
- 153. *Conversation with Oceania Dairy Limited.* (22 March 2024). Personal Communication.
- 154. Meat Industry Association. (2023). *Red meat sector trade summary.* https://www.mia.co.nz/assets/Uploads/Red-Meat-Sector-Trade-2023-Summary.pdf
- 155. New Zealand meat chicken production from June 2008 to June 2022. Poultry Industry Association of New Zealand. Retrieved 18 March 2024 from <u>https://www.pianz.org.nz/wp-</u> <u>content/uploads/2018/11/NZ-Meat-Chicken-Production-from-June-2008-to-June-2022.pdf</u>
- 156. Hamill, G. (2021). *Bobby calves: Industry benefits rather than wasteful perception.* Kellogg Rural Leadership Programme. <u>https://ruralleaders.co.nz/wp-content/uploads/2021</u> /10/Hamill-Greg_Bobby-calves-Industry-benefits-rather-than-wasteful-perception_K43.pdf

- 157. Edwards, J. P., Cuthbert, S., Pinxterhuis, J. B., & McDermott, A. (2021). The fate of calves born on New Zealand dairy farms and dairy farmer attitudes towards producing dairy-beef calves. *New Zealand Journal of Animal Science and Production*, *81*.
- 158. Matthews, J. P. f. W. R. C. (2014). *Rural waste surveys data analysis Waikato & Bay of Plenty.* GHD Ltd. <u>https://www.waikatoregion.govt.nz/assets/WRC/WRC-2019/TR201455.pdf</u>
- 159. Update on slink and casualty stock collections. Beef + Lamb New Zealand. Retrieved 18 March 2024 from <u>https://beeflambnz.com/news/update-slink-and-casualty-stock-collections</u>
- 160. Mazzetto, A. M., Falconer, S., & Ledgard, S. (2023). Carbon footprint of New Zealand beef and sheep meat exported to different markets. *Environmental Impact Assessment Review*, *98*. <u>https://doi.org/10.1016/j.eiar.2022.106946</u>
- 161. Mackintosh, C. G., & Wilson, P. R. (2003). Impact of diseases on the NZ deer industry. *New Zealand Society of Animal Production*, 63. <u>https://www.nzsap.org/system</u> /files/proceedings/2003/ab03063.pdf
- 162. *The NAIT programme*. OSPRI. Retrieved 18 March 2024 from <u>https://www.ospri.co.nz/our-programmes/nait-programme/</u>
- 163. *Animal database*. DairyNZ. Retrieved 18th April 2024 from <u>https://www.dairynz.co.nz</u> /animal/breeding-decisions/animal-database/
- 164. Hutching, G. (2018, October 21). Behind New Zealand's most popular meat. *Stuff*. <u>https://www.stuff.co.nz/business/farming/107888831/behind-new-zealands-most-popular-meat#:~:text=But%20trying%20to%20pin%20down,million%20die%20before%20their%20time</u>.
- 165. *Bird flu and diseases*. Poultry Industry Association of New Zealand. Retrieved 18 March 2024 from <u>https://www.pianz.org.nz/who-we-are/bird-flu-and-diseases/</u>
- 166. Weeks, J. (2019, December 3). Mass chicken deaths 'catastrophic' and unprecedented, experts say after nearly 200,000 birds die. *Stuff*. <u>https://www.stuff.co.nz</u> /business/farming/117848980/mass-chicken-deaths-catastrophic-and-unprecedentedexperts-say-after-nearly-200000-birds-die
- 167. NZ Pork. (2023). New Zealand pork industry board annual report 2023. <u>https://assets-global.website-files.com/5f87a6eb2f34a3a32b6165cf/65d523a56d07e898428f8838_Annual %20Report%202023%20(approved).pdf</u>
- 168. *Calves born alive on farms in New Zealand*. Figure.NZ. Retrieved 28 February 2024 from https://figure.nz/chart/yzDcBTRDjzZl1nL7-QSf5cJAIFPeeECr4
- 169. Beef + Lamb New Zealand. (2018). *Fact sheet April 2018*. <u>https://beeflambnz.com</u> /sites/default/files/factsheets/pdfs/fact-sheet-167-Calf-rearing.pdf
- 170. Agricultural production statistics: year to June 2023 (provisional). Stats NZ -Tatauranga Aotearoa. Retrieved 28 February 2024 from <u>https://www.stats.govt.nz/information-</u> <u>releases/agricultural-production-statistics-year-to-june-2023-provisional/#:~:</u> <u>text=Key%20facts,down%202%20percent%20from%202022</u>
- 171. *Liveweight targets*. DairyNZ. Retrieved 28 February 2024 from <u>https://www.dairynz.co.nz</u> /animal/heifers/lw-targets/#:~:text=Based%20on%20base%20cow%20liveweight %20of%20503kg
- 172. Greening, S., Mulqueen, K., Rawdon, T., French, N., & Gates, M. (2020). Estimating the level of disease risk and biosecurity on commercial poultry farms in New Zealand. *New Zealand Veterinary Journal*, *68*(5). https://doi.org/10.1080/00480169.2020.1746208
- 173. Hassan, F., Atallah, S., & Reda, R. (2021). Comparison of performance, meat quality, and profitability of Cobb, Hubbard, and Ross broiler strains. *European Poultry Science*, *85*. https://doi.org/10.1399/eps.2021.332
- 174. *Free-range deer farming*. Deer Industry New Zealand. Retrieved 28 February 2024 from <u>https://www.deernz.org/home/the-deer-difference/free-range-deer-farming/</u>

- 175. *Deer farming production targets*. Deer Industry New Zealand. Retrieved 28 February 2024 from <u>https://www.deernz.org/deer-hub/support-services/planning-and-financial-</u> management/deer-farming-production-targets/
- 176. Beef + Lamb New Zealand. (2009). *Lactation, lamb growth, and the lamb weaning decision*. <u>https://beeflambnz.com/sites/default/files/factsheets/pdfs/fact-sheet-94-lactation-lamb-growth-and-the-weaning-decision.pdf</u>
- 177. Beef + Lamb New Zealand. (2023). *Lamb crop 2023*. <u>https://beeflambnz.com/sites/default</u> /files/2023-12/Lamb-Crop-2023.pdf
- 178. Muir, P., & Thomson, B. (2010). *Better estimation of national ewe and beef cow liveweights*. Ministry of Agriculture and Forestry. <u>https://www.mpi.govt.nz/dmsdocument/3089-better-estimation-of-national-ewe-and-beef-cow-liveweights</u>
- 179. Game Animal Council New Zealand. (2023). *Wild deer management and meat recovery*. <u>https://nzgameanimalcouncil.org.nz/wp-content/uploads/2024/02/Wild-Deer-</u> <u>Management-and-Meat-Recovery-December-2023-final.pdf</u>
- 180. *Commercial processing data supplied by AssureQuality to the Department of Conservation and shared by Ben Reddiex.* (18 May 2024). Personal Communication.
- 181. Department of Conservation Te Papa Atawhai. (22 March 2024). *Workshop on meat recovery and utilisation of wild animals*. Wellington.
- 182. National wild animal recovery operations. Department of Conservation Te Papa Atawhai. Retrieved 10 April 2024 from <u>https://www.doc.govt.nz/get-involved/apply-for-</u> permits/business-or-activity/national-wild-animal-recovery-operations/
- 183. Ministry for Primary Industries Manatū Ahu Matua. (2023). *Donating meat guidelines*. https://www.mpi.govt.nz/dmsdocument/60913-Donating-meat-guidelines-factsheet
- 184. Kelly, R. (8 April 2024). Forest and Bird take aim at Fiordland Wapiti Hunting. *The Southland Times*. <u>https://www.stuff.co.nz/nz-news/350234555/forest-and-bird-take-aim-fiordland-wapiti-hunting</u>
- 185. Bolton, S. E., & von Keyserlingk, M. A. G. (2021). The dispensable surplus dairy calf: Is this issue a "wicked problem" and where do we go from here? *Frontiers in Veterinary Science*, 8. https://doi.org/10.3389/fvets.2021.660934
- 186. Animal welfare regulations. Bobby calves 2022. <u>https://www.mpi.govt.nz/dmsdocument</u> /31722-Bobby-calves-animal-welfare-leaflet#:~:text=Calves%20less%20than%2014 %20days,short%20and%20comfortable%20as%20possible
- 187. Mead, T. (2023, January 24). Fonterra tells suppliers they can no longer kill bobby calves. *1News*. https://www.1news.co.nz/2023/01/24/fonterra-tells-suppliers-they-can-no-longerkill-bobby-calves/
- 188. Burggraaf, V. (2016). *Beef + Lamb New Zealand dairy beef integration project: final report.* AgResearch. <u>https://beeflambnz.com/sites/default/files/factsheets/pdfs/dairy-beef-integration-project-final-report.pdf</u>
- 189. *Think different. Eat veal. Taste deliciousness*. Pearl Veal. Retrieved 18 March 2024 from https://www.pearlveal.nz/
- 190. *Sexed semen*. Livestock Improvement Corporation. Retrieved 18 March 2024 from <u>https://licnz.com/products-services/sexed/</u>
- 191. *Purposeful lives for calves*. Owl Farm. Retrieved 18 March 2024 from <u>https://owlfarm.nz/owl-farm-kpis/animal-wellbeing/purposeful-lives-for-calves-bobby-wagyu-beef-breeding-strategy/</u>
- 192. Welcome to South Pacific Sera. South Pacific Sera. Retrieved 18 March 2024 from https://www.southpacificsera.co.nz/
- 193. Kolver, E. S., Roche, J. R., Burke, C. R., Kay, J. K., & Aspin, P. W. (2007). Extending lactation in pasture-based dairy cows: I. Geontype and diet effect on milk and reproduction. *Journal of Dairy Science*(12). <u>https://doi.org/10.3168/jds.2007-0324</u>

- 194. *Cellular agriculture*. Office of the Prime Minister's Chief Science Advisor Kaitohutohu Mātanga Pūtaiao Matua ki te Pirimia. Retrieved 23 April 2024 from <u>https://www.pmcsa</u>.ac.nz/topics/cellular-agriculture/
- Hegarty, J. E., Harding, J. E., Oliver, M. H., Gamble, G., Dickson, J. L., Chase, G., & Jaquiery, A. L. (2017). Oral dextrose gel to improve survival in less vigorous newborn triplet lambs: A randomised controlled trial. *New Zealand Journal of Agricultural Research*, 60(1). https://doi.org/10.1080/00288233.2016.1240091
- 196. Bingham, C. M., & Hodge, A. (2024, March 18). Lamb mortality and clostridial deaths. *Zoetis*. <u>https://www.zoetis.co.nz/livestocksolutions/sheepsolutions/seasonal-</u> <u>spotlight/articles/lamb-mortality-and-clostridial-deaths.aspx</u>
- 197. Hunnam, J. C., Lawrence, K., Rashid, Z. B. A., Hitchcock, B., McDougall, S., Wehrle-Martinez, A., & Weston, J. F. (2024). An assessment of the epidemiology and herd-level impact of the fractured humerus epidemic in New Zealand dairy cattle, 2007-2015: Results from four studies. *Animals*, 14(3). <u>https://doi.org/10.3390/ani14030524</u>
- 198. *PigCare(TM) accredited pork*. NZPork. Retrieved 18 March 2024 from <u>https://www.nzpork</u>. .co.nz/farmers/pigcare-tm
- 199. *NZFAP and NZFAP Plus New Zealand's national farm assurance programmes*. New Zealand Farm Assured. Retrieved 18 March 2024 from https://www.nzfap.com/#accordion-container
- 200. *B+LNZ Wormwise programme*. Beef + Lamb New Zealand. Retrieved 26 April 2024 from <u>https://beeflambnz.com/wormwise</u>
- 201. Welcome to Ovis Management. OVIS Management. Retrieved 26 April 2024 from https://www.sheepmeasles.co.nz/
- 202. *Eliminating facial eczema impacts programme*. Beef + Lamb New Zealand. Retrieved 26 April 2024 from <u>https://beeflambnz.com/programmes-partnerships/current-programmes/facial-eczema-programme</u>
- 203. *Free range salt & black pepper lambs tails*. Woolworths (South Africa). Retrieved 18 March 2024 from <u>https://www.woolworths.co.za/prod/_/A-2572230000009</u>
- 204. Asure Quality Kaitiaki Kai. (2017). *New Zealand livestock transport assurance*. <u>https://www.asurequality.com/assets/Topstock/NZLTA-Programme-Manual-V1-July-2017.pdf</u>
- 205. United Fresh New Zealand Incorporated. (2023). *Fresh Facts 2023.* https://unitedfresh.co.nz/assets/site/images/images/Fresh-Facts-%E2%80%93-Online-version-2023.pdf
- 206. *Arable industry profile*. Foundation for Arable Research. Retrieved 10 April 2024 from <u>https://www.far.org.nz/arable-industry-profile</u>
- 207. Ekman, J. (2019). *Guidelines for fresh produce food safety 2019*. Fresh Produce Safety Centre Australia & New Zealand. <u>https://www.hortnz.co.nz/assets/Compliance/2019-07-24-</u> <u>Guidelines-for-Fresh-Produce-Food-Safety-2019-WEB.pdf</u>
- 208. Bidwell, H. (2023, May 2023). Bigger is better: Inside T&G's new \$100 million apple packhouse. *Hawkes Bay Today*. <u>https://www.nzherald.co.nz/hawkes-bay-today/news/bigger-is-better-inside-tgs-new-100-million-apple-packhouse/MHYEO2UPHRETFBULGNOALQOAVY/</u>
- 209. Wheeler, J. (2017, November 05). Leap forward in cherry grading revealed by Compac. *Pead PR for Compac*. <u>https://content.compacsort.com/blog/cherry-grading-revealed</u>
- 210. Love Food Hate Waste. (2018). *What is known about food waste in New Zealand*. <u>https://lovefoodhatewaste.co.nz/wp-content/uploads/2020/09/What-is-known-about-food-waste-in-New-Zealand.pdf</u>
- 211. WasteMINZ. (2018). What is known about food waste in New Zealand. WasteMINZ. https://lovefoodhatewaste.co.nz/wp-content/uploads/2020/09/What-is-known-about-foodwaste-in-New-Zealand.pdf

- 212. Thorsen, M., Mirosa, M., & Skeaff, S. (2021). A quantitative and qualitative study of food loss in glasshouse-grown tomatoes. *Horticulturae*, *8*(1),39. <u>https://doi.org/10.3390/horticulturae</u> 8010039
- 213. Thrive Consulting (Prepared for Central Otago District Council). (2021). Understanding fruit loss in Central Otago. <u>https://www.codc.govt.nz/repository/libraries/id:2apsqkk8g</u> <u>1cxbyoqohn0/hierarchy/services/economic-development/documents/Understanding</u> <u>%20Fruit%20Loss%20in%20Central%20Otago%20report%20final.pdf</u>
- 214. Conversation with Horticulture New Zealand. (1 March 2024). Personal Communication.
- 215. *Lift lock RPC 23L (16623)*. Chep. Retrieved 23 May 2024 from <u>https://www.chep.com</u> /product/lift-lock-rpc-23l-16623
- 216. Black foldable RPC 21L (for supply to Progressive Enterprises) (16421). Chep. Retrieved 23 May 2024 from <u>https://www.chep.com/nz/en/product/black-foldable-rpc-21l-supply-progressive-enterprises-16421</u>
- 217. *What is the RSE scheme?* Horticulture New Zealand Ahumāra Kai Aotearoa,. Retrieved 21 April 2024 from <u>https://www.hortnz.co.nz/people-jobs-and-labour/rse-scheme/</u>
- 218. *About TANZ*. Trust Alliance New Zealand. Retrieved 13 March 2024 from <u>https://trustalliance.co.nz/about/</u>
- 219. Bergin, J. (2023, February 20). What are the real opportunities for decentralisation in 5 years? *Xero Blog*. <u>https://blog.xero.com/news-events/opportunities-for-decentralisation/</u>
- 220. Moghimi, F., & Asiabanpour, B. (2023). Economics of vertical farming in the competitive market. *Clean Technologies and Environmental Policy*, *25*(6). https://doi.org/10.1007/s10098 -023-02473-8
- 221. *Welcome to a new era of pasture farming*. Halter. Retrieved 28 April 2024 from <u>https://www.halterhq.com/</u>
- 222. Horticulture New Zealand Ahumāra Kai Aotearoa. (2023). *Growing together 2035: Aotearoa horticulture action plan - strategy*. <u>https://www.mpi.govt.nz/dmsdocument/55309-Growing-together-2035-Aotearoa-Horticulture-Action-Plan-Strategy</u>
- 223. Plant Germplasm Import Council (prepared for the Ministry for Primary Industries Manatū Ahu Matua). (2022). *The New Zealand plant germplasm import pathway*. <u>https://www.mpi.govt.nz/dmsdocument/50558-Summary-of-Plant-Germplasm-Import-Pathway-strategy</u>
- Parmar, N., Singh, K. H., Sharma, D., Singh, L., Kumar, P., Nanjundan, J., Khan, Y. J., Chauhan, D. K., & Thakur, A. K. (2017). Genetic engineering strategies for biotic and abiotic stress tolerance and quality enhancement in horticultural crops: a comprehensive review.
 3Biotech, 7(4). https://doi.org/10.1007/s13205-017-0870-y
- 225. Ahmar, S., Hensel, G., & Gruszka, D. (2023). CRISPR/Cas9-mediated genome editing techniques and new breeding strategies in cereals current status, improvements, and perspectives. *Biotechnology Advances*, *69*Article 108248. <u>https://doi.org/10.1016</u> /j.biotechadv.2023.108248
- 226. *Our science*. BioLumic. Retrieved 20 May 2024 from <u>https://www.biolumic.com/when-where</u>
- 227. *We solve problems in agriculture with porprietary vision-based AI that adds value.* Farmwave. Retrieved 15 January 2024 from <u>https://farmwave.com/</u>
- 228. Hunter, Z. (2022, December 11). Kiwifruit e-Bin innovation to help 'exacerbated' labour shortage. *Bay of Plenty Times*. <u>https://www.nzherald.co.nz/bay-of-plenty-</u> <u>times/news/kiwifruit-e-bin-innovation-to-help-exacerbated-labour-</u> <u>shortage/ZTHUVC55RJDAPCZGIKODDLP4IA/</u>
- 229. Capozzi, F., Magkos, F., Fava, F., Milani, G., Agostoni, C., Astrup, A., & Saguy, I. (2021). A multidisciplinary perspective of ultra-processed foods and associated food processing technologies: A view of the sustainable road ahead. *Nutrients*, *5*(13). <u>https://doi.org/10.3390/nu13113948</u>
- 230. Torres-León, C., Ramírez-Guzman, N., Londoño-Hernandez, L., Martinez-Medina, G. A., Díaz-Herrera, R., Navarro-Macias, V., Alvarez-Pérez, O. B., Picazo, B., Villarreal-Vázquez, M.,

Ascacio-Valdes, J., & Aguilar, C. N. (2018). Food waste and byproducts: An opportunity to minimize malnutrition and hunger in developing countries. *Frontiers in Sustainable Food Systems*, 2. <u>https://doi.org/10.3389/fsufs.2018.00052</u>

- 231. Scoppola, M. (2021). Globalisation in agriculture and food: The role of multinational enterprises. *European Review of Agricultural Economics*, 48(4). <u>https://doi.org/10.1093/erae/jbab032</u>
- Monteiro, C. A., Cannon, G., Levy, R. B., Moubarac, J. C., Louzada, M. L., Rauber, F., Khandpur, N., Cediel, G., Neri, D., Martinez-Steele, E., Baraldi, L. G., & Jaime, P. C. (2019). Ultra-processed foods: What they are and how to identify them. *Public Health Nutrition*, 22(5). <u>https://doi.org/10.1017/S1368980018003762</u>
- 233. de Araujo, T. P., de Moraes, M. M., Afonso, C., Santos, C., & Rodrigues, S. S. P. (2022). Food processing: Comparison of different food classification systems. *Nutrients*, *14*(4). <u>https://doi.org/10.3390/nu14040729</u>
- 234. Crimarco, A., Landry, M. J., & Gardner, C. D. (2022). Ultra-processed foods, weight gain, and co-morbidity risk. *Current Obesity Reports*, *11*(3). <u>https://doi.org/10.1007/s13679-021-00460-y</u>
- 235. Van Tulleken, C. (2023). *Ultra-processed people: Why do we all eat stuff that isn't food... and why can't we stop*. Cornerstone Publishing.
- 236. Baker, P., Machado, P., Santos, T., Sievert, K., Backholer, K., Hadjikakou, M., Russell, C., Huse, O., Bell, C., Scrinis, G., Worsley, A., Friel, S., & Lawrence, M. (2020). Ultra-processed foods and the nutrition transition: Global, regional and national trends, food systems transformations and political economy drivers. *Obesity Reviews*, *21*(12). <u>https://doi.org</u> /10.1111/obr.13126
- 237. Mallinson, L. J., Russell, J. M., & Barker, M. E. (2016). Attitudes and behaviour towards convenience food and food waste in the United Kingdom. *Appetite*, *103*. <u>https://doi.org/10.1016/j.appet.2016.03.017</u>
- 238. Sapien Labs. (2023). Ultra-processed food consumption and mental wellbeing outcomes. https://sapienlabs.org/wp-content/uploads/2023/09/Sapien_Processed-Food-Rapid-Report_2023-2.pdf
- 239. Uys, G. (2023, March 16). Pork industry says cheap imported meat trumps our animal welfare standards. *Stuff*. <u>https://www.stuff.co.nz/business/farming/131457945/pork-industry-says-cheap-imported-meat-trumps-our-animal-welfare-standards</u>
- 240. Raak, N., Symmank, C., Zahn, S., Aschemann-Witzel, J., & Rohm, H. (2017). Processing- and product-related causes for food waste and implications for the food supply chain. *Waste Management*, 61. <u>https://doi.org/10.1016/j.wasman.2016.12.027</u>
- 241. Mena, C., Adenso-Diaz, B., & Yurt, O. (2011). The causes of food waste in the supplier– retailer interface: Evidences from the UK and Spain. *Resources, Conservation and Recycling*, 55(6). <u>https://doi.org/10.1016/j.resconrec.2010.09.006</u>
- 242. Dora, M., Wesana, J., Gellynck, X., Seth, N., Dey, B., & De Steur, H. (2020). Importance of sustainable operations in food loss: Evidence from the Belgian food processing industry. *Annals of Operations Research*, *290*(1). <u>https://doi.org/10.1007/s10479-019-03134-0</u>
- 243. Darlington, R., Staikos, T., & Rahimifard, S. (2009). Analytical methods for waste minimisation in the convenience food industry. *Waste Management*, *29*(4). <u>https://doi.org/10.1016/j.wasman.2008.08.027</u>
- 244. My Farm Investments. (2021). *Dairy Update*. https://www.myfarm.co.nz/images/assets /7dcc/1/summer%20series%20-%20dairy%20presentation%20-%201%20dec%202021.pdf
- 245. *Meat processing in New Zealand industry analysis*. IBISWorld. Retrieved 19 February 2024 from https://www.ibisworld.com/nz/industry/meat-processing/90/#:~:text=Huge %20exporting%20processors%20dominate%20the,all%20export%2Dfocused%20meat%20pr ocessors.

- 246. Coriolis (Prepared for the Ministry of Economic Development). (2010). *Moving to the centre: The future of the New Zealand food industry*. <u>https://www.mbie.govt.nz/dmsdocument</u> /2231-moving-to-the-centre-the-future-of-the-new-zealand-food-industry-pdf
- 247. Coriolis (Prepared for Ministry of Business, Innovation and Employment Hīkina Whakatutuki). (2012). *Driving growth in the processed foods sector*. <u>https://www.mbie.govt.nz/dmsdocument/2229-driving-growth-processed-foods-sector-pdf</u>
- 248. Coriolis (Prepared for Ministry of Business, Innovation and Employment Hīkina Whakatutuki). (2019). *Investor's guide to the New Zealand food and beverage industry*. <u>https://www.mbie.govt.nz/assets/coriolis-investors-guide-2020.pdf</u>
- 249. Adenso-Díaz, B., & Mena, C. (2013). Food industry waste management. In *Sustainable Food Processing* (pp. 435). <u>https://doi.org/10.1002/9781118634301.ch18</u>
- 250. Kantor, L. S., Lipton, K. L., Manchester, A. C., & Oliveira, V. H. d. (1997). Estimating and addressing America's food losses. *Food Review/ National Food Review, 20*(1). <u>https://doi.org/10.22004/ag.econ.234453</u>
- 251. Richter, B., & Bokelmann, W. (2016). Approaches of the German food industry for addressing the issue of food losses. *Waste Management*, 48. <u>https://doi.org/10.1016/j.wasman.2015</u>...<u>11.039</u>
- 252. Henningsson, S., Hyde, K., Smith, A., & Campbell, M. (2004). The value of resource efficiency in the food industry: A waste minimisation project in East Anglia, UK. *Journal of Cleaner Production*, *12*(5). <u>https://doi.org/10.1016/S0959-6526(03)00104-5</u>
- 253. *From grass to glass*. Fonterra. Retrieved 4 March 2024 from <u>https://www.fonterra.com</u> /nz/en/goodness-of-nz-milk/from-grass-to-glass.html
- 254. *Export*. Ministry for Primary Industries Manatū Ahu Matua. Retrieved 13 March 2024 from <u>https://www.mpi.govt.nz/export/</u>
- 255. New Zealand imports of residues and waste from food industries, and prepared animal fodder. Figure.NZ. Retrieved 13 March 2024 from https://figure.nz/chart/ttrpPeXnWw9UlYtS-rT9CoeChWBZKKCOn
- 256. Reynolds, C. J., Mirosa, M., & Clothier, B. (2016). New Zealand's food waste: Estimating the tonnes, value, calories and resources wasted. *Agriculture*, *6*(1). <u>https://doi.org/10.3390</u> /agriculture6010009
- 257. Priya, K. M., & Alur, S. (2023). Analyzing consumer behaviour towards food and nutrition labeling: A comprehensive review. *Heliyon*, *9*(9). <u>https://doi.org/10.1016/j.heliyon</u>..2023.e19401
- 258. Rösler, F., Kreyenschmidt, J., & Ritter, G. (2021). Recommendation of good practice in the food-processing industry for preventing and handling food loss and waste. *Sustainability*, *13*(17). <u>https://doi.org/10.3390/su13179569</u>
- Liu, C., Hotta, Y., Santo, A., Hengesbaugh, M., Watabe, A., Totoki, Y., Allen, D., & Bengtsson, M. (2016). Food waste in Japan: Trends, current practices and key challenges. *Journal of Cleaner Production*, 133(1). <u>https://doi.org/10.1016/j.jclepro.2016.06.026</u>
- 260. Garcia-Garcia, G., Woolley, E., & Rahimifard, S. (2017). Optimising industrial food waste management. *Procedia Manufacturing*, 8. <u>https://doi.org/10.1016/j.promfg.2017.02.055</u>
- 261. Robinson, B. (Unpublished) Annual production of solid biowastes from the food processing industry.
- 262. Auckland Council Te Kaunihera o Tāmaki Makaurau. (2023). *Auckland's waste assessment*. <u>https://www.aucklandcouncil.govt.nz/plans-projects-policies-reports-bylaws/our-plans-</u> <u>strategies/topic-based-plans-strategies/environmental-plans-</u> <u>strategies/docswastemanagementplan/waste-assessment-2023.pdf</u>
- 263. Redlingshöfer, B., Coudurier, B., & Georget, M. (2017). Quantifying food loss during primary production and processing in France. *Journal of Cleaner Production*, *164*. <u>https://doi.org/10.1016/j.jclepro.2017.06.173</u>

- 264. *Land based operations*. Sealord. Retrieved 22 April 2024 from <u>https://www.sealord.com</u> /about-us/our-operations/
- 265. Cao, G., Shah, P., & Ramanathan, U. (2020). Reducing edible food waste in the UK food manufacturing supply chain through collaboration. In *Sustainable Supply Chains: Strategies, Issues, and Models*. <u>https://link.springer.com/chapter/10.1007/978-3-030-48876-5_10</u>
- 266. Tan, Y., Hai, F., Popp, J., & Oláh, J. (2022). Minimizing waste in the food supply chain: Role of information system, supply chain strategy, and network design. *Sustainability*, *14*(18). <u>https://doi.org/10.3390/su141811515</u>
- 267. da Costa, T. P., Gillespie, J., Cama-Moncunill, X., Ward, S., Condell, J., Ramanathan, R., & Murphy, F. (2023). A systematic review of real-time monitoring technologies and its potential application to reduce food loss and waste: Key elements of food supply chains and IoT technologies. *Sustainability*, *15*(1). <u>https://doi.org/10.3390/su15010614</u>
- 268. Hong, J., Zhou, Z.-d., Li, X., & Lau, K. H. (2020). Supply chain quality management and firm performance in China's food industry—The moderating role of social co-regulation. *The International Journal of Logistics Management*, *31*. <u>https://doi.org/10.1108/IJLM-05-2018-0124</u>
- 269. Brennan, L., Francis, C., Jenkins, E. L., Schivinski, B., Jackson, M., Florence, E., Parker, L., Langley, S., Lockrey, S., Verghese, K., Phan-Le, N. T., Hill, A., & Ryder, M. (2023). Consumer perceptions of food packaging in its role in fighting food waste. *Sustainability*, *15*(3). https://doi.org/10.3390/su15031917
- 270. Han, J. W., Ruiz-Garcia, L., Qian, J. P., & Yang, X. T. (2018). Food packaging: A comprehensive review and future trends. *Comprehensive Reviews in Food Science and Food Safety*, *17*(4). https://doi.org/10.1111/1541-4337.12343
- 271. Poyatos-Racionero, E., Ros-Lis, J. V., Vivancos, J.-L., & Martínez-Máñez, R. (2018). Recent advances on intelligent packaging as tools to reduce food waste. *Journal of Cleaner Production*, *172*. <u>https://doi.org/10.1016/j.jclepro.2017.11.075</u>
- Tadei, R., Trubian, M., Avendano, J., Della Croce, F., & Menga, G. (1995). Aggregate planning and scheduling in the food industry: A case study. *European Journal of Operational Research*, 87(3). <u>https://doi.org/10.1016/0377-2217(95)00230-8</u>
- 273. Strotmann, C., Göbel, C., Friedrich, S., Kreyenschmidt, J., Ritter, G., & Teitscheid, P. (2017). A participatory approach to minimizing food waste in the food industry—A manual for managers. *Sustainability*, *9*(1). <u>https://doi.org/10.3390/su9010066</u>
- 274. Smith, R. (1994). Food hygiene training: The chance to create a coherent policy. *British Food Journal*, *96*(7). <u>https://doi.org/10.1108/00070709410076351</u>
- 275. Webel, S., Bockholt, U., Engelke, T., Gavish, N., Olbrich, M., & Preusche, C. (2013). An augmented reality training platform for assembly and maintenance skills. *Robotics and Autonomous Systems*, *61*(4). <u>https://doi.org/10.1016/j.robot.2012.09.013</u>
- 276. Rodrigues, M., Miguéis, V., Freitas, S., & Machado, T. (2024). Machine learning models for short-term demand forecasting in food catering services: A solution to reduce food waste. *Journal of Cleaner Production*, 435. <u>https://doi.org/10.1016/j.jclepro.2023.140265</u>
- 277. Asian Productivity Organization. (2006). *Postharvest management of fruit and vegetables in the Asia-Pacific region*. <u>https://www.apo-tokyo.org/wp-content/uploads/2015/11/e-edition_Postharvest-MGT-of-Fruit-and-Vegetables.pdf</u>
- 278. Pearson, D., Mirosa, M., Andrews, L., & Kerr, G. (2017). Reframing communications that encourage individuals to reduce food waste. *Communication Research and Practice*, *3*(2). https://doi.org/10.1080/22041451.2016.1209274
- 279. Reynolds, C., Goucher, L., Quested, T., Bromley, S., Gillick, S., Wells, V. K., Evans, D., Koh, L., Carlsson Kanyama, A., Katzeff, C., Svenfelt, Å., & Jackson, P. (2019). Review: Consumptionstage food waste reduction interventions – What works and how to design better interventions. *Food Policy*, 83. <u>https://doi.org/10.1016/j.foodpol.2019.01.009</u>

- 280. Flapper, S. D. P., Fransoo, J. C., Broekmeulen, R. A. C. M., & Inderfurth, K. (2002). Planning and control of rework in the process industries: A review. *Production Planning & Control*, 13(1). https://doi.org/10.1080/09537280110061548
- 281. *Global Data Synchronisation (GDS)*. IBM. Retrieved 13 March 2024 from <u>https://www.ibm.com/docs/en/product-master/12.0.0?topic=concepts-global-data-</u> <u>synchronization-gds</u>
- 282. *GS1 and Global Data Standards*. GS1 Hong Kong. Retrieved 13 March 2024 from <u>https://www.gs1hk.org/publications/casebook/gs1-hk-business-casebook-2019/gs1-and-global-data-standards#:~:text=Global%20Data%20Standards%20(GDS)%20enable,chain %2C%20enhancing%20supply%20chain%20visibility.</u>
- 283. *Global Data Synchronisation Network (GDSN)*. GS1 UK. Retrieved 13 March 2024 from <u>https://www.gs1uk.org/standards-services/data-services/global-data-synchronisation-network-gdsn</u>
- 284. Tekler, Z. D., Low, R., Chung, S. Y., Choong Low, J. S., & Blessing, L. (2019). A waste management behavioural framework of Singapore's food manufacturing industry using factor analysis. *Procedia Cooperative Institutional Research Program, 80*. <u>https://doi.org/10.1016/j.procir.2019.01.066</u>
- 285. Jaouhari, Y., Travaglia, F., Giovannelli, L., Picco, A., Oz, E., Oz, F., & Bordiga, M. (2023). From industrial food waste to bioactive ingredients: A review on the sustainable management and transformation of plant-derived food waste. *Foods*, *12*(11). <u>https://doi.org/10.3390</u>/foods12112183
- 286. Engelund, E. H., Breum, G., & Friis, A. (2009). Optimisation of large scale food production using lean manufacturing principles. *Journal of Foodservice*, *20*(1),4. <u>https://doi.org/10.1111</u> /j.1748-0159.2008.00109.x
- 287. Kennedy, I., Plunkett, A., & Haider, J. (2013). Implementation of lean principles in a food manufacturing company. In *Advances in Sustainable and Competitive Manufacturing Systems*. <u>https://doi.org/10.1007/978-3-319-00557-7_127</u>
- 288. Ciccullo, F., Fabbri, M., Abdelkafi, N., & Pero, M. (2022). Exploring the potential of business models for sustainability and big data for food waste reduction. *Journal of Cleaner Production*, *340*. <u>https://doi.org/10.1016/j.jclepro.2022.130673</u>
- 289. Karuppiah, K., Sankaranarayanan, B., & Ali, S. M. (2023). A systematic review of sustainable business models: Opportunities, challenges, and future research directions. *Decision Analytics Journal*, 8. <u>https://doi.org/10.1016/j.dajour.2023.100272</u>
- 290. What is CiRCLR? CiRCLR. Retrieved 25 March 2024 from https://circlr.nz/product/
- 291. Consumers care about sustainability and back it up with their wallets. McKinsey & Company. Retrieved 15 April 2024 from <u>https://www.mckinsey.com/industries/consumer-packaged-goods/our-insights/consumers-care-about-sustainability-and-back-it-up-with-their-wallets</u>
- 292. Martin-Rios, C., Hofmann, A., & Mackenzie, N. (2021). Sustainability-oriented innovations in food waste management technology. *Sustainability*, *13*(1). <u>https://doi.org/10.3390</u> /su13010210
- 293. Gould, G. W. (1996). Methods for preservation and extension of shelf life. *International Journal of Food Microbiology*, *33*(1). <u>https://doi.org/10.1016/0168-1605(96)01133-6</u>
- 294. Popescu, P., Popa, M., Mitelut, A., & Popa, E. (2018). State of the art on new processing techniques used for preservation of agricultural products A critical review. *Scientific Bulletin. Series F. Biotechnologies, XXII*.
- 295. Rahman, M., Islam, R., Hasan, S., Zzaman, W., Rana, M. R., Ahmed, S., Roy, M., Sayem, A., Matin, A., Raposo, A., Zandonadi, R. P., Botelho, R. B. A., & Sunny, A. R. (2022). A comprehensive review on bio-preservation of bread: An approach to adopt wholesome strategies. *Foods*, *11*(3). <u>https://doi.org/10.3390/foods11030319</u>
- 296. NanoPack. NanoPack. Retrieved 27 February 2024 from https://www.nanopack.eu/

- 297. Schleicher, A. (2019, November 18). NanoPack project delivers shelf-life extending film, prepares for commercial launch. *Packaging Insights*. <u>https://www.packaginginsights.com</u> /<u>news/nanopack-project-delivers-shelf-life-extending-film-prepares-for-commercial-launch.html</u>
- 298. Lalpuria, M., Anantheswaran, R., & Floros, J. (2012). 22 Packaging technologies and their role in food safety. In A. Demirci & M. O. Ngadi (Eds.), *Microbial Decontamination in the Food Industry* (pp. 701). Woodhead Publishing. <u>https://doi.org/10.1533/9780857095756</u>. <u>4.701</u>
- 299. García-García, R., & Searle, S. S. (2016). Preservatives: Food use. In B. Caballero, P. M. Finglas, & F. Toldrá (Eds.), *Encyclopedia of Food and Health* (pp. 505). Academic Press. https://doi.org/10.1016/B978-0-12-384947-2.00568-7
- 300. Masotti, F., Cattaneo, S., Stuknyte, M., & De Noni, I. (2023). Current insights into nonthermal preservation technologies alternative to conventional high-temperature short-time pasteurization of drinking milk. *Critical Review of Food Science and Nutrition*, 63(22). <u>https://doi.org/10.1080/10408398.2021.2022596</u>
- 301. Hernandez-Torres, C. J., Reyes-Acosta, Y. K., Chavez-Gonzalez, M. L., Davila-Medina, M. D., Kumar Verma, D., Martinez-Hernandez, J. L., Narro-Cespedes, R. I., & Aguilar, C. N. (2022). Recent trends and technological development in plasma as an emerging and promising technology for food biosystems. *Saudi Journal of Biological Sciences*, 29(4). <u>https://doi.org</u> /10.1016/j.sjbs.2021.12.023
- 302. Awalgaonkar, G., Beaudry, R., & Almenar, E. (2020). Ethylene-removing packaging: Basis for development and latest advances. *Comprehensive Reviews in Food Science and Food Safety*, 19(6). <u>https://doi.org/10.1111/1541-4337.12636</u>
- 303. Extending shelf life: ethylene absorption packaging strategies for produce transport. DeltaTrak. Retrieved 29 February 2024 from <u>https://www.deltatrak.com/support</u> /resources/white-papers/extending-shelf-life-ethylene-absorption-packaging-strategies-forproduce-transport
- 304. Nowak, D., & Jakubczyk, E. (2020). The freeze-drying of foods-the characteristic of the process course and the effect of its parameters on the physical properties of food materials. *Foods*, *9*(10). <u>https://doi.org/10.3390/foods9101488</u>
- 305. Application of high-pressure-based technologies in the food industry. Ohio State University Extension. Retrieved 28 February 2024 from https://ohioline.osu.edu/factsheet/fst-fabe-1001#:~:text=Deli%20meat%2C%20guacamole%2C%20seafood%2C,good%20candidates%20for%20HPP%20technology
- 306. *Modified atmosphere and vacuum packaging of food*. Ministry for Primary Industries -Manatū Ahu Matua. Retrieved 28 February 2024 from <u>https://www.mpi.govt.nz/food-</u> <u>safety-home/safe-eat/modified-atmosphere-and-vacuum-packaging-of-food/</u>
- Nowosad, K., Sujka, M., Pankiewicz, U., & Kowalski, R. (2021). The application of PEF technology in food processing and human nutrition. *Journal of Food Science and Technology*, 58(2). <u>https://doi.org/10.1007/s13197-020-04512-4</u>
- 308. Majid, I., Nayik, G. A., Nanda, V., & Yildiz, F. (2015). Ultrasonication and food technology: A review. *Cogent Food & Agriculture*, 1(1). <u>https://doi.org/10.1080/23311932.2015.1071022</u>
- 309. Koch Foods Opcenter planning & scheduling case study. Lean Scheduling International. Retrieved 27 February 2024 from <u>https://lean-scheduling.com/case-studies/koch-foods-opcenter-production-planning-scheduling-case-study/</u>
- 310. Blue Book. (2024, March 1). Packaging alliance says EU packaging study ignores food safety and food waste. <u>https://www.producebluebook.com/2024/03/01/packaging-alliance-says-eu-packaging-study-ignores-food-safety-and-food-waste/#</u>
- 311. Yan, M. R., Hsieh, S., & Ricacho, N. (2022). Innovative food packaging, food quality and safety, and consumer perspectives. *Processes*, *10*(4). <u>https://doi.org/10.3390/pr10040747</u>

- 312. Papetti, P., Costa, C., Antonucci, F., Figorilli, S., Solaini, S., & Menesatti, P. (2012). A RFID web-based infotracing system for the artisanal Italian cheese quality traceability. *Food Control*, *27*(1). https://doi.org/10.1016/j.foodcont.2012.03.025
- 313. *Radio frequency identification (RFID)*. GS1. Retrieved 7 March 2024 from <u>https://www.gs1nz</u>.org/services/radio-frequency-identification-rfid/
- 314. Molina-Besch, K., Wikström, F., & Williams, H. (2018). The environmental impact of packaging in food supply chains—Does life cycle assessment of food provide the full picture? *The International Journal of Life Cycle Assessment*, 24(1). <u>https://doi.org/10.1007/s11367-018-1500-6</u>
- 315. Silva, J. C., Figueiredo, M. C., & Braga, A. C. (2019). *Demand forecasting: A case study in the food industry*. Computational Science and its Applications International Conference on Computational Science and its Applications 2019, Saint Petersburg, Russia. <u>https://doi.org/10.1007/978-3-030-24302-9_5</u>
- 316. Nguyen, T. T. H. (2023). Applications of artificial intelligence for demand forecasting. *Operations and Supply Chain Management*, *16*(4). <u>https://doi.org/10.31387/oscm0550401</u>
- 317. Zhong, R. Y., & Zhang, A. (2021). Internet of things (IoT)-enabled agri-food supply chain management: A New Zealand case. In R. Agarwal, C. Bajada, R. Green, & K. Skellern (Eds.), *The Routledge Companion to Global Value Chains*. Routledge. <u>https://doi.org/10.4324</u> /9781315225661
- 318. Jagtap, S., & Rahimifard, S. (2019). The digitisation of food manufacturing to reduce waste Case study of a ready meal factory. *Waste Management*, 87. <u>https://doi.org/10.1016</u> /j.wasman.2019.02.017
- 319. Labs, W. (2022, August 22). Taking control of production through forecasting. *Food Engineering*. <u>https://www.foodengineeringmag.com/articles/100503-taking-control-of-</u> production-through-forecasting
- 320. Dilda, V., Driesen, A., Fenelli, P., & Fisher, J. (2023). The winning recipe for advanced planning systems. <u>https://www.mckinsey.com/capabilities/operations/our-insights/the-winning-recipe-for-transforming-advanced-planning-systems</u>
- 321. Van Wezel, W., Van Donk, D. P., & Gaalman, G. (2006). The planning flexibility bottleneck in food processing industries. *Journal of Operations Management*, *24*(3). <u>https://doi.org/10.1016/j.jom.2004.11.001</u>
- 322. Fan, B., Qian, J., Wu, X., Du, X., Li, W., Ji, Z., & Xin, X. (2019). Improving continuous traceability of food stuff by using barcode-RFID bidirectional transformation equipment: Two field experiments. *Food Control, 98*,449. <u>https://doi.org/10.1016/j.foodcont.2018.12.002</u>
- 323. Karadeniz, A. M., Arif, I., Kanak, A., & Ergün, S. (2019). Digital twin of eGastronomic things: A case study for ice cream machines. 2019 Institute of Electrical and Electronics Engineers International Symposium on Circuits and Systems (ISCAS). <u>https://doi.org/10.1109</u> /ISCAS.2019.8702679
- 324. Vidoni, M., & Vecchietti, A. (2015). A systemic approach to define and characterize advanced planning systems (APS). *Computers & Industrial Engineering*, *90*. <u>https://doi.org/10.1016</u> /j.cie.2015.10.006
- 325. Chen, J., Zhang, M., Xu, B., Sun, J., & Mujumdar, A. S. (2020). Artificial intelligence assisted technologies for controlling the drying of fruits and vegetables using physical fields: A review. *Trends in Food Science & Technology*, *105*. <u>https://doi.org/10.1016/j.tifs.2020.08.015</u>
- 326. Nguyen, V., Behdani, B., & Bloemhof-Ruwaard, J. (2019). Data-driven process redesign: Anticipatory shipping in agro-food supply chains. *International Journal of Production Research*, 58. <u>https://doi.org/10.1080/00207543.2019.1629673</u>
- 327. *Trend two: use of blockchain technology*. Ministry of Business, Innovation and Employment -Hīkina Whakatutuki. Retrieved 6 March 2024 from <u>https://www.mbie.govt.nz/business-and-</u> <u>employment/economic-development/our-long-term-insights-briefing-on-the-future-of-</u>

business-for-aotearoa-new-zealand/the-future-of-business-for-aotearoa-new-zealand/3-trend-two-use-of-blockchain-technology/

- 328. Henrichs, E., Noack, T., Pinzon Piedrahita, A. M., Salem, M. A., Stolz, J., & Krupitzer, C. (2022). Can a byte improve our bite? An analysis of digital twins in the food industry. *Sensors*, *22*(1). <u>https://doi.org/10.3390/s22010115</u>
- 329. Microsoft News Center. (2021, May 13). Mars and Microsoft work together to accelerate Mars' digital transformation and reimagine business operations, associate experience and consumer engagement. https://news.microsoft.com/2021/05/13/mars-and-microsoft-work-together-to-accelerate-mars-digital-transformation-and-reimagine-business-operations-associate-experience-and-consumer-engagement/
- 330. Lara, C., Leverenz, D., Hafner, G., Huang, J., Maurer, C., & Duelli, G. (2023). Comparative study of demand forecasting methods for efficient food waste reduction using real data from small and medium-sized bakeries in Germany. Waste and Resource Management, Technical University of Hamburg. <u>https://pureadmin.unileoben.ac.at/ws/portalfiles/portal</u> /24175723/TAGUNGSBAND_final.pdf#page=117
- 331. Spoiler alert. Spoiler Alert. Retrieved 27 February 2024 from https://www.spoileralert.com/
- 332. Saha, D., & Manickavasagan, A. (2021). Machine learning techniques for analysis of hyperspectral images to determine quality of food products: A review. *Current Research in Food Science*, 4. <u>https://doi.org/10.1016/j.crfs.2021.01.002</u>
- 333. What is the difference between 1D and 2D barcode scanning? Lowry solutions. Retrieved 7 March 2024 from <u>https://lowrysolutions.com/blog/what-is-the-difference-between-1d-and-2d-barcode-scanning/</u>
- 334. Lucio, R. (2019, May 29). Woolworths trials data barcodes. *Inside FMCG*. <u>https://insidefmcg</u>.com.au/2019/05/29/woolworths-trials-data-barcodes/
- 335. Yoha, K. S., & Moses, J. A. (2023). 3D printing approach to valorization of agri-food processing waste streams. *Foods*, *12*(1). <u>https://doi.org/10.3390/foods12010212</u>
- 336. Faculty of Engineering. (2020, September 3). Hopping into the world of 3D-printed food. *University of Auckland - Waipapa Taumata Rau*. <u>https://www.auckland.ac.nz/en</u> /news/2020/09/03/hopping-into-the-world-of-3d-printed-food.html
- 337. Koneva, V. (2023, December 1). How 3D printed food may be transforming your mealtime. <u>https://3dscanexpert.com/3d-printers/3d-printed-food/</u>
- 338. Askew, K. (2022, August 29). 'Flavornomics' is unpicking the complexity of vanilla and drivers of liking for quality boost. *Food Navigator*. <u>https://www.foodnavigator.com/Article</u> /2022/08/29/flavornomics-is-unpicking-the-complexity-of-vanilla-and-drivers-of-liking-forquality-boost
- 339. Ding, H., Tian, J., Yu, W., Wilson, D. I., Young, B. R., Cui, X., Xin, X., Wang, Z., & Li, W. (2023). The application of artificial intelligence and big data in the food industry. *Foods*, *12*(24). <u>https://doi.org/10.3390/foods12244511</u>
- 340. New Zealand Food Technology News. (2021, September 27). Fifty years of the standardised barcode. <u>https://www.foodtechnology.co.nz/content/fifty-years-of-the-standardised-barcode/</u>
- 341. Marker, A. (2019, August 01). The essential guide to barcode inventory management and inventory control systems. <u>https://www.smartsheet.com/content/barcode-inventory-management#:~:text=Businesses%20use%20barcode%20inventory%20management,how%20many%20are%20in%20stock</u>.
- 342. C3. (2023). Enterprise AI for demand forecasting and production scheduling. <u>https://c3.ai/wp-content/uploads/2023/04/C3-AI-Case-Study-AI-for-Demand-Forecasting.pdf?utmMedium=cpc&utmSource=google</u>
- 343. Singapore Ministry of Sustainability and the Environment, & Singapore National Evironment Agency. (2023). *Factsheet: Updates to Singapore's food waste managerment system*. <u>https://www.mse.gov.sg/cos/resources/cos-annex-e.pdf</u>

- 344. European Environment Agency. (2022). *Country profile: France*. <u>https://www.eea.europa.eu</u>/publications/many-eu-member-states/france
- 345. Zero Waste Europe. (2024, March 13). European Parliament gives green light to food waste targets: Prevent Waste Coalition welcomes targets, but is disappointed with low ambition. <u>https://zerowasteeurope.eu/press-release/european-parliament-gives-green-light-to-food-waste-targets-prevent-waste-coalition-welcomes-targets-but-is-disappointed-with-low-ambition/</u>
- 346. *Eco Choice Aotearoa*. Eco Choice Aotearoa. Retrieved 27 February 2024 from <u>https://www</u>.ecochoiceaotearoa.org.nz/
- 347. *Our members*. Ecolabelling Global Network. Retrieved 27 February 2024 from <u>https://globalecolabelling.net/organisations/</u>
- 348. Lee, E. J., Bae, J., & Kim, K. H. (2020). The effect of sustainable certification reputation on consumer behavior in the fashion industry: Focusing on the mechanism of congruence. *Journal of Global Fashion Marketing*, *11*(2). <u>https://doi.org/10.1080/20932685.2020</u>.<u>1726198</u>
- 349. Understanding sustainability certifications & accreditation options. New Zealand Trade and Enterprise - Te Taurapa Tūhono. Retrieved 27 February 2024 from https://my.nzte .govt.nz/article/sustainability-certification-and-accreditation-programmes
- 350. Launch of the national anti-waste label. République française. Retrieved 27 February 2024 from <u>https://entreprendre.service-public.fr/actualites/A16436?lang=en#:~:text=The</u> %20anti%2Dwaste%20circular%20economy,on%201er%20March%202023.
- 351. *National anti-food waste label*. Ministère de la Transition Écologique et de la Cohésion des Territoires. Retrieved 27 February 2024 from <u>https://www.ecologie.gouv.fr/label-national-anti-gaspillage-alimentaire</u>
- 352. Kjolberg, T. (2022). Voluntary climate labelling in Denmark. *Daily Scandinavian*. <u>https://www.dailyscandinavian.com/voluntary-climate-labelling-in-denmark/</u>
- 353. Sheldon, M. (2022, May 3). Denmark to become first country to develop climate label for food. *Hunter College New York City Food Policy Center*. <u>https://www.nycfoodpolicy.org/food-policy-snapshot-denmark-climate-label/</u>
- 354. *Labelling*. Food Standards Australia New Zealand. Retrieved 14 February 2024 from <u>https://www.foodstandards.govt.nz/business/labelling</u>
- 355. Food Standards Australia New Zealand Act 1991, Australian Government. https://www.legislation.gov.au/C2004A04193/latest/text
- 356. Australia New Zealand Food Standards Code Standard 1.2.5 Information requirements date marking of food for sale. Australian Government. Retrieved 9 May 2024 from https://www.legislation.gov.au/F2015L00401/latest/text
- 357. Nwadike, L. (2016, June 27). How do food manufacturers pick those dates on their product packaging And what do they mean? *The Conversation*. <u>https://theconversation.com/how-do-food-manufacturers-pick-those-dates-on-their-product-packaging-and-what-do-they-mean-60591</u>
- 358. Thompson, B., Toma, L., Barnes, A. P., & Revoredo-Giha, C. (2018). The effect of date labels on willingness to consume dairy products: Implications for food waste reduction. *Waste Management*, 78. https://doi.org/10.1016/j.wasman.2018.05.021
- 359. Kavanaugh, M., & Quinlan, J. J. (2020). Consumer knowledge and behaviors regarding food date labels and food waste. *Food Control*, *115*,107285. <u>https://doi.org/10.1016/j.foodcont</u>.2020.107285
- 360. Eveleth, R. (2014, March 28). "Sell by" and "best by" dates on food are basically made up— But hard to get rid of. *Smithsonian*. <u>https://www.smithsonianmag.com/smart-news/sell-and-best-dates-food-are-basically-made-hard-get-rid-180950304/</u>
- 361. Arumugam, N. (2010, February 10). Ignore expiration dates. *Slate*. <u>https://slate.com/human-interest/2010/02/expiration-dates-mean-very-little.html</u>

- 362. Møller, H., Nurttila, A., Lødrup, N., Madsen, P. L., & Rosengren, Å. (2015). *Date labelling in the Nordic countries.* TemaNord. <u>https://doi.org/10.6027/TN2015-504</u>
- 363. *Report food being sold past its use-by date*. Ministry for Primary Industries Manatū Ahu Matua. Retrieved 14 February 2024 from <u>https://www.mpi.govt.nz/contact-us/make-a-food-complaint/report-food-being-sold-past-its-use-by-date/#:~:text=Foods%20with%20best</u> %2Dbefore%20dates,lodge%20a%20complaint%20with%20them.
- 364. Theotokis, A., Pramatari, K., & Tsiros, M. (2012). Effects of expiration date-based pricing on brand image perceptions. *Journal of Retailing*, *88*(1). <u>https://doi.org/10.1016/j.jretai</u>.2011.06.003
- 365. Anderson, L., Grudnoff, M., Hawking, T., & Campbell, R. (2023). *Food waste in Australia and how supermarkets profit from it.* The Australia Institute. <u>https://australiainstitute.org.au/report/food-waste-in-australia/</u>
- 366. *Look, smell, taste, don't waste*. Too Good To Go. Retrieved 13 February 2024 from https://www.toogoodtogo.com/en-gb/initiative/look-smell-taste/look-smell-taste
- 367. Sustainable Packaging Solutions. (2019). *Relationship between shelf life modulation and food waste decrease for selected products*. <u>https://ec.europa.eu/research/participants</u>/documents/downloadPublic?documentIds=080166e5d34c9b32&appId=PPGMS
- 368. Bump. Mimica Lab. Retrieved 25 March 2024 from https://www.mimicalab.com/bump
- 369. Valant, J. (2015). 'Best before' date labels: Protecting consumers and limiting food waste. European Parliamentary Research Service. <u>https://www.europarl.europa.eu/RegData/etudes/BRIE/2015/548990/EPRS_BRI(2015)548990_REV1_EN.pdf</u>
- 370. Southey, F. (2019). 'Best before, often good after': Unilever adopts anti-food waste labels on food packaging. *Food Navigator Europe*. <u>https://www.foodnavigator.com/Article/2019</u> /09/13/Best-before-often-good-after-Unilever-adopts-anti-food-waste-labels-on-foodpackaging
- 371. Neerman, P. (2020, February 05). Too Good to Go mobilizes France against expiry dates. *Retail Detail*. <u>https://www.retaildetail.eu/news/food/too-good-go-mobilizes-france-against-expiry-dates/</u>
- 372. Madigan, L. (2019, September 06). 'Best before' or 'still good', a Danish response to food waste. *Bord Bia Irish Food Board*. <u>https://www.bordbia.ie/industry/news/food-alerts/best-before-or-still-good-a-danish-response-to-food-waste/</u>
- 373. George, S. (2022, August 30). Why supermarkets are removing best-before dates from packaging. *edie*. <u>https://www.edie.net/why-supermarkets-are-removing-best-before-dates-from-packaging/</u>
- 374. Wood, Z. (2023, June 06). M&S removes use-by dates from milk to reduce waste. *The Guardian*. <u>https://www.theguardian.com/business/2023/jun/06/ms-removes-use-by-dates-from-milk-to-reduce-waste</u>
- 375. Hanes, P. (2015). French senate abolishes best before (DLUO) label. *foodtank*. <u>https://foodtank.com/news/2015/03/french-senate-abolishes-best-before-dluo-label/</u>
- 376. Vanderschelden, D. (2022, November 03). 'Best before' food labels are changing in France to reduce waste. *The Connexion*. <u>https://www.connexionfrance.com/article/French-news/Best-before-food-labels-are-changing-in-France-to-reduce-waste</u>
- 377. *France's law for fighting food waste*. Zero Waste Europe. Retrieved 19 February 2024 from https://zerowasteeurope.eu/wp-content/uploads/2020/11/zwe 11 2020 factsheet france_en.pdf
- 378. Overview of the waste disposal levy. Ministry for the Environment Manatū Mō Te Taiao. Retrieved 27 February 2024 from <u>https://environment.govt.nz/what-government-is-doing/areas-of-work/waste/waste-disposal-levy/overview/</u>
- 379. *3R programmes and resources food waste management*. National Environmental Agency A Singapore Government Agency. Retrieved 27 February 2024 from <u>https://www.nea</u>

.gov.sg/our-services/waste-management/3r-programmes-and-resources/food-wastemanagement

- 380. Maile, K. (2020, March 23). Organic waste legislative update. *Waste Today Magazine*. <u>https://www.wastetodaymagazine.com/news/organic-waste-legislative-update-food/#:~:text=Some%20states%E2%80%94California%2C%20Connecticut%2C,or%20mandatory%20organics%20recycling%20laws</u>.
- 381. Danish Crown and Too Good To Go collaborate to combat food waste. euromeat. Retrieved 26 February 2024 from <u>https://euromeatnews.com/Article-Danish-Crown-and-Too-Good-To-Go-collaborate-to-combat-food-waste/5777</u>
- 382. Date label initiative cuts consumer food waste. Food Nation. Retrieved 26 February 2024 from <u>https://foodnationdenmark.com/cases/date-label-initiative-cuts-consumer-food-waste/</u>
- 383. Regulation 1169/2011 EN Food Information to Consumers Regulation EUR-Lex, European Union (2011). <u>https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=celex</u> <u>%3A32011R1169</u>
- 384. *Carrefour takes on the food waste mountain*. Carrefour. Retrieved 26 February 2024 from <u>https://www.carrefour.com/en/news/carrefour-takes-food-waste-mountain</u>
- 385. Plasil, T. (2020). "Best before, often good after" re-scripting the date label of food in Norway. Nordic Journal of Science and Technology Studies, 8(1). <u>https://doi.org/10.5324/njsts</u>...v8i1.3396
- 386. Nordic Council of Ministers. (2017). *Food waste and date labelling*. <u>https://norden.diva-portal.org/smash/get/diva2:1120163/FULLTEXT01.pdf</u>
- 387. George, S. (2018, October 08). Tesco removes fruit and veg 'best before' dates. *edie*. <u>https://www.edie.net/tesco-removes-fruit-and-veg-best-before-dates/</u>
- 388. Thomson, G. B. (2017). Food date labels and hunger in America. *Concordia Law Review*(2).
- 389. *Food Date Labelling Act*. ReFED. Retrieved 4 March 2024 from <u>https://policyfinder.refed</u>.org/federal-policy/food-date-labeling-act
- 390. Frohlich, X. (2023). Mandatory date labels on food could end confusion and prevent food waste. *The Washington Post*. <u>https://www.washingtonpost.com/made-by-history/2023/05/25/food-date-labels/</u>
- 391. ReFED. (2018). *Retail food waste action guide*. <u>https://refed.org/downloads/Retail_Guide</u> <u>Web.pdf</u>
- 392. Commerce Commission New Zealand Te Komihana Tauhokohoko. (2022). *Market study in the retail grocery sector*. <u>https://comcom.govt.nz/_data/assets/pdf_file/0024/278403</u> /Market-Study-into-the-retail-grocery-sector-Final-report-8-March-2022.pdf
- 393. End Food Waste Australia, Australian Department of Industry, Science and Resources, AusIndustry Cooperative Research Centres Program, & Australian Department of Climate Change, Energy, the Environment and Water. (2024). *Foodservice sector action plan overview*. <u>https://endfoodwaste.com.au/wp-content/uploads/2024/02/Foodservice_</u> SAP_Overview_FINAL.pdf
- 394. Ministry for the Environment Manatū Mō Te Taiao. (2022). *Reducing household and business food waste: Literature review*. <u>https://environment.govt.nz/assets/publications/Reducing-household-and-business-food-waste-literature-review.pdf</u>
- 395. End Food Waste Australia, Australian Department of Climate Change, Energy, the Environment and Water, & Green Industries South Australia. (2024). *Catering sector action plan report 2024*. <u>https://endfoodwaste.com.au/wp-content/uploads/2024/02/</u> <u>Catering SAP Report FINAL.pdf</u>
- 396. Supermarkets, grocery stores and convenience stores in New Zealand market size, industry analysis, trends and forecasts (2024-2029). IBISWorld. Retrieved 18 April 2024 from https://www.ibisworld.com/nz/industry/supermarkets-grocery-stores-convenience-stores/716/#FrequentlyAskedQuestions

- 397. Radio New Zealand. (2024, March 08). Foodstuffs and Countdown 'dominate market' but Commerce Commission stops short of radical shake-up. <u>https://www.rnz.co.nz/news</u> /<u>business/462908/foodstuffs-and-countdown-dominate-market-but-commerce-</u> <u>commission-stops-short-of-radical-shake-up</u>
- 398. *About Bidfood*. Bidfood. Retrieved 27 March 2024 from <u>https://www.bidfood.co.nz/about-bidfood</u>
- 399. Welcome to T&G Fresh. T&G. Retrieved 27 March 2024 from <u>https://tandg.global/tandg-fresh/</u>
- 400. Waste and Resources Action Programme. (2013). *Overview of waste in the UK hospitality and food service sector*. <u>https://wrap.org.uk/sites/default/files/2020-10/WRAP-</u> <u>Overview%20of%20Waste%20in%20the%20UK%20Hospitality%20and%20Food%20Service%</u> <u>20Sector%20FINAL.pdf</u>
- 401. United Nations Environment Programme. (2024). *Food Waste Index report 2024. Think eat save: Tracking progress to halve global food waste.* <u>https://wedocs.unep.org/20.500</u>..11822/45230
- 402. Dong, W., Armstrong, K., Jin, M., Nimbalkar, S., Guo, W., Zhuang, J., & Cresko, J. (2022). A framework to quantify mass flow and assess food loss and waste in the US food supply chain. *Communications Earth & Environment*, *3*(1). <u>https://doi.org/10.1038/s43247-022-00414-9</u>
- 403. Sunshine Yates Consulting (Prepared for WasteMINZ). (2018). *New Zealand food waste audits*. <u>https://lovefoodhatewaste.co.nz/wp-content/uploads/2019/02/Final-New-Zealand-Food-Waste-Audits-2018.pdf</u>
- 404. WasteMINZ. (2018). Food waste in the cafe and restaurant sector in New Zealand. <u>https://lovefoodhatewaste.co.nz/wp-content/uploads/2022/05/New-Zealand-cafe-and-resturant-food-waste-WasteMINZ-2018.pdf</u>
- 405. Environment Committee, House of Representatives. (2020). *Briefing to investigate food waste in New Zealand*. <u>https://selectcommittees.parliament.nz/v/0/74878e5b-da07-412a-a041-a6b2c528b274</u>
- 406. Goonan, S., Mirosa, M., & Spence, H. (2014). Getting a taste for food waste: A mixed methods ethnographic study into hospital food waste before patient consumption conducted at three New Zealand foodservice facilities. *Journal of the Academy of Nutrition and Dietetics*, 114(1). https://doi.org/10.1016/j.jand.2013.09.022
- 407. Bridson, L. (2023). *Measuring the lifecycle greenhouse gas emissions of patient meals in three Aotearoa New Zealand public hospitals*. Masters of Environmental Management, Massey University Te Kunenga Ki Pūrehuroa.
- 408. Sanford, D., & Nadeau, H. (2022). Hospital food waste: Reducing waste and cost to our health care system and environment. *Online Journal of Issues in Nursing*, *27*(2). https://doi.org/10.3912/OJIN.Vol27No01PPT33
- 409. Waste and Resources Action Programme. (2019). *UK progress against Courtauld 2025 targets and UN Sustainable Development Goal 12.3*. <u>https://wrap.org.uk/sites/default/files</u> /2020-11/WRAP-Progress against Courtauld 2025 targets and UN SDG 123.pdf
- 410. Parry, A., Harris, B., Fisher, K., & Forbes, H. (2020). *UK progress against Courtauld 2025 targets and UN Sustainable Development Goal 12.3.* Waste and Resources Action Plan. <u>https://wrap.org.uk/sites/default/files/2020-11/WRAP-Progress against Courtauld</u> <u>2025 targets and UN SDG 123.pdf</u>
- 411. Tusa, O. (2021). Development and analysis of static and dynamic pricing models for fresh fruit and vegetable retailers with food waste as a key consideration. MSc, Norwegian School of Economics. Bergen. <u>https://openaccess.nhh.no/nhh-xmlui/bitstream/handle</u> /11250/2786897/masterthesis.pdf?sequence=1
- 412. Riesenegger, L., & Hübner, A. (2022). Reducing food waste at retail stores An explorative study. *Sustainability*, *14*(5),2494. <u>https://doi.org/10.3390/su14052494</u>

- 413. De Moraes, C. C., de Oliveira Costa, F. H., Pereira, C. R., Da Silva, A. L., & Delai, I. (2020). Retail food waste: Mapping causes and reduction practices. *Journal of Cleaner Production*, 256. <u>https://doi.org/10.1016/j.jclepro.2020.120124</u>
- 414. Heikkila, L., Reinikainen, A., Katajajuuri, J. M., Silvennoinen, K., & Hartikainen, H. (2016). Elements affecting food waste in the food service sector. *Waste Management*, *56*. <u>https://doi.org/10.1016/j.wasman.2016.06.019</u>
- 415. *Waste tracking (foodservice)*. ReFED. Retrieved 27 March 2024 from <u>https://insights-engine.refed.org/solution-database/waste-tracking-foodservice</u>
- 416. ReFED. (2018). *Foodservice food waste action plan*. <u>https://refed.org/downloads</u> /Foodservice_Guide_Web.pdf
- 417. *First expired first out*. ReFED. Retrieved 27 March 2024 from <u>https://insights-engine.refed.org/solution-database/first-expired-first-out</u>
- 418. Inecta. (2023, April 18). Understanding FEFO in the food industry: Best practices and implementation. <u>https://www.inecta.com/blog/understanding-fefo-method-in-the-food-industry-importance-best-practices-and-implementation</u>
- 419. *FIFO vs FEFO: Inventory management and your bottom line*. Reliance Wholesale Inc. Retrieved 3 May 2024 from <u>https://reliancewholesale.com/blog/fifo-vs-fefo-inventory-management-and-your-bottom-line/</u>
- 420. *Better for our world, one meal at a time*. Foodprint. Retrieved 13 February 2024 from <u>https://foodprint.app/</u>
- 421. *Leave no meal behind*. ResQ Club. Retrieved 13 February 2024 from <u>https://www.resq-club.com/</u>
- 422. *Enhanced demand planning*. ReFED. Retrieved 7 March 2024 from <u>https://insights-engine.refed.org/solution-database/enhanced-demand-planning</u>
- 423. *ReFED insights engine*. ReFED. Retrieved 23 April 2024 from <u>https://insights.refed.org/</u>
- 424. *Explore solutions to food waste*. ReFED. Retrieved 13 March 2024 from <u>https://insights-engine.refed.org/solution-database?dataView=total&indicator=us-dollars-profit</u>
- 425. *Email correspondence with Jeffrey Constantino, ReFED*. (22 May 2024). Personal Communication.
- 426. *Metrics and management*. Food Waste Technical and Social Innovations Research Group -University of Otago. Retrieved 11 April 2024 from <u>https://foodwaste-otago.org/metrics-and-</u> <u>management#:~:text=Without%20robust%20measures%20of%20food,impactful%20areas%</u> <u>20to%20target%20work</u>.
- 427. De Laurentiis, V., Caldeira, C., & Sala, S. (2020). No time to waste: Assessing the performance of food waste prevention actions. *Resources, Conservation and Recycling, 161*. <u>https://doi.org/10.1016/j.resconrec.2020.104946</u>
- 428. *Decreased minimum order quantity*. ReFED. Retrieved 7 March 2024 from <u>https://insights-engine.refed.org/solution-database/decreased-minimum-order-quantity</u>
- 429. *Increased delivery frequency*. ReFED. Retrieved 12 March 2024 from <u>https://insights-engine.refed.org/solution-database/increased-delivery-frequency</u>
- 430. *Stakeholder recommendations: Retailers: Solutions for grocery waste*. ReFED. Retrieved 23 May 2024 from <u>https://refed.org/stakeholders/retailers/</u>
- 431. Loebnitz, N., Schuitema, G., & Grunert, K. G. (2015). Who buys oddly shaped food and why? Impacts of food shape abnormality and organic labeling on purchase intentions. *Pyschology and Marketing*, *32*(4). <u>https://doi.org/10.1002/mar.20788</u>
- 432. Aschemann-Witzel, J., de Hooge, I. E., & Almli, V. L. (2019). Suboptimal food? Food waste at the consumer–retailer interface. In *Saving Food* (pp. 347). Academic Press. <u>https://doi.org/10.1016/b978-0-12-815357-4.00012-2</u>
- 433. Aschemann-Witzel, J., de Hooge, I., Amani, P., Bech-Larsen, T., & Oostindjer, M. (2015). Consumer-related food waste: Causes and potential for action. *Sustainability*, 7(6),6457. <u>https://doi.org/10.3390/su7066457</u>

- 434. Bryce, E. (2021, February 26). 'Ugly produce' is having a marketing moment. Anthropocene. <u>https://www.anthropocenemagazine.org/2021/02/ugly-produce-is-having-a-marketing-</u> <u>moment/#:~:text=Researchers%20have%20discovered%20that%20it's,produce%20from%20</u> <u>a%20sad%20decline</u>
- 435. Diamante, L. (2024, March 24). The timely rise of imperfect produce. *foodunfolded*. <u>https://www.foodunfolded.com/article/the-timely-rise-of-imperfect-produce</u>
- 436. Mookerjee, S., Cornil, Y., & Hoegg, J. (2021). From waste to taste: How "ugly" labels can increase purchase of unattractive produce. *Journal of Marketing*, *85*(3),62. <u>https://doi.org/10.1177/0022242920988656</u>
- 437. Barrett, J. (2023, July 30). 'It's not even ugly': The Australian businesses rescuing 'imperfect' fruit and veg. *The Guardian*. <u>https://www.theguardian.com/australia-news/2023/jul</u> /27/farmers-pick-fruit-subscription-boxes-imperfect
- 438. Ying, R. (2022, March 18). Dine at these restaurants that minimise food waste. *The Peak*. <u>https://www.thepeakmagazine.com.sg/gourmet-travel/dine-at-these-four-restaurants-which-produce-minimum-food-waste/</u>
- 439. Daniels, J. (2016, July 22). 'Ugly' fresh produce gets hipper for grocers, but there's one problem supply. *CNBC*. <u>https://www.cnbc.com/2016/07/22/ugly-fresh-produce-gets-hipper-for-grocers-but-theres-one-problem--supply.html</u>
- 440. Deutsch, J. (2019, May 23). How can restaurants connect with an 'ugly produce' distributor? *Restaurant Business*. <u>https://www.restaurantbusinessonline.com/advice-guy/how-can-</u> restaurants-connect-ugly-produce-distributor
- 441. Aschemann-Witzel, J., Giménez, A., & Ares, G. (2020). Suboptimal food, careless store? Consumer's associations with stores selling foods with imperfections to counter food waste in the context of an emerging retail market. *Journal of Cleaner Production*, *262*,121252. <u>https://doi.org/10.1016/j.jclepro.2020.121252</u>
- 442. Hartmann, T., Jahnke, B., & Hamm, U. (2021). Making ugly food beautiful: Consumer barriers to purchase and marketing options for Suboptimal Food at retail level A systematic review. *Food Quality and Preference*, *90*,104179. <u>https://doi.org/10.1016/j.foodqual.2021.104179</u>
- 443. Pimentel, B. F., Misopoulos, F., & Davies, J. (2022). A review of factors reducing waste in the food supply chain: The retailer perspective. *Cleaner Waste Systems*, *3*. <u>https://doi.org</u>/10.1016/j.clwas.2022.100028
- 444. Yang, Y., Barnes, H., Yang, B., Onofrei, G., & Nguyen, H. (2023). Food waste management for the UK grocery retail sector—A supply chain collaboration perspective. *Production Planning & Control*, 1. <u>https://doi.org/10.1080/09537287.2023.2226622</u>
- 445. Kor, Y., Prabhu, J., & Esposito, M. (2017, December 19). How large food retailers can help solve the food waste crisis. *Harvard Business Review*. <u>https://hbr.org/2017/12/how-large-food-retailers-can-help-solve-the-food-waste-crisis</u>
- 446. Brancoli, P., Lundin, M., Bolton, K., & Eriksson, M. (2019). Bread loss rates at the supplierretailer interface – Analysis of risk factors to support waste prevention measures. *Resources, Conservation and Recycling*, 147,128. <u>https://doi.org/10.1016/j.resconrec.2019.04.027</u>
- 447. Evans, D., Welch, D., & Swaffield, J. (2017). Constructing and mobilizing 'the consumer': Responsibility, consumption and the politics of sustainability. *Environment and Planning A: Economy and Space*, 49(6),1396. https://doi.org/10.1177/0308518x17694030
- 448. van Giesen, R. I., & de Hooge, I. E. (2019). Too ugly, but I love its shape: Reducing food waste of suboptimal products with authenticity (and sustainability) positioning. *Food Quality and Preference*, *75*,249. <u>https://doi.org/10.1016/j.foodqual.2019.02.020</u>
- 449. Tesco. (2024). *Target, measure, act. Food waste report 2024*. https://www.tescoplc.com/media/sm2hn54l/tesco-food-waste-report-jan24-96.pdf
- 450. *Perfectly Imperfect initiative saves 50m packs of fruit and veg at risk of going to waste.* Tesco. Retrieved 15 April 2024 from <u>https://www.tescoplc.com/wonky-veg-5th-anniversary/</u>

- 451. *Tesco suppliers to use online swap-shop to cut costs and reduce waste*. Tesco. Retrieved 15 April 2024 from <u>https://www.tescoplc.com/tesco-suppliers-to-use-online-swap-shop-to-cut-costs-and-reduce-waste/</u>
- 452. Commerce Commission New Zealand Te Komihana Tauhokohoko. (2023). *Factsheet: The Grocery Supply Code*. <u>https://comcom.govt.nz/_data/assets/pdf_file/0022</u> /329710/Commerce-Commission-Grocery-supply-code-factsheet-28-September-2023.pdf
- 453. Young, C. W., Russell, S. V., Robinson, C. A., & Chintakayala, P. K. (2017). Sustainable retailing

 Influencing consumer behaviour on food waste. *Business Strategy and the Environment*, 27(1). https://doi.org/10.1002/bse.1966
- 454. Aschemann-Witzel, J., de Hooge, I., & Normann, A. (2016). Consumer-related food waste: Role of food marketing and retailers and potential for action. *Journal of International Food and Agribusiness Marketing*, 28(3). <u>https://doi.org/10.1080/08974438.2015.1110549</u>
- 455. Smithers, R. (2018, October 8). Tesco to scrap 'best before' dates from fruit and vegetable lines. *The Guardian*. <u>https://www.theguardian.com/environment/2018/oct/08/tesco-to-scrap-best-before-dates-from-fruit-and-vegetable-lines</u>
- 456. *Tesco changes yogurt labelling to cut down on food waste*. Tesco. Retrieved 27 March 2024 from <u>https://www.tescoplc.com/tesco-changes-yogurt-labelling-to-cut-down-on-food-waste/</u>
- 457. Williams, H., Lindström, A., Trischler, J., Wikström, F., & Rowe, Z. (2020). Avoiding food becoming waste in households The role of packaging in consumers' practices across different food categories. *Journal of Cleaner Production*, *265*,121775. <u>https://doi.org/10.1016/j.jclepro.2020.121775</u>
- 458. Aschemann-Witzel, J., Randers, L., & Pedersen, S. (2022). Retail or consumer responsibility? Reflections on food waste and food prices among deal-prone consumers and market actors. *Business Strategy and the Environment*, *32*(4). <u>https://doi.org/10.1002/bse.3202</u>
- 459. Heard, B. R., Bandekar, M., Vassar, B., & Miller, S. A. (2019). Comparison of life cycle environmental impacts from meal kits and grocery store meals. *Resources, Conservation and Recycling*, 147. <u>https://doi.org/10.1016/j.resconrec.2019.04.008</u>
- 460. Hirschberg, C., Rajko, A., Schumacher, T., & Wrulich, M. (2016, November 9). The changing market for food delivery. *McKinsey and Company*. <u>https://www.mckinsey.com/industries</u> /<u>technology-media-and-telecommunications/our-insights/the-changing-market-for-food-delivery</u>
- 461. Curry, D. (2024, April 24). Food delivery app revenue and usage statistics (2024). *Business of Apps*. <u>https://www.businessofapps.com/data/food-delivery-app-market/</u>
- 462. Ahuja, K., Chandra, V., Lord, V., & Peens, C. (2021, September 22). Ordering in: The rapid evolution of food delivery. *McKinsey and Company*. <u>https://www.mckinsey.com/industries</u> /<u>technology-media-and-telecommunications/our-insights/ordering-in-the-rapid-evolution-of-food-delivery</u>
- 463. *Meal delivery New Zealand*. statista. Retrieved 18 April 2024 from <u>https://www.statista</u>.com/outlook/emo/online-food-delivery/meal-delivery/new-zealand
- 464. Wynn, K. (2022, June 25). Food delivery wars: DoorDash, Uber Eats, Menulog, Delivereasy and Yummi compared. *The New Zealand Herald*. <u>https://www.nzherald.co.nz/nz/food-delivery-wars-doordash-uber-eats-menulog-delivereasy-and-yummi-compared/XS6DEEL2X6HPALTM4IQ6CH2VGQ/</u>
- 465. Trivedi, V., Trivedi, A., Pandey, K. K., & Chaurasia, S. S. (2023). Ordering the right quantity? Examining the impact of plate size vagueness on food waste in an online food delivery system. *Journal of Cleaner Production*, *391*. <u>https://doi.org/10.1016/j.jclepro.2023.136052</u>
- 466. Talwar, S., Kaur, P., Ahmed, U., Bilgihan, A., & Dhir, A. (2022). The dark side of convenience: How to reduce food waste induced by food delivery apps. *British Food Journal*, *125*(1),205. <u>https://doi.org/10.1108/bfj-02-2021-0204</u>

- 467. Li, C., Mirosa, M., & Bremer, P. (2020). Review of online food delivery platforms and their impacts on sustainability. *Sustainability*, *12*(14),5528. <u>https://doi.org/10.3390/su12145528</u>
- 468. Sharma, R., Dhir, A., Talwar, S., & Kaur, P. (2021). Over-ordering and food waste: The use of food delivery apps during a pandemic. *International Journal of Hospitality Management, 96*. https://doi.org/10.1016/j.ijhm.2021.102977
- 469. Waste and Resources Action Programme. (2022). *Food Waste Reduction Roadmap progress report*. <u>https://wrap.org.uk/sites/default/files/2022-12/WRAP_Food_Waste_</u> <u>Reduction Roadmap_Progress_Report_2022.pdf</u>
- 470. US Food Loss and Waste 2030 Champions. US Department of Agriculture. Retrieved 19 April 2024 from <u>https://www.usda.gov/foodlossandwaste/champions</u>
- 471. 2030 Champions brings together business and government to cut US food waste. Champions 12.3. Retrieved 19 April 2024 from <u>https://champions123.org/2030-champions-brings-together-business-and-government-cut-us-food-waste</u>
- 472. United States Department of Agriculture, United States Environmental Protection Agency. (2022). U.S. Food Loss & Waste 2030 Champions milestones report (September 2022). https://www.usda.gov/sites/default/files/documents/2030-Champions-2022.pdf
- 473. Leake, J., & Baker, N. (2024, February 23). How did Coles and Woolworths become so powerful? The story of the big two supermarkets. *ABC News*. <u>https://www.abc.net.au/news/2024-02-23/a-history-of-the-duopoly-coles-woolworths</u>/103494070
- 474. *Australian food pact*. End Food Waste Australia. Retrieved 11 February 2024 from <u>https://endfoodwaste.com.au/australian-food-pact/</u>
- 475. Stott, A. (2019, July 22). What we can learn from France's ban on food waste. *Beyond Food Waste*. <u>https://beyondfoodwaste.com/france_food_waste_ban/</u>
- 476. Zero Waste Europe. (2020). *Zero Waste Europe factsheet: France's law for fighting food waste*. <u>https://zerowasteeurope.eu/wp-content/uploads/2020/11/zwe_11_2020_factsheet_france_en.pdf</u>
- 477. *France's ban on food waste three years later*. Retrieved 19 April 2024 from <u>https://foodtank.com/news/2019/06/opinion-frances-ban-on-food-waste-three-years-later/</u>
- 478. *Households*. Stats NZ Tatauranga Aotearoa. Retrieved 23 April 2024 from <u>https://www.stats.govt.nz/topics/households</u>
- 479. Hebrok, M., & Boks, C. (2017). Household food waste: Drivers and potential intervention points for design An extensive review. *Journal of Cleaner Production*, 151. <u>https://doi.org/10.1016/j.jclepro.2017.03.069</u>
- 480. Boulet, M., Hoek, A. C., & Raven, R. (2021). Towards a multi-level framework of household food waste and consumer behaviour: Untangling spaghetti soup. *Appetite*, *156*. <u>https://doi.org/10.1016/j.appet.2020.104856</u>
- 481. Household waste prevention hub: Household waste preventions activities food. WRAP. Retrieved 16 May 2024 from <u>https://wrap.org.uk/resources/guide/waste-prevention-activities/food</u>
- 482. United Nations Environment Programme. (2021). *Food Waste Index report 2021.* https://wedocs.unep.org/handle/20.500.11822/35280
- 483. BioCycle. (2024, April 9). UN's 2024 food waste index attributes 60% of food waste to households. <u>https://www.biocycle.net/2024-food-waste-index/</u>
- 484. Waste Not Consulting (Prepared for WasteMINZ). (2015). *New Zealand food waste audits*. <u>https://lovefoodhatewaste.co.nz/wp-content/uploads/2016/03/New-Zealand-Food-Waste-Bin-Audit-Report-2015.pdf</u>
- 485. Li, X., Jiang, Y., & Qing, P. (2023). Estimates of household food waste by categories and their determinants: Evidence from China. *Foods*, *12*(4). <u>https://doi.org/10.3390/foods12040776</u>
- 486. Caldeira, C., De Laurentiis, V., Corrado, S., van Holsteijn, F., & Sala, S. (2019). Quantification of food waste per product group along the food supply chain in the European Union: A mass
flow analysis. *Resources, Conservation & Recycling, 149.* <u>https://doi.org/10.1016/j.resconrec</u>.2019.06.011

- 487. Shaw, P. J., Smith, M. M., & Williams, I. D. (2018). On the prevention of avoidable food waste from domestic households. *Recycling*, *3*(2). <u>https://doi.org/10.3390/recycling3020024</u>
- 488. Kantar, KiwiHarvest, & Rabobank. (2023). *Rabobank KiwiHarvest New Zealand food waste* survey results 2023. <u>https://www.rabobank.co.nz/foodwaste/</u>
- 489. Waste minimisation (information requirements) amendment regulations 2023, New Zealand Government. <u>https://www.legislation.govt.nz/regulation/public/2023/0262/latest/whole</u> <u>.html?search=qs_act%40bill%40regulation%40deemedreg_Waste+Minimisation+(Calculatio</u> <u>n+and+Payment+of+Waste+Disposal+Levy)+Amendment+Regulations+2023_resel_25_h&p=</u> 1#LMS904904
- 490. *Video call with Jenny Marshall, Minsitry for the Environment Manatū Mō Te Taiao.* (15 April 2024). Personal Communication.
- 491. Unpublished report shared by email by Brody Gilroy, Waste-Ed with Kate. (3 April 2024). Personal Communication.
- 492. Vittuari, M., Garcia Herrero, L., Masotti, M., Iori, E., Caldeira, C., Qian, Z., Bruns, H., van Herpen, E., Obersteiner, G., Kaptan, G., Liu, G., Mikkelsen, B. E., Swannell, R., Kasza, G., Nohlen, H., & Sala, S. (2023). How to reduce consumer food waste at household level: A literature review on drivers and levers for behavioural change. *Sustainable Production and Consumption*, 38,104. <u>https://doi.org/10.1016/j.spc.2023.03.023</u>
- 493. Parker, J. R., Umashankar, N., & Schleicher, M. G. (2019). How and why the collaborative consumption of food leads to overpurchasing, overconsumption, and waste. *Journal of Public Policy & Marketing*, *38*(2). https://doi.org/10.1177/0743915618823783
- 494. Wang, L., Yang, Y., & Wang, G. (2022). The clean your plate campaign: Resisting table food waste in an unstable world. *Sustainability*, *14*(8),4699. <u>https://doi.org/10.3390/su14084699</u>
- 495. Organisation for Economic Co-operation and Development. (2023). *How green is household behaviour?: Sustainable choices in a time of interlocking crises*. OECD Publishing. https://doi.org/10.1787/2bbbb663-en.
- 496. Borg, K., Boulet, M., Karunasena, G., & Pearson, D. (2022). Segmenting households based on food waste behaviours and waste audit outcomes: Introducing over providers, under planners and considerate planners. *Journal of Cleaner Production*, *351*. <u>https://doi.org/10.1016/j.jclepro.2022.131589</u>
- 497. Graham-Rowe, E., Jessop, D. C., & Sparks, P. (2014). Identifying motivations and barriers to minimising household food waste. *Resources, Conservation and Recycling, 84*,15. https://doi.org/10.1016/j.resconrec.2013.12.005
- 498. Aktas, E., Sahin, H., Topaloglu, Z., Oledinma, A., Huda, A. K. S., Irani, Z., Sharif, A. M., Van't Wout, T., & Kamrava, M. (2018). A consumer behavioural approach to food waste. *Journal of Enterprise Information Management*, *31*(5). <u>https://doi.org/10.1108/JEIM-03-2018-0051</u>
- 499. Aleshaiwi, A. (2023). Household food waste in domestic gatherings The negotiation between social and moral duties. *British Food Journal, 125*(9). <u>https://doi.org/10.1108/BFJ-02-2022-0166</u>
- 500. Evans, D. (2012). Beyond the throwaway society: Ordinary domestic practice and a sociological approach to household food waste. *Sociology*, *46*(1). <u>https://doi.org/10.1177</u> /0038038511416150
- 501. Visschers, V. H. M., Wickli, N., & Siegrist, M. (2016). Sorting out food waste behaviour: A survey on the motivators and barriers of self-reported amounts of food waste in households. *Journal of Environmental Psychology*, 45. <u>https://doi.org/10.1016/j.jenvp.2015.11.007</u>
- 502. Secondi, L., Principato, L., & Laureti, T. (2015). Household food waste behaviour in EU-27 countries: A multilevel analysis. *Food Policy*, *56*. https://doi.org/10.1016/j.foodpol.2015.07.007

- 503. van Geffen, L., van Herpen, E., & van Trijp, H. (2016). *Causes and determinants of consumer* food waste - A theoretical framework. <u>https://eu-refresh.org/sites/default/files</u> /Causes%20&%20Determinants%20of%20Consumers%20Food%20Waste 0.pdf
- 504. Principato, L., Mattia, G., Di Leo, A., & Pratesi, C. A. (2021). The household wasteful behaviour framework: A systematic review of consumer food waste. *Industrial Marketing Management*, *93*. https://doi.org/10.1016/j.indmarman.2020.07.010
- 505. Ganglbauer, E., Fitzpatrick, G., & Comber, R. (2013). Negotiating food waste. ACM Transactions on Computer-Human Interaction, 20(2),1. https://doi.org/10.1145 /2463579.2463582
- 506. Porpino, G., Parente, J., & Wansink, B. (2015). Food waste paradox: Antecedents of food disposal in low income households. *International Journal of Consumer Studies*, *39*(6). https://doi.org/10.1111/ijcs.12207
- 507. van Geffen, L., van Herpen, E., Sijtsema, S., & van Trijp, H. (2020). Food waste as the consequence of competing motivations, lack of opportunities, and insufficient abilities. *Resources, Conservation & Recycling: X, 5.* <u>https://doi.org/10.1016/j.rcrx.2019.100026</u>
- 508. Jackson, P., & Viehoff, V. (2016). Reframing convenience food. *Appetite*, *98*. <u>https://doi.org</u> /10.1016/j.appet.2015.11.032
- 509. Jörissen, J., Priefer, C., & Bräutigam, K.-R. (2015). Food waste generation at household level: Results of a survey among employees of two European research centers in Italy and Germany. *Sustainability*, 7(3). <u>https://doi.org/10.3390/su7032695</u>
- 510. Waste and Resources Action Programme. (2007). *Understanding food waste.* https://www.carbonindependent.org/files/foodwasteresearchsummaryfinaladp29_3_07_2 5a4c08b.b8683843.pdf
- 511. van Drie, E. (2022). *Reducing food waste: The influence of meal-kits on food waste compared to instant food and traditional cooking A quantitative analysis on minimising food waste.* Master of Business Administration, Radboud University. Nijmegen, The Netherlands. <u>https://theses.ubn.ru.nl/items/00818cdd-f2a0-47ff-b93a-39e539b243c1</u>
- 512. Schuster, S., Speck, M., van Herpen, E., Buchborn, F., Langen, N., Nikravech, M., Mullick, S., Eichstädt, T., Chikhalova, Y., Budiansky, E., Engelmann, T., & Bickel, M. (2022). Do meal boxes reduce food waste from households? *Journal of Cleaner Production*, *375*. https://doi.org/10.1016/j.jclepro.2022.134001
- 513. Styles, R. (2023, December 18). Are meal kits cheaper than the supermarket? *consumer*. <u>https://www.consumer.org.nz/articles/meal-kits</u>
- 514. *Bargain Box is the affordably delicious choice for every household*. Bargain Box. Retrieved 18 April 2024 from <u>https://www.bargainbox.co.nz/</u>
- 515. *Tomatoes*. StopFoodWaste. Retrieved 24 April 2024 from <u>https://stopfoodwaste.org</u> /<u>tips/store/tomatoes</u>
- 516. Aloysius, N., Ananda, J., Mitsis, A., & Pearson, D. (2023). Why people are bad at leftover food management? A systematic literature review and a framework to analyze household leftover food waste generation behavior. *Appetite*, *186*. <u>https://doi.org/10.1016/j.appet.2023</u>. <u>106577</u>
- 517. Unpublished report shared by Jenny Marshall, Ministry for the Environment Manatū Mō Te Taiao. (15 April 2024). Personal Communication.
- 518. WasteMINZ. (2014). *National food waste prevention study*. <u>https://lovefoodhatewaste.co.nz</u> /wp-content/uploads/2016/03/Online-Survey-of-attitudes-to-food-waste-.pdf
- 519. *Love Food Hate Waste New Zealand*. Love Food Hate Waste NZ. Retrieved 26 March 2024 from <u>https://lovefoodhatewaste.co.nz/</u>
- 520. Waste and Resources Action Programme. (2013). *The impact of Love Food Hate Waste*. https://wrap.org.uk/sites/default/files/2020-08/WRAP-West%20London%20LFHW %20Impact%20case%20study_0.pdf

- 521. Brown, S. (2009). The new deficit model. *Nature Nanotechnology*, *4*(10). <u>https://doi.org</u> /10.1038/nnano.2009.278
- 522. Abrahamse, W., & Matthies, E. (2018). Informational strategies to promote pro environmental behaviour. In *Environmental Psychology* (pp. 261). <u>https://doi.org/10.1002</u> /9781119241072.ch26
- 523. Yamakawa, H., Shaw, P. J., Williams, I. D., & Watabe, K. (2017). Food waste prevention: Lessons from the Love Food, Hate Waste campaign in the UK. Sixteenth International Waste Management and Landfill Symposium, Sardinia. <u>https://www.researchgate.net/profile</u> /Kohei-Watanabe-3/publication/320331025 Food waste prevention lessons from the Love Food Hate Waste campaign in the UK/links/5c08ca3a299bf139c741d082/Foodwaste-prevention-lessons-from-the-Love-Food-Hate-Waste-campaign-in-the-UK.pdf
- 524. van Geffen, L., van Herpen, E., & van Trijp, H. (2020). Household food waste—How to avoid it? An integrative review. In E. Närvänen, N. Mesiranta, M. Mattila, & A. Heikkinen (Eds.), *Food Waste Management: Solving the Wicked Problem* (pp. 27). Palgrave Macmillan. <u>https://doi.org/10.1007/978-3-030-20561-4_2</u>
- 525. Soma, T., Li, B., & Maclaren, V. (2020). Food waste reduction: A test of three consumer awareness interventions. *Sustainability*, *12*(3). <u>https://doi.org/10.3390/su12030907</u>
- 526. An initiative of the Federal Government to avoid food waste: too good for the bin. Bundesministeriums für Ernährung und Landwirtschaft - Federal Ministry of Food and Agriculture. Retrieved 13 March 2024 from <u>https://www.zugutfuerdietonne.de/</u> <u>navigation/sub-footer-navigation/english</u>
- 527. Sunshine Yates Consulting (Prepared for Ministry for the Environment Manatū Mō Te Taiao). (2023). *Research into barriers to use of food scraps collections*. <u>https://environment</u>.<u>govt.nz/assets/publications/Waste/Research-into-barriers-to-use-of-food-scrapscollections.pdf</u>
- 528. Candeal, T., Brüggemann, N., Bruns, H., Casonato, C., Diercxsens, C., Garcia Herrero, L., Gil, J. M., Haglund, Y., Kaptan, G., Kasza, G., Mikkelsen, B. E., Obersteiner, G., Pires, I. M., Swannell, R., Vainioranta, J., Van Herpen, E., Vittuari, M., Watanabe, K., & Sala, S. (2023). *Tools, best practices and recommendations to reduce consumer food waste A compendium*. <u>https://data.europa.eu/doi/10.2760/967005</u>
- 529. Boulet, M., Grant, W., Hoek, A., & Raven, R. (2022). Influencing across multiple levels: The positive effect of a school-based intervention on food waste and household behaviours. *J Environ Manage*, 308. <u>https://doi.org/10.1016/j.jenvman.2022.114681</u>
- 530. *Our kaupapa*. Garden to Table. Retrieved 01 May 2024 from https://gardentotable.org .nz/kaupapa
- 531. *About Enviroschools*. Enviroschools. Retrieved 01 May 2024 from <u>https://enviroschools</u>. .org.nz/about-us/
- 532. Lokhorst, A. M., Werner, C., Staats, H., van Dijk, E., & Gale, J. L. (2011). Commitment and behavior change. *Environment and Behavior*, 45(1),3. <u>https://doi.org/10.1177</u>/0013916511411477
- 533. *The food waste Warriors challenge*. Love Food Hate Waste New Zealand. Retrieved 7 March 2024 from <u>https://lovefoodhatewaste.co.nz/food-waste-warriors-challenge/</u>
- 534. Thaler, R., & Sunstein, C. (2009). *Nudge: Improving Decisions About Health, Wealth, and Happiness*. Penguin Books Ltd.
- 535. Tversky, A., & Kahneman, D. (1992). Advances in prospect theory: Cumulative representation of uncertainty. *Journal of Risk and uncertainty*, 5. <u>https://doi.org/10.1007/BF00122574</u>
- 536. Bovens, L. (2009). The ethics of nudge. In T. Grüne-Yanoff & S. O. Hansson (Eds.), *Preference Change: Approaches from Philosophy, Economics and Psychology* (pp. 207). Springer Netherlands. <u>https://doi.org/10.1007/978-90-481-2593-7_10</u>
- 537. Schmidt, A. T., & Engelen, B. (2020). The ethics of nudging: An overview. *Philosophy compass*, *15*(4). <u>https://doi.org/10.1111/phc3.12658</u>

- 538. Evers, C., Marchiori, D., Junghans, A., Cremers, J., & De Ridder, D. (2018). Citizen approval of nudging interventions promoting healthy eating: the role of intrusiveness and trustworthiness. *BMC Public Health*, *18*. https://doi.org/10.1186/s12889-018-6097-y
- 539. Thaler, R. H. (2018). Nudge, not sludge. *Science*, *361*(6401). <u>https://doi.org/10.1126/science</u>. .aau9241
- 540. *Eat me first*. Love Food Hate Waste New Zealand. Retrieved 8 May 2024 from <u>https://love foodhatewaste.co.nz/eat-me-first-stickers/</u>
- 541. *The new hub for your home*. Samsung. Retrieved 01 May 2024 from <u>https://www.samsung.</u> <u>com/nz/family-hub/</u>
- 542. Cappelletti, F., Papetti, A., Rossi, M., & Germani, M. (2022). Smart strategies for household food waste management. *Procedia Computer Science*, 200. <u>https://doi.org/10.1016/j</u>.<u>procs.2022.01.286</u>
- 543. Nguyen, V. N., Nguyen, T. H., Huynh, T. T., Nguyen, V. H., & Stigberg, S. K. (2015). Interactive fridge: A solution for preventing domestic food waste. Inclusive Smart Cities and e-Health, Cham. <u>https://doi.org/10.1007/978-3-319-19312-0_36</u>
- 544. *Spaghetti portioner*. Retrieved 16 May 2024 from <u>https://picosolutions.com/products</u> /spaghetti-portioner
- 545. Romani, S., Grappi, S., Bagozzi, R. P., & Barone, A. M. (2018). Domestic food practices: A study of food management behaviors and the role of food preparation planning in reducing waste. *Appetite*, *121*. <u>https://doi.org/10.1016/j.appet.2017.11.093</u>
- 546. Roe, B. E., Qi, D., Beyl, R. A., Neubig, K. E., Apolzan, J. W., & Martin, C. K. (2022). A randomized controlled trial to address consumer food waste with a technology-aided tailored sustainability intervention. *Resources, Conservation & Recycling, 179*. <u>https://doi.org/10.1016/j.resconrec.2021.106121</u>
- 547. Committee on world food security. (2014). *High level panel of experts report on food losses and waste in the context of sustainable food systems - Extract from the report: Summary and recommendations*. <u>https://www.fao.org/3/ml099e/ml099e.pdf</u>
- 548. Patra, D., Feng, S., & Howard, J. W. (2022). Confusion of food-date label with food safety Implications for food waste. *Current Opinion in Food Science*, 48. <u>https://doi.org/10.1016</u> /j.cofs.2022.100917
- 549. European Commission. (2018). *Market study on date marking and other information* provided on food labels and food waste prevention. <u>https://food.ec.europa.eu</u> /system/files/2018-07/fw lib srp date-marking.pdf
- 550. WasteMINZ. (2018). Love Food Hate Waste Campaign Evaluation. <u>https://lovefoodhate</u> waste.co.nz/wp-content/uploads/2019/02/FINAL-WasteMINZ-National-Food-Waste-Prevention-Study-2018.pdf
- 551. Badiger, A., Katz, T., Simons, C. T., & Roe, B. E. (2023). When considering whether to waste food, consumers focus attention on food label dates rather than phrases. *Waste Management*, *168*. <u>https://doi.org/10.1016/j.wasman.2023.06.006</u>
- 552. Gong, Z., Su, L. Y.-F., Zhang, J. S., Chen, T., & Wang, Y.-C. (2022). Understanding the association between date labels and consumer-level food waste. *Food Quality and Preference*, *96*. <u>https://doi.org/10.1016/j.foodqual.2021.104373</u>
- 553. Wilson, N. L. W., Rickard, B. J., Saputo, R., & Ho, S.-T. (2017). Food waste: The role of date labels, package size, and product category. *Food Quality and Preference*, *55*,35. <u>https://doi.org/10.1016/j.foodqual.2016.08.004</u>
- 554. Weis, C., Narang, A., Rickard, B., & Souza-Monteiro, D. M. (2021). Effects of date labels and freshness indicators on food waste patterns in the United States and the United Kingdom. *Sustainability*, *13*(14). <u>https://doi.org/10.3390/su13147897</u>
- 555. Turvey, C., Moran, M., Sacheck, J., Arashiro, A., Huang, Q., Heley, K., Johnston, E., & Neff, R. (2021). Impact of messaging strategy on consumer understanding of food date labels.

Journal of Nutrition Education and Behavior, *53*(5). <u>https://doi.org/10.1016/j.jneb</u>.2021.03.007

- 556. Ministry for the Environment Manatū Mō Te Taiao. (2023). *Te rautaki para | Waste strategy*. <u>https://environment.govt.nz/assets/publications/Te-rautaki-para-Waste-strategy.pdf</u>
- 557. Sustainable waste management: What the EU is doing. European Parliament. Retrieved 01 May 2024 from <u>https://www.europarl.europa.eu/topics/en/article/20180328STO00751</u> /sustainable-waste-management-what-the-eu-is-doing
- 558. *National waste policy*. Australian Department of Climate Change, Energy, the Environment and Water. Retrieved 01 May 2024 from <u>https://www.dcceew.gov.au/environment</u> /protection/waste/how-we-manage-waste/national-waste-policy
- 559. Singapore Ministry of the Environment and Water Resources, & Singapore National Environment Agency. *Zero waste masterplan Singapore*. <u>https://www.mse.gov.sg/resources</u> /zero-waste-masterplan.pdf
- 560. Waste and Resources Action Programme. (2011). *Literature review Relationship between household food waste collection and food waste prevention*. <u>https://wrap.org.uk/sites</u> /default/files/2020-12/Literature-Review-Relationship-between-Household-Food-Waste-Collection-and-Food-Waste-Prevention.pdf
- 561. Checkpoint. (2022, February 16). Auckland Council bins rubbish tag scheme. <u>https://www</u>.<u>rnz.co.nz/national/programmes/checkpoint/audio/2018830809/auckland-council-bins-rubbish-tag-scheme</u>
- 562. Schëtter. (2021). *Zero waste strategy*. <u>https://www.sias.lu/assets/files/SIAS_Faenk-vir-un-2021-Schuttrange-EN.pdf</u>
- 563. Dahlen, L., & Lagerkvist, A. (2010). Pay as you throw: Strengths and weaknesses of weightbased billing in household waste collection systems in Sweden. *Waste Management*, *30*(1). https://doi.org/10.1016/j.wasman.2009.09.022
- 564. Kim, M. (2023, August 24). How South Korea's composting system became a model for the world. *Los Angeles Times*. <u>https://www.latimes.com/world-nation/story/2023-08-24/how-south-koreans-composting-system-became-a-model-for-the-world</u>
- 565. *European citizens' food waste panel*. European Commission. Retrieved 01 May 2024 from https://citizens.ec.europa.eu/food-waste-panel_en
- 566. *Waste Minimisation Fund*. Ministry for the Environment Manatū Mō Te Taiao. Retrieved 21 March 2024 from <u>https://environment.govt.nz/what-you-can-do/funding/waste-</u> <u>minimisation-fund/</u>
- 567. Changes to DCC kerbside collection in 2024. Dunedin City Council Kaunihera a rohe o Ōtepoti. Retrieved 26 March 2024 from <u>https://www.dunedin.govt.nz/council/council-projects/waste-futures/the-future-of-rubbish-and-recycling-in-dunedin</u>
- 568. Addressing organic food waste in Whanganui. Ministry for the Environment Manatū Mō Te Taiao. Retrieved 01 May 2024 from <u>https://environment.govt.nz/what-you-can-</u><u>do/stories/addressing-organic-food-waste-in-whanganui/</u>
- 569. National food waste reduction programmes. Ministry for the Environment Manatū Mō Te Taiao. Retrieved 19 February 2024 from <u>https://environment.govt.nz/what-you-can-</u> <u>do/funding/resources-for-seekers-of-environmental-funding/national-food-waste-reduction-</u> <u>programmes/</u>
- 570. Datta, J., & Petticrew, M. (2013). Challenges to evaluating complex interventions: A content analysis of published papers. *BioMed Central Public Health*, *13*(568). <u>https://doi.org/10.1186</u>/1471-2458-13-568
- 571. Dunning, T. (2008). Improving causal inference: Strengths and limitations of natural experiments. *Political Research Quarterly*, *61*(2). <u>https://doi.org/10.1177</u>/1065912907306470

- 572. García-Herrero, L., Casonato, C., Caldeira, C., De Laurentiis, V., Bruns, H., & Sala, S. (2023). Scoping consumer food waste: an evaluation framework of prevention interventions. https://doi.org/10.2760/3128
- 573. van Dooren, C., Janmaat, O., Snoek, J., & Schrijnen, M. (2019). Measuring food waste in Dutch households: A synthesis of three studies. *Waste Management*, 94. <u>https://doi.org</u> /10.1016/j.wasman.2019.05.025
- 574. ReFED. (2018). *Restaurant food waste action guide*. <u>https://refed.org/downloads</u> /<u>Restaurant_Guide_Web.pdf</u>
- 575. OzHarvest. (2021). *Halving household foodwaste: Which behaviours matter*? <u>https://www</u>.ozharvest.org/halving-food-waste-report/
- 576. De Jager, J. N. W. (2019). Political consumerism in the South African and British antiapartheid movements: The historical role of consumer boycott campaigns. In M. Borstrom, M. Micheletti, & P. Oosterveer (Eds.), *The Oxford handbook of political consumerism*. Oxford University Press. <u>https://doi.org/10.1093/oxfordhb/9780190629038.001.0001</u>
- 577. Skinner, R. (2017). "Every bite buys a bullet": Sanctions, boycotts and solidarity in transnational anti-apartheid activism. *Moving the Social*, *57*. <u>https://doi.org/10.13154</u>/mts.57.2017.97-114
- 578. Webb, J. (2017). Seduced or sceptical consumers? Organised action and the case of fair trade coffee. *Sociological Research Online*, *12*(3),73. <u>https://doi.org/10.5153/sro.1536</u>
- 579. Funding for medium to long-term, industry-led research collaborations. Australian Government. Retrieved 14 May 2024 from <u>https://business.gov.au/grants-and-</u> programs/cooperative-research-centres-crc-grants#:~:text=Cooperative%20Research%20 <u>Centres%20(CRC)%20Grants%20provide%20funding%20for%20medium%20to,%2C%20indus</u> try%2Dled%20research%20collaborations
- 580. Cooperative Research Centres Association. (2019). 2020-2021 Pre-budget submission. https://treasury.gov.au/sites/default/files/2020-09/115786_COOPERATIVE_RESEARCH_ CENTRES_ASSOCIATION.pdf
- 581. Crown Research Institutes. Ministry of Business Innovation and Employment Hīkina Whakatutuiki, Retrieved 14 May 2014 from <u>https://www.mbie.govt.nz/science-and-technology/science-and-innovation/agencies-policies-and-budget-initiatives/research-organisations/cri</u>
- 582. *Current Cooperative Research Centres (CRCs)*. Australian Government. Retrieved 14 May 2024 from <u>https://business.gov.au/grants-and-programs/cooperative-research-centres-crc-grants/current-cooperative-research-centres-crcs</u>
- 583. *A growing network dedicated to rethinking the future of our food*. End Food Waste Australia. Retrieved 14 May 2024 from <u>https://endfoodwaste.com.au/our-partners/</u>
- 584. *The Partnerships Scheme*. Ministry of Business Innovation and Employment Hīkina Whakatutuiki. Retrieved 14 May 2024 from <u>https://www.mbie.govt.nz/science-and-</u> <u>technology/science-and-innovation/funding-information-and-opportunities/investment-</u> <u>funds/partnerships</u>
- 585. Research and development survey: 2023. Stats NZ -Tatauranga Aotearoa. Retrieved 29 April 2024 from https://www.stats.govt.nz/information-releases/research-and-development-survey-2023/#:~:text=the%20number%20of%20businesses%20performing,to%200.95%20percent%20in%202023
- 586. Ministry of Business, Innovation and Employment Hīkina Whakatutuiki. (2021). *The research, science and innovation report.* <u>https://mbienz.shinyapps.io/research-science-innovation-report/pdf/research-science-and-innovation-system-performance-report-2021.pdf</u>
- 587. About Sustainable Food and Fibre Futures. Ministry for Primary Industries Manatū Ahu Matua. Retrieved 21 March 2024 from <u>https://www.mpi.govt.nz/funding-rural-</u> <u>support/sustainable-food-fibre-futures/about-sustainable-food-and-fibre-futures/</u>

- 588. Ministry for Primary Industries Manatū Ahu Matua. *Guidelines for applicants: Sustainable food and fibre futures*. <u>https://www.mpi.govt.nz/dmsdocument/32656-Sustainable-Food-and-Fibre-Futures-Guidelines-for-Applicants</u>
- 589. *Ārohia trailblazer grant*. Callaghan Innovation Te Pokapū Auaha. Retrieved 09 April 2024 from <u>https://www.callaghaninnovation.govt.nz/products/fund/arohia-trailblazer-grant/</u>
- 590. *Deep tech incubators*. Callaghan Innovation Te Pokapū Auaha. Retrieved 09 April 2024 from <u>https://www.callaghaninnovation.govt.nz/products/fund/deep-tech-incubators/</u>
- 591. *R&D career grants*. Callaghan Innovation Te Pokapū Auaha. Retrieved 09 April 2024 from <u>https://www.callaghaninnovation.govt.nz/products/fund/r-and-d-career-grants/</u>
- 592. *R&D experience grants*. Callaghan Innovation Te Pokapū Auaha. Retrieved 09 April 2024 from <u>https://www.callaghaninnovation.govt.nz/products/fund/r-and-d-experience-grants/</u>
- 593. Ministry of Business, Innovation and Employment Hīkina Whakatutuiki. (2015). *National* statement of science investment. <u>https://www.mbie.govt.nz/assets/2eaba48268/national-</u> statement-science-investment-2015-2025.pdf
- 594. The Country. (2016, July 14). Innovation playing a bigger role. <u>https://www.nzherald.co.nz</u> /the-country/news/innovation-playing-a-bigger-role/C5M72WPET4RONNIUN7NUOBDUPE/
- 595. *Fonterra research and development centre*. Fonterra. Retrieved 08 April 2024 from <u>https://www.fonterra.com/nz/en/our-co-operative/research-and-development.html#stories</u>
- 596. *Food Waste Innovation Auahatanga Parakai*. University of Otago Ōtakou Whakaihu Waka. Retrieved 10 April 2024 from <u>https://foodwaste-otago.org/</u>
- 597. *Circular bioeconomy*. AgResearch. Retrieved 10 April 2024 from <u>https://www.agresearch</u>.co.nz/science-areas/transformational-initiatives/circular-bioeconomy/
- 598. *Sustainable, resilient food supply*. Plant & Food Research Rangahau Ahumāra Kai. Retrieved 10 April 2024 from <u>https://www.plantandfood.com/en-nz/sustainable-resilient-food-supply</u>
- 599. *Top of the South organic waste mapping study*. Plant & Food Research Rangahau Ahumāra Kai,. Retrieved 10 April 2024 from <u>https://www.plantandfood.com/en-nz/article/top-of-the-south-organic-waste-mapping-study</u>
- 600. *Growing the food and beverage sector*. Ministry of Business, Innovation and Employment -Hīkina Whakatutuki. Retrieved 15 April 2024 from <u>https://www.mbie.govt.nz/business-and-</u> <u>employment/economic-development/growing-the-food-and-beverage-sector/</u>
- 601. Focus area: growing export success. Ministry of Business, Innovation and Employment -Hīkina Whakatutuki. Retrieved 15 April 2024 from <u>https://www.mbie.govt.nz/business-and-</u> <u>employment/economic-development/industry-policy/industry-transformation-plans/digital-</u> <u>technologies/digital-technologies-industry-transformation-plan/focus-area-growing-export-</u> <u>success/</u>
- 602. About us. AgriTech. Retrieved 10 April 2024 from https://agritechnz.org.nz/about-us/
- 603. Ministry for Primary Industries Manatū Ahu Matua. (2012). *Survey of technology transfer services to farmers and growers in New Zealand*. <u>https://www.mpi.govt.nz/dmsdocument/</u> <u>3740-survey-of-technology-transfer-services-to-farmers-and-growers-in-new-zealand</u>
- 604. Mu, L., Liu, H., Cui, Y., Fu, L., & Gejima, Y. (2018). Mechanized technologies for scaffolding cultivation in the kiwifruit industry: A review. *Information Processing in Agriculture*, 5(4). https://doi.org/10.1016/j.inpa.2018.07.005
- 605. *KVH policy research and development*. Kiwifruit Vine Health. Retrieved 15 May 2024 from <u>https://kvh.org.nz/assets/documents/About-KVH-tab/KVH_Policy_Research_and_Development.pdf</u>
- 606. Parminter, T. G., & Max, S. (2004). *A learning model of technology transfer for the kiwifruit industry*. New Zealand Agricultural and Resource Economics Society June 25-26, Blenheim, New Zealand. <u>https://ageconsearch.umn.edu/record/97784/?v=pdf</u>
- 607. Plant & Food Research. (2021, August 12). New kiwifruit breeding centre driving greater innovation to commence in 2021. <u>https://www.plantandfood.com/en-nz/article/new-kiwifruit-breeding-centre-driving-greater-innovation-to-commence-in-2021</u>

- 608. Plant & Food Research Rangahau Ahumāra Kai. (2021, October 04). New kiwifruit breeding centre launches to boost innovation. <u>https://www.plantandfood.com/en-nz/article/new-kiwifruit-breeding-centre-launches-to-boost-innovation</u>
- 609. *Zespri Innovation Fund*. Zespri Kiwifruit. Retrieved 08 April 2024 from <u>https://www.zespri</u>.com/en-NZ/zagfund
- 610. *International partnerships INRAE*. Science New Zealand. Retrieved 09 April 2024 from <u>https://sciencenewzealand.org/our-member/international-partnerships/inrae/</u>
- 611. Plant & Food Research Rangahau Ahumāra Kai. (2023, March 20). New raspberry to extend harvest window. <u>https://www.plantandfood.com/en-nz/article/new-raspberry-to-extend-harvest-window</u>
- 612. New Zealanders' food waste bill continues to grow despite improvements in household food behaviours. Rabobank. Retrieved 09 April 2024 from https://www.rabobank.co.nz/media-releases/2021/210621-new-zealanders-food-waste-bill-continues-to-grow-despite-improvements-in-food-behaviours/
- 613. *Kiwis throwing away less of their grocery spend, but \$3.2 billion of wasted food still ending up in the bin.* Rabobank. Retrieved 9 April 2024 from <u>https://www.rabobank.co.nz/media-releases/2023/260923-rabobank-kiwiharvest-2023-food-waste-survey/</u>
- 614. Browne, S., Mullen, A., Mulholland, B., Lo, C., & Ruttledge, A. (2023). Nutrition quality and food and packaging waste associated with the school food system: A pilot, citizen science study in an Irish secondary school. *Journal of Human Nutrition and Dietetics*, *36*(6). https://doi.org/10.1111/jhn.13220
- 615. Pateman, R. M., De Bruin, A., Piirsalu, E., Reynolds, C., Stokeld, E., & West, S. E. (2020). Citizen science for quantifying and reducing food loss and food waste. *Frontiers in Sustainable Food Systems*, 4. <u>https://doi.org/10.3389/fsufs.2020.589089</u>
- 616. Reynolds, C., Oakden, L., West, S., Pateman, R. M., Elliott, C., Armstrong, B., Gillespie, R., & Patel, M. (2021). Citizen science for the food system. In K. Cohen & R. Doubleday (Eds.), *Future Directions for Citizen Science and Public Policy*. Centre for Science and Policy. <u>https://openaccess.city.ac.uk/id/eprint/26319/</u>
- 617. Emİnoğlu, M. B., & Yegül, U. (2022). Chapter 4: Smart farming application in fruit harvesting. In T. Akar (Ed.), *Research & reviews in agriculture, forestry and aquaculture* (pp. 45). Yaşar Hız. <u>https://www.gecekitapligi.com/Webkontrol/uploads/Fck/Agriculturearalik2022_2.pdf</u>
- 618. Zhang, Z., Ampatzidis, Y., Fu, L., & Zhang, Z. (2022). Economic analysis of an apple harvest and infield sorting machine. In Z. Zhang, Z. Zhang, C. Igathinathane, Y. Wang, Y. Ampatzidis, & G. Liu (Eds.), *Mechanical Harvest of Fresh Market Apples. Progress Over the Past Decades* (Vol. 1). Springer. <u>https://doi.org/10.1007/978-981-16-5316-2_7</u>
- 619. Frangoul, A. (2019, March 26). Automated robots are picking apples at an orchard in New Zealand. *CNBC*. <u>https://www.cnbc.com/2019/03/26/automated-robots-are-picking-apples-at-an-orchard-in-new-zealand.html</u>
- 620. Stowe, E., & Dhingra, A. (2021). Development of the Arctic® Apple. In I. Goldman (Ed.), *Plant Breeding Reviews* (pp. 273). John Wiley & Sons, Inc. <u>https://doi.org/10.1002/9781119717003</u> .ch8
- 621. Leasure-Earnhardt, A., Scrufari, C. A., & Valentine, R. (2017). The national gleaning project: The importance of gleaning and gresh food recovery in a sustainable and just food system. In I. Werkheiser & Z. Piso (Eds.), *Food justice in US and global contexts* (Vol. 24, pp. 171). Springer. <u>https://doi.org/10.1007/978-3-319-57174-4_15</u>
- 622. Fundraising and community events food safety rules. Ministry for Primary Industries -Manatū Ahu Matua. Retrieved 15 May 2024 from <u>https://www.mpi.govt.nz/food-</u> <u>business/starting-a-food-business/exemptions-from-the-food-act/fundraising-and-</u> <u>community-event-food-safety-rules/</u>
- 623. *Protecting your crops from frost, hail, wind*. NZ Canopies. Retrieved 14 May 2024 from <u>https://nzcanopies.co.nz/</u>

- 624. Nissen, R., Bound, S., Adhikari, R., & Cover, I. (2018). *Factors affecting post-harvest* management of apples: A guide to optimising quality. <u>https://www.resilience.com.au/wp-</u> content/uploads/2019/01/FGT-PostHarvestManual2018-LOWRES.pdf
- 625. Moraes, N. V., Lermen, F. H., & Echeveste, M. E. S. (2021). A systematic literature review on food waste/loss prevention and minimization methods. *Journal of Environmental Management*. <u>https://doi.org/10.1016/j.jenvman.2021.112268</u>
- 626. Uthairatanakij, A., Laohakunjit, N., Jitareerat, P., Cholmaitri, C., & Golding, J. (2023). Green technology for reducing postharvest losses and improving the nutritional quality of fresh horticultural produce. In K. İbrahim (Ed.), *New Advances in Postharvest Technology*. IntechOpen. <u>https://doi.org/10.5772/intechopen.109938</u>
- 627. *Longer shelf life, better quality*. Hazel Technologies. Retrieved 15 December 2023 from <u>https://www.hazeltechnologies.com/</u>
- 628. Lu, R., Zhang, Z., & Pothula, A. K. (2017). Innovative technology for apple harvest and in-field sorting. *Fruit Quarterly*, *25*(2). <u>https://nyshs.org/wp-content/uploads/2017/12/Lu-Pages-11-14-from-NYFQ-Book-Summer-2017-2.pdf</u>
- 629. Lee, H., Moon, A., Moon, K., & Lee, Y. (2017). *Disease and pest prediction IoT system in orchard: A preliminary study*. 2017 Ninth International Conference on Ubiquitous and Future Networks (ICUFN). <u>https://doi.org/10.1109/ICUFN.2017.7993840</u>
- 630. Ludwig-Ohm, S., Dirksmeyer, W., & Klockgether, K. (2019). Approaches to reduce food losses in German fruit and vegetable production. *Sustainability*, *11*(23). <u>https://doi.org/10.3390/su</u> <u>11236576</u>
- 631. *Protecting your orchard with an overhead canopy*. Growtech. Retrieved 5 December 2023 from <u>https://growtechgroup.co.nz/blog/protecting-your-orchard-overhead-canopy</u>
- 632. Law, T., & Mathewson, N. (2021, May 16). Hail-struck kiwifruit saved from going to waste and donated to people in need. *Stuff*. <u>https://www.stuff.co.nz/national/125148610</u> /hailstruck-kiwifruit-saved-from-going-to-waste-and-donated-to-people-in-need
- 633. Antunes, M. D. C., Panagopoulos, T., Rodrigues, S., Neves, N., & Curado, F. (2005). The effect of pre- and postharvest calcium applications on 'Hayward' kiwifruit storage ability. Leuven, Belgium. <u>https://doi.org/10.17660/ActaHortic.2005.682.118</u>
- 634. Ruiz-Aracil, M. C., Guillén, F., Ilea, M. I. M., Martínez-Romero, D., Lorente-Mento, J. M., & Valverde, J. M. (2023). Comparative effect of melatonin and 1-methylcyclopropene postharvest applications for extending 'Hayward' kiwifruit storage life. *Agriculture*, *13*(4). https://doi.org/10.3390/agriculture13040806
- 635. Chen, D., Tang, J. L., Xi, H. X., & Zhao, X. R. (2021). Image recognition of modern agricultural fruit maturity based on internet of things. *Traitment du Signal, 38*(4). https://doi.org/10.18280/ts.380435
- 636. Rojas, G., Fernandez, E., Whitney, C., Luedeling, E., & Cuneo, I. F. (2021). Adapting sweet cherry orchards to extreme weather events Decision analysis in support of farmers' investments in Central Chile. *Agricultural systems*, *187*. <u>https://doi.org/10.1016/j.agsy.2020</u>.103031
- 637. *Kiwifruit orchard monitoring*. Harvest.com. Retrieved 10 December 2023 from <u>https://</u><u>harvest.com/solutions/kiwifruit-orchards/</u>
- 638. Little, K., Patti, T., & Haritos, V. (2022). *Literature review: Valorisation options for stone fruit waste.* Monash University (Prepared for Sustainability Victoria and Cutri Fruit). <u>https://assets.sustainability.vic.gov.au/susvic/Report-Waste-Valorisation-options-for-stone-fruit-waste-Monash-University</u> 2022-07-18-051339_ehur.pdf
- 639. Murphy, S. (2024, January 31). Business playing a part in diverting cherry waste for use in smoothies, icecream. *RNZ*. <u>https://www.rnz.co.nz/news/country/507923/business-playing-a-part-in-diverting-cherry-waste-for-use-in-smoothies-icecream</u>
- 640. Jobbágy, J., Dočkalík, M., Krištof, K., & Burg, P. (2021). Mechanized grape harvest efficiency. *Applied Sciences*, *11*(10). <u>https://doi.org/10.3390/app11104621</u>

- 641. Nasrollahiazar, E., Basso, B., Sabbatini, P., & Reinke, M. *Opportunities for mechanization and precision viticulture in Michigan wine and juice grape vineyards*. Press Release. <u>https://www.canr.msu.edu/news/mechanization-and-precision-viticulture-in-michigan-wine-and-juice-grape-vineyards</u>
- 642. *Croptide*. Retrieved 15 December 2023 from <u>https://croptide.com/</u>
- 643. New Zealand Wine. (2023). *Vineyard report 2023 New Zealand wine growers*. https://www.nzwine.com/media/d41a2ayk/2023_vineyard-register-2023.pdf
- 644. Cayuela, J. A., Vázquez, A., Pérez, A., & García, J. (2009). Control of table grapes postharvest decay by ozone treatment and resveratrol induction. *Food Science and Technology International*, 15(5). <u>https://doi.org/10.1177/1082013209350539</u>
- 645. McKenzie, T. J., Singh-Peterson, L., & Underhill, S. J. R. (2017). Quantifying postharvest loss and the implication of market-based decisions: A case study of two commercial domestic tomato supply chains in Queensland, Australia. *Horticulturae*, *3*(3). <u>https://doi.org/10.3390</u>/horticulturae3030044
- 646. Nkolisa, N., Magwaza, L. S., Workneh, T. S., Chimphango, A., & Sithole, N. J. (2019).
 Postharvest quality and bioactive properties of tomatoes (*Solanum lycopersicum*) stored in a low-cost and energy-free evaporative cooling system. *Heliyon*, 5(8). <u>https://doi.org/10.1016</u>
 /i.heliyon.2019.e02266
- 647. Mesterházy, Á., Oláh, J., & Popp, J. (2020). Losses in the grain supply chain: Causes and solutions. *Sustainability*, *12*(6). <u>https://doi.org/10.3390/su12062342</u>
- 648. Western Cape to enforce 50% restriction on organic waste in landfills by 2022. Western Cape Government. Retrieved 26 March 2024 from <u>https://www.westerncape.gov.za/110green</u> /news/western-cape-enforce-50-restriction-organic-waste-landfills-2022
- 649. Western Cape Government, & Department of Environmental Affairs and Development Planning. (2021). *Reducing waste in Western Cape landfills.* <u>https://www.westerncape.gov.za/eadp/sites/eadp.westerncape.gov.za/files/atoms/files/Gr</u> <u>een%20Ambassador Special%20Edition%202021 Organic%20Waste 0.pdf</u>
- 650. *Waste not...* Western Cape Government. Retrieved 26 March 2024 from https://www.westerncape.gov.za/110green/news/waste-not
- 651. Pelarangan sampah organik ke TPA sebagai solusi krisis pengelolaan sampah regional sekaligus langkah mitigasi perubahan iklim. Yayasan Pecinta Budaya Bebali (Indonesia). Retrieved 26 March 2024 from <u>https://ypbb.web.id/resource/pelarangan-sampah-organikke-tpa-sebagai-solusi-krisis-pengelolaan-sampah-regional-sekaligus-langkah-mitigasiperubahan-iklim/</u>
- 652. West Java Public Relations. (2024, March 5). Greater Bandung agrees to reduce 50 percent of waste to Sarimukti TPA. <u>https://jabarprov.go.id/berita/bandung-raya-sepakat-kurangi-50persen-sampah-ke-tpa-sarimukti-10206</u>
- 653. Lee, E., Shurson, G., Oh, S. H., & Jang, J. C. (2024). The management of food waste recycling for a sustainable future: A case study on South Korea. *Sustainability*, *16*(2). https://doi.org/10.3390/su16020854
- 654. Yoo, K. Y., Horváth, B., Chowdhury, S., Adiwal, J., & UNDP Seoul Policy Centre. (2019). *Food waste management in Korea: Focusing on Seoul.* United Nations Development Programme. <u>https://www.undp.org/sites/g/files/zskgke326/files/migration/seoul_policy_center/USPC-</u> Policy-Brief-6-Food-Waste.pdf
- 655. Kim, H., & Park, J. (2023). Quantification of food loss and waste and its percentage estimation along the food supply chain in Korea. *Waste Management and Research*, 41(10). https://doi.org/10.1177/0734242X231178476
- 656. Yu, K. Y. (2017, January 13). Volume based waste fee (VBMF) system for municipal solid waste. *Seoul Solution*. <u>https://seoulsolution.kr/en/node/6326</u>

- 657. Solid Waste-Resource Management Regulations made under Section 102 of the Environment Act 1994-95, Province of Nova Scotia. <u>https://novascotia.ca/just/regulations</u> /regs/envsolid.htm
- 658. Giroux Environmental Consulting (Prepared for the Canadian Council of Ministers of the Environment Le Conseil canadien des ministres de l'environment.). (2021). Best management practices for disposal bans, levies, and incentives for end-of-life plastics. https://ccme.ca/en/res/finaldisposalbansbmps-ensecured.pdf
- 659. Solid waste by-law (Consolidated by the City of Vancouver for convenience only to December 12, 2023). <u>https://bylaws.vancouver.ca/8417c.pdf</u>
- 660. Metro Vancouver. (2021). *Metro Vancouver recycling and solid waste management*. <u>https://metrovancouver.org/services/solid-waste/Documents/solid-waste-management-annual-summary-2021.pdf</u>
- 661. *Commercial food material disposal ban*. Commonwealth of Massachusetts. Retrieved 28 March 2024 from <u>https://www.mass.gov/guides/commercial-food-material-disposal-ban</u>
- 662. Healey, M. T., Driscoll, K., Tepper, R. L., & Heiple, B. (2023). Organics action plan. Commonwealth of Massachusetts Exectuive Office of Energy and Environmental Affairs. <u>https://www.mass.gov/doc/massachusetts-organics-action-plan-november-</u> 2023/download#:~:text=Massachusetts%20food%20waste%20diversion%20has,tons%20on %20an%20annual%20basis.
- 663. *Solid waste master plan*. Commonwealth of Massachusetts. Retrieved 28 March 2024 from https://www.mass.gov/guides/solid-waste-master-plan
- 664. *Food waste requirements*. City of Seattle. Retrieved 28 March 2024 from <u>https://www.seattle.gov/utilities/your-services/collection-and-disposal/food-and-yard/food-waste-requirements</u>
- 665. City of Seattle. (2022). *Seattle's 2022 solid waste plan update*. <u>https://www.seattle.gov</u> /documents/departments/spu/documents/plans/solidwaste/2022solidwastedraftch3.pdf
- 666. *Food and yard rates*. City of Seattle. Retrieved 24 March 2024 from <u>https://www.seattle</u>.<u>.gov/utilities/your-services/accounts-and-payments/rates/collection-and-disposal/food-and-yard-rates</u>
- 667. Vermont Agency of Natural Resources Department of Environmental Conservation. (2020). Universal recycling: Food scrap ban guidance. <u>https://dec.vermont.gov/sites/dec</u> /files/wmp/SolidWaste/Documents/Universal-Recycling/Food-Scrap-Ban-Guidance.pdf
- 668. Waste Management and Prevention Division (Prepared for Department of Environmental Conservation, Vermont). (2021). 2020 diversion and disposal report. <u>https://dec.vermont</u> .gov/sites/dec/files/wmp/SolidWaste/Documents/2020-Diversion-and-Disposal-Report.pdf
- 669. European Commission. (2023). *Proposal for a directive of the European Parliament and of the Council amending Directive 2008/98/EC on waste*. <u>https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX:52023PC0420</u>
- 670. Favoino, E., Giavini, M., & del Parco di Monza, S. A. (2020). *Bio-waste generation in the EU: Current capture levels and future potential.* Zero Waste Europe (commissioned by the Biobased Industries Consortium). <u>https://zerowasteeurope.eu/wp-content/uploads</u> /2020/07/2020 07 06 bic zwe report bio waste.pdf
- 671. Paleari, S. (2016). *Municipal waste management: Austria*. European Environment Agency. <u>https://www.eionet.europa.eu/etcs/etc-ce/products/country-profiles-on-the-management-of-municipal-waste-1/austria_msw_2016.pdf</u>
- 672. Ettlinger, S., & Bapasola, A. (2016). *Landfill tax, incineration tax and landfill ban in Austria.* Eunomia, Institute for European Environmental Policy. <u>https://ieep.eu/wp-content/uploads</u> /2022/12/AT-Landfill-Tax-final.pdf
- 673. VLAREMA. Vlaanderen Flemish Government. Retrieved 27 March 2024 from <u>https://</u> <u>navigator.emis.vito.be/detail?wold=43991&woLang=en</u>

- 674. Wallonie. (2017). *Waloon Waste-Respources Plan: non-technical summary.* http://environnement.wallonie.be/dechetsressources/docs/WWRP-NTS-EN.pdf
- 675. European Environment Agency. (2022). *Country profile: Belgium*. https://www.eea.europa.eu/publications/many-eu-member-states/belgium
- 676. European Environment Agency. (2022). *Country profile: Czechia.* https://www.eea.europa.eu/publications/many-eu-member-states/czechia
- 677. European Environment Agency. (2022). *Country profile: Denmark.* https://www.eea.europa.eu/publications/many-eu-member-states/denmark
- 678. European Environment Agency. (2013). *Municipal waste management in Denmark*. <u>https://www.eea.europa.eu/publications/managing-municipal-solid-waste/denmark-municipal-waste-management</u>
- 679. European Commission. (2011). *Country factsheet Estonia (EE).* <u>https://ec.europa.eu/environment/pdf/waste/framework/EE%20factsheet_FINAL.pdf</u>
- 680. European Environment Agency. (2022). *Country profile: Estonia.* https://www.eea.europa.eu/publications/many-eu-member-states/estonia/view
- 681. Papineschi, J., Hogg, D., Chowdhury, T., Durrant, C., & Thomson, A. (2019). *Analysis of Nordic regulatory framework and its effect on waste prevention and recycling in the region*. Nordic Council of Ministers. <u>https://doi.org/10.6027/TN2019-522</u>
- 682. European Environment Agency. (2022). *Country profile: Finland.* https://www.eea.europa.eu/publications/many-eu-member-states/finland/view
- 683. European Environment Agency. (2022). *Country profile: Germany*. <u>https://www.eea.europa.eu/themes/waste/waste-prevention/countries/2023-waste-prevention-country-fact-sheets/germany_waste_prevention_2023</u>
- 684. European Environment Agency. (2022). *Country profile: Hungary*. https://www.eea.europa.eu/publications/many-eu-member-states/hungary/view
- 685. *Law on Waste Management (No. VIII-787)*. Food and Agriculture Organisation of the United Nations. Retrieved 20 May 2024 from <u>https://www.fao.org/faolex/results/details/en/c/LEX-FAOC028121/</u>
- 686. European Environment Agency. (2023). *Country profile: Lithuania.* <u>https://www.eea.europa.eu/publications/many-eu-member-states/lithuania/view</u>
- 687. Lieten, S. H., & Dijcker, R. (2018). *Landfill management in the Netherlands*. Rijkswaterstaat Water, Verkeer en Leefomgeving, Netherlands. <u>https://rwsenvironment.eu/publish/pages</u> /126540/landfill_management_in_the_netherlands_cocoon_20180503.pdf
- 688. European Environment Agency. (2023). *Country profile: Netherlands.* <u>https://www.eea.europa.eu/publications/many-eu-member-states/netherlands/view</u>
- 689. European Environment Agency. (2013). *Municipal waste management in Norway*. <u>https://www.eea.europa.eu/publications/managing-municipal-solid-waste/norway-</u>municipal-waste-management
- 690. European Environment Agency. (2022). *Country profile: Poland.* https://www.eea.europa.eu/publications/many-eu-member-states/poland/view
- 691. European Environment Agency. (2015). *Factsheet Slovenia*. <u>https://ec.europa.eu/environment/pdf/waste/framework/facsheets%20and%20roadmaps/Factsheet_Slovenia.pdf</u>
- 692. Ordinance (2001:512) on the disposal of waste, Ministry of Climate and Business Sweden. http://rkrattsbaser.gov.se/sfst?bet=2001:512
- 693. *Sweden is aiming for zero waste*. Swedish Institute. Retrieved 3 April 2024 from <u>https://sweden.se/climate/sustainability/swedish-recycling-and-beyond</u>
- 694. Merta, E., Milios, L., & Reichel, A. (2016). *Municipal waste management: Sweden*. European Environment Agency. <u>https://www.eionet.europa.eu/etcs/etc-ce/products/country-profiles-on-the-management-of-municipal-waste-1/sweden_msw_2016.pdf</u>

- 695. European Environment Agency. (2022). *Country profile: Sweden.* https://www.eea.europa.eu/publications/many-eu-member-states/sweden/view
- 696. Elander, M., Sternmarck, A., & Ostergren, K. (2016). *Sweden Country report on national food waste policy*. Fusions. <u>https://www.eu-fusions.org/phocadownload/country-report/SWEDEN%2023.02.16.pdf</u>
- 697. Ordinance on the Avoidance and the Disposal of Waste 2015, Schweizerische Eidgenossenschaft Switzerland. <u>https://www.fedlex.admin.ch/eli/cc/2015/891/en</u>
- 698. Herczeg, M. (2013). *Municipal waste management in Switzerland*. European Environment Agency. <u>https://www.eea.europa.eu/publications/managing-municipal-solid-</u> waste/switzerland-municipal-waste-management
- 699. The landfill (Scotland) regulations 2003: Prohibition of acceptance of certain wastes at landfills 2003. <u>https://www.legislation.gov.uk/ssi/2003/235/regulation/11</u>
- 700. Scottish Environment Protection Agency. (2024). *The ban on landfilling biodegradable municipal waste (scope and testing guidance)*. <u>https://www.sepa.org.uk/media/vknblvy3/bmw-testing-guidance.docx</u>
- 701. APS Group Scotland (Prepared for the Scottish Government). (2011). *Zero waste regulations: Policy statement*. <u>https://www.gov.scot/binaries/content/documents/govscot</u> /publications/corporate-report/2011/10/policy-statement-zero-waste-regulations /documents/0121809-pdf/0121809-pdf/govscot%3Adocument/0121809.pdf
- 702. Valentic, S. (2022, December 6). Zero waste Scotland examines biostabilization of residual waste. *Waste 360*. <u>https://www.waste360.com/organic-waste/zero-waste-scotland-examines-biostabilization-of-residual-waste</u>
- 703. *Assisted distressed sales*. ReFED. Retrieved 12 March 2024 from <u>https://insights-engine.refed.org/solution-database/assisted-distressed-sales</u>
- 704. *Masters of overstock food*. Natural Choice Foods. Retrieved 12 March 2024 from https://naturalchoicefoods.com/
- 705. *Surplus foods delivered with exceptional service*. Marvell Foods. Retrieved 12 March 2024 from <u>https://www.marvellfoods.com/</u>
- 706. *Anything you make, we can help make better*. Mid America Food Sales Ltd. Retrieved 12 March 2024 from <u>https://midamfoodsales.com/</u>
- 707. *Welcome to Surplus Direct*. Surplus Direct. Retrieved 12 March 2024 from <u>https://www.surplusdirect.co.nz/about-us</u>
- 708. *Dynamic pricing*. ReFED. Retrieved 6 March 2024 from <u>https://insights-engine.refed</u>.org/solution-database/dynamic-pricing
- 709. Barriola, X., Holmstrom, J., Loikkanen, L., & Saarinen, L. (2023). *Happy hour: markdown pricing to reduce food waste*. <u>https://papers.ssrn.com/sol3/papers.cfm?abstract_id=4687311</u>
- 710. *Resources: success stories and whitepapers*. Wasteless. Retrieved 6 March 2024 from https://www.wasteless.com/resources#:~:text=Fresh%20format%20retailer%20slashes%20 waste%20by%2039%25&text=By%20implementing%20Wasteless%20end%2Dto,freshness% 20at%20the%20optimal%20level.
- 711. Harvey, L. (2017, March 28). Using data analytics to reduce food waste. *MathWorks*. <u>https://blogs.mathworks.com/headlines/2017/03/28/using-data-analytics-to-reduce-food-waste/</u>
- 712. *Markdown alert applications*. ReFED. Retrieved 6 March 2024 from <u>https://insights-engine.refed.org/solution-database/markdown-alert-applications</u>
- 713. Too Good To Go. (2021). *Impact report*. <u>https://tgtg-mkt-cms-prod.s3.eu-west-</u> <u>1.amazonaws.com/27100/USImpactReport.pdf</u>
- 714. Mirosa, M., Liu, Y., & Mirosa, R. (2018). Consumers' behaviors and attitudes toward doggy bags: Identifying barriers and benefits to promoting behavior change. *Journal of Food Products Marketing*, *24*(5),563. <u>https://doi.org/10.1080/10454446.2018.1472699</u>

- 715. van Herpen, E., De Hooge, I. E., de Visser-Amundson, A., & Kleijnen, M. P. (2021). Take it or leave it: How an opt-out strategy for doggy bags affects consumer food waste behavior and restaurant evaluations. *Journal of Cleaner Production*, 325. <u>https://doi.org/10.1016/j.jclepro.2021.129199</u>
- 716. Vizzoto, F., Tessitore, S., Testa, F., & Iraldo, F. (2021). Plate waste in foodservice outlets: Revealing customer profiles and their support for potentially contentious measures to reduce it in Italy. *Resources, Conservation and Recycling, 174*. <u>https://doi.org/10.1016/j</u> .resconrec.2021.105771
- 717. Bozzola, M., Dal Palu, D., & De Giorgi, C. (2017). *Design for leftovers From food waste to social responsibility*. Design for Next 12th EAD Conference, Rome. <u>https://doi.org/10.1080</u> /14606925.2017.1352692
- 718. Billore, S., Kobayashi, T., & Wang, O. (2021). Consumer attitudes towards leftover food takeout interventions: a case study of the doggy bag in Japan. *British Food Journal*, *123*(9). https://doi.org/10.1108/BFJ-03-2021-0231
- 719. *Food Saver Challenge*. Nashville.gov. Retrieved 27 March 2024 from <u>https://www.nashville</u>.gov/departments/mayor/sustainability-resilience-environment/food-saver-challenge
- 720. Keating, M. (2019, November 1). Denver restaurants agree: Wasting food is nuts! *Natural Resources Defense Council*. <u>https://www.nrdc.org/bio/madeline-keating/denver-restaurants-agree-wasting-food-nuts</u>
- 721. *Food waste challenge*. World Chefs. Retrieved 13 February 2024 from <u>https://feedtheplanet</u> .worldchefs.org/fwc/
- 722. Commission for Environmental Cooperation, Canada (2019). *Why and how to measure food loss and waste*. <u>http://www.cec.org/files/documents/publications/11814-why-and-how-</u> <u>measure-food-loss-and-waste-practical-guide-en.pdf</u>
- 723. *Buffet signage*. ReFED. Retrieved 20 March 2024 from <u>https://insights-engine.refed.org</u> /solution-database/buffet-signage
- 724. Richardson, R., Prescott, M. P., & Ellison, B. (2021). Impact of plate shape and size on individual food waste in a university dining hall. *Resources, Conservation and Recycling, 168*. https://doi.org/10.1016/j.resconrec.2020.105293
- 725. Zhang, P., Fu, S., & Liu, X. (2022). The effect of plate and decoration color on consumer food waste in restaurants: A case of four Chinese cities. *Sustainability*, *14*(6). <u>https://doi.org/10</u>.3390/su14063479
- 726. Mills, B., O'Brien, R., Mirosa, M., & Skeaff, S. (2023). Food waste initiatives in the tertiary sector in Aotearoa New Zealand: The challenge of meeting SDG 12.3. *Pūhau ana te rā: Tailwinds*, 1(2). <u>https://doi.org/10.11157/patr.v1i1.23</u>
- 727. *Portion sizes*. ReFED. Retrieved 20 March 2024 from <u>https://insights-engine.refed.org</u> /solution-database/portion-sizes
- 728. Berkowitz, S., Marquart, L., Mykerezi, E., Degeneffe, D., & Reicks, M. (2016). Reducedportion entrees in a worksite and restaurant setting: Impact on food consumption and waste. *Public Health Nutrition*, *19*(16). <u>https://doi.org/10.1017/S1368980016001348</u>
- 729. Visschers, V. H. M., Gundlach, D., & Beretta, C. (2020). Smaller servings vs. information provision: Results of two interventions to reduce plate waste in two university canteens. *Waste Management*, *103*. <u>https://doi.org/10.1016/j.wasman.2019.12.046</u>
- 730. Ofei, K. T., Holst, M., Rasmussen, H. H., & Mikkelsen, B. E. (2015). Effect of meal portion size choice on plate waste generation among patients with different nutritional status: An investigation using Dietary Intake Monitoring System (DIMS). *Appetite*, *91*. <u>https://doi.org</u> /10.1016/j.appet.2015.04.043
- 731. Freedman, M. R., & Brochado, C. (2010). Reducing portion size reduces food intake and plate waste. *Obesity Journal*, *18*(9). <u>https://doi.org/10.1038/oby.2009.480</u>
- 732. Vermote, M., Versele, V., Stok, M., Mullie, P., D'Hondt, E., Deforche, B., Clarys, P., & Deliens, T. (2018). The effect of a portion size intervention on French fries consumption, plate waste,

satiety and compensatory caloric intake: An on-campus restaurant experiment. *Nutrition Journal*, *17*(1). <u>https://doi.org/10.1186/s12937-018-0352-z</u>

- 733. *Small plates*. ReFED. Retrieved 20 March 2024 from <u>https://insights-engine.refed.org</u> /solution-database/small-plates
- 734. *Trayless*. ReFED. Retrieved 20 March 2024 from <u>https://insights-engine.refed.org/solution-database/trayless</u>
- 735. Rajbhandari-Thapa, J., Ingerson, K., & Lewis, K. H. (2018). Impact of trayless dining intervention on food choices of university students. *Archives of Public Health*, *76*(1). https://doi.org/10.1186/s13690-018-0301-5
- 736. Sarjahani, A., Serrano, E. L., & Johnson, R. (2009). Food and non-edible, compostable waste in a university dining facility. *Journal of Hunger & Environmental Nutrition*, 4(1),95. <u>https://doi.org/10.1080/19320240802706874</u>
- 737. Zhang, W., & Kwon, J. (2022). The impact of trayless dining implementation on university diners' satisfaction, food selection, consumption, and waste behaviors. *Sustainability*, 14(24),16669. <u>https://doi.org/10.3390/su142416669</u>
- 738. Thiagarajah, K., & Getty, V. M. (2013). Impact on plate waste of switching from a tray to a trayless delivery system in a university dining hall and employee response to the switch. *Journal of the Academy of Nutrition and Dietetics*, *113*(1). <u>https://doi.org/10.1016/j.jand</u>.2012.07.004
- 739. *The game-changing advantages of AI in grocery*. Afresh. Retrieved 7 March 2024 from https://www.afresh.com/resources/the-game-changing-advantages-of-ai-in-grocery
- 740. Felix, P. (2018). Data-driven retail food waste reduction: A comparison of demand forecasting techniques and dynamic pricing strategies. University of Twente. https://essay.utwente.nl/76380/1/Felix_MA_EEMCS.pdf
- 741. Clarke, L. (2023, October 10). Analysis: How Tesco and Otto are using data to forecast demand. *RetailWeek*. <u>https://www.retail-week.com/analysis-how-tesco-and-otto-are-using-data-to-forecast-demand-/5053784.article</u>
- 742. *Temperature monitoring (foodservice)*. ReFED. Retrieved 27 March 2024 from <u>https://insights-engine.refed.org/solution-database/temperature-monitoring-foodservice</u>
- 743. Eriksson, M., Strid, I., & Hansson, P. (2016). Food waste reduction in supermarkets Net costs and benefits of reduced storage temperature. *Resources, Conservation and Recycling, 107*,73. <u>https://doi.org/10.1016/j.resconrec.2015.11.022</u>
- 744. Chabada, L., Dreyer, H. C., Hvolby, H. H., & Kiil, K. (2015). Linkage between delivery frequency and food waste: Multiple case studies of a Norwegian retail chain. In S. Umeda, M. Nakano, H. Mizuyama, H. Hibino, D. Kiritsis, & G. von Cieminski, Advances in production management systems: Innovative production management towards sustainable growth Cham, Germany. <u>https://doi.org/10.1007/978-3-319-22759-7_13</u>
- 745. 2D in retail: next generation barcodes. One scan. Infinite possibilities. GS1. Retrieved 7 March 2024 from <u>https://www.gs1.org/industries/retail/2D-barcodes</u>
- 746. GS1. (2022). Woolworths Australia seeing multiple benefits from 2D barcodes. https://www.gs1.org/sites/gs1/files/case_study_library_item/case-study-woolworthsgs1australia.pdf
- 747. *Minimized on hand inventory*. ReFED. Retrieved 11 March 2024 from <u>https://insights-engine.refed.org/solution-database/minimized-on-hand-inventory</u>
- 748. Ministry for Primary Industries Manatū Ahu Matua. (2017). *Food control plan: Food service and retail*. <u>https://www.mpi.govt.nz/dmsdocument/11800-food-control-plan-food-service-and-food-retail-retail-basics-suitable-for-commercial-printing</u>
- 749. *Restaurant minimises food waste with a digital application*. European Cluster Collaboration Platform. Retrieved 03 February 2024 from <u>https://clustercollaboration.eu/content</u> /<u>restaurant-minimises-food-waste-digital-application</u>

- 750. *Food waste prevention pioneers*. Leanpath. Retrieved 13 February 2024 from <u>https://www.leanpath.com/about/</u>
- 751. *Food waste tracker: digital solution to reduce food waste*. Smart Kitchen. Retrieved 13 February 2024 from <u>https://smartkitchen.solutions/en/food-waste-reducing/</u>
- 752. *The global AI leader in commercial food waste solutions*. Winnow. Retrieved 13 February 2024 from <u>https://www.winnowsolutions.com/</u>
- 753. Eriksson, M., Malefors, C., Callewaert, P., Hartikainen, H., Pietiläinen, O., & Strid, I. (2019). What gets measured gets managed Or does it? Connection between food waste quantification and food waste reduction in the hospitality sector. *Resources, Conservation and Recycling: X, 4*,100021. <u>https://doi.org/10.1016/j.rcrx.2019.100021</u>
- 754. *Wise up on waste' toolkit*. Unilever food solutions. Retrieved 13 February 2024 from <u>https://www.unileverfoodsolutions.com.au/chef-training-academy/managing-food-waste/wise-up-on-waste-toolkit.html</u>
- 755. *Your business is food*. New South Wales Environmental Protection Agency. Retrieved 13 February 2024 from <u>https://www.lovefoodhatewaste.nsw.gov.au/business</u>
- 756. *Smart kitchen initiative*. Stop Waste. Retrieved 13 February 2024 from <u>https://www.stopwaste.org/at-work/reduce-and-reuse/reduce-wasted-food/smart-kitchen-initiative</u>
- 757. Grootenhuis, G. (2020, August 11). 15 institutions reduce food waste by 42% during Smart Kitchens San Diego pilot. *San Diego Food System Alliance*. <u>https://sdfsa.org/blog/15-</u> institutions-reduce-food-waste-smart-kitchens-san-diego
- 758. Elnakib, S. A., Quick, V., Mendez, M., Downs, S., Wackowski, O. A., & Robson, M. G. (2021). Food waste in schools: A pre-/post-test study design examining the impact of a food service training intervention to reduce food waste. *International Journal of Environmental Research and Public Health*, *18*(12). <u>https://doi.org/10.3390/ijerph18126389</u>
- 759. Alcorn, M. R., Vega, D., Irvin, R., & Paez, P. (2020). Reducing food waste: An exploration of a campus restaurant. *British Food Journal*, *123*(4),1546. <u>https://doi.org/10.1108/bfj-03-2020-0165</u>
- Yona, O., Goldsmith, R., & Endevelt, R. (2020). Improved meals service and reduced food waste and costs in medical institutions resulting from employment of a food service dietitian A case study. *Israel Journal of Health Policy Research*, 9(1). <u>https://doi.org/10.1186/s13584-020-0362-0</u>
- 761. Lebersorger, S., & Schneider, F. (2011). Discussion on the methodology for determining food waste in household waste composition studies. *Waste Management*, 31(9). <u>https://doi.org</u> /10.1016/j.wasman.2011.05.023
- 762. Commission Delegated Decision (EU) 2019/1597 of 3 May 2019 supplementing Directive 2008/98/EC of the European Parliament and of the Council as regards a common methodology and minimum quality requirements for the uniform measurement of levels of food waste. <u>https://eur-lex.europa.eu/legal-content/EN/ALL/?uri=CELEX:32019D1597</u>
- 763. Environment and Climate Change Canada. (2020). *National waste characterization report: The composition of Canadian residual municipal solid waste*. <u>https://publications.gc.ca/site/eng/9.884760/publication.html</u>
- 764. Edjabou, M. E., Petersen, C., Scheutz, C., & Astrup, T. F. (2016). Food waste from Danish households: Generation and composition. *Waste Management*, *52*. <u>https://doi.org/10.1016</u> /j.wasman.2016.03.032
- 765. Western Australia Waste Authority. (2021). *Food waste: In-home research on food waste in Western Australian households*. <u>https://www.wasteauthority.wa.gov.au/publications</u> /view/food-waste-reports/food-waste-in-home-research-summary-on-food-waste-in-wahouseholds-2021

- 766. *Food waste, 2020.* Statistisches Bundesamt (Destatis). Retrieved 23 February 2024 from <u>https://www.destatis.de/EN/Themes/Society-Environment/Environment/_</u> <u>Graphic/_Interactive/food-waste.html</u>
- 767. German Federal Ministry for the Environment, Nature Conservation, Nuclear Safety, and Consumer Protection. (2023). *Waste management in Germany 2023: Facts, data figures*. <u>https://www.bmuv.de/fileadmin/Daten_BMU/Pools/Broschueren/abfallwirtschaft_2023_en_bf.pdf</u>
- 768. Miezah, K., Obiri-Danso, K., Kádár, Z., Fei-Baffoe, B., & Mensah, M. Y. (2015). Municipal solid waste characterization and quantification as a measure towards effective waste management in Ghana. Waste Management, 46. <u>https://doi.org/10.1016</u> /j.wasman.2015.09.009
- 769. Malta Ministry of Sustainable Development, the Environment and Climate Change. (2014). *Waste management plan for the Maltese islands*. <u>https://10.2760/967005</u>
- 770. Caldeira, C., Cobalea, H. B., De Laurentiis, V., & Sala, S. (2019). *Review of studies on food waste accounting at member state level*. European Commission Joint Research Centre. https://data.europa.eu/doi/10.2760/340637
- Hanssen, O. J., Syversen, F., & Stø, E. (2016). Edible food waste from Norwegian households -Detailed food waste composition analysis among households in two different regions in Norway. *Resources, Conservation and Recycling, 109*. <u>https://doi.org/10.1016/j.resconrec.</u> 2016.03.010
- 772. Naturvårdsverket. (2022). *Matavfall i sverige*. <u>https://www.naturvardsverket.se/4ac386/</u>globalassets/media/publikationer-pdf/8800/978-91-620-8861-3.pdf
- 773. *Member state page: Sweden*. European Commission. Retrieved 23 February 2024 from <u>https://ec.europa.eu/food/safety/food_waste/eu-food-loss-waste-prevention-hub/eu-member-state-page/show/SE</u>
- 774. Waste and Resources Action Programme. (2016). *Synthesis of food waste compositional data* 2014 & 2015. <u>https://wrap.org.uk/sites/default/files/2020-12/Synthesis-of-Food-Waste-</u> <u>Compositional-Data-2014-and-2015.pdf</u>
- 775. United States Office of Resource Conservation and Recovery. (2020). 2018 Wasted food report. <u>https://www.epa.gov/sites/default/files/2020-11/documents/2018_wasted_food_report-11-9-20_final_.pdf</u>
- 776. United States Environmental Protection Agency. (2020). *Wasted food measurement methodology scoping memo*. <u>https://www.epa.gov/sites/default/files/2020-</u>06/documents/food measurement methodology scoping memo-6-18-20.pdf
- 777. Schanes, K., Dobernig, K., & Gözet, B. (2018). Food waste matters A systematic review of household food waste practices and their policy implications. *Journal of Cleaner Production*, *182*. <u>https://doi.org/10.1016/j.jclepro.2018.02.030</u>
- 778. Swannell, R., Bruns, H., Brüggemann, N., Candeal, T., Casonato, C., Diercxsens, C., Garcia Herrero, L., Gil Roig, J. M., Haglund, Y., Van Herpen, E., Kaptan, G., Kasza, G., Mikkelsen, B. E., Miranda Pires, I. M., Obersteiner, G., Vainioranta, J., Vittuari, M., Watanabe, K., & Sala, S. (2023). *Evaluation of consumer food waste prevention interventions*. Publications Office of the European Union. <u>https://publications.jrc.ec.europa.eu/repository/handle/JRC133003</u>
- 779. Stöckli, S., Niklaus, E., & Dorn, M. (2018). Call for testing interventions to prevent consumer food waste. *Resources, Conservation and Recycling*, *136*. <u>https://doi.org/10.1016/j.resconrec.2018.03.029</u>
- 780. Ovenden, K., Allpress, J., & Prakash, A. (2023). *A literature review of interventions to reduce household waste*. <u>https://www.knowledgeauckland.org.nz/media/2567/tr2023-02-literature-review-of-interventions-to-reduce-household-waste.pdf</u>
- 781. Soma, T., Li, B., & Maclaren, V. (2021). An evaluation of a consumer food waste awareness campaign using the motivation opportunity ability framework. *Resources, Conservation and Recycling, 168.* <u>https://doi.org/10.1016/j.resconrec.2020.105313</u>

- 782. *Food waste Warriors*. Pead. Retrieved 7 March 2024 from <u>https://www.pead.co.nz/case-studies/food-waste-warriors/</u>
- 783. Save money save food challenge. Brisbane City Council. Retrieved 08 February 2024 from https://www.brisbane.qld.gov.au/clean-and-green/rubbish-tips-and-bins/reducing-waste-athome/love-food-hate-waste/save-money-save-food-challenge
- 784. Van der Werff, E., Vrieling, L., Van Zuijlen, B., & Worrell, E. (2019). Waste minimization by households A unique informational strategy in the Netherlands. *Resources, Conservation and Recycling*, 144. <u>https://doi.org/10.1016/j.resconrec.2019.01.032</u>
- 785. Dyen, M., & Sirieix, L. (2016). How does a local initiative contribute to social inclusion and promote sustainable food practices? Focus on the example of social cooking workshops. *International Journal of Consumer Studies*, *40*(6). <u>https://doi.org/10.1111/ijcs.12281</u>
- 786. Young, W., Russell, S. V., Robinson, C. A., & Barkemeyer, R. (2017). Can social media be a tool for reducing consumers' food waste? A behaviour change experiment by a UK retailer. *Resources, Conservation and Recycling*, *117*. <u>https://doi.org/10.1016/j.resconrec.2016.10.016</u>
- 787. van Herpen, E., van der Lans, I., Nijenhuis-de Vries, M., Holthuysen, N., Kremer, S., & Stijnen, D. (2016). Consumption life cycle contributions: Assessment of practical methodologies for inhome food waste measurement. <u>https://eu-refresh.org/sites/default/files/D1.3%</u> 20final%20report%20Nov%202016.pdf
- Quested, T. E., Palmer, G., Moreno, L. C., McDermott, C., & Schumacher, K. (2020).
 Comparing diaries and waste compositional analysis for measuring food waste in the home.
 Journal of Cleaner Production, 262. <u>https://doi.org/10.1016/j.jclepro.2020.121263</u>
- 789. Löchtefeld, M., Møller, F., Kyster, L., Petersen, A. N., Jaqué, S. N., Larsen, A., & Ziadeh, H. (2023). FridgeSort Improving fridge sorting behaviour to reduce food waste through a mobile serious game. MUM '23: Proceedings of the 22nd International Conference on Mobile and Ubiquitous Multimedia. <u>https://doi.org/10.1145/3626705.3627801</u>
- 790. Nisa, C. F., Bélanger, J. J., & Schumpe, B. M. (2022). Assessing the effectiveness of food waste messaging. *Environmental Science and Policy*, *132*. <u>https://doi.org/10.1016/j.envsci.2022.02.020</u>
- 791. Karunasena, G. G., Ananda, J., & Pearson, D. (2023). *Case studies on household food waste reduction interventions from Australia*. Fight Food Waste Cooperative Research Centre. <u>https://endfoodwaste.com.au/wp-content/uploads/2023/11/Full-report_Case-studies-on-household-food-waste-reduction-interventions.pdf</u>
- 792. *Meeting with Kate Mead and Brody Gilroy, Waste-Ed with Kate.* (28 February 2024). Personal Communication.
- 793. Devaney, L., & Davies, A. R. (2017). Disrupting household food consumption through experimental HomeLabs: Outcomes, connections, contexts. *Journal of Consumer Culture*, 17(3). <u>https://doi.org/10.1177/1469540516631153</u>
- 794. van der Werf, P., Seabrook, J. A., & Gilliland, J. A. (2019). "Reduce food waste, save money": Testing a novel intervention to reduce household food waste. *Environment and Behavior*, 53(2). <u>https://doi.org/10.1177/0013916519875180</u>
- 795. Wharton, C., Vizcaino, M., Berardy, A., & Opejin, A. (2021). Waste watchers: A food waste reduction intervention among households in Arizona. *Resources, Conservation and Recycling, 164*. <u>https://doi.org/10.1016/j.resconrec.2020.105109</u>
- 796. *Waste Watchers*. Arizona State University. Retrieved 08 February 2024 from <u>https://globalfutures.asu.edu/waste-watchers/</u>
- 797. Laila, A., Gallant, M., Bain, M., Alexander, C., Reis, L., Welboren, A., von Massow, M., Parizeau, K., Walton, K., Ma, D. W. L., & Haines, J. (2024). Household food waste intervention is feasible, acceptable, and effective. *Journal of Nutrition Education and Behavior*, *56*(2). <u>https://doi.org/10.1016/j.jneb.2023.11.004</u>

- 798. *Savey Meal-Bot*. PAK'nSAVE. Retrieved 14 February 2024 from <u>https://saveymeal-bot.co.nz/ingredients</u>
- 799. Newshub. (2024, May 2). How to save money and waste less food. <u>https://www.newshub</u>.co.nz/home/lifestyle/2024/02/how-to-save-money-and-waste-less-food.html
- 800. McClure, T. (2023, October 08). Supermarket AI meal planner app suggests recipe that would create chlorine gas. *The Guardian*. <u>https://www.theguardian.com/world/2023</u> /aug/10/pak-n-save-savey-meal-bot-ai-app-malfunction-recipes
- 801. Mariah, M. A. A., Vonnie, J. M., Erna, K. H., Nur'Aqilah, N. M., Huda, N., Abdul Wahab, R., & Rovina, K. (2022). The emergence and impact of ethylene scavengers techniques in delaying the ripening of fruits and vegetables. *Membranes (Basel)*, *12*(2). <u>https://doi.org/10.3390/</u> membranes12020117
- 802. Bluapple[®]. Bluapple[®]. Retrieved 29 February 2024 from <u>https://thebluapple.com/</u>
- 803. *Make your fruit & veg last longer than ever before!* Fridge Friend[®]. Retrieved 29 February 2024 from <u>https://fridgefriend.com.au/</u>
- 804. *Purifie® FridgeMate*. Purifie®. Retrieved 29 February 2024 from <u>https://purifie.com.au/</u> product/purifie-fridgemate/
- 805. Robinson, D., & Collinson, P. (2013, October 12). The food gadgets that could save you money. *The Guardian*. <u>https://www.theguardian.com/money/2013/oct/12/food-gadgets-save-money-extend-shelf-life</u>
- Shu, Y., Booker, A., Karetny, J., O'Keefe, K., Rees, K., Schroder, L., & Roe, B. E. (2023).
 Evaluation of a community-based food waste campaign using a national control group.
 Waste Management, 160. <u>https://doi.org/10.1016/j.wasman.2023.02.011</u>
- 807. Comber, R., & Thieme, A. (2013). Designing beyond habit: Opening space for improved recycling and food waste behaviors through processes of persuasion, social influence and aversive affect. *Personal and Ubiquitous Computing*, *17*(6). <u>https://doi.org/10.1007/s00779-012-0587-1</u>
- 808. *Welcome to the waste-free future*. Mill Industries Inc. Retrieved 7 March 2024 from https://www.mill.com/food-recycler#top
- 809. Samsung Electronics to showcase how enhanced AI and connectivity enable expansive kitchen experiences at CES 2024. Samsung Newsroom. Retrieved 29 February 2024 from <u>https://news.samsung.com/global/samsung-electronics-to-showcase-how-enhanced-ai-andconnectivity-enable-expansive-kitchen-experiences-at-ces-2024</u>
- 810. *Making better policies for food systems*. Organisation for Economic Co-operation and Development. Retrieved 17 May 2024 from <u>https://www.oecd.org/food-systems/</u>